$\begin{array}{c} \mathbb{F} \text{ischer } \mathbb{C} \text{lifford } \mathbb{M} \text{atrices and } \mathbb{C} \text{haracter} \\ \mathbb{T} \text{ables of } \mathbb{C} \text{ertain } \mathbb{G} \text{roups } \mathbb{A} \text{ssociated with } \mathbb{S} \text{imple} \\ \mathbb{G} \text{roups } \mathbb{O}_{10}^+(2), \ \mathbb{H} S \text{ and } \mathbb{L} y \end{array}$

 $\mathbf{B}\mathbf{y}$

Thekiso Trevor Seretlo Thekiso.Seretlo@unw.ac.za 207525917@ukzn.ac.za

Supervisor : Professor Jamshid Moori moori@ukzn.ac.za



School of Mathematical Sciences University of KwaZulu-Natal, Pietermaritzburg, South Africa.

A project submitted in the fulfillment of the requirements for PhD School of Mathematical Sciences, University of KwaZulu-Natal, Pietermaritzburg. December 2011

Abstract

The character table of any finite group provides a considerable amount of information about a group and the use of character tables is of great importance in Mathematics and Physical Sciences. Most of the maximal subgroups of finite simple groups and their automorphisms are extensions of elementary abelian groups. Various techniques have been used to compute character tables, however Bernd Fischer came up with the most powerful and informative technique of calculating character tables of group extensions. This method is known as the Fischer-Clifford Theory and uses Fischer-Clifford matrices, as one of the tools, to compute character tables. This is derived from the Clifford theory. Here \overline{G} is an extension of a group N by a finite group G, that is $\overline{G} = N.G$. We then construct a non-singular matrix for each conjugacy class of $\overline{G}/N \cong G$. These matrices, together with *partial* character tables of certain subgroups of G, known as the *inertia* groups, are used to compute the full character tables of \overline{G} .

In this dissertation, we discuss Fischer-Clifford theory and apply it to both split and non-split extensions. We first, under the guidance of Dr Mpono, studied the group $2^7:S_8$ as a maximal subgroup of $2^7:SP(6,2)$, to familiarize ourselves to Fischer-Clifford theory. We then looked at $2^6:A_8$ and $2^8:O_8^+(2)$ as maximal subgroups of $2^8:O_8^+(2)$ and $O_{10}^+(2)$ respectively and these were both split extensions. Split extensions have also been discussed quite extensively, for various groups, by different researchers in the past. We then turned our attention to non-split extensions. We started with $2^4 \cdot S_6$ and $2^5 \cdot S_6$ which were maximal subgroups of HS and HS:2 respectively. Except for some negative signs in the first column of the Fischer-Clifford matrices we used the Fisher-Clifford theory as it is. The Fischer-Clifford theory, is also applied to $5^3 \cdot L(3,5)$, which is a maximal subgroup of the Lyon's group Ly. To be able to use the Fisher-Clifford theory we had to consider *projective* representations and characters of inertia factor groups. This is <u>not</u> a simple method and quite some smart computations were needed but we were able to determine the character table of $5^3 \cdot L(3,5)$. All character tables computed in this dissertation will be sent to GAP for incorporation.

Preface

The work described in this thesis was carried out under the supervision and direction of Professor Jamshid Moori, Department of Mathematics and Applied Mathematics, University of KwaZulu-Natal, Pietermaritzburg, from August 2007 to October 2011.

The thesis represents original work by the author and has not otherwise been submitted in any form for any degree or diploma to any other University. Where use has been made of the work of others it is duly acknowledged in the text.

Signed:

.....

T.T. Seretlo (Student)

.....

Prof J. Moori (Supervisor)

Dedication

I together with my dear wife Malehlohonolo and my three sons Mafa, Tefo and Kutlwano dedicate this dissertation to my mother Madinko, as our present, in appreciation of her kindness and steadfast support, for her 75th birthday. This is just but, a humble effort, to follow the tracks of the footsteps left by my father Bobby.

Acknowledgements

I first would like to thank the following who, each made significant contribution in one way or the other in my work, Prof T.A. Dube, Daluvuyo Makunga, Prof R.J. Moitsheki and Dr C.A. Pooe. I would like to thank the campus rector of NWU Maheking campus Prof D. Kwgadi for allowing me to go on a sabbatical leave so that I could concentrate on my work. I am most grateful to my colleagues in the maths department NWU Maheking especially Prof C.M. Khalique who has always been very supportive of my efforts. I am very thankful and appreciative of my collegues in the maths department at UKZN Pietermaritzburg campus who provided me with conducive conditions for study, more especially, Dr P. Horton and Ayoub Basheer, I am not sure they realize the importance of their contributions, in compiling this dissertation. I am also very thankful of the help I received from Prof F. Ali, Dr T. Breuer, Prof R.T. Curtis and Prof R.A. Wilson the first two mentioned really helped me a lot with GAP. I am deeply and overall very indebted to Dr Z.E. Mpono, who not only introduced me to Prof Moori and to Fischer-Clifford theory, but also brought back a sense of self belief in me. No words can justifiably express my gratitude to Prof Moori, he not only tutored me, monitored me, gave me extremely helpful advice but also discovered some capabilities in me, that I was not aware of.

Notation and conventions

Throughout this thesis all groups will be assumed to be finite, unless otherwise stated. We will use the notation and terminology from the ATLAS [23] and ATLAS $\forall 3$ [124].

\mathbb{N}	natural numbers
\mathbb{Z}	integers
\mathbb{Q}	rational numbers
\mathbb{R}	real numbers
\mathbb{C}	complex numbers
G, N, H, K	groups
1_G	the identity element of G
$H \leq G$	H is a subgroup of G
$H \cong G$	H is isomorphic to G
\mathbb{F}	a field
\mathbb{F}^*	$\mathbb{F}-\{0\}$
$\langle x,y \rangle$	the subgroup generated by x and y
N.G	an extension of N by G
N:G	a split extension of N by G
$N^{\cdot}G$	a non-split extension of N by G
h^g	conjugation of h by g
nX	a general conjugacy class of ${\cal G}$ with representatives of order n
$g_1 \sim g_2$	g_1 is conjugate to g_2
o(g)	order of $g \in G$

$C_G(g)$	the centralizer of g in G
[g]	a conjugacy class of G with representative g
$N_G(H)$	the normalizer of a subgroup H in G
Hg	the right coset of H in G
gH	the left coset of H in G
X,Y,Ω	sets
$ \Omega $	the cardinality of the set Ω
$1^{\alpha}2^{\beta}3^{\gamma}\dots$	cycle structure of a permutation
Irr(G)	the set of ordinary irreducible characters of G
I_G	the identity character of G
$\chi(G H)$	the permutation character of G on H
χ_H	the restriction of the character χ of G to the subgroup H
ψ^G	the induction of the character ψ of subgroup H to G
na, nb, \ldots	irreducible characters of G of degree n
$\langle \chi_i, \chi_j angle$	the inner product of the characters χ_i and χ_j
dim(V)	the dimension of a vector space V
A_n	the alternating group on n symbols
S_n	the symmetric group on n symbols
GF(q)	the Galois field of q elements
V(n,q)	a vector space of dimension n over $GF(q)$
$Sp_{2n}(q)$	symplectic group of dimension $2n$ over $GF(q)$
$O_{2n}^+(q)$	the orthogonal group consistent with the form f^+ on $V = V(2n,q)$ invariant
$O_8^+(2)$	the simple orthogonal group of dimension 8 over $GF(2)$, $ O_8^+(2) = 2^{12} \times 3^5 \times 5^2 \times 7$
$O_{10}^+(2)$	the simple orthogonal group of dimension 10 over $GF(2)$, $ O_{10}^+(2) = 2^{20} \times 3^5 \times 5^2 \times 7 \times 17 \times 31$
L(n,q)	the projective special linear group $(PSL(n,q))$ on $V = V(n,q)$
L(3, 5)	the projective special linear group of dimension 3 over $GF(5)$,
	$\left L(3,5)\right = 2^5 \times 3 \times 5^3 \times 31$
HS	the Higman-Sims group
Ly	the Lyons group
p^n	an elementary abelian group of order p^n

Contents

A	bstra	\mathbf{ct}	i
P	refac	2	ii
D	edica	tion	iii
A	cknov	wledgements	\mathbf{iv}
N	otati	ons and conventions	v
1	Introduction		1
2 Coset Analysis			
	2.1	Prologue	5
	2.2	Group Extensions	5
	2.3	Conjugacy Classes of Group Extensions	7
	2.4	Representations and Characters	12
	2.5	Induced Characters	16
	2.6	Permutation Characters	18
	2.7	Orbit Lengths	22
3	Pro	jective Representation	24
	3.1	Prologue	24
	3.2	Schur Multiplier	24
	3.3	Projective Representations	26

	3.4	Projective Characters	30
4	Clifford Theory		
	4.1	Prologue	32
	4.2	Clifford Theory and Normal Subgroups	33
	4.3	Clifford Theory and Projective Representations	39
	4.4	Irreducible Constituents and Conjugacy Classes	41
5	Fischer - Clifford Matrices		
	5.1	Prologue	44
	5.2	Definition and General Theory	45
		5.2.1 Properties of Fischer-Clifford Matrices	46
		5.2.2 Fischer-Clifford Matrices (Special Case)	50
	5.3	Split Cosets	51
	5.4	Non-Split Extensions	53
	5.5	Character Table and GAP	55
6	A g	roup $2^7:S_8$ in \overline{Fi}_{22}	56
	6.1	Introduction	56
	6.2	The action of S_8 on 2^7	57
	6.3	The conjugacy classes of $2^7:S_8$	58
	6.4	The Fischer-Clifford matrices of $2^7:S_8$	62
	6.5	The power maps of $2^7:S_8$	65
	6.6	The fusion of $2^7:S_8$ into $2^7:SP_6(2)$	66
7	A g	roup of the form $2^6:A_8$ as an inertia factor group of $2^8:O_8^+(2)$	83
	7.1	Introduction	83
	7.2	The Combinatorics Method	84
	7.3	The GAP Method	88
	7.4	The Conjugacy Classes of $2^6: A_8 \dots \dots$	92

	7.5	The F	ischer-Clifford matrices of $2^6: A_8$	95
	7.6	The C	haracter Table of $2^6: A_8$	96
8	A G	roup o	of the Form $2^8:O_8^+(2)$ as a maximal subgroup of $O_{10}^+(2)$	104
	8.1	3.1 Bilinear Forms		
	8.2	2 Orthogonal Groups		105
	8.3	B The action of $O_8^+(2)$ on 2^8		107
	8.4	The	Character Table of $2^8:O_8^+(2)$	116
		8.4.1	Inertia factor groups $Sp(6,2)$ and $2^6:A_8$ and their fusion into $O_8^+(2)$	116
		8.4.2	The Fischer-Clifford Matrices of \overline{G}	119
		8.4.3	The Character Table of \overline{G}	127
9	$2^{4} \cdot S_{6}$	$_3$ and 2	$2^{5} S_6$ as maximal subgroups of HS and HS :2 respectively	151
	9.1	Introd	uction	151
		9.1.1	The Conway Groups	152
		9.1.2	The Higman-Sims Group	153
		9.1.3	The Groups $2^{4} S_6$ and $2^{5} S_6 \ldots \ldots$	153
	9.2	The G	roup $\overline{G_1} = 2^4 \cdot S_6 \cdot \ldots \cdot $	154
		9.2.1	Construction of $G_1 \cong S_6$	154
		9.2.2	Conjugacy Classes and Inertia Factors of $\overline{G_1}$	157
		9.2.3	Fischer-Clifford Matrices of $\overline{G_1}$	157
	9.3	The G	roup $\overline{G} = 2^{5} \cdot S_6$	161
		9.3.1	Construction of $G \cong S_6$	161
		9.3.2	Conjugacy Classes and Inertia Factors of \overline{G}	162
		9.3.3	Fischer Clifford Matrices of \overline{G}	163
	9.4	Fusion	of $2^4 S_6$ into $2^5 S_6$	170
10	A gi	roup o	f the form $5^{3} L(3,5)$ as a maximal subgroup of the Lyons Group Ly	171
	10.1	Introd	uction	171
	10.2	Constr	ruction of $\overline{G} \cong 5^{3} \cdot L(3,5)$	172

	10.3 Construction of $G \cong L(3,5)$	173		
	10.4 Inertia Factors of \overline{G}	174		
	10.5 Projective Character Table of $5^2:2.A_5$	175		
	10.6 Fischer-Clifford Matrices of \overline{G}	176		
	10.7 PowerMaps of \overline{G}	178		
	10.8 Character Table of \overline{G}	179		
Appendix 182				
\mathbf{A}	Programmes	182		
в	Character Tables of S2 and S3	187		
Bił	3ibliography 1			

1 Introduction

The classification of finite simple groups theorem states that every finite simple group is isomorphic to exactly one of the following :

- cyclic groups of prime order,
- alternating groups A_n , where $n \ge 5$,
- groups of Lie type and
- the 26 sporadic groups.

Groups of Lie type can also be divided into two types namely the classical groups and exceptional groups that include twisted groups. Classical groups are the projective special linear groups, the sympletic groups, the unitary groups and the orthogonal groups. Groups of exceptional type are groups of the form $G_n(q), F_n(q), E_n(q)$ and the Tits group ${}^2F_4(2)'$. The sporadic groups are the five Mathieu groups, seven Leech lattice groups, three Fischer groups, five Monster groups and the six pariahs.

The completion of the classification of all simple groups might have led to some naive thought of the death of group theory. On the contrary this has led to very serious studies on the classification of maximal subgroups of simple groups. One way of classifying finite groups was by character theory. Isaacs provides an extensive study of character theory in [60]. In the classification of maximal subgroups of simple groups, calculating character tables of these groups, has been one of the methods used a lot. Quite a large number of maximal subgroups of simple groups are extensions of elementary abelian groups. In this dissertation all groups studied are extensions of elementary abelian groups. In the past most of these character tables were calculated using methods that did not provide a lot of insight about the structure of the group concerned. In this dissertation we use the Fischer-Clifford theory, for computing character tables, which is a method used by Bernd Fischer. In this dissertation we apply the Fischer-Clifford method to both split and, with some amendments if needed, to non-split extensions. We follow from Ali [1] also Ali and Moori [2] who not only applied it to split extensions but also to non-split extensions. This follows on the tracks of a large number of researchers who applied this method to split extensions. For further reading, on Fischer-Clfford theory, one can go to Ali and Moori [3], Almestady [4], Darafsheh and Iranmanesh ([26], [27]), Fischer ([34], [36], [37]), List [75], List and Mohammed [76], Moori and Mpono ([90], [91], [92]), Mpono [99], Pahlings [103], Saleh [112], Schiffer [113] and Whitely [120].

In Chapter 2 we give some preliminary results on group extensions and group characters which will be used in later chapters. In section 2.2 we define group extensions and discuss some basic results. In section 2.3 we discuss the conjugacy classes of group extensions. We briefly discuss the technique of coset analysis for computing conjugacy classes of a group extension \overline{G} of N by G where N is an elementary abelian normal subgroup of \overline{G} . The technique of coset analysis was developed by Moori [81] which he also used in [82] and this has been widely used for computing conjugacy classes of group extensions. Analogous to the programmes developed in MAGMA by Ali [1] and in Cayley by Mpono [99] we developed Programmes A and B in GAP [41] which we used to compute the conjugacy classes of the groups $2^7:S_8$, $2^6:A_8$ and $2^8:O_8^+(2)$ that we studied in this dissertation in Chapters 6, 7 and 8 respectively. In section 2.4 we studied preliminary results on representation and characters which is used in later chapters. In section 2.5 we look at induced characters, section 2.6 deals with permutation characters and in section 2.7 we develop Programme C in GAP [41] that we use to compute orbit lengths of orbits of conjugacy classes and irreducible characters. For further reading on group extensions, representations and character of groups readers can also look at the following [5], [7], [10], [11], [14], [20], [21], [30], [48], [56], [60], [61], [62], [63], [71], [74], [100], [105], [110], [116] and [117].

In Chapter 3 we discuss projective representations and projective characters. The first step in obtaining the projective representations of a group G is to compute its *Schur multiplier*. In section 3.2 we discuss the Schur multiplier of a group and methods of computing the Schur multiplier. In section 3.3 we discuss the projective representations of G. We prove that for a projective representation P with factor set α of degree n, then $o([\alpha]) \mid n$. We show how a projective representation of G can be obtained from an ordinary representation of a "representation group" of G. We also look at three different methods of constructing a projective representation of a group G. We then study the projective characters of G in section 3.4. For further reading on projective representations and projective characters one can read [11], [47], [51], [55], [58], [60], [95], [96], [97], [98], [100], [107], [108] and [109].

In Chapter 4 we study the Clifford theory for ordinary and projective representations of a group $\overline{G} = N.G$, where $N \leq G$. Here we discuss how the groups N and G are related to \overline{G} and the consequences thereof. In section 4.2 we study Clifford Theory and normal subgroups. In section 4.3 we discuss Clifford Theory and projective representations. In Section 4.4 we look at group action and how to use GAP in our computations where the dimension of N is not the same as the permutation degree of G. Last in Section 4.5 we look at irreducible constituents and conjugacy classes. This is all required for the construction of Fischer-Clifford matrices which we discuss

in Chapter 5. For further reading on Clifford Theory, one can go to [11], [40], [58], [60], [66], [67] and [100].

Chapter 5 is devoted to one of the most important tools used in this dissertation namely the Fischer-Clifford matrices. If $\overline{G} = N.G$ is an extension of N by G where N, which is normal in \overline{G} , is an elementary abelian group. We compute a non-singular matrix for each conjugacy class of $\overline{G}/N \cong G$. Then we use these matrices, fusion maps and the partial character tables of inertia factor subgroups to compute the full character table of \overline{G} . In this dissertation we apply this technique to both split and non-split extensions. This technique has been used mostly in split extensions by Almestady [4], Darafsheh and Iranmanesh ([26], [27]), Fischer ([34], [35], [36]), [37], List [75], List and Mohammed [76], Moori and Mpono ([90], [91], [92]), Mpono [99], Pahlings [103], Saleh [112], Schiffer [113] and Whitely [120]. With, the necessary adjustments, Ali [1] used this method for both split and non-split extensions. Ali and Moori [2, 3] also used it for non-split extensions.

In Chapter 6 we look at the group $2^7:S_8$ as a maximal subgroup of $2^7:SP(6,2)$ which in turn is a maximal subgroup of \overline{Fi}_{22} , the full automorphism group of the smallest Fischer sporadic simple group Fi_{22} . In section 6.1 we compute the generators of Sp(6,2) which we use in section 6.2 to compute the generators of S_8 . In section 6.3 we compute and discuss the conjugacy classes of $2^7:S_8$ and then in section 6.5 we look at the power maps of $2^7:S_8$ which we use in computing the fusion of $2^7:S_8$ into $2^7:SP(6,2)$. We conclude this chapter by determining the character table of $2^7:S_8$.

Chapter 7 is based on a group of the form $2^6:A_8$ as an inertia factor group of $2^8:O_8^+(2)$. Here we use two methods to compute A_8 with 6×6 matrix generators. In section 7.2 we use combinatorics to compute these generators. In section 7.3 we use GAP to compute the generators of A_8 inside $O_8^+(2)$. In section 7.4 we compute and discuss the conjugacy classes of $2^6:A_8$. In 7.5 we compute and discuss the Fischer-Clifford matrices of $2^7:A_8$. In section 7.6 we finish off the chapter by calculating the character table of $2^6:A_8$.

Chapter 8 is concerned with a group of the form $2^8:O_8^+(2)$ as a maximal subgroup of $O_{10}^+(2)$. The group $2^{10+16}O_{10}^+(2)$ in turn is a maximal subgroup of $F_1 = M$, the monster. In section 8.1 we define the bilinear forms from which we use to define orthogonal forms which are in turn used to define orthogonal groups in section 8.2. In section 8.3 we look at the action of $O_8^+(2)$ on the elementary abelian group 2^8 . We then compute the inertia factor groups of $2^8:O_8^+(2)$ and their fusion into $O_8^+(2)$ in section 8.4.1. In section 8.4.2 we compute the Fischer-Clifford matrices while section 8.4.3 deals the power maps and the character table of $2^8:O_8^+(2)$.

Chapter 9 is focused on groups of the form $2^{4} \cdot S_6$ and $2^5 \cdot S_6$ which are both non-split extensions, with $2^{4} \cdot S_6$ and $2^5 \cdot S_6$ maximal subgroups of the Higman-Sims HS, and its full automorphism group HS:2 respectively. In order to be able to use the Fischer-Clifford theory on these extensions, we use the methods that are used by Ali [1]. In section 9.2 we discuss the Higman Sims group in relation to the Conway groups. We use the ATLAS \vee 3 to construct HS in section 9.1.2. In section 9.1.3 again using programmes from the ATLAS \vee 3 we construct the subgroups $2^{4} \cdot S_6$ and $2^5 \cdot S_6$ of HS and HS:2 respectively. In the group $2^{5} \cdot S_6$ we constructed, we found that there were three distinct groups of order 11 520. One of them was $2^{4} \cdot S_6$, the second was the split extension $2^4 \cdot S_6$ and the third we determined as $2^{5} \cdot A_6$. In section 9.2.1, inside HS:2, we constructed $2^4 \cdot S_6$ isomorphic to the one which was a maximal subgroup of HS. We then used the GAP method used in Chapter 7 to construct the generators of S_6 as 4×4 matrices. In section 9.2.2 and we let S_6 act on 2^4 and we then computed the conjugacy classes and inertia factor groups of $2^{4} \cdot S_6$. In section 9.2.3 we compute the Fischer-Clifford matrices and full character table of $2^{4} \cdot S_6$. In section 9.3 we look at the group $2^5 \cdot S_6$, constructed as an automorphism group of $2^4 \cdot S_6$ in section 9.3.1, similar to the methods in section 9.2.1, we construct the generators of S_6 as 5×5 generators in GF(2). In section 9.3.2 we let S_6 act on 2^5 and we then compute the conjugacy classes and inertia factor groups of $2^{5} \cdot S_6$. We finished off the chapter in section 9.3.3 by computing the Fischer-Clifford matrices and character table of $2^{5} \cdot S_6$.

The main focus of Chapter 10 is on a group of the form $5^3 \cdot L(3,5)$ as a maximal subgroup of the Lyons group Ly. Again as in Chapter 9, using the ATLAS $\forall 3$ we construct the group $5^3 \cdot L(3,5)$ inside the group Ly. This was done in section 10.2. In section 10.3 we use the methods of Chapter 6, which were used again in Chapter 9, to construct the group L(3,5). In 10.6 we let L(3,5) act on 5^3 and then we compute the inertia factor groups of $5^3 \cdot L(3,5)$. To be able to adopt the Fischer-Clifford method we compute the Schur multiplier and to do this we need the projective character table of $5^2:2.A_5$. In section 10.6 we compute the Fischer matrices, in section 10.7 we deal with the power maps of $5^3 \cdot L(3,5)$. The computation of the character table of $5^3.L(3,5)$ is done in section 10.8. A programme from GAP, I developed with the help of F. Ali, was used to check the accuracy and consistency of the character tables computed.

2

Coset Analysis

2.1. Prologue

In this chapter we give preliminary results on group extensions and group characters that will be required in later chapters. In section 2.2 we present definitions and some basic results on group extensions. In section 2.3 we discuss the conjugacy classes of elements of group extensions. We describe the technique of coset analysis for computing the conjugacy classes of group extension \overline{G} of N by G where N is an abelian normal subgroup of \overline{G} . This technique was developed and first used by Moori in [81, 82] and has since been widely used for computing the conjugacy classes of group extensions in all cases where it is applicable. For example, it has been used in Ali [1], Mpono [99], Saleh [112] and Whitely [120]. We also develop two GAP Programmes A and B analogous to the programmes developed by Mpono [99] for CAYLEY and Ali [1] for MAGMA to compute the conjugacy classes and the orders of the class representatives for the split extensions $\bar{G} = N:G$ where N is an elementary abelian p-group. We use these programmes to compute the conjugacy classes of the group extensions $2^7:S_8$, $2^6:A_8$ and $2^8:O_8^+(2)$ which will be studied in Chapters 6, 7 and 8 respectively. In Section 2.4 we present some theory on representations and characters of groups by concentrating on those results which would be useful in later chapters. Section 2.5 deals with the relationship between the characters of a group G and the characters of a subgroup H of G. In this section we will first study restriction of characters and then go on to study induced characters. Finally in Section 2.6 we give some results on permutation characters. For further information and readings on group extensions, group representations and characters readers are encouraged to consult [5, 7, 10, 11, 14, 56, 60, 61, 62, 63, 71, 74, 100, 105, 110, 116, 117] and many other relevant sources.

2.2. Group Extensions

Definition 2.2.1. Let N and G be groups. An extension of N by G is a group \overline{G} that satisfies the following properties

- (i) $N \leq \overline{G}$,
- (ii) $\overline{G}/N \cong G$.

We say that \overline{G} is a split extension of N by G if \overline{G} contains subgroups N and G_1 with $G_1 \cong G$ such that

- (i) $N \leq \overline{G}$,
- (*ii*) $NG_1 = \overline{G}$,
- (*iii*) $N \cap G_1 = \{1_{\bar{G}}\}.$

In this case \overline{G} is also called a semi-direct product of N and G, and identify G_1 and G.

Following ATLAS [23], we denote an arbitrary extension of N by G by N.G. A split extension of N by G is denoted by N:G and a non-split extension is denoted by N.G.

Definition 2.2.2. The automorphism group of a group G, denoted by Aut(G), is the set of all automorphisms of G under the binary operation of composition.

For \overline{G} , a semidirect product of N by G, every element in \overline{G} can be uniquely expressed in the form ng, where $n \in N$ and $g \in G$ and the multiplication of elements of \overline{G} is given by

$$(n_1g_1)(n_2g_2) = n_1 n_2^{g_1} g_1 g_2$$

where $n^g = gng^{-1}$. Also there is a homomorphism $\theta : G \longrightarrow Aut(N)$ given by $\theta(g) = \theta_g$, where $g \in G, \theta_g : N \longrightarrow N$ is defined by $\theta_g(n) = gng^{-1}$ and θ_g is an automorphism of N. Hence G acts on N.

Definition 2.2.3. Let \overline{G} , N and G be as defined above and $\theta : G \longrightarrow Aut(N)$. The semidirect product \overline{G} of N by G is said to realize θ if $\theta_g(n) = n^g \forall n \in N, g \in G$.

Remark 2.2.4. For \overline{G} a semidirect product of N by G, then \overline{G} is isomorphic to a semidirect product of N by G that realizes θ for some $\theta : G \longrightarrow Aut(N)$.

If \bar{G} is a split extension of N by G, then $\bar{G} = NG = \bigcup_{g \in G} Ng$. So G may be regarded as a right transversal for N in \bar{G} (that is, a complete set of right coset representatives of N in \bar{G}). Now suppose \bar{G} is any extension of N by G, not necessarily split, then since $\bar{G}/N \cong G$, there is a homomorphism $\lambda : \bar{G} \longrightarrow G$ that is onto, with kernel N. For $g \in G$ define a lifting of g as an element $\bar{g} \in \bar{G}$ such that $\lambda(\bar{g}) = g$. By choosing a lifting of each element of G, we get the set $\{\bar{g} : g \in G\}$ that is a transversal for N in \bar{G} .

We now show that even for a non-split extension of N by G, if N is abelian, G acts on N.

Lemma 2.2.5. ([1, 99, 111, 120]) Let \overline{G} be an extension of N by G where N is abelian. Then there is a homomorphism $\theta: G \longrightarrow Aut(N)$ such that $\theta_g(n) = \overline{g}n(\overline{g})^{-1}, n \in N$ and θ is independent of the choice of liftings $\{\overline{g} : g \in G\}$.

PROOF. Let $a \in \overline{G}$ and γ_a denote conjugation by a. Since N is a normal subgroup of \overline{G} , $(\gamma_a)_N \in Aut(N)$ and the function $\mu : \overline{G} \longrightarrow Aut(N)$ defined by $\mu(a) = (\gamma_a)_N$ is a homomorphism. If $a \in N$, then since N is abelian we have $\mu(a) = I_N$. Thus there is a homomorphism $\mu^* : \overline{G}/N \longrightarrow Aut(N)$ which is given by $\mu^*(Na) = \mu(a)$. However $G \cong \overline{G}/N$ and for any lifting $\{\overline{g} : g \in G\}$, the function $\phi : G \longrightarrow \overline{G}/N$ defined by $\phi(g) = N\overline{g}$ is an isomorphism. If $\{\overline{g}_1 : g \in G\}$ is another choice of liftings, then $\overline{g} \ \overline{g}_1^{-1} \in N$ for every $g \in G$ and thus $N\overline{g} = N\overline{g}_1$. Therefore the isomorphism ϕ is independent of the choice of liftings. Let $\theta : G \longrightarrow Aut(N)$ be the composition $\mu^* \circ \phi$. For $g \in G$ and \overline{g} a lifting of g, then $\theta(g) = \mu^*(\phi(g)) = \mu^*(N\overline{g}) = \mu(\overline{g}) \in Aut(N)$ and thus for $n \in N$, we have $\theta_g(n) = \mu(\overline{g})(n) = \overline{g}n(\overline{g})^{-1}$. Hence the result.

Remark 2.2.6. [120] Let \overline{G} be an extension of N by G where N is abelian and for each $g \in G$ let \overline{g} be a lifting of g. We identify G with \overline{G}/N under the isomorphism $g \mapsto N\overline{g}$. Thus $\{\overline{g} \mid g \in G\}$ is a right transversal for N in \overline{G} and thus every $x \in \overline{G}$ has a unique expression of the form $x = n\overline{g}$ where $n \in N$ and $g \in G$.

2.3. Conjugacy Classes of Group Extensions

In this section we discuss the technique of *coset analysis*, which was initially introduced by Moori [81], to determine the conjugacy classes of group extensions but first we state the following two results.

Theorem 2.3.1. Let G be a finite group

(i) Suppose that C_1 and C_2 are two conjugacy classes of G such that $C_1 \neq [1_G]$ and $C_1^n = C_2$ for some integer $n \geq 2$, where

$$C_1^n = \{x_1 x_2 \cdots x_n \mid x_i \in C_1, \ 1 \le i \le n\}$$
.

Then there exists some normal subgroup N of G and $g \in G - N$ such that C_1 is the coset Ng and the map $x \mapsto x^n$ is a bijection from C_1 onto C_2 .

(ii) If G has a normal subgroup N and $g \in G - N$ such that the coset Ng is a single conjugacy class of G, and such that for some $n \in \mathbb{Z}$ the map $x \mapsto x^n$ for $x \in Ng$ is a monomorphism, then Ng^n is a conjugacy class of G and $(Ng)^n = Ng^n$.

PROOF. See [12].

Proposition 2.3.2. Let $\overline{G} = N.G$, $\overline{g} \in \overline{G}$ a lifting of $g \in G$, C be the centralizer of $N\overline{g}$ in G and \overline{C} be the complete pre-image in \overline{G} of C. Then

- (i) the union of the cosets $N\bar{x}$ which are conjugate in G to $N\bar{g}$, is the union of the conjugacy classes L_1, L_2, \ldots, L_r of \bar{G} ,
- (ii) \bar{C} acts on the coset $N\bar{g}$ by conjugation,
- (iii) \bar{C} has r orbits in its action on $N\bar{g}$ and the orbit representatives $\bar{g}_1, \bar{g}_2, \ldots, \bar{g}_r$ are representatives of the conjugacy classes L_1, L_2, \ldots, L_r of \bar{G} ,
- (iv) the centralizer $C_{\bar{G}}(\bar{g}_i)$ for $1 \leq i \leq r$ is the stabilizer of \bar{g}_i in \bar{C} in its action on $N\bar{g}$.

PROOF. See [16].

We now briefly discuss the technique of *coset analysis* to determine the conjugacy classes of elements of group extensions $\bar{G} = N.G$ where N is an abelian normal subgroup of \bar{G} . For detailed information about this technique we encourage readers to consult F Ali [1], Moori [81, 82] and Mpono [99].

For each conjugacy class [g] in G with representative $g \in G$, we analyze the coset $N\overline{g}$, where \overline{g} is a lifting of g in \overline{G} and

$$\bar{G} = \bigcup_{q \in G} N\bar{g}$$

To each class representative $g \in G$ with lifting $\bar{g} \in \bar{G}$, we define

$$C_{\bar{q}} = \{ x \in \bar{G} : x(N\bar{g}) = (N\bar{g})x \}$$

Then $C_{\bar{g}}$ is the stabilizer of $N\bar{g}$ in \bar{G} under the action by conjugation of \bar{G} on $N\bar{g}$, and hence $C_{\bar{g}}$ is a subgroup of \bar{G} .

Remark 2.3.3. It is not difficult to see that N is a normal subgroup of $C_{\bar{q}}$.

Lemma 2.3.4. [120] $C_{\bar{g}}/N = C_{\bar{G}/N}(N\bar{g})$.

PROOF. Consider Nk, where $k \in \overline{G}$. Then

$$\begin{split} Nk \in C_{\bar{G}/N}(N\bar{g}) & \Leftrightarrow \quad Nk(N\bar{g})(Nk)^{-1} = N\bar{g} \\ & \Leftrightarrow \quad NkN\bar{g}Nk^{-1} = N\bar{g} \\ & \Leftrightarrow \quad NkN\bar{g}k^{-1} = N\bar{g} \\ & \Leftrightarrow \quad NkNn\bar{g}k^{-1} = N\bar{g} \quad \forall n \in N \\ & \Leftrightarrow \quad Nkn\bar{g}k^{-1} = N\bar{g} \ , \ \forall n \in N \\ & \Leftrightarrow \quad kn\bar{g}k^{-1} \in N\bar{g} \ , \ \forall n \in N \\ & \Leftrightarrow \quad k \in C_{\bar{g}} \\ & \Leftrightarrow \quad Nk \in C_{\bar{g}}/N \ . \end{split}$$

Thus we obtain that $C_{\bar{g}}/N = C_{\bar{G}/N}(N\bar{g}).$

8

Remark 2.3.5. Using Remark 2.3.4 and Lemma 2.3.5 we deduce that $C_{\bar{g}} = N.C_{\bar{G}/N}(N\bar{g})$. For \bar{g} a lifting of $g \in G$ in \bar{G} , we can identify $C_{\bar{G}/N}(N\bar{g})$ with $C_G(g)$ and write $C_{\bar{g}} = N.C_G(g)$ in general. If $\bar{G} = N:G$ then we can identify $C_{\bar{g}}$ with $C_g = \{x \in \bar{G} : x(Ng) = (Ng)x\}$, where the lifting of g in \bar{G} is g itself since $G \leq \bar{G}$ in the case of a split extension.

Corollary 2.3.6. If $\overline{G} = N:G$, then $C_g = N:C_G(g)$.

PROOF. We have that N is a normal subgroup of C_g . Now we show that $C_G(g) \leq C_g$ and that $N \cap C_G(g) = \{1\}$. Let $x \in C_G(g)$. Then we obtain $(Ng)^x = x(Ng)x^{-1} = xNgx^{-1} = Nxgx^{-1} = Ng$. Thus $x \in C_g$ and hence $C_G(g) \leq C_g$. Since $N \cap C_G(g) \leq N \cap G = \{1_G\}$, then we have that $N \cap C_G(g) = \{1_G\}$. Hence the result.

The conjugacy classes of \overline{G} (where N is abelian) will be determined by the action by conjugation of $C_{\overline{g}}$, for each conjugacy class [g] of G, on the elements of $N\overline{g}$. To act $C_{\overline{g}}$ on the elements of $N\overline{g}$, we first act N and then act $\{\overline{h} : h \in C_G(g)\}$, where \overline{h} is a lifting of h in G. We outline this action in two steps as follows:

STEP 1: The action of N on $N\bar{g}$: Let $C_N(\bar{g})$ be the stabilizer of \bar{g} in N. Then for any $n \in N$ we have $x \in C_N(n\bar{g}) \iff x \in C_N(\bar{g})$. Thus $C_N(\bar{g})$ fixes every element of $N\bar{g}$. Now let $|C_N(\bar{g})| = k$. Then under the action of N, $N\bar{g}$ splits into k orbits Q_1, Q_2, \ldots, Q_k , where

$$|Q_i| = [N: C_N(\bar{g})] = \frac{|N|}{k}$$
,

for $i \in \{1, 2, \dots, k\}$.

STEP 2: The action of $\{\bar{h} \mid h \in C_G(g)\}$ on $N\bar{g}$: Since the elements of $N\bar{g}$ are now in the orbits Q_1, Q_2, \ldots, Q_k from Step 1 above, we need only act $\{\bar{h} \mid h \in C_G(g)\}$ on these k orbits. Suppose that under this action f_j of these orbits Q_1, Q_2, \ldots, Q_k fuse together to form one orbit Δ_j , then the f_j 's obtained this way must satisfy

$$\sum_{j} f_j = k$$

and we have

$$|\Delta_j| = f_j \times \frac{|N|}{k}$$

Thus for $x = d_j \overline{g} \in \Delta_j$, we obtain that

$$|[x]_{\bar{G}}| = |\Delta_j| \times |[g]_G| = f_j \times \frac{|G|}{k|C_G(g)|}$$

and thus we obtain that

$$C_{\bar{G}}(x)| = \frac{|\bar{G}|}{|[x]_{\bar{G}}|} = |\bar{G}| \times \frac{k|C_G(g)|}{f_j|\bar{G}|} = \frac{k|C_G(g)|}{f_j}$$

Thus to calculate the conjugacy classes of $\overline{G} = N.G$, we need to find the values of k and the f_j 's for each class representative $g \in G$.

Remark 2.3.7. However in the case of $\overline{G} = N:G$ a split extension, we analyze the coset Ng instead of $N\overline{g}$ since in this case $G \leq \overline{G}$. Under the action of N on Ng, we always assume that $g \in Q_1$. Also instead of acting $\{\overline{h} : h \in C_G(g)\}$ on the k orbits Q_1, Q_2, \ldots, Q_k we just act $C_G(g)$ on these orbits. Since $g \in Q_1$, then $C_G(g)$ always fixes Q_1 and thus we will always have $f_1 = 1$. Hence

$$k = \sum_{j} f_j = 1 + \sum_{m} f_m$$

where the sum is taken over all m such that $g \notin Q_m$.

We now prove and discuss techniques that are useful in the determination of the orders of the elements of $\bar{G} = N:G$.

Theorem 2.3.8. Let $\overline{G} = N:G$ and $dg \in \overline{G}$ where $d \in N$ and $g \in G$ such that o(g) = m and o(dg) = k. Then m divides k.

PROOF. We have that

$$1_{\bar{G}} = (dg)^k = dd^g d^{g^2} d^{g^3} \dots d^{g^{k-1}} g^k$$

Since G acts on N and $d \in N$, we have $d, d^g, d^{g^2}, \ldots, d^{g^{k-1}} \in N$. Hence $dd^g d^{g^2} \ldots d^{g^{k-1}} \in N$. Thus we must have that $dd^g d^{g^2} \ldots d^{g^{k-1}} = 1_N$ and $g^k = 1_G$. Hence m divides k.

Theorem 2.3.9. Let $\overline{G} = N:G$ such that N is an elementary abelian p-group, where p is prime. Let $dg \in \overline{G}$ where $d \in N$ and $g \in G$ such that o(g) = m and o(dg) = k. Then either k = m or k = pm.

PROOF. See ([1],[99]).

Remark 2.3.10. Let $\overline{G} = N:G$, where N is an elementary abelian p-group. Let $dg \in \overline{G}$ with $d \in N, g \in G$ such that o(g) = m and o(dg) = k, then we observe that

$$(dg)^m = d.d^g.d^{g^2}.\ldots.d^{g^{m-1}}g^m .$$

Since $g^m = 1_G$, we obtain that $(dg)^m = w$, where $w \in N$ and it is given by

$$w = d.d^g.\ldots.d^{g^{m-1}}$$

By Theorem 2.3.9 above, we have that if $w = 1_N$ then k = m and if $w \neq 1_N$ then k = pm.

We use the method of coset analysis discussed above (outlined in Steps 1 and 2) together with Theorems 2.3.8 and 2.3.9 and Remark 2.3.10 in developing Programmes A and B in GAP [41] (analogous to the programmes developed by Ali [1] for MAGMA and Mpono [99] for CAYLEY) which are applied for the computation of conjugacy classes and the orders of the class representatives of the extension $\bar{G} = N:G$ where N is an elementary abelian p-group for prime p on which a linear group G acts.

Programme A

$$\label{eq:constraint} \begin{split} & \text{V:=FullRowSpace}(GF(q),n);\\ & \text{gr}[1]{:=}(\text{OneGF}(q))^*[n \times n \text{ matrix group generators}];\\ & \text{gr}[2]{:=}(\text{OneGF}(q))^*[n \times n \text{ matrix group generators}]; \end{split}$$

÷

 $gr[k]:=(OneGF(q))*[n \times n matrix group generators];$ $grp:=Group(gr[1], gr[2], \cdots, gr[k]);$ Ccl:=ConjugacyClasses(m); O:=Union(Orbits(grp,V));for i in [1..n(Ccl)] do Print(Representative(Ccl[i])); $w{:=}One(GF(q))^*[0,0,\cdots,0];$ $e{:=}[];$ while Difference (O,e) $<>[\]$ do d := [];for x in O do; $y := [x + w + (x^*(Representative(Ccl)[i]))];$ d:=Union(d,y);od; Print(d); e:=Union(d,e);if Difference(O,e) <> [] then w:=Representative(Difference(O,e)); fi; od; r:=[]; $u:=One(GF(q))^*[0,0,\cdots,0];$ while Difference (O,e) $<>[\]$ do m:=[]; for g in Centralizer (grp,Representative(Ccl)[i]) do $l:=[u^*g];$ m:=Union(m,l);od; Print("A block for the vectors under the action of a centralizer"); Print(m); r:=Union(m,r);if Difference(O,r) <> [] then u:=Representative(Difference(O,r)); fi; od;Print("********* od;

Programme B

V:=FullRowSpace(GF(q), n);m[1]:=(OneGF(q))*[$n \times n$ matrix group generators]; m[2]:=(OneGF(q))*[$n \times n$ matrix group generators];

÷

 $m[k]:=(OneGF(q))*[n \times n matrix group generators];$

 $m:=\operatorname{Group}(m[1], m[2], \cdots, m[k]);$ $c:=\operatorname{ConjugacyClasses}(m);$ $g:=\operatorname{Representative}(c[i]);$ $d:=\operatorname{One}(\operatorname{GF}(q))^*[\alpha_1, \alpha_2, \cdots, \alpha_n];$ $w:=d+d*g+d*g^2+\cdots+d*g^{m-1};$ $\operatorname{Print}(w);$

In Programme B we have o(g) = m and $g \in S$ is a class representative, for $1 \leq j \leq n, \alpha_j \in GF(q)$, $d * g = d^g$, and + signifies the operation in V and $dg \in \overline{G}$ is a class representative from the coset Ng.

2.4. Representations and Characters

In this section we give some preliminary results on representations and characters of groups which will be needed in later chapters. For further reading one can go to [1, 99].

Definition 2.4.1. Let G be a group, \mathbb{F} a field and $GL(n,\mathbb{F})$ the general linear group which is the multiplicative group of all nonsingular $n \times n$ matrices over F for some integer n. Then a homomorphism $\rho : G \longrightarrow GL(n,\mathbb{F})$ is called a **representation** of G over \mathbb{F} or simply an \mathbb{F} representation. The representation ρ is said to have degree n. The function $\chi : G \longrightarrow \mathbb{F}$ given by $\chi(g) = trace(\rho(g))$ is called the \mathbb{F} -character of G afforded by the F-representation ρ . The degree of χ is the same as that of ρ .

Two \mathbb{F} -representations ρ_1 and ρ_2 of G are said to be **equivalent** if there exists $P \in GL(n, \mathbb{F})$ such that $\rho_1(g) = P\rho_2(g)P^{-1}$ for all $g \in G$. An \mathbb{F} -representation ρ of G is said to be **reducible** if it is equivalent to a representation α which is given by

$$\alpha(g) = \left(\begin{array}{cc} \beta(g) & \gamma(g) \\ 0 & \delta(g) \end{array} \right)$$

for all $g \in G$, where β, γ, δ are F-representations of G. If ρ is not reducible, then it is said to be **irreducible**. Since similar matrices have the same trace, then it follows that equivalent representations afford the same character. The character afforded by an irreducible representation is called an **irreducible character**. Sums and products of characters are themselves characters.

We now give a famous result of Schur [114] which provides an assessable approach to group characters.

Theorem 2.4.2. (Schur's Lemma) Let $\rho_1 : G \longrightarrow GL(n, \mathbb{F})$ and $\rho_2 : G \longrightarrow GL(m, \mathbb{F})$ be two irreducible representations of a group G over a field \mathbb{F} . Assume that there exists a matrix P such that $P\rho_1(g) = \rho_2(g)P$ for all $g \in G$. Then either P is the zero matrix or P is nonsingular so that $\rho_1(g) = P^{-1}\rho_2(g)P$.

PROOF. See Theorem 1.8 of [89].

Corollary 2.4.3. [89] If $\rho: G \longrightarrow GL(n, \mathbb{F})$ is an irreducible representation of a group G over an algebraically closed field \mathbb{F} , then the only matrices which commute with all matrices $\rho(g), g \in G$ are scalar matrices aI_n , where $a \in \mathbb{F}$ and I_n is the $n \times n$ identity matrix.

PROOF. Let P be an $n \times n$ matrix such that $P\rho(g) = \rho(g)P$ for all $g \in G$. Then for any $a \in F$ we have that

$$(aI_n - P) \cdot \rho(g) = \rho(g) \cdot (aI_n - P), \ \forall g \in G \ . \ (1)$$

Let $m(x) = det(xI_n - P)$ be the characteristic polynomial of P. Since m(x) is a polynomial over F and F is algebraically closed, then there exists $a_1 \in F$ such that $m(a_1) = 0_F$. Hence $det(a_1I_n - P) = 0_F$ and thus $a_1I_n - P$ is singular. Then from relation (1) above and Schur's Lemma, we obtain that $a_1I_n - P = 0$ and hence $a_1I_n = P$.

Definition 2.4.4. Let G be a group, \mathbb{F} a field and $\phi : G \longrightarrow \mathbb{F}$ be a function which is constant on conjugacy classes of G. Then ϕ is called a class function of G.

From the above definition, we observe that every character is a class function. From now on, we will consider representations and characters of a finite group G over the complex field \mathbb{C} . We shall use the notation Irr(G) to denote the set of all irreducible characters of the group G. These irreducible characters are presented in a table, called the **character table** of G. In this table, the columns correspond to the conjugacy classes of G and the rows to the irreducible characters, with entry a_{ij} being the value of the *i*-th irreducible character on an element of the *j*-th conjugacy class.

We can show that every class function ϕ of G can be uniquely expressed in the form $\phi = \sum_{\chi \in Irr(G)} b_{\chi}\chi$, where $b_{\chi} \in \mathbb{C}$. Moreover ϕ is a character if and only if all $b_{\chi} \in \mathbb{N} \cup \{0\}$ and $\phi \neq 0$. We can also show that the following properties hold:

- (i) Two representations of G have the same character if and only if they are equivalent.
- (ii) The number of irreducible characters of G is equal to the number of conjugacy classes of elements of G.
- (iii) Any character of G can be written as a sum of irreducible characters.

Definition 2.4.5. Let G be a group, χ a character of G and $Irr(G) = {\chi_1, \chi_2, ..., \chi_r}$ such that $\chi = \sum_{i=1}^r n_i \chi_i$, where $n_i \in \mathbb{N} \cup {0}$. Then those χ_i for which $n_i > 0$ are called the **irreducible** constituents of χ . In general, if ψ is a character of G such that $\chi - \psi$ is a character or is zero, then ψ is a constituent of χ .

Orthogonality relations for characters are the cornerstone of character theory. Among other applications, they allow us to express an arbitrary class function in terms of irreducible characters and to determine instantaneously whether or not any given character is irreducible. **Theorem 2.4.6.** (Generalized Orthogonality Relation) Let G be a group and $Irr(G) = \{\chi_1, \chi_2, \ldots, \chi_r\}$. Then the following holds for every $h \in G$:

$$\frac{1}{|G|} \sum_{g \in G} \chi_i(gh) \chi_j(g^{-1}) = \delta_{ij} \frac{\chi_i(h)}{\chi_i(1_G)}$$

PROOF. See Theorem 2.13 of [60].

Theorem 2.4.7. Let χ be a character of G afforded by a representation ρ of degree n. Then for $g \in G$, $\rho(g)$ is similar to a diagonal matrix $diag(\varepsilon_1, \varepsilon_2, \ldots, \varepsilon_r)$ where each ε_i is a complex root of unity. Then $\chi(g) = \sum_{i=1}^{r} \varepsilon_i$ and $\chi(g^{-1}) = \overline{\chi(g)}$, where $\overline{\chi(g)}$ is the complex conjugation of $\chi(g)$.

PROOF. This is the Lemma 2.15 in [60].

Definition 2.4.8. Let χ and ψ be class functions of a group G. Then the inner product of χ and ψ is defined by

$$\langle \chi, \psi \rangle = \frac{1}{|G|} \sum_{g \in G} \chi(g) \overline{\psi(g)}$$

The following theorems are derived from the generalized orthogonality relation and are called the first and second orthogonality relations respectively.

Theorem 2.4.9. [60](First Orthogonality Relation) Let G be a group and $Irr(G) = \{\chi_1, \chi_2, \dots, \chi_r\}$. Then

$$\frac{1}{|G|} \sum_{g \in G} \chi_i(g) \overline{\chi_j(g)} = \delta_{ij} = \langle \chi_i, \chi_j \rangle \quad .$$

PROOF. Using the generalized orthogonality relation and taking $h = 1_G$, then the result follows immediately.

Theorem 2.4.10. [60](Second Orthogonality Relation) Let G be a group and $Irr(G) = \{\chi_1, \chi_2, \ldots, \chi_r\}$ and $\{g_1, g_2, \ldots, g_r\}$ be a set of representatives of the conjugacy classes of elements of G. Then

$$\sum_{\chi \in Irr(G)} \chi(g_i) \overline{\chi(g_j)} = \delta_{ij} |C_G(g_i)| \quad .$$

PROOF. Let X be the character table of G. Then viewed as a matrix, X is an $r \times r$ matrix whose (i, j)-th entry is given by $\chi_i(g_j)$. Let C_i be the conjugacy class which contains g_i and D be the diagonal matrix with entries $\delta_{ij}|C_i|$. Then by the first orthogonality relation, we obtain that

$$|G|\delta_{ij} = \sum_{g \in G} \chi_i(g) \overline{\chi_j(g)} = \sum_{t=1}^r |C_t| \chi_i(g_t) \overline{\chi_j(g_t)}$$

Then we obtain a system of r^2 equations which can be written as a single matrix equation as follows

$$|G|I = XD\overline{X}^T$$

where I is the identity $r \times r$ matrix and \overline{X}^T is the transpose of \overline{X} . Since X is a nonsingular matrix, then we obtain that

$$|G|I = D\overline{X}^T X$$

Rewriting the above matrix system as a system of equations yields

$$|G|\delta_{ij} = \sum_{t=1}^{r} |C_i| \overline{\chi_t(g_i)} \chi_t(g_j) \quad .$$

Hence we obtain that

$$\sum_{\chi \in Irr(G)} \chi(g_j) \overline{\chi(g_i)} = |C_G(g_i)| \delta_{ij}$$

Let G be a group and χ be a character of G afforded by a representation ρ . Then we define

$$ker(\chi) = \{g \in G \mid \chi(g) = \chi(1_G)\}$$

It can be shown that $ker(\chi) = ker(\rho)$ and hence $ker(\chi)$ is a normal subgroup of G. If $Irr(G) = \{\chi_1, \chi_2, \ldots, \chi_r\}$, then every normal subgroup of G is the intersection of some of the $ker(\chi_i)$.

Theorem 2.4.11. Let G be a group and N be a normal subgroup of G. Then

- (a) If χ is a character of G and $N \subseteq ker(\chi)$, then χ is constant on the cosets of N in G and the function $\hat{\chi}$ defined on G/N by $\hat{\chi}(Ng) = \chi(g)$ is a character of G/N.
- (b) If $\hat{\chi}$ is a character of G/N, then the function χ defined by $\chi(g) = \hat{\chi}(Ng)$ is a character of G.
- (c) In both (a) and (b) above, $\chi \in Irr(G)$ if and only if $\hat{\chi} \in Irr(G/N)$.

PROOF. See Theorem 2.2.2. of [120].

If N is a normal subgroup of G and ρ is a representation of G such that $N \subseteq ker(\rho)$, then there exists a unique representation $\hat{\rho}$ of G/N defined by $\hat{\rho}(Ng) = \rho(g)$. Thus knowing ρ , we can obtain $\hat{\rho}$ and vice versa. We also obtain that ρ is irreducible if and only if $\hat{\rho}$ is irreducible. Hence ρ and $\hat{\rho}$ can be identified. If ρ affords a character χ of G, then $\hat{\rho}$ affords a character $\hat{\chi}$ of G/N and also χ and $\hat{\chi}$ can be identified. Under this identification, we obtain that

$$Irr(G/N) = \{\chi \in Irr(G) \mid N \subseteq ker(\chi)\}$$

Thus the irreducible characters of G/N are precisely those irreducible characters of G which contain N in their kernels.

Definition 2.4.12. Let G be a group, N a normal subgroup of G and $\hat{\chi}$ be a character of G/N. Then the character χ of G defined by

$$\chi(g) = \hat{\chi}(Ng)$$

is called a lifting of $\hat{\chi}$ to G.

Thus given characters of G/N, we can obtain some characters of G by the lifting process. The character $\hat{\chi}$ and its lifting χ have the same degree.

2.5. Induced Characters

In this section we look at the ways of relating the representations of a group to the representations of its subgroups.

Definition 2.5.1. Let G be a finite group and $H \leq G$. If ρ is a representation of G, then the **restriction** of ρ to H is a representation of H. This representation is denoted by ρ_H . If χ is a character of G afforded by ρ , then the restriction of χ to H is denoted by χ_H and is a character of H afforded by the representation ρ_H such that

$$\chi_H = \sum_{\psi \in Irr(H)} k_{\psi} \psi \quad ,$$

where $k_{\psi} \in \mathbb{N} \cup \{0\}$.

The characters χ_H and χ take on the same values on the elements of H. If χ_H is irreducible, then χ is irreducible in G but the converse is not true in general.

Karpilovsky in [67] proves a theorem (Theorem 23.1.4) due to Gallagher that if $H \leq G$, $\chi \in Irr(G)$ such that $\chi(g) \neq 0 \forall g \in G - H$, then χ_H is irreducible, and for any $g \in G - H$, $\chi(g)$ is a root of unity. We also observe that (see [63]) every irreducible character of H is a constituent of some irreducible character of G restricted to H.

Theorem 2.5.2. [63] Let G be a group, $H \leq G$, $\chi \in Irr(G)$ and $Irr(H) = \{\psi_1, \psi_2, ..., \psi_r\}$. Then

$$\chi_H = \sum_{i=1}^r k_i \psi_i \quad ,$$

where $k_i \in \mathbb{N} \cup \{0\}$ satisfy the following relation

$$\sum_{i=1}^r k_i^2 \le [G:H]$$

Moreover, equality in the above relation holds if and only if $\chi(g) = 0$ for all $g \in G - H$.

PROOF. See [99]

Theorem 2.5.3. Let G be a group, H be a normal subgroup of G and $\chi \in Irr(G)$. Then all the constituents of χ_H have the same degree.

PROOF. See Proposition 20.7 of [63].

Let G be a group and $H \leq G$ such that the set $\{x_1, x_2, \ldots, x_r\}$ is a transversal for H in G. Let ϕ be a representation of H of degree n. Then we define ϕ^* on G as follows:

$$\phi^*(g) = \begin{pmatrix} \phi(x_1gx_1^{-1}), \phi(x_1gx_2^{-1}), \dots, \phi(x_1gx_r^{-1}) \\ \phi(x_2gx_1^{-1}), \phi(x_2gx_2^{-1}), \dots, \phi(x_2gx_r^{-1}) \\ \vdots \\ \phi(x_ngx_1^{-1}), \phi(x_ngx_2^{-1}), \dots, \phi(x_ngx_r^{-1}) \end{pmatrix}$$

where $\phi(x_i g x_j^{-1})$ are $n \times n$ sub-matrices of $\phi^*(g)$ satisfying the property that

$$\phi(x_i g x_j^{-1}) = 0_{n \times n} \quad \forall \ x_i g x_j^{-1} \notin H$$

Then we can show that ϕ^* is a representation of G of degree nr.

Definition 2.5.4. Let G, H, ϕ and ϕ^* be as above. Then the representation ϕ^* is called the representation of G induced from the representation ϕ of H and we denote this by writing $\phi^* = \phi^G$.

If ψ is a representation of H which is equivalent to ϕ , then it can be shown that ψ^G is equivalent to ϕ^G . Thus the induction process preserves equivalence between representations.

Definition 2.5.5. Let G be a group and $H \leq G$. Let χ be a class function of H. Then we define χ^G as follows:

$$\chi^{G}(g) = \frac{1}{|H|} \sum_{x \in G} \chi^{\circ}(xgx^{-1}) ,$$

where

$$\chi^{\circ}(h) = \begin{cases} \chi(h) & \text{if } h \in H \\ 0 & \text{otherwise} \end{cases}$$

Then χ^G is a class function of G, called the **induced class function** of G induced from χ . Also we have that $deg(\chi^G) = [G:H]deg(\chi)$.

Theorem 2.5.6. [60](Frobenius Reciprocity Theorem) Let G be a group, $H \leq G$ and suppose that χ is a class function of H and ϕ is a class function of G. Then

$$\langle \chi, \phi_H \rangle = \langle \chi^G, \phi \rangle.$$

PROOF. We obtain that

$$\langle \chi^G, \phi \rangle = \frac{1}{|G|} \sum_{g \in G} \chi^G(g) \overline{\phi(g)} = \frac{1}{|G|} \frac{1}{|H|} \sum_{g \in G} \sum_{x \in G} \chi^\circ(xgx^{-1}) \overline{\phi(g)}$$

Putting $y = xgx^{-1}$ and since ϕ is a class function, then we obtain that $\phi(y) = \phi(g)$. Hence we have

$$\begin{aligned} \langle \chi^G, \phi \rangle &= \frac{1}{|G|} \frac{1}{|H|} \sum_{g \in G} \sum_{x \in G} \chi^{\circ}(xgx^{-1}) \overline{\phi(g)} = \frac{1}{|G|} \frac{1}{|H|} \sum_{y \in G} \sum_{x \in G} \chi^{\circ}(y) \overline{\phi(y)} \\ &= \frac{1}{|H|} \sum_{y \in H} \chi(y) \overline{\phi(y)} = \langle \chi, \phi_H \rangle. \end{aligned}$$

Hence the result.

Let $H \leq G$ and ϕ be a representation of H that affords a character χ of H. Then χ^G is a character of G afforded by the induced representation ϕ^G of G. The character χ^G is called the *induced character* of G. The induction and restriction processes do not necessarily preserve irreducibility of characters. For further reading on induced characters, readers are encouraged to consult [8], [9], [62], [101] and many other relevant sources.

Theorem 2.5.7. Let G be a group and $H \leq G$. Let χ be a character of H, $g \in G$ and $\{x_1, x_2, \ldots, x_m\}$ be a set of representatives of the conjugacy classes of elements of H which fuse into [g] in G. Then we obtain that

$$\chi^G(g) = |C_G(g)| \sum_{i=1}^m \frac{\chi(x_i)}{|C_H(x_i)|}$$
,

where we have that $\chi^G(g) = 0$ whenever $H \cap [g] = \emptyset$.

PROOF. We have that

$$\chi^G(g) = \frac{1}{|H|} \sum_{x \in G} \chi^\circ(xgx^{-1}) \quad .$$

If $H \cap [g] = \emptyset$, then $xgx^{-1} \notin H$ and thus $\chi^{\circ}(xgx^{-1}) = 0 \quad \forall x \in G$ and hence $\chi^{G}(g) = 0$. Now if $H \cap [g] \neq \emptyset$, then let $h \in H \cap [g]$. Then as x runs over elements of G, we have $xgx^{-1} = h$ for exactly $|C_G(g)|$ values of x. Hence we obtain that

$$\chi^{G}(g) = \frac{1}{|H|} \sum_{x \in G} \chi(xgx^{-1}) = \frac{|C_{G}(g)|}{|H|} \sum_{h \in H \cap [g]} \chi(h) = |C_{G}(g)| \sum_{i=1}^{m} \frac{\chi(x_{i})}{|C_{H}(x_{i})|}$$

Hence the result.

2.6. Permutation Characters

Knowledge of the permutation characters of a group leads to information about the subgroup structure of the group. In this section we discuss permutation characters.

We say that a group G acts on a set X if there is a homomorphism $\phi : G \longrightarrow S_X$, where S_X is the symmetric group on X. We say that G acts *faithfully* on X if ϕ is a monomorphism. In this case G can be identified with a subgroup of S_X and G becomes a *permutation group* on X. In this section we assume that X is a finite set.

Definition 2.6.1. Let G be a group acting on a set X such that for any two k-tuples $(x_1, x_2, ..., x_k)$ and $(y_1, y_2, ..., y_k)$ of k distinct elements of X, there exists $g \in G$ for which $x_i^g = y_i$ for i = 1, 2, ..., k. Then we say that G is k-transitive on X. If G is 1-transitive on X, then we say that G is *transitive*. In this case G has only one *orbit* on X. If G acts on X, we define a representation $\pi : G \longrightarrow GL(n, \mathbb{C})$, where n = |X|. Let $X = \{x_1, x_2, \ldots, x_n\}$. For each $g \in G$ we define $\pi_g = (a_{ij})$ by

$$a_{ij} = \begin{cases} 1 & \text{if } x_i^g = x_j \\ 0 & \text{otherwise} \end{cases}$$

Then π_g is a permutation matrix of the action of g. The representation π defined above is called the *permutation representation* of G obtained from the action of G on X.

Definition 2.6.2. Let G be a group and X be a set such that G acts on X. Then we denote the character afforded by the permutation representation π by $\chi(G|X)$. This character is called the **permutation character** of G associated with the action of G on X. It is not difficult to show that for $g \in G$ we have

 $\chi(G|X)(g) = |\{x \in X \mid x^g = x\}| = \text{ the number of points of } X \text{ fixed by } g.$

Suppose that G acts transitively on X and G_x is the stabilizer of $x \in X$. Then the action of G on X is the same as the action of G on the cosets of $H = G_x$. Hence $\forall g \in G, \chi(G|X)(g)$ also gives the number of cosets of $H = G_x$ that are fixed by $g \in G$ and in this case we denote this number by $\chi(G|H)(g)$. Due to the fact that the action of G on X is the same as the action of G on the cosets of H, then we can write $\chi(G|H) = \chi(G|X)$.

Theorem 2.6.3. Let G be a group acting transitively on a set X. Let $\alpha \in X$, $H = G_{\alpha}$ and $\chi(G|H)$ be the permutation character of this action. If I_H is the identity character of H, then $\chi(G|H) = (I_H)^G$.

PROOF. We have that

$$(I_H)^G(g) = \frac{1}{|H|} \sum_{x \in G, xgx^{-1} \in H} I_H(xgx^{-1}) = \frac{1}{|H|} \sum_{x \in G, xgx^{-1} \in H} 1$$

Now if $xgx^{-1} \in H$, then $xg \in Hx$. Thus Hxg = Hx and hence Hx is fixed by $g \in G$. However the summation is taken over all $x \in G$ such that $xgx^{-1} \in H$. Hence the summation is taken over all $x \in G$ for which the coset Hx is fixed by $g \in G$. But $\forall y \in Hx$, Hx = Hy and thus we obtain that

$$\sum_{x \in G, xgx^{-1} \in H} 1 = |H| |\{Hx \mid Hxg = Hx\}|$$

so that

$$(I_H)^G(g) = \frac{1}{|H|} |H| |\{Hx \mid Hxg = Hx\}| = |\{Hx \mid Hxg = Hx\}| = \chi(G|H)(g).$$

Hence the result.

Theorem 2.6.4. [60] Let G be a group acting on a set X with $\chi(G|X)$ as the permutation character of the action. If X splits into k orbits under the action of G, then

$$\langle \chi(G|X), I_G \rangle = k.$$

PROOF. Suppose that the k orbits of X under the action of G are $\{X_1, \ldots, X_k\}$. Then we obtain that

$$X = \bigcup_{i=1}^{k} X_i.$$

Let $x_i \in X_i$ and H_i be the stabilizer of $x_i \in X_i$. Also let $\chi_i(G|H_i)$ be the permutation character of G on the cosets of H_i . Then we obtain that

$$\chi(G|X) = \sum_{i=1}^{k} \chi_i(G|H_i) \text{ where } \chi_i(G|H_i) = (I_{H_i})^G.$$

By the Frobenius reciprocity theorem, we obtain that

$$\langle \chi_i(G|H_i), I_G \rangle = \langle (I_{H_i})^G, I_G \rangle = \langle I_{H_i}, I_{H_i} \rangle = 1$$
.

Hence we obtain that

$$\langle \chi(G|X), I_G \rangle = \sum_{i=1}^k \langle \chi_i(G|H_i), I_G \rangle = \sum_{i=1}^k 1 = k.$$

Hence the result.

The following result will be used in later calculations to determine the conjugacy class fusions of subgroups of G.

Corollary 2.6.5. Let $H \leq G$. Let $g \in G$ and let x_1, x_2, \ldots, x_m be representatives of the conjugacy classes of H that fuse to [g]. Then

$$\chi(G|H)(g) = \sum_{i=1}^{m} \frac{|C_G(g)|}{|C_H(x_i)|}$$

PROOF. This follows from Theorem 2.4.7.

In the following we present some properties of the permutation characters.

Theorem 2.6.6. Let G be a group, $H \leq G$ and $\chi = \chi(G|H)$.

- (i) $deg(\chi)$ divides |G|.
- (ii) $\langle \chi, \psi \rangle \leq deg(\psi)$ for all $\psi \in Irr(G)$.
- (iii) $\langle \chi, I_G \rangle = 1.$

- (iv) $\chi(g) \in \mathbb{N} \cup \{0\}$ for all $g \in G$.
- (v) $\chi(g) \leq \chi(g^m)$ for all $g \in G$ and $m \in \mathbb{N} \cup \{0\}$.
- (vi) $\chi(g) = 0$ if o(g) does not divide $|G|/deg(\chi)$.
- (vii) $\chi(g) \frac{|[g]|}{deg(\chi)}$ is an integer for all $g \in G$.

PROOF. This is Theorem 2.5.6 in [120].

Let ϕ be a representation of G and α an automorphism of G. Then ϕ^{α} is a representation of G given by

$$\phi^{\alpha}(x) = \phi(x^{\alpha})$$
 and $\phi^{\alpha}(xy) = \phi^{\alpha}(x)\phi^{\alpha}(y)$

for $x, y \in G$. If the representation ϕ affords a character χ of G, then the representation ϕ^{α} affords a character χ^{α} of G which is given by $\chi^{\alpha}(x) = \chi(x^{\alpha})$ for $x \in G$. Then the representation ϕ^{α} and the character χ^{α} are called the *algebraic conjugates* of ϕ and χ respectively induced by the automorphism α . Let $X = (\chi_i(x_j))$ be the character table of G, where $\chi_i \in Irr(G)$, $1 \leq i \leq n$ and $x_j, 1 \leq j \leq n$ are representatives of the conjugacy classes of elements of G. Then the automorphism α of G induces a permutation on the conjugacy classes of G and thus induces a permutation on the columns of X. For each $\chi_i \in Irr(G)$, we deduce that $\chi_i^{\alpha} \in Irr(G)$. Hence α induces a permutation on the irreducible characters χ_i of G and thus induces a permutation on the rows of X. Moreover since $\chi_i^{\alpha}(x_j) = \chi_i(x_j^{\alpha})$, then the matrices obtained from X by these two operations are identical. Hence we obtain the following theorem known as Brauer's Theorem.

Theorem 2.6.7. [43](Brauer's Theorem) Let N be a group and $G \leq Aut(N)$. Then the number of orbits of G as a group of permutations acting on the irreducible characters of N is the same as the number of orbits of G as a group of permutations acting on the conjugacy classes of N.

PROOF. Let X be the character table of N. Then as a matrix, X is square and nonsingular. Let α be an automorphism of N such that $\alpha \in G$. Then α induces a permutation on the conjugacy classes of N and thus induces a permutation on the columns of X. Hence G acts on the conjugacy classes of N. Since $\alpha \in G$, then to each character χ of N, we obtain a character χ^{α} of N such that $\chi^{\alpha} \in Irr(N)$ whenever $\chi \in Irr(N)$. For $y \in N$, we obtain that $\chi^{\alpha}(y) = \chi(y^{\alpha})$. Thus α induces a permutation on the rows of X. Hence G acts on the irreducible characters of N. Let X^{α} denote the image of X under α . Then we obtain that

$$P(\alpha)X = X^{\alpha} = XQ(\alpha),$$

where $P(\alpha), Q(\alpha)$ are appropriate permutation matrices which are uniquely determined by $\alpha \in G$. Suppose that $\alpha, \beta \in G$. Then we obtain that $X^{\alpha\beta} = (X^{\alpha})^{\beta}$. Also we have that

$$P(\alpha\beta)X = X^{\alpha\beta} = (X^{\alpha})^{\beta} = (P(\alpha)X)^{\beta} = P(\beta)P(\alpha)X$$

and hence $P(\alpha\beta) = P(\beta)P(\alpha)$, since X is non-singular. We also have that $X^{\alpha\beta} = XQ(\alpha\beta)$ and $(X^{\alpha})^{\beta} = (XQ(\alpha))^{\beta} = XQ(\alpha)Q(\beta)$. Since $X^{\alpha\beta} = (X^{\alpha})^{\beta}$, we obtain that $XQ(\alpha\beta) = XQ(\alpha)Q(\beta)$. Again the non-singularity of X implies that $Q(\alpha\beta) = Q(\alpha)Q(\beta)$. Define mappings π_1 and π_2 on N by $\pi_1(\alpha) = (P(\alpha))^t$ and $\pi_2(\alpha) = Q(\alpha)$, where t denotes the transpose operation on matrices. Then π_1 and π_2 are permutation representations of N. Let θ_1 and θ_2 be the permutation characters afforded by π_1 and π_2 respectively. Since $X^{-1}P(\alpha)X = Q(\alpha)$, $P(\alpha)$ and $Q(\alpha)$ are similar and thus have the same trace. Since $trace(P(\alpha))^t = trace(P(\alpha))$, we have that $trace(P(\alpha))^t = trace(Q(\alpha))$. Hence $\theta_1 = \theta_2$ and π_1 and π_2 are equivalent. Let d_1, d_2 be the number of orbits of G on the irreducible characters and on the conjugacy classes of N respectively. Thus we observe that d_1 is the number of orbits of $\pi_1(G)$ in its action as a group of permutations. Also d_2 is the number of orbits of $\pi_2(G)$ in its action as a group of permutations. Since $\theta_1 = d_2$. Hence the result.

2.7. Orbit Lengths

Brauer's theorem states that when G acts on an automorphism group N in our case an elementary abelian group, then the number of orbits of G as a group of permutations on the conjugacy classes of N is equal to the number of orbits on the irreducible characters of N. However Brauer's theorem does not apply to orbit lengths, as orbit lengths of permutations on conjugacy classes may not be the same as orbit lengths of permutation on irreducible classes. To get the orbit lengths of the irreducible character and the conjugacy classes of N, we use Programme C which we developed in GAP.

```
Programme C
```

$$\label{eq:V:=FullRowSpace} \begin{split} & \text{V:=FullRowSpace}(GF(q),n); \\ & \text{gen}[1]{:=}(\text{OneGF}(\mathbf{q}))^*[n \times n \text{ matrix group generators}]; \\ & \text{gen}[2]{:=}(\text{OneGF}(\mathbf{q}))^*[n \times n \text{ matrix group generators}]; \end{split}$$

 $gen[k]:=(OneGF(q))^{*}[n \times n \text{ matrix group generators}];$ $G:=Group(gen[1], gen[2], \cdots, gen[k]);$ O:=Orbits(G,V);; k:=OrbitLengths(G,V); l:=OrbitLengths(Group(List(G,TransposedMat)),V); Print(k); Print(l);

We use Programme C to compute the orbit lengths of both the conjugacy classes and the irreducible characters. If G is an $n \times n$ matrix group, we would wish to let G act on our elementary abelian

group N. To be able to do this we rewrite N as an n-dimensional row vector space V over GF(q). The action of G on V is multiplication of V from the right by G. This gives us the orbits of G as a group of permutations on the conjugacy classes of N. The action of G^t , the transpose of G, on V, is multiplication on the right of V by G^t . This is equivalent to the multiplication of the column vectors of V from the left by G. This multiplication gives us the orbits of G as a group of permutations on the irreducible characters of N. We give an example that shows that the orbit lengths of the conjugacy classes need not be equal to the orbit lengths of irreducible classes.

Example 2.7.1. Let $\overline{G} = 2^5: S_6$ and V = FullRowSpace(GF(2), 5) where

 $G = S_6 = < \text{matrix group with } 2 \text{ generators} > .$

Using Programme C above, we get

OrbitLengthsOfConjugacyClasses = [1,6,15,10]

OrbitLengthsOfIrreducibleCharacters = [1, 15, 15, 1].

We have four orbits for both the irreducible characters and the conjugacy classes but the orbit lengths differ.

3

Projective Representation

3.1. Prologue

In this chapter we study the projective representations and characters which will be required in the subsequent chapters. We follow very closely the work of Ali [1]. We refer to the group representations and group characters that we defined in Chapter 2 as ordinary representations and ordinary characters respectively. The Schur multiplier of a group G plays an important role in the study of projective representations of G. We have therefore devoted Section 3.2 to the study of Schur multiplier of G. In Section 3.3 we are dealing with projective representations of G. We study the relationship of projective representations with the ordinary representations. We discuss that how projective representations of G can be constructed using three different approaches. We also show that how projective representation group of G. Finally in Section 3.4 we discuss projective characters and study the orthogonality relations analogous to the ones for ordinary characters. For further readings on projective representations and projective characters readers are referred to [11, 47, 51, 55, 58, 29, 60, 95, 96, 97, 98, 100, 107, 108, 109].

3.2. Schur Multiplier

The first step in obtaining the projective representations of a group G is to compute its Schur multiplier. In this section we discuss results useful in finding the Schur multiplier of a group.

Definition 3.2.1. A function $\alpha : G \times G \longrightarrow \mathbb{C}^*$ is called a factor set of G if

$$\alpha(xy,z)\alpha(x,y) = \alpha(x,yz)\alpha(y,z) \quad for \ all \ x,y,z \in G.$$

Two factor sets α and α' are said to be *equivalent* if there exists a function $\rho : G \longrightarrow \mathbb{C}^*$ such that $\alpha'(x,y) = \frac{\rho(x)\rho(y)}{\rho(xy)}\alpha(x,y)$ for all $x, y \in G$. This is an equivalence relation and we denote the equivalence class of the factor set α by $[\alpha]$. For factor sets α and α' , let $(\alpha\alpha')(x,y) = \alpha(x,y)\alpha'(x,y)$ for all $x, y \in G$. Then $\alpha\alpha'$ is a factor set, as is α^{-1} defined by $\alpha^{-1}(x,y) = (\alpha(x,y))^{-1}$.

Definition 3.2.2. The set of all equivalence classes of factor sets forms a group by defining $[\alpha][\alpha'] = [\alpha \alpha']$. The identity of this group is [1] where 1 is the factor set 1(x, y) = 1 for all $x, y \in G$, and $[\alpha]^{-1} = [\alpha^{-1}]$. This group is called the **Schur multiplier** of G and we denote it by M(G).

Theorem 3.2.3. (i) M(G) is a finite abelian group.

(ii) If G is a cyclic group, then M(G) = 1.

PROOF. See [100].

Lemma 3.2.4. Suppose that N is a normal subgroup of a finite group G. If M(G) = 1, then $M(G/N) \cong (N \cap G')/[N,G]$. In general, $|(N \cap G')/[N,G]|$ divides |M(G/N)|.

PROOF. See [66].

Theorem 3.2.5. Let G be a finite group and H be a subgroup of index n. Then the group $(M(G))^n$ of all n-th powers of M(G) is isomorphic to a subgroup of M(H).

PROOF. See [66].

Schur [114] reduced the problem of finding M(G) to obtaining the Schur multiplier of the Sylow *p*-subgroups of *G*. The following theorem describes the Schur multiplier of *G* in terms of the subgroup structure of *G*.

Theorem 3.2.6. [114] Let S be a Sylow p-subgroup of G. Then the Sylow p-subgroup of M(G) is isomorphic to a subgroup of M(S).

PROOF. See [66].

Theorem 3.2.7. A group G has trivial Schur multiplier if and only if it has a set of subgroups with trivial Schur multipliers and relatively prime indices.

PROOF. See [66]

Schur investigated the group G = SL(2,q). He discovered that M(G) is trivial unless q = 4 or q = 9. In Chapter 10 we look at the case $5^2:2.A_5$. Noting that $2.A_5 \cong SL(2,5)$, the Schur multiplier for $2.A_5$ is trivial but for $5^2:2.A_5$ we have that M(G) is a cyclic group of order 5.

For any group G we follow the methods of Ali [1] to test if we need an ordinary representation or we need a projective representation. In chapter 10 we need a projective representation for $5^2:2.A_5$ which is a split extension. To get the Schur multiplier for this group we use MAGMA for Programme J.
```
Programme J
```

```
> G:=Group;
> M:=GModule(G);
> X:=CohomologyModule(G,M);
> E:=SplitExtension(X);
> Eperm:=DegreeReduction(CosetImage(E,sub<E|>));
> pMultiplicator(Eperm,$p_i$);
> exit;
```

However the group $5^2:2.A_5$ is a unique *perfect* group, recall a group is said to be perfect if it is equal to its derived group, that is G' = G. We use the following Programme J' in GAP.

 ${\rm Programme}~J'$

```
gap> gg:=PerfectGroup(|G|,1);
```

```
gap> AbelianInvariantsMultiplier(gg);
```

3.3. Projective Representations

The notion of projective representation, due to Schur, was suggested by the study of relations between linear representations of a group and its factor groups over a central subgroup.

Definition 3.3.1. Let G be a group and \mathbb{F} be a field. Consider the map $P: G \longrightarrow GL(n, \mathbb{F})$ such that

- (i) $P(1_G) = I_n$, where I_n is the identity $n \times n$ matrix.
- (ii) For all $x, y \in G$, there exists a map $\alpha : G \times G \longrightarrow \mathbb{F}^*$ such that

 $P(x)P(y) = \alpha(x,y)P(xy)$ where $\alpha(x,y) \in \mathbb{F}^*$.

Then P is called a **projective representation** of G over \mathbb{F} of degree n. The map α is called the **factor set** associated with P.

From the above definition, we observe that

$$\alpha(x,y) = P(x)P(y)(P(xy))^{-1}$$

Thus for the factor set α associated with P, if $\alpha(x, y) = 1_F$ for all $x, y \in G$, then we obtain that P(xy) = P(x)P(y) and hence P becomes an ordinary representation of G. Sometimes a pair (P, α) is used to indicate a projective representation P and its associated factor set α .

There is another way of looking at projective representations. The group $PGL_n(\mathbb{F}) = GL_n(\mathbb{F})/Z(GL_n(\mathbb{F}))$ is called the projective general linear group where $Z(GL_n(\mathbb{F}))$ is the center of $GL_n(\mathbb{F})$ which consists of all non-zero scalar matrices. If P is a projective \mathbb{F} -representation of G then the composition of Pwith the natural homomorphism $G \longrightarrow PGL_n(\mathbb{F})$ is a homomorphism $G \longrightarrow PGL_n(\mathbb{F})$. Conversely, if $\pi : G \longrightarrow PGL_n(\mathbb{F})$ is any homomorphism, a projective representation P of G can be defined by setting P(g) equal to any element of the coset $\pi(g)$ of $Z(GL_n(\mathbb{F}))$ in $GL_n(\mathbb{F})$. Thus the projective \mathbb{F} -representations of G can be identified with the homomorphisms of G into the projective general linear group.

We now consider the associated factor sets of the projective representations.

Lemma 3.3.2. Let α be the associated factor set of a projective representation P of G. Then α satisfies $\alpha(xy, z)\alpha(x, y) = \alpha(x, yz)\alpha(y, z)$ for all $x, y, z \in G$.

PROOF. By associativity we have

$$P(x)P(y)P(z) = \alpha(x,y)P(xy)P(z) = \alpha(x,y)\alpha(xy,z)P(xyz)$$

and

$$P(x)P(y)P(z) = \alpha(y, z)P(x)P(yz) = \alpha(y, z)\alpha(x, yz)P(xyz).$$

Now the result follows since P(xyz) is invertible.

As with ordinary representations, we now define equivalence and irreducibility of projective representations. We will consider projective representations over the complex field \mathbb{C} from now on.

Definition 3.3.3. Two projective representations P_1 and P_2 of G are equivalent if there is a non-singular matrix T such that for all $g \in G$, $P_1(g) = c(g)TP_2(g)T^{-1}$ for some $c(g) \in \mathbb{C}^*$. If c(g) = 1 for all $g \in G$ then P_1 and P_2 are linearly equivalent. A projective representation P is irreducible if it is not linearly equivalent to a projective representation of the form

$$\left(\begin{array}{cc} * & * \\ 0 & * \end{array}\right).$$

Lemma 3.3.4. If two projective representations are equivalent then they have equivalent factor sets; if they are linearly equivalent they have equal factor sets.

PROOF. Let P_1 and P_2 be equivalent projective representations with factor sets α_1 and α_2 respectively. Suppose T is a non-singular matrix and $c: G \longrightarrow \mathbb{C}^*$ such that $P_1(g) = c(g)TP_2(g)T^{-1}$ for all $g \in G$. Now for $g, h \in G$,

$$\begin{aligned} \alpha_1(g,h) &= P_1(g)P_1(h)(P_1(gh))^{-1} \\ &= c(g)TP_2(g)T^{-1}c(h)TP_2(h)T^{-1}(c(gh))^{-1}T(P_2(gh))^{-1}T^{-1} \\ &= c(g)c(h)(c(gh))^{-1}TP_2(g)P_2(h)(P_2(gh))^{-1}T^{-1} \\ &= c(g)c(h)(c(gh))^{-1}\alpha_2(g,h), \end{aligned}$$

so α_1 and α_2 are equivalent. If P_1 and P_2 are linearly equivalent, then c(g) = 1 for all $g \in G$ in the above expressions, so $\alpha_1 = \alpha_2$.

Let $F[G, \mathbb{C}]$ be the set of all functions $\lambda : G \longrightarrow \mathbb{C}$. If P is a projective representation of G with factor set α and $\lambda \in F[G, \mathbb{C}]$, then $P' = \lambda P$, where $P'(g) = \lambda(g)P(g)$ for all $g \in G$, is a projective representation of G with factor set α' , and

$$\alpha'(x,y) = \lambda(x)\lambda(y)(\lambda(xy))^{-1}\alpha(x,y)$$
(3.1)

for all $x, y \in G$.

Remark 3.3.5. It follows from (3.1) that $\alpha \sim 1$ if and only if there exists $\lambda \in F[G, \mathbb{C}]$ such that for all $x, y \in G$

$$\alpha(x,y) = \lambda(x)\lambda(y)(\lambda(xy))^{-1}.$$

The following result provides a close connection between the degrees of the irreducible projective characters with factor set α and the $o([\alpha])$.

Lemma 3.3.6. [11] Let P be a projective representation of G with factor set α and deg(P) = n. If $o([\alpha]) = m$ then m divides n.

PROOF. We know that

$$P(x)P(y) = \alpha(x, y)P(xy).$$

Taking determinant we obtain

$$\begin{split} det(P(x))det(P(y)) &= det(\alpha(x,y)P(xy)) \\ &= \alpha(x,y)^n det(P(xy)) \end{split}$$

which implies

$$\alpha(x,y)^n = det(P(x))det(P(y))(det(P(xy))^{-1})$$

By Remark 3.3.5 we obtain $[\alpha]^n = 1$. Hence *m* divides *n*.

Projective representations of a group G can be obtained by three different ways. Firstly, we may obtain the projective representations of a group G by considering a central extension of G. Now we show that how the projective representations of a group G can be constructed from the ordinary representations of a so-called representation group of G.

Definition 3.3.7. A central extension of G is a group H together with a homomorphism π of H onto G such that $ker(\pi)$ lies in the center of H.

Lemma 3.3.8. Let (H, π) be a central extension of G with $A = ker(\pi)$. Let X be a set of coset representatives for A in H, and write $X = \{x_g : g \in G\}$, where $\pi(x_g) = g$. Define $\alpha : G \times G \longrightarrow A$ by $x_g x_h = \alpha(g, h) x_{gh}$. Then α is an A-factor set of G and the equivalence class of α is independent of the choice of X.

PROOF. See Isaacs [60].

Corollary 3.3.9. Let H be a central extension of G with A, X and α as in the previous lemma. Let T be an ordinary representation of H such that the restriction T_A is the scalar representation λI for some $\lambda \in Hom(A, \mathbb{C}^*)$, that is

$$T(a) = \begin{pmatrix} \lambda(a) & & \\ & \lambda(a) & & \\ & & \ddots & \\ & & & \lambda(a) \end{pmatrix}_{n \times n} \forall a \in A,$$

where n = deg(T). Define $P(g) = T(x_g)$ for $g \in G$. Then P is a projective representation of G with factor set $\lambda(\alpha)$, where $\lambda(\alpha)(g,h) = \lambda(\alpha(g,h))$. Furthermore, P is irreducible if and only if T is and the equivalence class of P is independent of the choice of coset representatives X.

PROOF. See [60].

Remark 3.3.10. Note that if T is an ordinary irreducible representation of H then the condition that T_A be scalar representation is satisfied by the Schur's lemma (Theorem 3.3.2), since A lies in the center of H.

Definition 3.3.11. A projective representation of G that can be constructed from an ordinary representation of a central extension H of G as in Corollary 3.3.8 is said to be **lifted** to H. A **representation group** of G is a central extension H of G such that every projective representation of G can be lifted to H.

Every group has a representation group by the following result which is due to Schur [114].

Theorem 3.3.12. Let G be a finite group of order n. Then G has at least one representation group H of order mn where m = |M(G)| and the kernel of the extension is isomorphic to the Schur multiplier M(G) of G.

PROOF. See, for example, [60].

Secondly, projective representations of G can also be obtained by the generalization of Clifford's method of constructing representations of G using representations of a normal subgroup N of G.

Finally, third approach to obtain projective representations involve a natural generalization of the group algebra which plays such an important role in ordinary representation theory. Interested readers are encouraged to consult Morris [95] and other relevant sources.

The projective representations of a group are often constructed by using a combination of the above mentioned three techniques. Interested readers are referred to a series of articles by Morris [96, 97, 98] and Read [107, 108, 109].

3.4. Projective Characters

Definition 3.4.1. Let P be a projective representation of G with factor set α . Define $\xi(g) = Trace(P(g))$ for all $g \in G$. Then ξ is called a **projective character** of G. We say that ξ is **irreducible** if P is, and ξ has factor set α , where α is the factor set of P.

Definition 3.4.2. Given a factor set α of G, an element $g \in G$ is said to be α -regular if $\alpha(g, x) = \alpha(x, g)$ for all $x \in C_G(g)$.

If g is α -regular, so is every conjugate of g, and an element g is α -regular if and only if g is α' -regular for every factor set α' equivalent to α . So we can define a conjugacy class of G to be α -regular if each of its elements is α -regular.

An important feature of ordinary characters is that they are class functions. However, this no longer true for projective characters. For projective characters we have

Proposition 3.4.3. Let ξ be the projective character of G with factor set α . If for any α -regular element x in G and for any y in G, $\alpha(x, y) = \alpha(y, y^{-1}xy)$ then ξ is a class function.

PROOF. This is Proposition 2.2(iii) in [66].

Theorem 3.4.4. Two projective representations P_1 and P_2 with factor set α are linearly equivalent if and only if they have the same projective character.

PROOF. See Theorem 4.4 in [95].

The projective characters of G can be determined from the ordinary characters of a representation group (H, π) of G. Let $\pi : H \longrightarrow G$ be defined by the extension H of G, and let $\{x_g : g \in G\}$ be a set of coset representatives for $ker(\pi)$ in H. If P is a projective representation of G with projective character ξ then there is an ordinary representation T of H such that $P(g) = T(x_g)$ for $g \in G$. Let χ be the character of H afforded by T, then $\xi(g) = \chi(x_g)$ for all $g \in G$.

Projective characters also satisfy the usual orthogonality relations. We have analogues to ordinary characters.

- **Theorem 3.4.5.** (i) The number of irreducible projective characters of G with factor set α is equal to the number of α -regular conjugacy classes of G.
 - (ii) Let $\xi_1, \xi_2, \ldots, \xi_t$ be the projective characters of G with factor set α , and let C_1, C_2, \ldots, C_t be the α -regular conjugacy classes of G with g_i a representative of C_i for $i = 1, 2, \ldots, t$. Then

$$\sum_{i=1}^{t} \xi_i(g_j) \overline{\xi_i(g_k)} = \delta_{jk} |C_G(g_j)| \quad for \ j,k \in \{1,2,\ldots t\}$$

(iii) An element g of G is α -regular if and only if there is an irreducible projective character ξ of G with factor set α such that $\xi(g) \neq 0$.

PROOF. See [47].

Let G^0 be the set of all α -regular elements of the group G. Then we have the following.

Theorem 3.4.6. Let Let $\xi_1, \xi_2, \ldots, \xi_t$ be the projective characters of G with factor set α , and let C_1, C_2, \ldots, C_t be the α -regular conjugacy classes of G with g_i a representative of C_i for $i = 1, 2, \ldots, t$. Then

$$\sum_{g \in G^0} \xi_i(g) \overline{(\xi_j(g))} = |G| \delta_{ij}.$$

PROOF. See [66]

Haggarty and Humphreys [47] showed that it is possible to determine the projective characters of G with a given factor set without the full representation group G. Suppose α is a factor set of G, with $[\alpha]$ having order e in the Schur multiplier M(G). Let ω be an e^{th} root of unity and let α' be a representative of $[\alpha]$ whose values are powers of ω . For $g, h \in G$ define $\alpha'(g, h)$ by $\alpha'(g, h) = \omega^{a(g,h)}$. Let L be the group generated by an element x of order e and elements x_g ($g \in G$) with multiplication $x^i x_g x^j x_h = x^{i+j} x^{a(g,h)} x_{gh}$. Then L is a quotient of the representation group Hand any projective representation of G with factor set α can be lifted to an ordinary representation of L. Thus the projective characters of G with factor set α can be determined from the ordinary character table of L.

Clifford Theory

4.1. Prologue

An important method for constructing irreducible representations of groups consists in the application of three basic operations:

- (i) Restriction to a subgroup,
- (ii) Extension from a subgroup,
- (iii) Induction from a subgroup.

The theory attains particular richness when the underlying subgroup is a normal subgroup of its extension. This is the content of the Clifford theory, originally developed by *Clifford* in 1937 [20] for ordinary representations and extended by *Mackey* in 1958 [79] to projective representations. In this chapter, we study the Clifford theory and its related consequences which are required to describe the Fischer-Clifford matrices in the next chapter. In Section 4.2, we study the relation between the characters of a group \bar{G} and its normal subgroup N. We will give various sufficient conditions for the extendibility of an irreducible character θ of N to \bar{G} . In Section 4.3, we engage the Clifford theory of projective representations. We will study, how it is always possible to extend an irreducible character of a normal subgroup N to a projective character of its inertia group \bar{H} . Finally in Section 4.4, we will study the problem which asserts that if χ is a \bar{G} -invariant irreducible character of a normal subgroup N of a finite group \bar{G} , then the number of distinct irreducible constituents of $\chi^{\bar{G}}$ is equal to the number of χ -regular conjugacy classes of \bar{G}/N . We also show that the number of irreducible constituents of $\chi^{\bar{G}}$ is equal to a character of each subgroup N < x, y > of \bar{G} with $[x, y] \in N$.

4.2. Clifford Theory and Normal Subgroups

In this section we study the important connection between characters of group \bar{G} and a normal subgroup N of \bar{G} .

Definition 4.2.1. Let \overline{G} be a group, $N \leq \overline{G}$ and θ be a character of N. Then for $\overline{g} \in \overline{G}$, we define $\theta^{\overline{g}} : \overline{g}^{-1}N\overline{g} \longrightarrow \mathbb{C}$ by $\theta^{\overline{g}}(t) = \theta(\overline{g}t\overline{g}^{-1})$ for all $t \in \overline{g}^{-1}N\overline{g}$. Then $\theta^{\overline{g}}$ is said to be a \overline{G} -conjugate of θ . If N is a normal subgroup of \overline{G} and $\theta^{\overline{g}} = \theta$ for all $\overline{g} \in \overline{G}$, then θ is said to be \overline{G} -invariant.

Theorem 4.2.2. Let G be a group, $K, H \leq G$ such that $K \leq H \leq G$ and χ be a character of K. Then for all $g \in G$ we have

(i) $(\chi^H)^g = (\chi^g)^{g^{-1}Hg}$

(*ii*)
$$(\chi^g)^G = \chi^G$$
.

PROOF. See [67]

Remark 4.2.3. If $N \leq \overline{G}$ and $\overline{g} \in \overline{G}$, then $\theta^{\overline{g}}$ is a character of $\overline{g}^{-1}N\overline{g}$. However if N is normal in \overline{G} , $\theta^{\overline{g}}$ becomes a character of N.

Let \overline{G} be a group, N a normal subgroup of \overline{G} and $\theta \in Irr(N)$ then we define

$$Irr(\bar{G},\theta) = \{\chi \mid \chi \in Irr(\bar{G}), \langle \chi_N, \theta \rangle > 0\}$$
.

Observe that $\langle \chi_N, \theta \rangle_N = \langle \chi, \theta^{\bar{G}} \rangle_{\bar{G}}.$

Definition 4.2.4. Let \overline{G} be a group, N a normal subgroup of \overline{G} and $\theta \in Irr(N)$. Then

$$I_{\bar{G}}(\theta) = \{ \bar{g} \in \bar{G} \mid \theta^{\bar{g}} = \theta \}$$

is the inertia group of θ in \overline{G} .

Since $I_G(\theta)$ is the stabilizer of θ in the action of \overline{G} on Irr(N), it follows that it is a subgroup and that $I_G(\theta) \supseteq N$.

Lemma 4.2.5. [58] Let \overline{G} be a group, N a normal subgroup of \overline{G} and $\theta \in Irr(N)$. Then

(a) For $\bar{g_1}, \bar{g_2} \in \bar{G}$ we have $\theta^{\bar{g_1}\bar{g_2}} = (\theta^{\bar{g_1}})^{\bar{g_2}}$. In particular $N \leq I_{\bar{G}}(\theta) \leq \bar{G}$. If

$$\bar{G} = \bigcup_{i=1}^{m} I_{\bar{G}}(\theta) \bar{g}_i$$

with $[\bar{G} : I_{\bar{G}}(\theta)] = m$, then $\{\theta^{\bar{g}} | \bar{g} \in \bar{G}\} = \{\theta^{\bar{g_1}}, \theta^{\bar{g_2}}, \dots, \theta^{\bar{g_m}}\}$, and $\theta^{\bar{g_1}}, \theta^{\bar{g_2}}, \dots, \theta^{\bar{g_m}}$ are pairwise distinct.

- (b) If ψ_1 , ψ_2 are any characters of N and $\bar{g} \in \bar{G}$, then $\langle \psi_1^{\bar{g}}, \psi_2^{\bar{g}} \rangle_N = \langle \psi_1, \psi_2 \rangle_N$. In particular $\theta^{\bar{g}} \in Irr(N)$ if $\theta \in Irr(N)$.
- (c) If ψ is a character of \overline{G} and θ of N, then $\langle \psi_N, \theta \rangle_N = \langle \psi_N, \theta^{\overline{g}} \rangle_N$ for all $\overline{g} \in \overline{G}$.

PROOF. See [58].

We now state a fundamental theorem, which is due to Clifford [20] but we give a proof from Huppert [58].

Theorem 4.2.6. [58] (Clifford Theorem) Suppose \overline{G} is a group, N a normal subgroup of \overline{G} , $\theta \in Irr(N)$ and $\chi \in Irr(\overline{G}, \theta)$. Let $\langle \chi_N, \theta \rangle = e > 0$. Assume also that

$$\bar{G} = \bigcup_{i=1}^{m} I_{\bar{G}}(\theta) \bar{g}_i \quad and \quad m = [\bar{G} : I_{\bar{G}}(\theta)].$$

Then we have

(a)
$$(\theta^{\bar{G}})_N = |I_{\bar{G}}(\theta)/N| \sum_{i=1}^m \theta^{\bar{g}_i}.$$

- (b) $\langle \theta^{\bar{G}}, \theta^{\bar{G}} \rangle_{\bar{G}} = |I_{\bar{G}}(\theta)/N|$. In particular $\theta^{\bar{G}} \in Irr(\bar{G})$ if and only if $I_{\bar{G}}(\theta) = N$.
- (c) $\chi_N = e \sum_{i=1}^m \theta^{\bar{g}_i}$. In particular,

$$\chi(1) = em\theta(1)$$
 and $\langle \chi_N, \chi_N \rangle_N = e^2 m.$

Also

$$e^2 \leq |I_{\bar{G}}(\theta)/N|$$
 and $e^2m \leq |\bar{G}/N|.$

PROOF. (a) For $x \in N$ we have by the previous lemma

$$\theta^{\bar{G}}(x) = \frac{1}{|N|} \sum_{\bar{g} \in \bar{G}} \theta(x^{\bar{g}^{-1}}) = \frac{1}{|N|} \sum_{\bar{g} \in \bar{G}} \theta^{\bar{g}}(x) = \frac{|I_{\bar{G}}(\theta)|}{|N|} \sum_{i=1}^{m} \theta^{\bar{g}_i}(x).$$

(b) By Frobenius reciprocity (Theorem 2.5.6) and part (a) we obtain

$$\langle \theta^{\bar{G}}, \theta^{\bar{G}} \rangle_{\bar{G}} = \langle (\theta^{\bar{G}})_N, \theta \rangle_N = |I_{\bar{G}}(\theta)/N|.$$

(c) For all $\bar{g} \in \bar{G}$, we have by Lemma 4.2.4(c)

$$\langle \chi_N, \theta^{\bar{g}} \rangle_N = \langle \chi_N, \theta \rangle_N = e.$$

Hence

$$\chi_N = e \sum_{i=1}^m \theta^{\bar{g}_i} + \psi,$$

where ψ is a character of N or zero. As

$$\langle \theta^G, \chi \rangle_{\bar{G}} = \langle \theta, \chi_N \rangle_N = e,$$

we obtain

$$\theta^G = e\chi + \dots$$

Restriction to N shows by part (a)

$$e\chi_N + \ldots = (\theta^{\bar{G}})_N = |I_{\bar{G}}(\theta)/N| \sum_{i=1}^m \theta^{\bar{g}_i}.$$

Hence in χ_N there does not appear any irreducible character of N different from the $\theta^{\bar{g}_j}$. Therefore

$$\chi_N = e \sum_{i=1}^m \theta^{\bar{g}_i},$$

which implies immediately

$$\chi(1) = em\theta(1)$$
 and $\langle \chi_N, \chi_N \rangle_N = e^2 m.$

By part (b) we have

$$|I_{\bar{G}}(\theta)/N| = \langle \theta^{\bar{G}}, \theta^{\bar{G}} \rangle_{\bar{G}} = \langle e\chi + \dots, e\chi + \dots \rangle_{\bar{G}} \ge e^2$$

and hence

$$|\bar{G}/N| = |\bar{G}: I_{\bar{G}}(\theta)| |I_{\bar{G}}(\theta)/N| \ge me^2.$$

Remark 4.2.7. It can be shown that the number e in the above theorem is the degree of an irreducible projective representation of \bar{G}/N , hence it divides $|\bar{G}/N|$. See Huppert [58].

As a consequence of Clifford theorem we have the following result, which is of fundamental importance in the character theory of normal subgroups.

Theorem 4.2.8. [60] Let \overline{G} be a group, N a normal subgroup of \overline{G} , $\theta \in Irr(N)$ and $T = I_{\overline{G}}(\theta)$. Let

$$A = \{ \psi \in Irr(T) \mid \langle \psi_N, \theta \rangle \neq 0 \},\$$
$$B = \{ \chi \in Irr(\bar{G}) \mid \langle \chi_N, \theta \rangle \neq 0 \}.$$

Then

(a) If ψ ∈ A, then ψ^Ḡ ∈ Irr(Ḡ).
(b) If ψ^Ḡ = χ and ψ ∈ A, then ⟨ψ_N, θ⟩ = ⟨χ_N, θ⟩.

- (c) If $\psi^{\bar{G}} = \chi$ and $\psi \in A$, then ψ is the unique irreducible constituent of χ_T which sits in A.
- (d) The map $\psi \longmapsto \psi^{\bar{G}}$ is a bijection of A to B.

PROOF. See Isaacs [60].

Remark 4.2.9. From the previous theorem we deduce that induction to \overline{G} maps the irreducible characters of T that contain θ in their restriction to N faithfully onto the irreducible characters of \overline{G} that contain θ in their restriction to N.

An important task of the Clifford theory is to examine when irreducible characters of normal subgroups are extendible to their respective inertia groups.

Definition 4.2.10. Let \overline{G} be a group, H a subgroup of \overline{G} , $\theta \in Irr(H)$ and $\chi \in Irr(\overline{G})$ such that $\chi_H = \theta$. Then θ is said to be **extendible** to an irreducible character of \overline{G} .

If θ is extendible to an irreducible character of \overline{G} , we will simply say that θ is extendible to \overline{G} . There are various conditions which have to be satisfied in order that θ can be extended to \overline{G} . Readers can also consult [11, 38, 39, 60, 65, 58] for further reading and information on extendibility of characters.

Theorem 4.2.11. [67] Let N a normal subgroup of \overline{G} , $\chi \in Irr(N)$, where χ is \overline{G} -invariant and let Γ be a matrix representation of N which affords χ . Then

- (i) there exists a projective representation ρ of \bar{G} such that $\Gamma(n) = \rho(n)$ and $(\rho(\bar{g}))^{o(\bar{g})} = I$, for all $n \in N, \bar{g} \in \bar{G}$ where I is the identity matrix,
- (ii) if $\bar{G} = NH$ for some $H \leq \bar{G}$ and if ρ_H is an ordinary representation of H, then χ can be extended to \bar{G} .

PROOF. (i) Let $\bar{g} \in \bar{G}$. Since χ is \bar{G} -invariant, then the representations Γ and $\Gamma^{\bar{g}}$ of N are equivalent. Hence there is an invertible matrix $\theta(\bar{g})$ such that $(\theta(\bar{g}))^{-1}\Gamma(n)\theta(\bar{g}) = \Gamma^{\bar{g}}(n)$, for all $n \in N$. We may assume that $\theta(n) = \Gamma(n)$ for all $n \in N$. We have that $\theta : \bar{G} \longrightarrow GL(k, \mathbb{F})$, where $k = deg(\Gamma)$, and that $\theta_N = \Gamma$. Now let $\bar{g}_1, \bar{g}_2 \in \bar{G}$. Then we obtain that

$$(\theta(\bar{g}_1\bar{g}_2))^{-1}\Gamma(n)\theta(\bar{g}_1\bar{g}_2) = \Gamma^{\bar{g}_1\bar{g}_2}(n) = (\Gamma^{\bar{g}_1})^{\bar{g}_2}(n) = (\theta(\bar{g}_2))^{-1}\Gamma^{\bar{g}_1}(n)\theta(\bar{g}_2) = (\theta(\bar{g}_2))^{-1}(\theta(\bar{g}_1))^{-1}\Gamma(n)\theta(\bar{g}_1)\theta(\bar{g}_2).$$

So that

$$\theta(\bar{g}_1)\theta(\bar{g}_2)(\theta(\bar{g}_1\bar{g}_2))^{-1}\Gamma(n) = \Gamma(n)\theta(\bar{g}_1)\theta(\bar{g}_2)(\theta(\bar{g}_1\bar{g}_2))^{-1}$$

Thus for all $n \in N$, $\theta(\bar{g}_1)\theta(\bar{g}_2)(\theta(\bar{g}_1\bar{g}_2))^{-1}$ commutes with $\Gamma(n)$ and thus by the Corollary 2.3.3, we can define a function $\alpha : \bar{G} \times \bar{G} \longrightarrow \mathbb{C}^*$ such that $\theta(\bar{g}_1)\theta(\bar{g}_2) = \alpha(\bar{g}_1, \bar{g}_2)\theta(\bar{g}_1\bar{g}_2)$. Since Γ

is a representation of N, then we obtain that $\theta(1_N) = \Gamma(1_N) = I$. Hence θ is a projective representation of \overline{G} with associated factor set α . Let $o(\overline{g}) = m$ and if $\overline{g} \in N$, then we obtain that $(\theta(\overline{g}))^m = I$. However if $\overline{g} \in \overline{G} - N$, then since $\theta(\overline{g}^m) = \theta(1_{\overline{G}}) = I$, there exists $\lambda(\overline{g}) \in \mathbb{C}^*$ such that $(\theta(\overline{g}))^m = \lambda(\overline{g})I$. Now let $\mu(\overline{g}) \in \mathbb{C}^*$ such that $(\mu(\overline{g}))^m = (\lambda(\overline{g}))^{-1}$ and let $\mu(n) = 1$ for all $n \in N$. Then the projective representation ρ of \overline{G} given by $\rho(\overline{g}) = \mu(\overline{g})\theta(\overline{g})$ is such that $\rho(n) = \mu(n)\theta(n) = \theta(n) = \Gamma(n)$ for all $n \in N$. Also we have that

$$(\rho(\bar{g}))^m = (\mu(\bar{g})\theta(\bar{g}))^m = (\mu(\bar{g}))^m (\theta(\bar{g}))^m = (\lambda(\bar{g}))^{-1}\lambda(\bar{g})I = I$$

Hence property (i) is established.

(ii) Let T be a transversal for $N \cap H$ in H containing 1_H . Then every $\bar{g} \in \bar{G}$ has a unique expression of the form $\bar{g} = tn$, where $t \in T, n \in N$. Now let $\bar{g_1} \in \bar{G}, \ \bar{g_1} \neq \bar{g}$ be given by $\bar{g_1} = t_1n_1$, where $t_1 \in T, n_1 \in N$. Since $t, t_1 \in T$, then $t, t_1 \in H$ and hence $tt_1 \in H$. Now let $tt_1 = t_2n_2$, where $t_2 \in T$ and $n_2 \in N \cap H$. Define ψ on \bar{G} by $\psi(\bar{g}) = \rho(t)\rho(n)$. Since $n_2t_1^{-1}nt_1n_1 \in N$, we obtain that

$$\psi(\bar{g}\bar{g}_1) = \psi(tnt_1n_1) = \psi(tt_1t_1^{-1}nt_1n_1) = \psi(t_2n_2t_1^{-1}nt_1n_1) = \rho(t_2)\rho(n_2t_1^{-1}nt_1n_1)$$

Also we have

$$\psi(\bar{g})\psi(\bar{g}_1) = \rho(t)\rho(n)\rho(t_1)\rho(n_1) = \rho(t)\rho(t_1)(\rho(t_1))^{-1}\rho(n)\rho(t_1)\rho(n_1)$$

= $\rho(t)\rho(t_1)[(\rho(t_1))^{-1}\rho(n)\rho(t_1)]\rho(n_1).$

However from the proof of part(i) above we have that $(\rho(\bar{g}))^{-1}\Gamma(n)\rho(\bar{g}) = \Gamma^{\bar{g}}(n)$ and $\rho(n) = \Gamma(n)$ for all $n \in N, \bar{g} \in \bar{G}$. Since $t_1^{-1}nt_1 \in N$, then we obtain that

$$\rho(t_1^{-1}nt_1) = \Gamma(t_1^{-1}nt_1) = \Gamma^{t_1}(n) = (\rho(t_1))^{-1}\Gamma(n)\rho(t_1) = (\rho(t_1))^{-1}\rho(n)\rho(t_1).$$

Since by the assumption ρ is an ordinary representation on H we have $\rho(tt_1) = \rho(t)\rho(t_1)$ since $tt_1 \in H$. We deduce that

$$\psi(\bar{g})\psi(\bar{g}_1) = \rho(t)\rho(t_1)\rho(t_1^{-1}nt_1)\rho(n_1) = \rho(tt_1)\rho(t_1^{-1}nt_1)\rho(n_1)$$

= $\rho(t_2n_2)\rho(t_1^{-1}nt_1)\rho(n_1) = \rho(t_2)\rho(n_2t_1^{-1}nt_1n_1).$

Hence we obtain that $\psi(\bar{g}\bar{g}_1) = \psi(\bar{g})\psi(\bar{g}_1)$. Therefore ψ is an ordinary representation of \bar{G} . However $\forall n \in N$, we obtain that $\psi(n) = \rho(n) = \Gamma(n)$ and thus the character afforded by the representation ψ of \bar{G} , extends χ to \bar{G} . Hence the result.

Theorem 4.2.12. [67] Let $\overline{G} = NG$ where N is a normal subgroup of \overline{G} , and $G \leq \overline{G}$ such that $N \cap G \subseteq N'$. If θ is an irreducible \overline{G} -invariant character of N such that $(deg(\theta), |G|) = 1$, then θ can be extended to \overline{G} .

PROOF. For a detailed proof which uses the previous theorem, see Corollary 27.1.2 of [67]

Theorem 4.2.13. [24, 120, 99] (Mackey's Theorem) Let N be a normal subgroup of \overline{G} and θ be a \overline{G} -invariant irreducible character of N. If N is abelian and \overline{G} splits over N, then θ can be extended to \overline{G} .

PROOF. Let $\bar{G} = N:G$. Since \bar{G} is a semidirect product of N by G, then any $x \in \bar{G}$ can be expressed uniquely as x = ng, where $n \in N, g \in G$. Define χ on \bar{G} by $\chi(ng) = \theta(n)$. Since N is abelian, θ has degree 1 and thus is linear. The invariance of θ in \bar{G} implies that $\theta(n) = \theta(xnx^{-1})$ for all $x \in \bar{G}$. Now let $x_1 = n_1g_1, x_2 = n_2g_2$ be elements of \bar{G} . Then we obtain that

$$\begin{aligned} \chi(x_1x_2) &= \chi(n_1g_1n_2g_2) = \chi(n_1n_2^{g_1}g_1g_2) = \theta(n_1n_2^{g_1}) \\ &= \theta(n_1)\theta(n_2^{g_1}) = \theta(n_1)\theta(n_2) = \chi(x_1)\chi(x_2). \end{aligned}$$

Therefore χ is a linear character of \overline{G} such that $\chi_N = \theta$.

Remark 4.2.14. Mackey's theorem has been proved differently in Ali [1] and Mpono [99] by applying Theorem 4.2.11.

Theorem 4.2.15. Let N be a normal subgroup of a finite group \overline{G} and θ be an irreducible character of N which is invariant in \overline{G} , then θ is extendible to a character of \overline{G} if $([\overline{G}:N], \frac{|N|}{deq(\theta)}) = 1$.

PROOF. See [38].

Theorem 4.2.16. Suppose \overline{G} is a splitting extension of a normal subgroup N, then any linear character $\theta \in Irr(N)$ can be extended to its inertia group $I_{\overline{G}}(\theta)$.

PROOF. See [1, 99].

Note that Mackey's theorem is reinforced by the Theorem 4.2.15 since for N abelian, all its irreducible characters are linear and hence are extendible to their respective inertia groups.

Theorem 4.2.17. [39, 59, 120](**Gallagher's Theorem**) Let N a normal subgroup of \overline{G} , $\theta \in Irr(N)$ and $\overline{H} = I_{\overline{G}}(\theta)$. If θ can be extended to $\psi \in Irr(\overline{H})$ then as β ranges over all the irreducible characters of \overline{H} which contain N in their kernels, $\beta\psi$ ranges over all the irreducible characters of \overline{H} which contain θ in their restriction to N.

PROOF. Since $\bar{H} = I_{\bar{G}}(\theta)$, then θ is self-conjugate in \bar{H} and thus by Clifford's theorem we obtain that $(\theta^{\bar{H}})_N = f\theta$ for some positive integer f. Comparing degrees we have $(\theta^{\bar{H}})_N = [\bar{H} : N]\theta$ and so $\langle \theta^{\bar{H}}, \theta^{\bar{H}} \rangle = \langle \theta, (\theta^{\bar{H}})_N \rangle = [\bar{H} : N]$. Now we claim that $\theta^{\bar{H}} = \sum_{\beta} \beta(1_{\bar{G}})\beta\psi$, where β ranges over all the irreducible characters of \bar{H} that contain N in their kernels. Both $\theta^{\bar{H}}$ and $\sum_{\beta} \beta(1_{\bar{G}})\beta\psi$ are zero off N since for $g \notin N, xgx^{-1} \notin N$ for all $x \in \bar{G}$ and thus $\theta^{\bar{H}}(g) = 0$. Also for $g \notin N$, by the orthogonality of the columns of the character table of \bar{H}/N we have that $\sum_{\beta} \beta(1_{\bar{G}})(\beta\psi)(g) =$ $[\sum_{\beta} \beta(1_{\bar{G}})\beta(g)]\psi(g) = 0$. We also have that $(\theta^{\bar{H}})_N = [\bar{H} : N]\theta = (\sum_{\beta} \beta(1_{\bar{G}})\beta\psi)_N$ since for $g \in N$, $\sum_{\beta} \beta(1_{\bar{G}})\beta(g)\psi(g) = \sum_{\beta} (\beta(1_{\bar{G}}))^2\psi(g) = [\bar{H} : N]\psi(g) = [\bar{H} : N]\theta(g).$ Hence we obtain that $\theta^{\bar{H}} = \sum_{\beta} \beta(1_{\bar{G}})\beta\psi.$ So we have

$$[\bar{H}:N] = \langle \theta^{\bar{H}}, \theta^{\bar{H}} \rangle = \langle \sum_{\beta} \beta(1_{\bar{G}}) \beta \psi, \sum_{\tau} \tau(1_{\bar{G}}) \tau \psi \rangle = \sum_{\beta,\tau} \beta(1_{\bar{G}}) \tau(1_{\bar{G}}) \langle \beta \psi, \tau \psi \rangle$$

The diagonal terms contribute at least $\sum (\beta(1_{\bar{G}}))^2 = [\bar{H} : N]$, so the $\beta \psi$ are irreducible and distinct, and are all the irreducible constituents of $\theta^{\bar{H}}$ and so are all the irreducible characters of \bar{H} that contain θ in their restriction to N. For $\phi \in Irr(\bar{H})$ such that $\langle \phi_N, \theta \rangle \neq 0$, we obtain that $\langle \phi_N, \theta \rangle = \langle \phi, \theta^{\bar{H}} \rangle$ which implies that ϕ is an irreducible constituent of $\theta^{\bar{H}}$ and hence is of the form $\beta \psi$.

Remark 4.2.18. Let \overline{G} be an extension of N by G. If every irreducible character of N can be extended to its inertia group in \overline{G} , then by application of Theorem 4.2.7 and Remark 4.2.8, the characters of \overline{G} can be obtained as follows:

Let $\theta_1, \theta_2, \ldots, \theta_t$ be representatives of the orbits of \overline{G} on Irr(N). For each i, let $\overline{H}_i = I_{\overline{G}}(\theta_i)$ and let $\psi_i \in Irr(\overline{H}_i)$ with $(\psi_i)_N = \theta_i$. Now each irreducible character of \overline{G} contains some θ_i in its restriction to N by Clifford's theorem. So by Theorem 4.2.7 and Remark 4.2.8 we have

$$Irr(\bar{G}) = \bigcup_{i=1}^{t} \{ (\beta\psi_i)^{\bar{G}} : \beta \in Irr(\bar{H}_i), N \subset ker(\beta) \}.$$

Hence the characters of \overline{G} fall into t blocks, with each block corresponding to an inertia group.

Finally in this section, we give a result due to *Isaacs* about the value of an extension χ of θ to G. For $N \leq G$, $\theta \in Irr(N)$ has an extension χ to G if $I_G(\theta) = G$. We prove that the values of χ are equally distributed over the cosets of N.

Theorem 4.2.19. Suppose $N \leq G$, $\chi \in Irr(G)$ with $\chi_N \in Irr(N)$. Then

$$\frac{1}{|N|} \sum_{y \in Ng} |\chi(y)|^2 = 1$$

for all $g \in G$.

PROOF. See Theorem 21.5 of [58].

4.3. Clifford Theory and Projective Representations

The projective representations of a group are closely related to Clifford theory. In this section we study the Clifford theory for projective representations.

Definition 4.3.1. Let $N \leq \overline{G}$. If Y is an irreducible (ordinary) representation of N then for $\overline{g} \in \overline{G}$, $Y^{\overline{g}}$ defined by $Y^{\overline{g}}(n) = Y(\overline{g}n\overline{g}^{-1}), n \in N$, is a representation of N, called a **conjugate** of Y. The **inertia group** of Y, T(Y), is the set of all $\overline{g} \in \overline{G}$ such that Y is equivalent to $Y^{\overline{g}}$. Note that $T(Y) = I_{\overline{G}}(\theta)$ where θ is the character of N afforded by Y.

Now let Y be an irreducible representation of N, where $N \leq \overline{G}$ and let $\overline{H} = T(Y)$, so Y is equivalent to all its conjugates in \overline{H} . The following theorem shows that Y can always be extended to a projective representation of \overline{H} and gives a necessary and sufficient condition for Y to be extendible to an ordinary representation of \overline{H} .

Theorem 4.3.2. Let $N \leq \overline{G}$, Y an irreducible representation of N and \overline{H} be as above. Then Y extends to a projective representation X of \overline{H} with factor set $\overline{\alpha}$ such that $\overline{\alpha}$ is constant on cosets of N in \overline{H} . Therefore $\overline{\alpha}$ can be regarded as a factor set α of $H = \overline{H}/N$ defined by $\alpha(Nh, Nk) = \overline{\alpha}(h, k)$. Also, α satisfies $\alpha^{d|N|} \sim 1$ where d is the degree of Y. Furthermore, Y extends to a linear representation of \overline{H} if and only if $\alpha \sim 1$. In particular, if $H^2(\overline{G}, \mathbb{C}^*) = 1$, then Y always extends to a linear representation of G.

PROOF. See Nagao and Tsushima [100].

Theorem 4.3.3. Let $N \leq \overline{G}$, Y be an irreducible representation of N with $\overline{H} = T(Y)$ and $H = \overline{H}/N$. Extend Y to a projective representation X of \overline{H} as in Theorem 4.3.2 with factor set $\overline{\alpha}$. Then

- 1. If W is an irreducible representation of H that has Y as one of its irreducible constituents in its restriction to N then there exists an irreducible projective representation Z of H with factor set α^{-1} such that W is equivalent to the representation $\overline{Z} \otimes X$ of \overline{H} , where α is the factor set of H obtained from $\overline{\alpha}$, and \overline{Z} is the representation of \overline{H} obtained naturally from Z.
- 2. If, conversely, Z is any irreducible projective representation of H with factor set α^{-1} , then $\overline{Z} \otimes X$ is an irreducible representation of \overline{H} which is equivalent to some representation that contains Y in its restriction to N.

PROOF. See Nagao and Tsushima [100].

Theorem 4.3.4. [113] Let $N \triangleleft \overline{H}$, $\varphi \in Irr(N)$ be invariant under \overline{H} and let $\overline{\varphi}$ be a projective extension of φ to \overline{H} with factor set α . Then

 $Irr(\bar{H},\varphi) = \{\bar{\varphi}\psi \mid \psi \text{ is an irreducible } \alpha^{-1} - \text{projective character of } \bar{H}/N\}.$

In particular, the number of irreducible α^{-1} - projective characters of \bar{H}/N is equal to the number of α - regular classes of \bar{H} .

PROOF. See [113].

Now we restate the results from Theorems 4.3.2 and 4.3.3 in the form in which we will be using them, in terms of projective and ordinary characters.

Corollary 4.3.5. [82] Let $\overline{G} = N \cdot G$, where $N \leq \overline{G}$ and $\overline{G}/N \cong G$. Let $\theta \in Irr(N)$ and $\overline{H} = I_{\overline{G}}(\theta)$.

(i) There exists a projective character φ of \overline{H} with factor set $\overline{\alpha}$ such that $\varphi_N = \theta$ and $\overline{\alpha}$ is constant on cosets of N, so $\overline{\alpha}$ can be regarded as a factor set α of $H = \overline{H}/N$.

- (ii) If $\theta(1_N) = d$, then $\alpha^{d|N|} \sim 1$.
- (iii) If η runs over all the irreducible projective characters of H with factor set α^{-1} , then $\varphi \bar{\eta}$ runs over all irreducible characters of \bar{H} that contains θ in their restrictions to N where $\bar{\eta}$ is the projective character of \bar{H} obtained naturally from η .

PROOF. See [82].

Remark 4.3.6. In the above theorem, if θ extends to an ordinary character of \overline{H} , then we show that $\alpha \sim 1$. In this case η 's are the ordinary irreducible characters of H. Hence Theorem 4.2.16 is a special case of the above corollary.

Remark 4.3.7. Now by Remark 4.2.8 and Corollary 4.3.5, the characters of $\overline{G} = N \cdot G$ can be obtained as follows:

Let $\theta_1, \theta_2, \ldots, \theta_t$ be the representatives of the orbits of \overline{G} on the set Irr(N). Let $\overline{H}_i = I_{\overline{G}}(\theta_i), \varphi_i$ be a projective character of \overline{H}_i with factor set $\overline{\alpha}_i$ such that $\theta_i = \varphi_N$. Then

$$Irr(\bar{G}) = \bigcup_{i=1}^{t} \{ (\eta \varphi_i)^{\bar{G}} \mid \eta \in IrrProj(\bar{H}_i), \text{ with factor set } \alpha_i^{-1} \},$$

where α_i is obtained from $\bar{\alpha}_i$ as in Corollary 4.3.5.

Hence the characters table of \overline{G} is partitioned into t blocks $\triangle_1, \triangle_2, \ldots, \triangle_t$ where \triangle_i is produced from the inertia subgroup \overline{H}_i .

4.4. Irreducible Constituents and Conjugacy Classes

This section treats two topics. The first concerns the number of irreducible constituents of induced characters, and the second the number of conjugacy classes. Using some properties of extensions of characters, we will study the problem which asserts that if χ is a \bar{G} -invariant irreducible character of a normal subgroup N of a finite group \bar{G} , then the number of distinct irreducible constituents of $\chi^{\bar{G}}$ is equal to the number of χ -regular conjugacy classes of \bar{G}/N . We also show that the number of irreducible constituents of $\chi^{\bar{G}}$ is equal to the number of conjugacy classes of \bar{G}/N . We also show that the number of irreducible constituents of $\chi^{\bar{G}}$ is equal to the number of conjugacy classes of \bar{G}/N if and only if χ extends to a character of each subgroup N < x, y > of \bar{G} with $[x, y] \in N$.

Most of the results in this section are from *Gallagher* [40] but we give proofs from [67].

Lemma 4.4.1. Let N be a normal subgroup of \overline{G} such that \overline{G}/N is cyclic of order n. If χ is \overline{G} -invariant irreducible characters of N, then there exists precisely n irreducible characters of \overline{G} extending χ and their sum is $\chi^{\overline{G}}$.

PROOF. See Lemma 23.3.2 of [67].

Let N be a normal subgroup of a group \overline{G} and, for each $\overline{g} \in \overline{G}$, let the group $C_{\overline{g}}$ containing N be as defined in section 2.3, where

$$C_{\bar{q}}/N = C_{\bar{G}/N}(N\bar{g}).$$

Let χ be a \overline{G} -invariant irreducible character of N. From the proof of Lemma 4.3.1, χ extends to a character $\chi_{\overline{g}}$ of the subgroup $N < \overline{g} >$ with $\overline{g} \in \overline{G}$. We say that \overline{g} is χ -regular if $(\chi_{\overline{g}})^x = \chi_{\overline{g}}$ for all $x \in C_{\overline{g}}$. Note that Gallagher [40] uses the term goodness instead of χ -regular.

Remark 4.4.2. In [67] it was proved that the notion of χ -regularity is independent of the choice of $\chi_{\bar{g}}$ and depends only on χ and the conjugacy class of $N\bar{g}$ in \bar{G}/N .

We say that the conjugacy class of $N\bar{g}$ in \bar{G}/N is χ -regular if \bar{g} is χ -regular. By the above Remark, this notion is well defined.

Theorem 4.4.3. [40]. Let N be a normal subgroup of a group \overline{G} and let χ be a \overline{G} -invariant irreducible character of N. Then the number of distinct irreducible constituents of $\chi^{\overline{G}}$ is equal the number of χ -regular conjugacy classes of \overline{G}/N .

PROOF. See [67].

Corollary 4.4.4. [40] Let N be a normal subgroup of a group \overline{G} and let χ be a \overline{G} -invariant irreducible character of N. Then the number of distinct irreducible constituents of $\chi^{\overline{G}}$ is at most the number of conjugacy classes of \overline{G}/N with equality if and only if χ extends to a character of each subgroup N < x, y > with $[x, y] \in N$.

PROOF. See [67].

Now we provide some information on the number of conjugacy classes of G by using certain character-theoretic facts. In what follows r(G) denotes the number of conjugacy classes of G. Then r(G) is also the number of irreducible complex characters of G.

Lemma 4.4.5. [67] The following formula holds:

$$r(G) = \frac{1}{|G|} \sum_{g \in G} |C_G(g)|.$$

PROOF. The group G acts on itself by conjugation. If χ is the corresponding permutation character, then

$$\chi(g) = |C_G(g)| \quad \text{for all} \quad g \in G$$

and the G-orbits are precisely the conjugacy classes of G. Hence,

$$r(G) = \langle \chi, 1_G \rangle = \frac{1}{|G|} \sum_{g \in G} \chi(g) = \frac{1}{|G|} \sum_{g \in G} |C_G(g)|,$$

as desired.

Theorem 4.4.6. [40]. Let N be a normal subgroup of \overline{G} . Then

(i)
$$r(\bar{G}) \leq r(\bar{G}/N)r(N);$$

- (ii) The following conditions are equivalent:
- (a) $r(\bar{G}) = r(\bar{G}/N)r(N),$
- (b) $C_{\bar{g}} = C_{\bar{G}}(\bar{g})N$ for all $\bar{g} \in \bar{G}$,
- (c) each irreducible character of N extends to a character of each subgroup N < x, y > with $[x, y] \in \overline{G}$.

PROOF. See Theorem 28.2.3 of [67].

Fischer - Clifford Matrices

5.1. Prologue

The character table of a group provides considerable information about the group, and hence it is of importance in the physical sciences as well as in pure mathematics. Character tables of finite groups can be constructed using various techniques. For example, the Schreier-Sims algorithm, Todd-Coxeter coset enumeration method, the Burnside-Dixion algorithm and various other techniques. However *Bernd Fischer* studied a technique for constructing the character tables of group extensions. This technique, which is known as the technique of *Fischer-Clifford Matrices*, derives its fundamentals from the Clifford theory and provides very powerful information for constructing character tables. If $\bar{G} = N.G$ is an appropriate extension of N by G, the method involves the construction of a nonsingular matrix for each conjugacy class of \overline{G}/N . In this dissertation we apply this technique to both split and non-split extensions. This technique has also been discussed and used (mainly to split extensions) in Almestady [4], Darafsheh and Iranmanesh [26, 27], Fischer [34, 35, 36, 37], List [75], List and Mohammed [76], Lux and Pahlings [77], Moori and F. Ali [2], Moori and Mpono [90, 91, 92], Mpono [99], Pahlings [103], Saleh [112], Schiffer [113] and Whitely [120]. For the Fischer-Clifford matrices and their properties, although we shall note the work of Mpono [99], Schiffer [113] and Whitely [120], we follow the work of F. Ali [1] closely as he discussed both split and non-split extensions.

In Section 5.2 we define Fischer-Clifford matrices in general. In Subsection 5.2.1 we shall discuss the properties of the Fischer-Clifford matrices which are helpful in their computation. In Subsection 5.2.2 we study a special case of Fischer-Clifford matrices of a $\bar{G} = N.G$ with the property that every irreducible character of N can be extended to an irreducible character of its inertia group in \bar{G} . Sections 5.3 and 5.4 deal with the Fischer-Clifford matrices for the split cosets and non-split extensions respectively.

5.2. Definition and General Theory

Let $\bar{G} = N \cdot G$ be an extension of N by G, where N is normal subgroup of \bar{G} and $\bar{G}/N \cong G$. Let $\bar{g} \in \bar{G}$ be a lifting of $g \in G$ under the natural homomorphism $\bar{G} \longrightarrow G$ and [g] be a conjugacy class of elements of G with representative g. Let $X(g) = \{x_1, x_2, \ldots, x_{c(g)}\}$ be a set of representatives of the conjugacy classes of \bar{G} from the coset $N\bar{g}$ whose images under the natural homomorphism $\bar{G} \longrightarrow G$ are in [g] and we take $x_1 = \bar{g}$. Let $\{\theta_1, \theta_2, \ldots, \theta_t\}$ be a set of representatives of \bar{G} on Irr(N) such that for $1 \leq i \leq t$, we have $\bar{H}_i = I_{\bar{G}}(\theta_i)$ with the corresponding inertia factors H_i and let ψ_i be a projective character of \bar{H}_i with factor set $\bar{\alpha}_i$ such that $(\psi_i)_N = \theta_i$. By Remark 4.3.7 we have

$$Irr(\bar{G}) = \bigcup_{i=1}^{t} \{ (\psi_i \bar{\beta})^{\bar{G}} \mid \beta \in IrrProj(H_i), \text{ with factor set } \alpha_i^{-1} \},\$$

where α_i is obtained from $\bar{\alpha}_i$ and $\bar{\beta}$ from β as in Remark 4.3.6. Without loss of generality suppose that $\theta_1 = 1_N$ is the identity character of N. Then $\bar{H}_1 = \bar{G}$ and $H_1 = G$. Now choose y_1, y_2, \ldots, y_r to be the representatives of the α_i^{-1} -conjugacy classes of elements of H_i that fuse to [g] in G. Since $y_k \in H_i$ for $1 \leq k \leq r$, then we define $y_{lk} \in \bar{H}_i$ such that y_{lk} ranges over all representatives of the conjugacy classes of elements of \bar{H}_i which map to y_k under the homomorphism $\bar{H}_i \longrightarrow H_i$ whose kernel is N. Now by using the formula for induced characters given in Theorem 2.5.7, we have

$$\begin{aligned} (\psi_i \bar{\beta})^{\bar{G}}(x_j) &= \sum_{1 \le k \le r} \sum_{\ell}' \frac{|C_{\bar{G}}(x_j)|}{|C_{\bar{H}i}(y_{\ell_k})|} \psi_i \bar{\beta}(y_{\ell_k}) \\ &= \sum_{1 \le k \le r} \sum_{\ell}' \frac{|C_{\bar{G}}(x_j)|}{|C_{\bar{H}i}(y_{\ell_k})|} \psi_i(y_{\ell_k}) \bar{\beta}(y_{\ell_k}) \\ &= \sum_{1 \le k \le r} (\sum_{\ell}' \frac{|C_{\bar{G}}(x_j)|}{|C_{\bar{H}i}(y_{\ell_k})|} \psi_i(y_{\ell_k})) \beta(y_k) \end{aligned}$$

where $\sum_{\ell} i$ is the summation over all ℓ for which $y_{\ell_k} \sim x_j$ in \overline{G} . Now we define a matrix $M_i(g)$ by $M_i(g) = (a_{uv})$, where $1 \le u \le r$ and $1 \le v \le c(g)$, and

$$a_{uv} = \sum_{\ell}' \frac{|C_{\overline{G}}(x_j)|}{|C_{\overline{H}_i}(y_{\ell_k})|} \psi_i(y_{\ell_k}) \quad .$$

Then we obtain that

$$(\psi_i \beta)^{\overline{G}}(x_j) = \sum_{1 \le k \le r} a_{uv} \hat{\beta}(y_k)$$

By doing so for all $1 \le i \le t$ such that H_i contains an element in [g] we obtain the matrix M(g) given by

$$M(g) = \begin{bmatrix} M_1(g) \\ M_2(g) \\ \vdots \\ M_t(g) \end{bmatrix}$$

where $M_i(g)$ is the submatrix corresponding to the inertia group $\bar{H}i$ and its inertia factor H_i . If $H_i \cap [g] = \emptyset$, then $M_i(g)$ will not exist and M(g) does not contain $M_i(g)$. The size of the matrix M(g) is $l \times c(g)$ where l is the number of α_i^{-1} -regular conjugacy classes of elements of the inertia factors H_i 's for $1 \leq i \leq t$ which fuse into [g] in G and c(g) is the number of conjugacy classes of elements of elements of \bar{G} which correspond to the coset $\bar{g}N$. Then M(g) is the Fischer-Clifford matrix of \bar{G} corresponding to the coset $\bar{g}N$. We will see later that M(g) is a $c(g) \times c(g)$ nonsingular matrix. Let

$$R(g) = \{(i, y_k) \mid 1 \le i \le t , H_i \cap [g] \neq \emptyset , 1 \le k \le r\}$$

and we note that y_k runs over representatives of the α_i^{-1} -conjugacy classes of elements of H_i which fuse into [g] in G. Following the notation used in Fischer [33], Mpono [99] and Whitely [120] we denote M(g) by writing $M(g) = Cl(N\bar{g}) = (a_j^{(i,y_k)})$, where

$$a_{j}^{(i,y_{k})} = \sum_{\ell}' \frac{|C_{\bar{G}}(x_{j})|}{|C_{\bar{H}i}(y_{\ell_{k}})|} \psi_{i}(y_{\ell_{k}}) \quad ,$$

with columns indexed by X(g) and rows indexed by R(g). Then the partial character table of \overline{G} on the classes $\{x_1, x_2, \ldots, x_{c(g)}\}$ is given by

$$\begin{bmatrix} C_1(g)M_1(g) \\ C_2(g)M_2(g) \\ \vdots \\ C_t(g)M_t(g) \end{bmatrix}$$

where the Fischer-Clifford matrix M(g) is divided into blocks with each block corresponding to an inertia group \bar{H}_i and $C_i(g)$ is the partial projective character table of H_i with factor set α_i^{-1} consisting of the columns corresponding to the α_i^{-1} -regular classes that fuse into [g] in G. We obtain the characters of \bar{G} by multiplying the relevant columns of the projective characters of H_i with factor set α_i^{-1} by the rows of M(g). We can also observe that the number of irreducible characters of \bar{G} is the sum of numbers of projective characters of the inertia factors H_i 's with factor set α_i^{-1} , for all $i, 1 \leq i \leq t$.

5.2.1 Properties of Fischer-Clifford Matrices

In this section we shall discuss some properties of the Fischer-Clifford matrices which are useful in their computation. These properties have been discussed in [1, 26, 27, 35, 36, 75, 76, 90, 91, 92, 99, 112, 120].

Let K be a group and $A \leq Aut(K)$. Then by Brauer's theorem (Theorem 2.6.7) A acts on the conjugacy classes of elements of K and on the irreducible characters of K resulting in the same number of orbits.

Lemma 5.2.1. Suppose we have the following matrix describing the above actions:

 $1 = l_1 \quad l_2 \quad \cdots \quad l_j \quad \cdots \quad l_t$ $s_1 \\ s_2 \\ \vdots \\ s_i \\ \vdots \\ s_i \\ \vdots \\ s_t \end{pmatrix} \begin{pmatrix} 1 & 1 & \cdots & 1 & \cdots & 1 \\ a_{21} & a_{22} & \cdots & a_{2j} & \cdots & a_{2t} \\ \vdots & \vdots & & \vdots & & \vdots \\ a_{i1} & a_{i2} & \cdots & a_{ij} & \cdots & a_{it} \\ \vdots & \vdots & & \vdots & & \vdots \\ a_{t1} & a_{t2} & \cdots & a_{tj} & \cdots & a_{tt} \end{pmatrix}$

where $a_{1j} = 1$ for $j \in \{1, 2, ..., t\}$, l_j 's are lengths of orbits of A on the conjugacy classes of K, s_i 's are lengths of orbits of A on Irr(K) and a_{ij} is the sum of s_i irreducible characters of K on the element x_j , where x_j is an element of the orbit of length l_j . Then the following relation holds for $i, i' \in \{1, 2, ..., t\}$: $\sum_{j=1}^{t} a_{ij}\overline{a_{i'j}}l_j = |K|s_i\delta_{ii'}$.

PROOF. This result has been proved as Lemma 2.3.2 in [112] and as Lemma 4.3.2 in [120].

For arithmetical properties weights are important. We present M(g) with corresponding weights. Let $x_j \in X(g)$. For a fixed coset $X = \bar{g}N \in \bar{G}/N$, we define $m_j = [N_{\bar{G}}(X) : C_{\bar{G}}(x_j)]$.

The Fischer-Clifford matrix M(g) is partitioned row-wise into blocks, where each block corresponds to an inertia group. The columns of M(g) are indexed by X(g) and for each $x_j \in X(g)$, at the top of the columns of M(g), we write $|C_{\bar{G}}(x_j)|$ and at the bottom we write m_j . The rows of M(g) are indexed by R(g) and on the left of each row we write $|C_{H_i}(y_k)|$, where y_k fuses into [g] in G. Then in general we can write M(g) with corresponding weights for rows and columns as follows, where blocks corresponding to the inertia groups are separated by horizontal lines.

	$ C_{\overline{G}}(x_1) $	$ C_{\overline{G}}(x_2) $		$ C_{\overline{G}}(x_{c(g)}) $
$ C_G(g) $	$\begin{pmatrix} a_1^{(1,g)} \end{pmatrix}$	$a_2^{(1,g)}$		$a_{c(g)}^{(1,g)}$
$ C_{H_2}(y_1) $ $ C_{H_2}(y_2) $:	$\begin{array}{c} a_{1}^{(2,y_{1})} \\ a_{1}^{(2,y_{2})} \\ \vdots \end{array}$	$a_2^{(2,y_1)} \\ a_2^{(2,y_2)} \\ \vdots$	···· ··· :	$a^{(2,y_1)}_{c(g)} a^{(2,y_2)}_{c(g)} = \vdots$
$\begin{aligned} C_{H_i}(y_1) \\ C_{H_i}(y_2) \end{aligned}$	$a_1^{(i,y_1)} \ a_1^{(i,y_2)}$	$a_2^{(i,y_1)} \ a_2^{(i,y_2)}$		$a^{(i,y_1)}_{c(g)} \ a^{(i,y_2)}_{c(g)}$
	:	•	•	
$\begin{aligned} C_{H_t}(y_1) \\ C_{H_t}(y_2) \end{aligned}$	$a_1^{(t,y_1)} \ a_1^{(t,y_2)}$	$a_2^{(t,y_1)} \\ a_2^{(t,y_2)}$	· · · ·	$a^{(t,y_1)}_{c(g)} \ a^{(t,y_2)}_{c(g)}$
:	:	:	:	:
	m_1	m_2		$m_{c(g)}$

Table 5.1

Remark 5.2.2. Fischer [36] has shown that the Fischer-Clifford matrix M(g) satisfies complex conjugation.

The following result gives the orthogonality relation for M(g). Its proof was obtained from Whitley [120], Proposition 4.3.3.

Proposition 5.2.3. [1, 99, 120] (Column orthogonality) Let $\overline{G} = N \cdot G$, then

$$\sum_{(i,y_k)\in R(g)} |C_{H_i}(y_k)| a_j^{(i,y_k)} \overline{a_{j'}^{(i,y_k)}} = \delta_{jj'} |C_{\overline{G}}(x_j)| \quad .$$

PROOF. The partial character table of \overline{G} at classes $x_1, \ldots, x_{c(g)}$ is given by

$$\begin{bmatrix} C_1(g)M_1(g) \\ C_2(g)M_2(g) \\ \vdots \\ C_t(g)M_t(g) \end{bmatrix}$$

By column orthogonality of the character table of \overline{G} , we have

$$\begin{split} |C_{\overline{G}}(x_{j})|\delta_{jj'} &= \sum_{i=1}^{t} \sum_{\beta_{i} \in IrrProj(H_{i})} (\sum_{y_{k}:(i,y_{k}) \in R(g)} a_{j}^{(i,y_{k})}\beta_{i}(y_{k}))(\overline{\sum_{y_{k}':(i,y_{k}') \in R(g)} a_{j'}^{(i,y_{k}')}\beta_{i}(y_{k}')}) \\ &= \sum_{i=1}^{t} \sum_{\beta_{i} \in IrrProj(H_{i})} (\sum_{y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k}')}}\beta_{i}(y_{k})\overline{\beta_{i}(y_{k})} + \sum_{y_{k}} \sum_{y_{k}' \neq y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}}\beta_{i}(y_{k})\overline{\beta_{i}(y_{k}')}) \\ &= \sum_{i=1}^{t} (\sum_{y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}}\sum_{\beta_{i} \in IrrProj(H_{i})} \beta_{i}(y_{k})\overline{\beta_{i}(y_{k})} + \sum_{y_{k} \neq y_{k}} \sum_{y_{k}' \neq y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}}\sum_{\beta_{i} \in IrrProj(H_{i})} \beta_{i}(y_{k})\overline{\beta_{i}(y_{k})} + \sum_{y_{k} \neq y_{k}} \sum_{y_{k}' \neq y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}}\sum_{\beta_{i} \in IrrProj(H_{i})} \beta_{i}(y_{k})\overline{\beta_{i}(y_{k})}) \\ &= \sum_{i=1}^{t} (\sum_{y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}} \sum_{\beta_{i} \in IrrProj(H_{i})} \beta_{i}(y_{k})\overline{\beta_{i}(y_{k})}) \\ &= \sum_{i=1}^{t} (\sum_{y_{k}} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}} |C_{H_{i}}(y_{k})| + 0) \\ &= \sum_{(i,y_{k}) \in R(g)} a_{j}^{(i,y_{k})}\overline{a_{j'}^{(i,y_{k})}} |C_{H_{i}}(y_{k})|. \end{split}$$

Theorem 5.2.4. $a_j^{(1,g)} = 1$ for all $j \in \{1, 2, ..., c(g)\}$.

PROOF. For $y_{\ell_k} \sim x_j$ in \overline{G} , we have $|C_{\overline{G}}(x_j)| = |C_{\overline{H}_1}(y_{\ell_k})|$. Thus we obtain that

$$a_j^{(1,g)} = \sum_{\ell}' \frac{|C_{\overline{G}}(x_j)|}{|C_{\overline{H}_1}(y_{\ell_k})|} \psi_1(y_{\ell_k}) = \sum_{\ell}' 1 = 1$$

Hence the result.

Proposition 5.2.5. [75, 120] The matrix $M(1_G)$ is the matrix with rows equal to the orbit sums of the action of \overline{G} on Irr(N) with duplicate columns discarded. For this matrix we have $a_j^{(i,1_G)} = [G:H_i]$, and an orthogonality relation for rows:

$$\sum_{j=1}^{t} \frac{1}{|C_{\overline{G}}(x_j)|} a_j^{(i,1_G)} a_j^{(i',1_G)} = \frac{1}{|C_{H_i}(1_G)|} \delta_{ii'} = \frac{1}{|H_i|} \delta_{ii'}$$

PROOF. See [99].

As a consequence of Lemma 5.3.1, Proposition 5.3.3 and from Fischer [36], we have the following properties:

(a)
$$|X(g)| = |R(g)|$$

(b)
$$\sum_{j=1}^{c(g)} m_j a_j^{(i,y_k)} \overline{a_j^{(i',y_k')}} = \delta_{(i,y_k),(i',y_k')} \frac{|C_G(g)|}{|C_{H_i}(y_k)|} |N| ,$$

(c)
$$\sum_{(i,y_k)\in R(g)} a_j^{(i,y_k)} \overline{a_{j'}^{(i,y_k)}} |C_{H_i}(y_k)| = \delta_{jj'} |C_{\overline{G}}(x_j)|$$

(d) M(g) is square and nonsingular.

5.2.2 Fischer-Clifford Matrices (Special Case)

Let $\bar{G} = N.G$ be an extension of N by G such that every irreducible character θ of N can be extended to its inertia group $\bar{H} = I_{\bar{G}}(\theta)$. Now we define the Fischer-Clifford matrices in the same way as the general case. Let $\bar{g} \in \bar{G}$ be a lifting of $g \in G$ under the natural homomorphism $\bar{G} \longrightarrow G$ and [g] be a conjugacy class of elements of G with representative g. Let $X(g) = \{x_1, x_2, \ldots, x_{c(g)}\}$ be a set of representatives of the conjugacy classes of \bar{G} from the coset $N\bar{g}$ whose images under the natural homomorphism $\bar{G} \longrightarrow G$ are in [g] and we take $x_1 = \bar{g}$. Let $\{\theta_1, \theta_2, \ldots, \theta_t\}$ be a set of representatives of the orbits of \bar{G} on Irr(N) such that for $1 \leq i \leq t$, we have $H_i = I_{\bar{G}}(\theta_i)$ with $H_i = \bar{H}_i/N \leq G$ and that $\psi_i \in Irr(\bar{H}_i)$ is an extension of θ_i to \bar{H}_i . Then without loss of generality suppose that $\theta_1 = I_N$ is the identity character of N. Then $\bar{H}_1 = \bar{G}$ and $H_1 = G$. Now choose y_1, y_2, \ldots, y_r to be the representatives of the conjugacy classes of elements of H_i which fuse into [g] in G. Since $y_k \in H_i$ for $1 \leq k \leq r$, then we define $y_{\ell_k} \in \bar{H}_i$ such that y_{ℓ_k} ranges over all the representatives of the conjugacy classes of elements of \bar{H}_i which map to y_k under the homomorphism $\bar{H}_i \longrightarrow H_i$ whose kernel is N. Let $\beta \in Irr(\bar{H}_i)$ such that $N \subseteq ker(\beta)$. Then β is a lifting of $\hat{\beta} \in Irr(H_i)$ such that $\beta(y_{\ell_k}) = \hat{\beta}(y_k)$ for any lifting $y_{\ell_k} \in \bar{H}_i$ of $y_k \in H_i$. Now by using Theorem 2.4.7, as in the general case, we obtain that

$$(\psi_i \beta)^{\bar{G}}(x_j) = \sum_{1 \le k \le r} (\sum_{\ell}' \frac{|C_{\bar{G}}(x_j)|}{|C_{\bar{H}_i}(y_{\ell_k})|} \psi_i(y_{\ell_k})) \hat{\beta}(y_k)$$

where \sum_{ℓ}' is the summation over all ℓ for which $y_{\ell_k} \sim x_j$ in \overline{G} . We define a matrix $M_i(g)$ by $M_i(g) = (a_{uv})$, where $1 \leq u \leq r$ and $1 \leq v \leq c(g)$, and

$$a_{uv} = \sum_{\ell} \frac{|C_{\overline{G}}(x_j)|}{|C_{H_i}(y_{\ell_k})|} \psi_i(y_{\ell_k})$$

Then we obtain that

$$(\psi_i\beta)^{\overline{G}}(x_j) = \sum_{1 \le k \le r} a_{uv}\hat{\beta}(y_k)$$

By doing so for all $1 \le i \le t$ such that H_i contains an element in [g] we obtain the matrix M(g) given by

$$M(g) = \begin{bmatrix} M_1(g) \\ M_2(g) \\ \vdots \\ M_t(g) \end{bmatrix}$$

where $M_i(g)$ is the submatrix corresponding to the inertia group \overline{H}_i and its inertia factor H_i . Then as in the previous section, M(g) is the *Fischer-Clifford matrix* of \overline{G} corresponding to the coset $\overline{g}N$. Let

$$R(g) = \{(i, y_k) \mid 1 \le i \le t , H_i \cap [g] \ne \emptyset , 1 \le k \le r\}$$

and we note that y_k runs over representatives of the conjugacy classes of elements of H_i which fuse into [g] in G. Again we denote M(g) by writing $M(g) = (a_j^{(i,y_k)})$, where

$$a_{j}^{(i,y_{k})} = \sum_{\ell} \frac{|C_{\bar{G}}(x_{j})|}{|C_{\bar{H}i}(y_{\ell_{k}})|} \psi_{i}(y_{\ell_{k}})$$

with columns indexed by X(g) and rows indexed by R(g). Then we obtain the irreducible characters of \overline{G} by multiplying the relevant columns of the irreducibles characters of H_i by the rows M(g).

Remark 5.2.6. All our results of Section 5.2.1 are applicable with irreducible projective characters are replaced by ordinary irreducible characters.

5.3. Split Cosets

From now on suppose that N is an elementary abelian normal p-subgroup of \overline{G} and $\overline{g}N = X$ is a fixed coset of $\overline{G}/N \cong G$. Let $M = C_{\overline{g}} = N_{\overline{G}}(X)$. We define

$$N_{\bar{g}} := \langle [\bar{g}, n], n \in N \rangle$$
.

With these notations we have the following lemma.

- **Lemma 5.3.1.** (i) $N_x = N_{\bar{g}}$ for all $x \in X$ and $[\bar{g}, u_1].[\bar{g}, u_2] = [\bar{g}, u_1 u_2]$ for all $u_1, u_2 \in N$.
 - (ii) $N_{\bar{g}} \triangleleft M$ and $N_{\bar{g}} \leq N$.
- (iii) If $\varphi \in Irr(N)$, then $N_{\bar{g}} \leq ker(\varphi)$ or $I_{\bar{G}}(\varphi) \cap \bar{g}N = \emptyset$.

PROOF.

(i) Let $x = \bar{g}n \in \bar{g}N$ and $u \in N$, then

$$[x, u] = [\bar{g}n, u] = n^{-1} (u^{-1})^{\bar{g}} n u$$
$$= (u^{-1})^{\bar{g}} u \quad \text{since } N \text{ is abelian}$$
$$= \bar{g}^{-1} u^{-1} \bar{g} u = [\bar{g}, u]$$

which implies that

$$N_x = N_{\bar{g}}$$
 for all $x \in \bar{g}N$.

Also since N is abelian, we obtain for all $u_1, u_2 \in N$

$$\begin{split} [\bar{g}, u_1].[\bar{g}, u_2] &= (u_1^{-1})^{\bar{g}} u_1 (u_2^{-1})^{\bar{g}} u_2 \\ &= (u_1^{-1} u_2^{-1})^{\bar{g}} u_1 u_2 \\ &= [\bar{g}, u_1 u_2] \,. \end{split}$$

Hence

$$[\bar{g}, u_1].[\bar{g}, u_2] = [\bar{g}, u_1 u_2] \text{ for } u_1, u_2 \in N.$$

(ii) Since $[\bar{g}, u] = (u^{-1})^{\bar{g}} u \in N$, we obtain $N_{\bar{g}} \leq N \leq M$. Conversely, let $m \in M$ then

$$\begin{split} [\bar{g}, u]^m &= m^{-1} [\bar{g}, u] m \\ &= (\bar{g}^{-1})^m (u^{-1})^m \bar{g}^m u^m \\ &= (\bar{g}^m)^{-1} (u^m)^{-1} \bar{g}^m u^m \\ &= [\bar{g}^m, u^m] \in N_{\bar{g}} \; . \end{split}$$

Hence $N_{\bar{g}} \lhd M$.

(iii) Let $\varphi \in Irr(N)$ be fixed. Then

$$\begin{split} N_{\bar{g}} &\leq Ker(\varphi) &\Leftrightarrow \varphi([\bar{g}, u]) = \varphi(1) = 1 \quad \text{for all} \quad u \in N \\ &\Leftrightarrow \varphi(\bar{g}^{-1}u^{-1}\bar{g}u) = \varphi((u^{-1})^{\bar{g}}u) = 1 \\ &\Leftrightarrow \varphi((u^{-1})^{\bar{g}}) = (\varphi(u))^{-1} = \varphi(u^{-1}) \\ &\Leftrightarrow \varphi^{\bar{g}}(u^{-1}) = \varphi(u^{-1}) \\ &\Leftrightarrow \varphi^{\bar{g}} = \varphi \\ &\Leftrightarrow \bar{g}N \subseteq I_{\bar{G}}(\varphi) \\ &\Leftrightarrow \bar{g}N \cup I_{\bar{G}}(\varphi) \neq \emptyset \,. \end{split}$$

Remark 5.3.2. We can easily show that $\langle X \rangle / N_{\bar{g}}$ is abelian and $X / N_{\bar{g}}$ is a coset of $\langle X \rangle / N_{\bar{g}}$.

Lemma 5.3.3. [36] The rows of the Fischer-Clifford matrix Cl(X) can be identified with restrictions of *M*-invariant characters of $\langle X \rangle / N_{\bar{g}}$ to $X / N_{\bar{g}}$.

PROOF. This is Lemma 5.3 in [36].

Remark 5.3.4. In the above lemma, the rows of Cl(X) will be an independent set of orbit sums, under the action of M on $\langle X \rangle / N_{\bar{g}}$. This observation was first given in Fischer [34].

Definition 5.3.5. A coset X is said to be a split coset if it contains an element x such that $M = N.C_{\bar{G}}(x)$.

Note that we do not require $\langle x \rangle \cap N = \langle 1 \rangle$ in the above definition.

Lemma 5.3.6. [113] If the extension split, then every coset is a split coset.

PROOF. Let $X = \bar{g}N$ and $h \in C_{\bar{G}}(\bar{g})$ then $h(\bar{g}n)h^{-1} = (h\bar{g}h^{-1})(hnh^{-1}) = \bar{g}hnh^{-1} = \bar{g}n^h \in \bar{g}N$. Now since $N \leq M$ and $C_{\bar{G}}(\bar{g}) \leq M$ then $M \geq N.C_{\bar{G}}(\bar{g})$. Let C be the complement of N in \bar{G} such that $\bar{g} \in C$. Let $m \in M$ then m = n.k, for some $k \in C$. Since $M = N_{\bar{G}}(\bar{g}N)$, $(\bar{g}N)^m = \bar{g}N$. Hence

$$\bar{g}N = (\bar{g}N)^m = m(\bar{g}N)m^{-1}$$

= $n(k\bar{g}Nk^{-1})n^{-1} = n(k\bar{g}Nk^{-1})n^{-1} = n(\bar{g}N)^k n^{-1}.$

So that $n^{-1}(\bar{g}N)n = (\bar{g}N)^k$ and $n^{-1}\bar{g}N = (\bar{g}N)^k$. Hence $\bar{g}N = (\bar{g}N)^k$. It follows that $\bar{g}N = (\bar{g}N)^k = \bar{g}^k N$, which implies that $\bar{g}^k \in \bar{g}N$. Hence $\bar{g}^k \in C \cap \bar{g}N = \{\bar{g}\}$ and so $k \in C_{\bar{G}}(\bar{g})$, which implies that $m = n.k \in N.C_{\bar{G}}(\bar{g})$ and so $M \leq N.C_{\bar{G}}(\bar{g})$. Thus $M = N.C_{\bar{G}}(\bar{g})$.

The following result is of fundamental importance and very helpful to fill the entries of Fischer-Clifford matrices.

Lemma 5.3.7. [36] Let X be a split coset then the rows of Cl(X) can be identified with M-invariant characters of $N/N_{\bar{g}}$ multiplied by a p-th root of unity.

PROOF. See [36] and [113].

Lemma 5.3.8. Let $X = \bar{g}N$ be a split coset and $N_{\bar{G}}(X) = NC_{\bar{G}}(x)$ for $x \in X(g)$. Then we have the following:

$$\begin{aligned} (i) \ a_1^{(i,y_k)} &= \frac{|C_G(g)|}{|C_{H_i}(y_k)|} ,\\ (ii) \ |a_j^{(i,y_k)}| &\le |a_1^{(i,y_k)}| \text{ for all } 1 \le j \le r ,\\ (iii) \ If \ |N| &= p^w, \text{ then } a_j^{(i,y_k)} \equiv a_1^{(i,y_k)} (mod \ p) \end{aligned}$$

PROOF. See [113].

5.4. Non-Split Extensions

Let $\bar{G} = N^{\cdot}G$ be a non-split extension, where N is an elementary abelian normal p-subgroup of \bar{G} . Let $\bar{g}N$ be a conjugacy class representative of \bar{G}/N and φ be a representative of \bar{G} -orbit irreducible characters of N with the projective extension $\bar{\varphi}$ to \bar{G} . We consider the groups $\langle \bar{g} \rangle N \leq \bar{G}$ and $\langle \bar{g}N \rangle \leq \bar{G}/N$. Lemma 5.4.1.

$$\langle \bar{g} \rangle N/N = \langle \bar{g}N \rangle$$
.

PROOF. Let $x \in \langle \bar{g} \rangle N/N$, then $x = \bar{g}^m nN = \bar{g}^m N$ for some $m \in \mathbb{Z}$. So that $x = (\bar{g}N)^m \in \langle \bar{g}N \rangle$. Hence $\langle \bar{g} \rangle N/N \leq \langle \bar{g}N \rangle$. Conversely, let $x \in \langle \bar{g}N \rangle$. Then $x = (\bar{g}N)^m = \bar{g}^m N$ for some $m \in \mathbb{Z}$. Hence $x = (\bar{g}^m N) \in \langle \bar{g} \rangle N/N$. Thus $\langle \bar{g}N \rangle \leq \langle \bar{g} \rangle N/N$. Therefore $\langle \bar{g} \rangle N/N = \langle \bar{g}N \rangle$.

Lemma 5.4.2. With the above notations, we have the following:

- (a) $<\bar{g}>N \leq M$.
- (b) $(\langle \bar{g} \rangle N)' = N_{\bar{g}}$ where $(\langle \bar{g} \rangle N)'$ denotes the commutator subgroup of $\langle \bar{g} \rangle N$.
- (c) $\langle \bar{g} \rangle N \leq I_M(\varphi)$ where $\varphi \in Irr(N)$.
- (d) Given $\varphi \in Irr(N)$ there exists an extension $\eta\beta$ to $\langle \bar{g} \rangle N$ where $\eta = (\bar{\varphi})_{\langle \bar{g} \rangle N}$ and β is a projective character of $\langle \bar{g}N \rangle$.

PROOF.

(a) Let $x \in \langle \bar{g} \rangle N$ then $x = \bar{g}^m N$ for some $m \in \mathbb{Z}$. Now

$$\begin{aligned} x(\bar{g}N) &= \bar{g}^m n(\bar{g}N) = \bar{g}^m n \bar{g}N = \bar{g}^m n N \bar{g} \quad (\text{since} \quad N \leq \bar{G}) \\ &= \bar{g}^m N \bar{g} = N \bar{g}^{m+1}. \end{aligned}$$

Similarly, $(\bar{g}N)x = N\bar{g}^{m+1}$. Hence $x \in M = N_{\bar{G}}(\bar{g}N)$ and so $\langle \bar{g} \rangle N \leq M$.

(b) First suppose that $[\bar{g}, n] \in N_{\bar{g}}$ then $[\bar{g}, n] \in (\langle \bar{g} \rangle N)'$ and thus $N_{\bar{g}} \leq (\langle \bar{g} \rangle N)'$. Also, for $n \in N$, by the definition of $N_{\bar{g}}$, we have

$$(\bar{g}N_{\bar{g}})(nN_{\bar{g}}) = (nN_{\bar{g}})(\bar{g}N_{\bar{g}}).$$

Therefore $(\langle \bar{g} \rangle N/N_{\bar{g}})$ is abelian, and hence $(\langle \bar{g} \rangle N)' \leq N_{\bar{g}}$ and we deduce that $(\langle \bar{g} \rangle N)' = N_{\bar{g}}$.

- (c) Let $\varphi \in Irr(N)$ then $N_{\bar{g}} \leq Ker(\varphi)$. Now by Lemma 5.3.1, we have $\bar{g}N \cap I_M(\varphi) \neq \emptyset$. Therefore \bar{g} lies in $I_M(\varphi)$ and so $\langle \bar{g} \rangle \leq I_M(\varphi)$. Hence $\langle \bar{g} \rangle N \leq I_M(\varphi)$.
- (d) Notice that by part (c), $W = \langle \bar{g} \rangle N$ is a subgroup of $I_M(\varphi)$. Hence φ is invariant under W. So we can apply the Theorem 4.3.3 to φ and W (see Theorem 5.8 in [100]). Let $\chi \in Irr(\langle \bar{g} \rangle N, \varphi)$ then by the Clifford theorem (Theorem 4.3.3) we obtain $\chi = ((\bar{\varphi})_{\langle \bar{g} \rangle N})\beta = \eta\beta$ where β is an $\bar{\alpha}^{-1}$ -projective character of $\langle \bar{g} \rangle N/N = \langle \bar{g}N \rangle$ and $\bar{\alpha}$ is the factor set of $\langle \bar{g} \rangle N \times \langle \bar{g} \rangle N$ obtained from α . If N is abelian, then χ is linear since $\chi_N = \varphi$ is linear (because $deg(\chi) = deg(\varphi) = 1$).

Theorem 5.4.3. [35] Let $\bar{g} \in \bar{G}$ so that $\langle \bar{g} \rangle N$ is abelian. Then $\langle \bar{g}N \rangle \leq Z(\bar{G}/N)$ and the rows of Fischer-Clifford matrix of $\bar{g}N$ for regular classes of the inertia group of φ in \bar{G} can be regarded as restrictions to $\bar{g}N$ of the \bar{G} -orbit sums of the (projective) extension $\eta\beta$ to $\langle \bar{g} \rangle N$ of φ .

PROOF. See [35] and [113].

5.5. Character Table and GAP

Using Fischer-Clifford matrices and partial character tables, we are able to compute the full character table of \overline{G} . Since the character tables have been computed manually, in order to detect errors we tested their validity using GAP. For doing so we developed and used Programme E to rewrite the character tables in GAP format.

Programme E

gap>ct:=fuction()local ct;ct:=rec();

>ct.SizesCentralizers:=[n Centralizer Orders];

>ct.OrdersClassRepresentatives:=[n Class Representatives Orders];

>ct.Irr:=[[$n \times n$ irreducibles]];

>ct.UnderlyingCharacteristic:=0;ct.Id:= \overline{G} ;

>ConvertToLibraryCharacterTable NC(ct);return ct;end;ct:=ct();

gap>SetInfoLevel(InfoCharacterTable,2);

gap>IsInternallyConsistent(ct);

gap>PossiblePowerMaps(ct, p); (p-prime divisor of \overline{G}).

I would like to acknowledge F. Ali who helped me to develop Programme E. The Programme E also computes the power maps.

6 A group $2^7:S_8$ in \overline{Fi}_{22}

Prologue

The group $S_8 \cong O_6^+(2)$ is a maximal subgroup of $SP_6(2)$ of order 40320 and index 36. It can be generated by two elements of orders 2 and 7 respectively inside of $SP_6(2)$. The group $2^7:S_8$ is a maximal subgroup of $2^7:SP_6(2)$ of index 36. However $2^7:SP_6(2)$ is itself a maximal subgroup of the full automorphism group \overline{Fi}_{22} of the smallest Fischer sporadic simple group Fi_{22} , of index 694980. The object of this chapter is to compute the Fischer-Clifford matrices of $2^7:S_8$ which can then be used together with the ordinary character tables of the inertia factors of S_8 to compute its full ordinary character table. One can look at [1, 18, 84, 85, 90, 91, 99] for further reading. The notation used is taken from the ATLAS of finite groups [23] which we denoted ATLAS and ATLAS of finite group representation [124] denoted ATLAS $\mathbb{V}3$.

6.1. Introduction

The group $\overline{Fi}_{22} = Fi_{22}.2$ is the full automorphism group of the smallest Fischer sporadic simple group Fi_{22} . It has a maximal subgroup $2^7:SP_6(2)$ of order 185794560 and index 694980 which has been discussed in [91]. More details about this maximal subgroup can be obtained from [99]. This maximal subgroup contains a group of the form $2^7:S_8$ as a subgroup of order 5160960 and index 36. The group S_8 is a maximal subgroup of $SP_6(2)$ of order 40320 and index 36. There are two orthogonal groups sitting maximally in $SP_6(2)$ viz. $O_6^+(2) \cong S_8$ and $O_6^-(2) \cong U_4(2):2$ of orders 40320 and 51840 and indices 36 and 28 respectively.

We used the computer algebra system of GAP [41] running on a SUN GX2 computer at the University of KwaZulu-Natal in Pietermaritzburg. The naming of the conjugacy classes of elements will be consistent with that used in the ATLAS [23].

We generate $SP_6(2)$ as a matrix group by two 7 × 7 matrices α and β over GF(2), where α and β

are given by

α –	0 0 0	1 1 1	0 1 1	1 1 1	1 0 0	1 1 0	0 7 0 0 0	β-	$\begin{bmatrix} 0\\1\\1\\0 \end{bmatrix}$	0 1 1	1 1 0	1 1 0	0 0 1	0 0 1	0 0 0	
	1 0 0	1 1 0	1 1 0	0 0 0	1 1 0	0 0 0	0 0 1		$\begin{bmatrix} 0\\1\\1 \end{bmatrix}$	1 0 0	$\begin{array}{c} 1 \\ 0 \\ 0 \end{array}$	1 1 1	1 1 1	$\begin{array}{c} 0 \\ 0 \\ 1 \end{array}$	0 0 1	

with $o(\alpha) = 5$ and $o(\beta) = 2$.

6.2. The action of S_8 on 2^7

We generate S_8 as a matrix group inside of $SP_6(2)$ by two 7×7 matrices α_1 and β_1 of orders 2 and 7 respectively as follows:

	Γ0	1	0	1	1	1	0 -		ΓΟ	0	1	1	1	0	0]
	1	0	0	1	1	1	0		0	1	1	0	0	0	0
	1	1	1	1	1	1	0		0	1	1	0	0	1	0
$\alpha_1 =$	0	0	0	1	0	0	0	$, \beta_1 =$	1	0	1	1	1	1	0
	1	1	0	1	0	1	0		1	1	1	0	1	1	0
	1	1	0	1	1	0	0		0	1	0	0	0	1	0
	Lο	0	0	0	0	0	1 _		Lο	0	1	1	0	1	1

When S_8 acts on 2^7 , we obtain six orbits of lengths 1, 1, 28, 28, 35, 35 with corresponding point stabilizers $S_8, S_8, S_6 \times 2, S_6 \times 2, (S_4 \times S_4):2, (S_4 \times S_4):2$ of orders 40320, 40320, 1440, 1440, 1152, 1152 respectively. These point stabilizers can be generated inside of S_8 as groups of 7×7 matrices over GF(2) as follows:

$$S_6 \times 2 = \langle \alpha_2, \beta_2 \rangle$$
, $(S_4 \times S_4): 2 = \langle \alpha_3, \beta_3 \rangle$

where

$$\alpha_{2} = \begin{bmatrix} 1 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 & 1 \end{bmatrix} , \quad \beta_{2} = \begin{bmatrix} 1 & 1 & 0 & 1 & 1 & 0 & 0 \\ 1 & 1 & 1 & 0 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 & 1 & 0 & 0 \\ 1 & 0 & 1 & 1 & 1 & 0 & 0 \\ 1 & 0 & 1 & 1 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 1 & 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 & 1 \end{bmatrix} , \quad \beta_{3} = \begin{bmatrix} 1 & 1 & 1 & 0 & 1 & 1 & 0 \\ 1 & 1 & 0 & 0 & 1 & 0 & 0 \\ 1 & 0 & 1 & 0 & 1 & 0 & 0 \\ 1 & 0 & 1 & 0 & 1 & 0 & 0 \\ 1 & 0 & 1 & 0 & 1 & 0 & 0 \\ 1 & 0 & 1 & 0 & 1 & 1 & 1 \end{bmatrix}$$

We have that $o(\alpha_2) = 2$, $o(\beta_2) = 6$, $o(\alpha_3) = 2$, $o(\beta_3) = 12$. The six orbits resulting from the action of S_8 on 2^7 have the following representatives (0, 0, 0, 0, 0, 0, 0), (1, 1, 0, 1, 0, 0, 1), (1, 0, 1, 0, 1, 0, 1), (0, 0, 0, 1, 0, 0, 0), (1, 1, 1, 1, 1, 1, 1), (1, 0, 0, 0, 0, 0, 0) respectively.

6.3. The conjugacy classes of $2^7:S_8$

The action of S_8 on 2^7 produces six orbits of lengths 1, 1, 28, 28, 35, 35 with corresponding point stabilizers $S_8, S_8, S_6 \times 2, S_6 \times 2, (S_4 \times S_4):2, (S_4 \times S_4):2$ of orders 40320, 40320, 1440, 1440, 1152, 1152 respectively. Let $\chi(S_8 \mid 2^7)$ be the permutation character of S_8 acting on 2^7 . Then, from methods that were developed in [99] and also used in [1], we obtain that

$$\chi(S_8 \mid 2^7) = 1 + 1 + 2I_{S_6 \times 2}^{S_8} + 2I_{(S_4 \times S_4):2}^{S_8}$$

= 1a + 1a + 2(1a + 7a + 20a) + 2(1a + 14a + 20a)
= 6 × 1a + 2 × 7a + 2 × 14a + 4 × 20a

where $I_{S_6 \times 2}^{S_8}$, $I_{(S_4 \times S_4):2}^{S_8}$ are the identity characters of $S_6 \times 2$ and $(S_4 \times S_4):2$ induced to S_8 respectively. Thus $\chi(S_8 \mid 2^7)$ will give the number k of points of 2^7 fixed by each $g \in S_8$ such that $k = 2^m$, where $m \in \mathbb{N}$ satisfies $1 \leq m \leq 7$. These are given in Table 6.1 below.

$[g]_{S_8}$	1a	. 2a	2ł	0 20	2d	. 3a	31	b 4a	a 4b	4c	4d
k	128	8 64	32	2 32	2 16	32	8	3 16	5 16	8	8
$[g]_{S_8}$	5a	6a	6b	6c	6d	6e	7a	8a	10a	12a	15a
k	8	16	4	4	8	4	2	4	4	4	2

Table 6.1:

We used GAP for programmes A and B (see Appendix A), which can also be found in [1] and [99], written in MAGMA and CAYLEY. We also used coset analysis, which is also discussed in chapter 2, to compute the conjugacy classes of elements of $2^7:S_8$. These conjugacy classes are given in Table 6.2 and the descriptions of the parameters used can also be found in [1, 99, 120]. We give programmes A and B and the table for conjugacy classes:

PROGRAMME A for $2^7:S_8$

$$\begin{split} & gap > V:= FullRowSpace(GF(2),7); \\ gap > gr1:=(OneGF(2))^*[7 \times 7 matrix group generators]; \\ & gap > gr2:=(OneGF(2))^*[7 \times 7 matrix group generators]; \\ & gap > grp:=Group(gr1,gr2); \\ & gap > Ccl:=ConjugacyClasses(grp); \\ & gap > O:=Union(Orbits(grp,V)); \\ & gap > for i in [1..22] do \\ & > Print(Representative(Ccl[i])); \\ & >w:=One(GF(q))^*[0,0,\cdots,0]; \\ & >e:=[]; \\ & >while Difference(O,e) <>[] do \\ & >d:=[]; \\ & >for x in O do; \\ >y:=[x+w+(x^*(Representative((Ccl)[i]))]; \end{split}$$

>d:=Union(d,y); > od;>Print(d); >e:=Union(d,e); >if Difference(O,e) <> [] then >w:=Representative(Difference(O,e));>fi; >od; >r:=[]; $>u:=One(GF(2))^{*}[0, 0, \cdots, 0];$ >while Difference(O,e) <> [] do >m:=[]; >for g in Centralizer(grp,Representative(Ccl[i])) do $>l:=[u^*g];$ >m:=Union(m,l); >od; >Print("A block for the vectors under the action of a centralizer"); >Print(m); >r:=Union(m,r); $>\!\!\mathrm{if}$ Difference(O,r) <> [] then >u:=Representative(Difference(O,r));>fi; > od;>Print("************** (**********************): > od;

PROGRAMME B for $2^7:S_8$

$$\begin{split} & \text{gap>V:=FullRowSpace}(GF(2),7);\\ & \text{gap>m1:=}(\text{OneGF}(2))^*[7\times7\text{ matrix group generators}];\\ & \text{gap>m2:=}(\text{OneGF}(q))^*[7\times7\text{ matrix group generators}];\\ & \text{gap>m2:=}(\text{Group}(m1,m2);\\ & \text{gap>c:=}\text{ConjugacyClasses}(m);\\ & \text{gap>g:=}\text{Representative}(\mathbf{c}[\mathbf{i}]);\\ & \text{gap>d:=}\text{One}(\text{GF}(2))^*[\alpha_1,\alpha_2,\cdots,\alpha_7];\\ & \text{gap>w:=}d+d*g+d*g^2+\cdots+d*g^{k-1}; \end{split}$$

gap>Print(w);

Table 6.2: Conjugacy Classes of $2^7:S_8$

$g \in S_8$	k	f_j	d_j	w	$[x]_{2^7:S_8}$	$ C_{2^7:S_8}(x) $
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	1A	$5\ 160\ 960$
		1	(1, 1, 0, 1, 0, 0, 1)	$\left(1,1,0,1,0,0,1\right)$	2A	$5\ 160\ 960$
1A	2^{7}	28	(1, 0, 1, 0, 1, 0, 1)	$\left(1,0,1,0,1,0,1 ight)$	2B	184 320
		28	(0, 0, 0, 1, 0, 0, 0)	$\left(0,0,0,1,0,0,0 ight)$	2C	184 320
		35	(1, 1, 1, 1, 1, 1, 1, 1)	(1, 1, 1, 1, 1, 1, 1)	2D	$147 \ 456$
		35	(1, 0, 0, 0, 0, 0, 0, 0)	$\left(1,0,0,0,0,0,0\right)$	2E	$147 \ 456$
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2F	92 160
		1	(1, 0, 0, 1, 1, 0, 1)	$\left(1,1,0,0,1,1,0 ight)$	4A	92 160
		6	(1, 0, 0, 0, 0, 1, 1)	$\left(0,1,0,0,0,0,0\right)$	4B	15 360
2A	2^{6}	6	(1, 0, 1, 1, 1, 1, 0)	$\left(0,0,0,0,0,0,0,0 ight)$	2G	15 360
		10	(1, 1, 1, 1, 1, 1, 1, 1)	$\left(0,0,1,0,0,0,0\right)$	4C	9 216
		10	(0, 0, 1, 1, 1, 0, 0)	$\left(0,0,0,0,0,0,0,0 ight)$	2H	9 216
		15	(1,0,1,0,1,0,1)	(0, 0, 0, 0, 0, 0, 0, 0)	2I	6 144
			<u>.</u>		continued	on next page

$g \in S_8$	k	f_j	d_j	w	$[x]_{2^7:S_8}(x) $	
		15	(0, 0, 0, 1, 0, 0, 0)	(0, 1, 0, 1, 1, 0, 0)	4D	6 144
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2J	12 288
		1	(0, 1, 1, 1, 0, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	2K	$12\ 288$
		1	(1, 1, 0, 1, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	2L	12 288
2B	2^{5}	1	(1, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2M	12 288
		2	(0, 0, 1, 1, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	2N	$6\ 144$
		2	(1, 1, 1, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2O	$6\ 144$
		12	(1, 0, 1, 0, 1, 0, 1)	(0, 1, 1, 0, 1, 1, 0)	4E	1 024
		12	(0, 0, 0, 1, 0, 0, 0)	(0, 1, 1, 0, 1, 1, 0)	4F	1 024
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2P	6 144
		1	(1, 1, 0, 0, 1, 0, 1)	(1, 1, 1, 1, 1, 1, 0)	4G	$6\ 144$
		1	(1, 0, 0, 0, 0, 1, 1)	(1, 0, 0, 0, 0, 1, 0)	4H	$6\ 144$
		1	(0, 1, 0, 0, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2Q	$6\ 144$
2C	2^{5}	3	(1, 0, 1, 0, 1, 0, 1)	(1, 1, 1, 1, 1, 1, 0)	4I	2048
		3	(1, 0, 0, 0, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2R	2048
		3	(0, 1, 0, 1, 0, 0, 1)	(1, 1, 0, 1, 1, 0, 0)	4J	2 048
		3	(1, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2S	2048
		8	(0, 1, 1, 1, 0, 1, 1)	(1, 1, 0, 1, 1, 0, 0)	4K	768
		8	(0, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 1, 0)	4L	768
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2T	1 536
		1	(1, 0, 0, 0, 0, 0, 1)	(0, 1, 0, 0, 0, 0, 0)	4M	1 536
		1	(0, 0, 0, 1, 1, 1, 1)	((0, 0, 1, 1, 0, 0, 0))	4N	1 536
2D	2^{4}	1	(0, 1, 0, 1, 1, 1, 0)	(0, 1, 1, 1, 1, 1, 0)	4O	1 536
		3	(1, 0, 1, 0, 1, 0, 1)	(0, 1, 1, 0, 0, 0, 0)	4P	512
		3	(0, 1, 0, 0, 1, 0, 1)	(0, 1, 1, 1, 0, 0, 0)	4Q	512
		3	(0, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	2U	512
		3	(0, 1, 0, 0, 1, 0, 0)	(1, 1, 0, 0, 1, 1, 0)	4R	512
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	3A	11 520
		1	(1, 0, 1, 0, 1, 0, 1)	(1, 1, 0, 0, 1, 0, 1)	6A	11 520
3A	2^{5}	5	(0, 1, 1, 1, 0, 1, 1)	(1, 1, 1, 1, 0, 0, 0)	6B	2 304
		5	(0, 0, 0, 1, 0, 0, 0)	(0, 1, 1, 1, 0, 1, 0)	6C	2 304
		10	(1, 0, 0, 0, 0, 1, 1)	$\left(0,1,1,0,0,0,1 ight)$	6D	$1\ 152$
		10	$\left(0,1,0,0,1,1,0\right)$	$\left(0,1,1,0,1,0,0 ight)$	6E	1 152
		1	$\left(0,0,0,0,0,0,0 ight)$	$\left(0,0,0,0,0,0,0 ight)$	3B	288
		1	(1, 0, 1, 0, 1, 0, 1)	$\left(1,0,1,0,1,0,1 ight)$	6F	288
3B	2^{3}	1	(0, 1, 1, 1, 0, 1, 1)	(0, 1, 1, 1, 0, 1, 1)	6G	288
		1	$\left(0,0,0,1,0,0,0 ight)$	$\left(1,0,0,1,0,1,0 ight)$	6H	288
		2	(1, 0, 0, 0, 0, 1, 1)	(0, 1, 1, 0, 0, 0, 1)	6I	144
		2	(1, 0, 1, 1, 1, 1, 0)	(1, 0, 1, 1, 1, 1, 0)	6J	144
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4S	1 536
		1	(1, 0, 0, 0, 0, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4T	1 536
4A	2^{4}	3	(1, 0, 1, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4U	512
		3	(1, 0, 0, 0, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4V	512
		4	(0, 1, 1, 1, 0, 1, 1)	(1, 0, 1, 0, 0, 1, 0)	8A	384
		4	(0, 0, 0, 1, 0, 0, 0)	(1, 1, 0, 1, 1, 1, 0)	8B	384
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4W	512
		1	(1, 0, 0, 0, 0, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4X	512
		1	(1, 1, 0, 0, 1, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4Y	512
4B	$ 2^4$	1	(0, 1, 0, 0, 1, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4Z	512
		2	(1,0,1,1,1,0,1)	(0, 0, 0, 0, 0, 0, 0, 0)	4AA	256
		2	(1, 1, 0, 1, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4AB	256
		4	(1, 0, 1, 0, 1, 0, 1)	(1, 1, 0, 1, 1, 1, 0)	8C	128
					continued	on next page

Table 6.2 (continued from previous page)

				F	I 87	
$g \in S_8$	k	f_j	d_j	w	$[x]_{2^7:S_8}(x) $	
		4	(0, 0, 0, 1, 0, 0, 0)	$\left(1,0,1,0,0,1,0\right)$	8D	128
		1	(0, 0, 0, 0, 0, 0, 0, 0)	$\left(0,0,0,0,0,0,0,0 ight)$	4AC	256
		1	(0, 1, 1, 1, 0, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4AD	256
4C	2^{3}	1	(1, 1, 0, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4AE	256
		1	(1,0,1,1,1,1,0)	(0, 0, 0, 0, 0, 0, 0, 0)	4AF	256
		2	(1, 0, 1, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4AG	128
		2	(0, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4AH	128
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4AI	128
		1	(1, 0, 1, 0, 1, 0, 1)	$\left(0,0,0,1,0,0,1\right)$	8E	128
		1	(0, 1, 1, 1, 0, 1, 1)	(0, 1, 0, 1, 0, 0, 0)	8F	128
4D	2^{3}	1	(0, 1, 0, 0, 1, 0, 1)	$\left(0,0,0,0,0,0,0,0 ight)$	4AJ	128
		1	(0, 1, 1, 0, 0, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	4AK	128
		1	(0, 0, 0, 1, 0, 0, 0)	$\left(0,1,0,0,0,0,0\right)$	8G	128
		1	(1, 0, 1, 1, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4AL	128
		1	(1, 0, 0, 0, 0, 1, 0)	(0, 0, 1, 0, 1, 1, 0)	8H	128
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	5A	240
5A	2^{3}	1	(0, 1, 0, 0, 1, 0, 1)	(0, 0, 0, 1, 1, 0, 0)	10A	240
		3	(1,0,1,0,1,0,1)	(1, 0, 1, 1, 1, 0, 0)	10B	80
		3	(0, 0, 0, 1, 0, 0, 0)	(1, 0, 0, 0, 1, 1, 0)	10C	80
		1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0)	6K	576
		1	(1,0,1,1,1,0,1)	(1, 0, 0, 0, 0, 1, 0)	12A	576
		1	(0, 0, 0, 0, 1, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6L	576
6A	2^{4}	1	(1, 0, 0, 1, 0, 0, 0)	(1, 1, 1, 1, 1, 0, 0)	12B	576
		3	(1, 0, 1, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6M	192
		3	(1,0,0,0,0,1,1)	(1, 1, 1, 0, 0, 0, 0)	12C	192
		3	(0, 0, 0, 1, 0, 0, 0)	(1, 0, 0, 0, 0, 1, 0)	12D	192
		3	(1,0,0,0,1,1,0)	(0, 0, 0, 0, 0, 0, 0, 0)	6N	192
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	6 <i>O</i>	192
		1	(1, 0, 1, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6P	192
6B	2^{3}	1	(1, 1, 1, 1, 1, 1, 1, 1)	(1, 1, 1, 1, 1, 0, 0)	12E	192
		1	(0, 1, 0, 0, 1, 1, 0)	(1, 1, 1, 1, 1, 1, 0)	12F	192
		2	(0, 1, 1, 1, 0, 1, 1)	(1, 0, 1, 0, 0, 1, 0)	12G	96
		2	(0, 0, 0, 1, 0, 0, 0)	(1, 1, 1, 1, 1, 0, 0)	12H	96
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	6Q	144
6C	2^{2}	1	(1, 0, 1, 0, 1, 0, 1)	(1, 1, 0, 0, 1, 1, 0)	12I	144
		1	(1, 0, 0, 0, 0, 1, 1)	(1, 1, 1, 0, 0, 0, 0)	12J	144
		1	(1, 0, 1, 1, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	6R	144
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	6S	96
		1	(0, 1, 1, 1, 0, 1, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6T	96
6D	2^{3}	1	(1, 1, 0, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6U	96
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	6V	96
		2	(1, 0, 1, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6W	48
		2	(1, 0, 0, 0, 1, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	6X	48
		1	(0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0)	6Y	48
6E	2^{2}	1	(1,0,1,0,1,0,1)	(1, 1, 0, 0, 0, 1, 0)	12K	48
		1	(0, 1, 0, 0, 1, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0)	6Z	48
		1	(0,0,0,1,0,0,0)	(0, 0, 1, 0, 0, 1, 0)	12L	48
		1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	7A	14
7A	2	1	(1,1,1,1,1,1,1)	(1, 1, 0, 1, 0, 0, 1)	14A	14
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	81	32
8A	2^{2}	1	(1,0,1,0,1,0,1)	(0, 0, 0, 0, 0, 0, 0, 0)	8J	32
		1	(0,1,1,1,0,1,1)	(0, 0, 0, 0, 0, 0, 0, 0)	8K	32
					continued	on nort noro

Table 6.2 (continued from previous page)
$g \in S_8$	k	f_j	d_j	w	$[x]_{2^7:S_8}(x) $	
		1	(0, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	8L	32
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	10D	40
10A	2^{2}	1	(1, 0, 1, 0, 1, 0, 1)	(1, 1, 0, 1, 0, 0, 1)	20A	40
		1	(0, 1, 0, 0, 1, 0, 1)	(0, 0, 1, 0, 0, 1, 0)	20B	40
		1	(0, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	10E	40
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	12M	48
12A	2^{2}	1	(1, 0, 1, 0, 1, 0, 1)	(1, 1, 0, 0, 1, 0, 1)	24A	48
		1	(0, 1, 1, 1, 0, 1, 1)	(1, 1, 0, 1, 0, 0, 1)	24B	48
		1	(0, 0, 0, 1, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	12N	48
		1	(0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	15A	30
15A	2	1	(1,1,1,1,1,1,1)	(1,1,1,1,1,0,0)	30A	30

Table 6.2 (continued from previous page)

We obtain that $2^7:S_8$ has altogether 128 conjugacy classes of elements.

6.4. The Fischer-Clifford matrices of $2^7:S_8$

From the above section, we obtained 128 conjugacy classes of elements of $2^7:S_8$. From [89] since we have 128 conjugacy classes of elements, we have 128 irreducible characters of $2^7:S_8$. By [14] when S_8 acts on $Irr(2^7)$ we get six orbits and we use programme C (see Appendix A) to show that these orbits are of lengths 1, 1, 28, 28, 35 and 35 respectively. The inertia factor groups H_i , $i = 1, \dots, 6$ are subgroups of S_8 of index 1, 1, 28, 28, 35 and 35 respectively and from the ATLAS [23] we get that $H_1 = S_8, H_2 = S_8, H_3 = S_6 \times 2, H_4 = S_6 \times 2, H_5 = (S_4 \times S_4):2$ and $H_6 = (S_4 \times S_4):2$. The fusions of the inertia factor groups $S_6 \times 2$ and $(S_4 \times S_4):2$ into S_8 are given in Table 6.3 below

$[g]_{S_6 \times 2}$	\longrightarrow	$[y]_{S_8}$	$[g]_{(S_4 \times S_4):2}$	\longrightarrow	$[y]_{S_8}$
1A		1A	1A		1A
2A		2A	2A		2B
2B		2A	2B		2D
2C		2B	2C		2A
2D		2B	2D		2C
2E		2C	2E		2B
2F		2C	2F		2D
2G		2D	3A		3A
3A		3A	3B		3B
3B		3B	4A		4A
4A		4A	4B		4C
4B		4B	4C		4B
4C		4B	4D		4D
4D		4C	4E		4D
5A		5A	4F		4C
6A		6A	6A		6B
			continue	d on ne	xt page

Table 6.3: The fusion of $S_6 \times 2$ and $(S_4 \times S_4)$:2 into S_8

Tabl	e 0.3 (continue	a nom previ	ous pa	ge)
$[g]_{S_6 \times 2}$	\longrightarrow	$[y]_{S_8}$	$[g]_{(S_4 \times S_4):2}$	\longrightarrow	$[y]_{S_8}$
6B		6A	6B		6A
6C		6B	6C		6E
6D		6C	8A		8A
6E		6D	10A		10A
6F		6E	12A		12A

Table 6.3 (continued from previous page)

From the above fusions, we are now able to obtain the Fischer-Clifford matrices of $2^7:S_8$. According to [99], all these Fischer-Clifford matrices will have integer entries. These Fischer-Clifford matrices are thus given in Table 6.4 below

Table 6.4: The Fischer-Clifford Matrices of $2^7:S_8$

M(g)	M(g)				
$M(1A) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 & 1 & -1 \\ 28 & -28 & 4 & -4 & -4 & 4 \\ 28 & 28 & 4 & 4 & -4 & -4 \\ 35 & 35 & -5 & -5 & 3 & 3 \\ 35 & -35 & -5 & 5 & 3 & -3 \end{bmatrix}$	$M(4A) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 & 1 & -1 \\ 6 & -6 & -2 & 2 & 0 & 0 \\ 6 & 6 & -2 & -2 & 0 & 0 \\ 1 & 1 & 1 & 1 & -1 & -1 \\ 1 & -1 & 1 & -1 & -$				
$M(2A) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1$	$M(3A) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 & -1 & 1 \\ 5 & -5 & 3 & -3 & -1 & 1 \\ 5 & 5 & -3 & -3 & 1 & 1 \\ 10 & -10 & -2 & 2 & 2 & -2 \\ 10 & 10 & 2 & 2 & -2 & -2 \end{bmatrix}$				
$M(2B) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & -1 & -$	$M(4C) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & 1 & -1 & -$				
$M(2C) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1 & 1$	$M(3B) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & -1 & -1$				

	M(g)	M(g)
	1 -1 1 -1 1 1 1 -1 -1	
	3 3 3 3 -1 -1 -1 -1	Γ1 1 1 1 1 1
M(2D) =		$M(5A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 3 & 3 & 1 & 1 \end{bmatrix}$
	3 -3 3 -3 -1 -1 1 1	
	2 -2 -2 2 2 2 -2 0 0	
M(4B) =	2 2 2 2 2 -2 -2 0 0	$M(6C) = \begin{bmatrix} 1 & 1 & -1 & -1 \end{bmatrix}$
	1 1 1 1 1 1 -1 -1	
	4 -4 4 -4 0 0 0 0	
	1 -1 -1 1 -1 1 1 -1	
	1 -1 1 -1 1 -1 1 -1 1 -1	
	1 1 -1 -1 1 1 -1 -1	
M(4D) =	1 - 1 - 1 1 1 -1 -1 1	$M(6B) = \begin{bmatrix} 2 & -2 & 2 & -2 & 0 & 0 \end{bmatrix}$
	1 1 1 1 -1 -1 -1 -1	$\begin{bmatrix} m(0D) = \\ 2 & 2 & -2 & -2 & 0 & 0 \end{bmatrix}$
	1 -1 1 -1 -1 -1 1 -1 1	
	1 1 -1 -1 -1 1 1	
	1 -1 -1 1 1 -1 1 -1	
	1 -1 1 -1 -1 1 1	
	3 -3 3 -3 1 1 -1 -1	1 -1 -1 1 1 -1
M(6A) =	1 1 -1 -1 -1 1 1 -1	$M(6D) = \begin{bmatrix} 1 & 1 & 1 & 1 & -1 & -1 \end{bmatrix}$
	3 3 -3 -3 1 -1 -1 1	
	3 -3 -3 3 -1 1 -1 1	
	3 3 3 3 -1 -1 -1 -1	
	1 1 1 1]	
	1 1 -1 -1	
M(6E) =	1 -1 -1 1	$M(7A) = \begin{bmatrix} 1 & -1 \end{bmatrix}$
	1 -1 1 -1	
		Γ1 1 1 1 1
	1 1 -1 -1	
M(8A) =		$M(10A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \end{bmatrix}$
M(12A) =		$ \begin{vmatrix} M(15A) = \\ 1 & -1 \end{vmatrix} $

Table 6.4 (continued from previous page)

6.5. The power maps of $2^7:S_8$

The power maps of $2^7:S_8$ are given in Table 6.5 below

$[g]_{S_8}$	$[x]_{2^7:S_8}$	2	3	5	7	$[g]_{S_8}$	$[x]_{2^7:S_8}$	2	3	5	7
	1A						2F	1A			
	2A	1A					4A	2C			
	2B	1A					4B	2C			
1A	2C	1A				2A	2G	1A			
	2D	1A					4C	2C			
	2E	1A					2H	1A			
							2I	1A			
							4D	2C			
	2J	1A					2P	1A			
	2K	1A					4G	2C			
	2L	1A					4H	2C			
	2M	1A					2Q	1A			
2B	2N	1A				2C	4I	2C			
	2O	1A					2R	1A			
	4E	2C					4J	2C			
	4F	2C					2S	1A			
							4K	2C			
							4L	2C			
	2T	1A					3A		1A		
	4M	2C					6A	3A	2A		
	4N	2C					6B	3A	2A		
2D	4O	2C				3A	6C	3A	2C		
	4P	2C					6D	3A	2B		
	4Q	2C					6E	3A	2C		
	2U	1A									
	4R	20					8 D		1.4		
	45	2P					3B 8 F	0 D			
	41	2P				20	6F	3B 9 D	2B		
4A	40 4V	2P				3B	0G CH	35	20		
	4V	2P 4 I					0H C I	35	2A		
	8A 8 D	4J 4 J					01 G I	3B 2D	20		
	0D 4W	4J 9 D						3D 9 I	2A		
	4W 4 X	$\frac{21}{2P}$					4AC 4AD	25 21			
	4Y	$\frac{21}{2P}$					4AE	21			
4B	47	$\frac{21}{2P}$				4C	4AF	$\frac{25}{2L}$			
1D	4 A A	$\frac{21}{2P}$				10	4AG	$\frac{2L}{2L}$			
	4AB	$\frac{21}{2P}$					4 <i>AH</i>	$\frac{2L}{2L}$			
	8C	4.J						20			
	8D	4J									
	4AI	2P					6K	3A	2F		
	8E	4J					12A	6C	4C		
	8F	4J					6L	3A	2H		
4D	4AJ	2R				6A	12B	6C	4A		
	4AK	2R					6M	3A	2I		
						1		conti	nued or	next	page

Table 6.5: The Power Maps of elements of $2^7: S_8$

$[g]_{S_8}$	$[x]_{2^7:S_8}$	2	3	5	7	$[g]_{S_8}$	$[x]_{2^7:S_8}$	2	3	5	7
	8G	4J					12C	6C	4B		
	4AL	2P					12D	6C	4A		
	8H	4J					6N	3A	2G		
	5A			1A			6O	3A	2P		
	10A	5A		2A			6P	3A	2Q		
5A	10C	5A		2C		6B	12E	6C	4H		
	10C	5A		2C			12F	6C	4G		
							12G	6C	4K		
							12H	6C	4L		
	6Q	3B	2F				6S	3B	2J		
	12I	6I	4D				6T	3B	2K		
6C	12J	6I	4C			6D	6U	3B	2L		
	6R	3B	2H				6V	3B	2M		
							6W	3B	2L		
							6X	3B	2M		
	6Y	3B	2T				7A				1A
6E	12K	6G	4L			7A	14A	7A			2A
	6Z	3B	2U								
	12L	6G	4M								
	8I	4AD					10D	5A		2F	
8A	8J	4AE				10A	20A	10B		4D	
	8K	4AE					20B	10B		4B	
	8L	4AD					10E	5A		2H	
	12M	6Q	4S				15A		5A	3A	
12A	24A	12E	8A			15A	30A	15A	10A	6A	
	24B	12E	8B								
	12N	6O	4S								

Table 6.5 (continued from previous page)

6.6. The fusion of $2^7:S_8$ into $2^7:SP_6(2)$

The group $2^7:SP_6(2)$ is a maximal subgroup of \overline{Fi}_{22} containing a maximal subgroup of the form $2^7:S_8$. To determine the fusion of $2^7:S_8$ into $2^7:SP_6(2)$, we shall use the fusion of S_8 into SP(6,2), Theorem 7.5.1 from [99] and the power maps of both groups, listed in Table 6.1 above and in [41] respectively. The fusion of S_8 into SP(6,2) is given in Table 6.6.

		(<i>,</i>	
$[g]_{S_8}$	$\longrightarrow [y]_{SP(6,2)}$	$[g]_{S_8}$	\longrightarrow	$[y]_{SP(6,2)}$
1A	1A	2A		2A
2B	2C	2C		2D
2D	2B	3A		3A
3B	3C	4A		4C
4B	4E	4C		4D
4D	4B	5A		5A
6A	6A	6B		6D
6C	6E	6D		6F
6E	6G	7A		7A
8A	8B	10A		10A
12A	12B	15A		15A

Table 6.6: The Fusion of S_8 into SP(6,2)

Thus the fusion of $2^7:S_8$ into $2^7:SP_6(2)$ is given in Table 6.7 below.

$[g]_{S_8}$	$[x]_{2^7:S_8}$	\longrightarrow $[y]_{2^7:SP(6,2)}$	$[g]_{S_8}$	$[x]_{2^7:S_8}$	\longrightarrow $[y]_{2^7:SP(6,2)}$
	1A	1A		2F	2D
	2A	2B		4A	4C
	2B	2A		4B	4A
1A	2C	2C	2A	2G	2F
	2D	2B		4C	4B
	2E	2C		2H	2F
				2I	2E
				4D	4C
	2J	2G		2P	2K
	2K	2H		4G	4F
	2L	2I		4H	4H
	2M	2J		2Q	2M
2B	2N	2I	2C	4I	4G
	2O	2J		2R	2L
	4E	4D		4J	4H
	4F	4E		2S	2M
				4K	4I
				4L	4J
	2T	2N		3A	3A
	4M	4K		6A	6A
	4N	4L		6B	6A
2D	40	4O	3A	6C	6C
	4P	4M		6D	6B
	4Q	4N		6E	6C
	2U	2O			
	4R	4P			
	3B	3C		4S	4Y
	6F	6E		4T	4Z
3B	6G	6F	4A	4U	4AA
	6H	6G		4V	4AB
				C	ontinued on next page

Table 6.7: The fusion of $2^7:S_8$ into $2^7:SP(6,2)$

$[g]_{S_8}$	$[x]_{2^7:S_8}$	\longrightarrow $[y]_{2^7:SP(6,2)}$	$[g]_{S_8}$	$[x]_{2^7:S_8}$	\longrightarrow	$[y]_{2^7:SP(6,2)}$
	6I	6F		8A		8C
	6J	6G		8B		8D
	4W	4U		4AC		4AC
	4X	4V		4AD		4AD
	4Y	4W		4AE		4AE
4B	4Z	4X	4C	4AE		4AF
	4AA	4W		4AG		4AG
	4AB	4X		4AH		4AH
	8C	8A				
	8D	8B				
	4AI	4AI		5A		5A
	8E	8E		10A		10A
	8F	8F		10B		10B
4D	4AJ	4AJ	5A	10C		10C
	4AK	4AK				
	8G	8G				
	4AL	4AL				
	8H	8H				
	6K	6H		6 <i>O</i>		6O
	12A	12A		6P		6P
	6L	6J		12E		12F
6A	12B	12C	6B	12F		12G
	6M	6I		12G		12H
	12C	12B		12H		12I
	12D	12C				
	6N	6J				
	6Q	6U		6R		6Q
	12I	12J		6S		6R
6C	12J	12K	6D	6T		6S
	6R	6V		6U		6T
				6V		6S
				6W		6T
	6X	6W		7A		7A
6E	12K	12L	7A	14A		14A
	6Y	6X				
	12M	12M				
	8I	8I		10D		10D
8A	8J	8J	10A	20A		20A
	8K	8K		20B		20B
	8L	8L		10E		10E
	12N	12P		15A		15A
12A	24A	24C	15A	30A		30A
	24B	24D				
	120	12Q				

Table 6.7 (continued from previous page)

We use Fisher-Clifford matrices and partial character tables of inertia factor groups to compile the character table. We rewrite the character table in the GAP format and we then use programme E (see Appendix A) to check its validity and its consistency concerning the character table orthogonalities. This character table is given in Table 6.8

PROGRAMME E for $2^7:S_8$

$$\begin{split} & gap>ct:=fuction()local ct;ct:=rec(); \\ > ct.SizesCentralizers:=[128 Centralizer Orders]; \\ > ct.OrdersClassRepresentatives:=[128 Class Representatives Orders]; \\ > ct.Irr:=[[128 \times 128 irreducibles]]; \\ > ct.UnderlyingCharacteristic:=0;ct.Id:=\overline{G}; \\ > ConvertToLibraryCharacterTable NC(ct);return ct;end;ct:=ct(); \\ & gap>SetInfoLevel(InfoCharacterTable,2); \\ & gap>IsInternallyConsistent(ct); \\ & gap>PossiblePowerMaps(ct,p); (p-prime divisor of \overline{G}). \end{split}$$

Note: Although this was a good exercise to begin with, on closer inspection, we found out that the group $2^7:S_8$ was actually the group $(2^6:S_8) \times 2$ which is a direct product.

				1A						2A				
	1A	2A	2B	2C	2D	2E	2F	4A	4B	2G	4C	2H	2I	4D
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1
χ_3	7	7	7	7	7	7	-5	-5	-5	-5	-5	-5	-5	-5
χ_4	7	7	7	7	7	7	5	5	5	5	5	5	5	5
χ_5	14	14	14	14	14	14	-4	-4	-4	-4	-4	-4	-4	-4
χ_6	14	14	14	14	14	14	4	4	4	4	4	4	4	4
χ_7	20	20	20	20	20	20	-10	-10	-10	-10	-10	-10	-10	-10
χ_8	20	20	20	20	20	20	10	10	10	10	10	10	10	10
χ_9	21	21	21	21	21	21	-9	-9	-9	-9	-9	-9	-9	-9
χ_{10}	21	21	21	21	21	21	9	9	9	9	9	9	9	9
χ_{11}	28	28	28	28	28	28	-10	-10	-10	-10	-10	-10	-10	-10
χ_{12}	28	28	28	28	28	28	10	10	10	10	10	10	10	10
χ_{13}	35	35	35	35	35	35	-5	-5	-5	-5	-5	-5	-5	-5
χ_{14}	35	35	35	35	35	35	5	5	5	5	5	5	5	5
χ_{15}	42	42	42	42	42	42	0	0	0	0	0	0	0	0
χ_{16}	56	56	56	56	56	56	-4	-4	-4	-4	-4	-4	-4	-4
χ_{17}	56	56	56	56	56	56	4	4	4	4	4	4	4	4
χ_{18}	64	64	64	64	64	64	-16	-16	-16	-16	-16	-16	-16	-16
χ_{19}	64	64	64	64	64	64	16	16	16	16	16	16	16	16
χ_{20}	70	70	70	70	70	70	-10	-10	-10	-10	-10	-10	-10	-10
χ_{21}	70	70	70	70	70	70	10	10	10	10	10	10	10	10
χ_{22}	90	90	90	90	90	90	0	0	0	0	0	0	0	0
χ_{23}	1	-1	1	-1	1	-1	1	-1	1	-1	1	-1	-1	1
χ_{24}	1	-1	1	-1	1	-1	-1	1	-1	1	-1	1	1	-1
χ_{25}	7	-7	7	-7	7	-7	-5	5	-5	5	-5	5	5	-5
χ_{26}	7	-7	7	-7	7	-7	5	-5	5	-5	5	-5	-5	5
χ_{27}	14	-14	14	-14	14	-14	-4	4	-4	4	-4	4	4	-4
χ_{28}	14	-14	14	-14	14	-14	4	-4	4	-4	4	-4	-4	4
χ_{29}	20	-20	20	-20	20	-20	-10	10	-10	10	-10	10	10	-10
χ_{30}	20	-20	20	-20	20	-20	10	-10	10	-10	10	-10	-10	10
χ_{31}	21	-21	21	-21	21	-21	-9	9	-9	9	-9	9	9	-9
χ_{32}	21	-21	21	-21	21	-21	9	-9	9	-9	9	-9	-9	9
χ_{33}	28	-28	28	-28	28	-28	-10	10	-10	10	-10	10	10	-10
χ_{34}	28	-28	28	-28	28	-28	10	-10	10	-10	10	-10	-10	10
χ_{35}	35	-35	35	-35	35	-35	-5	5	-5	5	-5	5	5	-5
χ_{36}	35	-35	35	-35	35	-35	5	-5	5	-5	5	-5	-5	5
χ_{37}	42	-42	42	-42	42	-42	0	0	0	0	0	0	0	0
χ_{38}	56	-56	56	-56	56	-56	-4	4	-4	4	-4	4	4	-4
χ_{39}	56	-56	56	-56	56	-56	4	-4	4	-4	4	-4	-4	4
χ_{40}	64	-64	64	-64	64	-64	-16	16	-16	16	-16	16	16	-16
χ_{41}	64	-64	64	-64	64	-64	16	-16	16	-16	16	-16	-16	16
χ_{42}	70	-70	70	-70	70	-70	-10	10	-10	10	-10	10	10	-10
χ_{43}	70	-70	70	-70	70	-70	10	-10	10	-10	10	-10	-10	10
χ_{44}	90	-90	90	-90	90	-90	0	0	0	0	0	0	0	0

Table 6.8: Character table of $2^7:S_8$

		2B				2C												
	2J	2K	2L	2M	2N	20	4E	4F	2P	4G	4H	2Q	4I	2R	4J	2S	4K	4L
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_3	-1	-1	-1	-1	-1	-1	-1	-1	3	3	3	3	3	3	3	3	3	3
χ_4	-1	-1	-1	-1	-1	-1	-1	-1	3	3	3	3	3	3	3	3	3	3
χ_5	6	6	6	6	6	6	6	6	2	2	2	2	2	2	2	2	2	2
χ_6	6	6	6	6	6	6	6	6	2	2	2	2	2	2	2	2	2	2
χ_7	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4
χ_8	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4	4
χ_9	-3	-3	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1	1	1	1	1
χ_{10}	-3	-3	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1	1	1	1	1
χ_{11}	-4	-4	-4	-4	-4	-4	-4	-4	4	4	4	4	4	4	4	4	4	4
χ_{12}	-4	-4	-4	-4	-4	-4	-4	-4	4	4	4	4	4	4	4	4	4	4
χ_{13}	3	3	3	3	3	3	3	3	-5	-5	-5	-5	-5	-5	-5	-5	-5	-5
χ_{14}	3	3	3	3	3	3	3	3	-5	-5	-5	-5	-5	-5	-5	-5	-5	-5
χ_{15}	-6	-6	-6	-6	-6	-6	-6	-6	2	2	2	2	2	2	2	2	2	2
χ_{16}	8	8	8	8	8	8	8	8	0	0	0	0	0	0	0	0	0	0
χ_{17}	8	8	8	8	8	8	8	8	0	0	0	0	0	0	0	0	0	0
χ_{18}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{19}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{20}	-2	-2	-2	-2	-2	-2	-2	-2	2	2	2	2	2	2	2	2	2	2
χ_{21}	-2	-2	-2	-2	-2	-2	-2	-2	2	2	2	2	2	2	2	2	2	2
χ_{22}	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6	-6
χ_{23}	1	-1	-1	1	-1	1	-1	1	1	-1	-1	1	-1	1	-1	1	1	-1
χ_{24}	1	-1	-1	1	-1	1	-1	1	1	-1	-1	1	-1	1	-1	1	1	-1
χ_{25}	-1	1	1	-1	1	-1	1	-1	3	-3	-3	3	-3	3	-3	3	3	-3
χ_{26}	-1	1	1	-1	1	-1	1	-1	3	-3	-3	3	-3	3	-3	3	3	-3
χ_{27}	6	-6	-6	6	-6	6	-6	6	2	-2	-2	2	-2	2	-2	2	2	-2
χ_{28}	6	-6	-6	6	-6	6	-6	6	2	-2	-2	2	-2	2	-2	2	2	-2
χ_{29}	4	-4	-4	4	-4	4	-4	4	4	-4	-4	4	-4	4	-4	4	4	-4
χ_{30}	4	-4	-4	4	-4	4	-4	4	4	-4	-4	4	-4	4	-4	4	4	-4
χ_{31}	-3	3	3	-3	3	-3	3	-3	1	-1	-1	1	-1	1	-1	1	1	-1
χ_{32}	-3	3	3	-3	3	-3	3	-3	1	-1	-1	1	-1	1	-1	1	1	-1
χ_{33}	-4	4	4	-4	4	-4	4	-4	4	-4	-4	4	-4	4	-4	4	4	-4
χ_{34}	-4	4	4	-4	4	-4	4	-4	4	-4	-4	4	-4	4	-4	4	4	-4
χ_{35}	3	-3	-3	3	-3	3	-3	3	-5	5	5	-5	5	-5	5	-5	-5	5
χ_{36}	3	-3	-3	3	-3	3	-3	3	-5	5	5	-5	5	-5	5	-5	-5	5
χ_{37}	-6	6	6	-6	6	-6	6	-6	2	-2	-2	2	-2	2	-2	2	2	-2
χ_{38}	8	-8	-8	8	-8	8	-8	8	0	0	0	0	0	0	0	0	0	0
χ_{39}	8	-8	-8	8	-8	8	-8	8	0	0	0	0	0	0	0	0	0	0
χ_{40}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{41}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{42}	-2	2	2	-2	2	-2	2	-2	2	-2	-2	2	-2	2	-2	2	2	-2
χ_{43}	-2	2	2	-2	2	-2	2	-2	2	-2	-2	2	-2	2	-2	2	2	-2
χ_{44}	-6	6	6	-6	6	-6	6	-6	-6	6	6	-6	6	-6	6	-6	-6	6

The character table of $2^7:S_8(\text{continued})$

				2D							3A				3B	
	2T	4M	4N	4O	4P	4Q	2U	4R	3A	6A	6B	6C	6D	6E	3B	6F
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	-1	-1	-1	-1	-1	-1	-1	-1	1	1	1	1	1	1	1	1
χ_3	-1	-1	-1	-1	-1	-1	-1	-1	4	4	4	4	4	4	1	1
χ_4	1	1	1	1	1	1	1	1	4	4	4	4	4	4	1	1
χ_5	0	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1	2	2
χ_6	0	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1	2	2
χ_7	-2	-2	-2	-2	-2	-2	-2	-2	5	5	5	5	5	5	-1	-1
χ_8	2	2	2	2	2	2	2	2	5	5	5	5	5	5	-1	-1
χ_9	3	3	3	3	3	3	3	3	6	6	6	6	6	6	0	0
χ_{10}	-3	-3	-3	-3	-3	-3	-3	-3	6	6	6	6	6	6	0	0
χ_{11}	-2	-2	-2	-2	-2	-2	-2	-2	1	1	1	1	1	1	1	1
χ_{12}	2	2	2	2	2	2	2	2	1	1	1	1	1	1	1	1
χ_{13}	3	3	3	3	3	3	3	3	5	5	5	5	5	5	2	2
χ_{14}	-3	-3	-3	-3	-3	-3	-3	-3	5	5	5	5	5	5	2	2
χ_{15}	0	0	0	0	0	0	0	0	-6	-6	-6	-6	-6	-6	0	0
χ_{16}	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-1	-1
χ_{17}	4	4	4	4	4	4	4	4	-4	-4	-4	-4	-4	-4	-1	-1
χ_{18}	0	0	0	0	0	0	0	0	4	4	4	4	4	4	-2	-2
χ_{19}	0	0	0	0	0	0	0	0	4	4	4	4	4	4	-2	-2
χ_{20}	2	2	2	2	2	2	2	2	-5	-5	-5	-5	-5	-5		1
χ_{21}	-2	-2	-2	-2	-2	-2	-2	-2	-5	-5	-5	-5	-5	-5		1
χ_{22}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{23}		-1	1	-1	1	1	-1	-1	1	-1	-1	1	-1	1		-1
χ_{24}	-1	1	-1	1	-1	-1	1	1	1	-1	-1	1	-1	1		-1
χ_{25}	-1	1	-1	1	-1	-1	1	1	4	-4	-4	4	-4	4		-1
χ_{26}		-1	1	-1	1	1	-1	-1	4	-4	-4	4	-4	4		-1
χ_{27}	0	0	0	0	0	0	0	0	-1	1	1	-1	1	-1		-2
χ_{28}		0	0	0	0	0	0	0	-1	1	1	-1	1	-1		-2
χ_{29}	-2	2	-2	2	-2	-2	2	2	5	-5	-5	5	-5 -	5	-1	1
X30	2	-2	2	-2	2	2	-2	-2	D C	-5	-5	о С	-5 C	о С	-1	1
X31	3 9	-3 9	ა ი	-3 9	ა ი	ა ი	-3 9	-3 9	0	-0 6	-0 6	0	-0 6	0		0
X32	-0 -0	ა	-ə 0	ა	-ə 0	-ა ი	ა	ა	1	-0 1	-0 1	1	-0 1	1		1
X33	-2	∠ _?	-2 2	_2 _2	-2 2	-2 2	∠ _9	⊿ _ว	1	-1 _1	-1 _1	1	-1 _1	1		-1 _1
X34	2	-2	2 9	-2	2 9	⊿ 9	-2 9	-2 9	L K	-1 K	-1 K	E I	-1	L K	1	-1 0
X35	3 2	-ə 2	ა _ვ	-ə 2	ა _?	ა _ვ	-ə 2	-0 2	5	-ə _5	-ə _5	0 5	-ə _5	0 5		-2 _9
X36	-3	о 0	-3 0	О		-ა 0	ა ი	О	6	-0 6	-0 6	5	-0 6	5		-2
X37		4	4	4	4	4	4	4	-0	4	4	-0	4	-0		1
X38	-4	-4 _1	-4 1	-4 _1	-4 1	-4 1	4 _1	4 _1	-4	4	4	-4 _1	4 1	-4 _1	-1 _1	1
X39	4	-4 0	4 0	-4	4 0	4 0	-4 0	-4 0	-4	4	4 1	-4	- 1	-4	-1 0	1 9
X40		0	0	0	0	0	0	0	4	-4 _1	-4 _1	4	-4 _1	4 1	-2	∠ ງ
X41	9	_0	0 9	_9	0 0	บ จ	_0	ບ _ງ	4 _5	-4 5	-4	4 _5	-4	-4 _5	1	∠ _1
X42	_2	-2 2	_2 _2	-2 2	_2 _2	_2 _2	-2 9	-4 2	-5	5	5	-0 _5	5	-0 -5		-1 _1
X43	-2	∠ 0	-2	2 0	-2	-2	2 0	∠ 0	-5	0 0	0	-0 0	0	-0 0		-1
X44	U	U	U	U	U	U	U	U	U	0	U	U	U	U		U

The character table of $2^7:S_8(\text{continued})$

		3B					4A						4B			
	6G	6H	6I	6J	4S	4T	4U	V	8A	8B	4W	4X	4Y	4Z	4AA	4AB
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
<i>χ</i> 3	1	1	1	1	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1
χ_4	1	1	1	1	3	3	3	3	3	3	-1	-1	-1	-1	-1	-1
χ_5	2	2	2	2	2	2	2	2	2	2	-2	-2	-2	-2	-2	-2
χ_6	2	2	2	2	-2	-2	-2	-2	-2	-2	2	2	2	2	2	2
χ_7	-1	-1	-1	-1	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2
χ_8	-1	-1	-1	-1	2	2	2	2	2	2	2	2	2	2	2	2
χ_9	0	0	0	0	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1
χ_{10}	0	0	0	0	3	3	3	3	3	3	-1	-1	-1	-1	-1	-1
χ_{11}	1	1	1	1	2	2	2	2	2	2	2	2	2	2	2	2
χ_{12}	1	1	1	1	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2
χ_{13}	2	2	2	2	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{14}	2	2	2	2	1	1	1	1	1	1	1	1	1	1	1	1
χ_{15}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{16}	-1	-1	-1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{17}	-1	-1	-1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{18}	-2	-2	-2	-2	0	0	0	0	0	0	0	0	0	0	0	0
χ_{19}	-2	-2	-2	-2	0	0	0	0	0	0	0	0	0	0	0	0
χ_{20}	1	1	1	1	4	4	4	4	4	4	0	0	0	0	0	0
χ_{21}	1	1	1	1	-4	-4	-4	-4	-4	-4	0	0	0	0	0	0
X22	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{23}	-1	1	-1	1		-1	1	-1	1	-1	1	-1	-1	1	-1	1
χ_{24}	-1	1	-1	1	-1	1	-1	1	-1	1	-1	1	1	-1	1	-1
χ_{25}	-1	1	-1	1	-3	3	-3	3	-3	3		-1	-1	1	-1	1
χ_{26}	-1	1	-1	1	3	-3	3	-3	3	-3	-1	1	1	-1	1	-1
χ_{27}	-2	2	-2	2	2	-2	2	-2	2	-2	-2	2	2	-2	2	-2
χ_{28}	-2	2	-2	2	-2	2	-2	2	-2	2	2	-2	-2	2	-2	2
X29	1	-1	1	-1	-2	2	-2	2	-2	2	-2	2	2	-2	2	-2
X30	1	-1	L L	-1		-2 2	∠ _?	-2 2	2 _3	-2 2		-2 _1	-2 _1	2 1	-2 _1	∠ 1
X31	0	0	0	0	נ- ק	ა _ვ	-ა ა	ວ _ຈ	-0 2	ວ _?	_1	-1 1	-1 1	_1	-1 1	_1
X32	_1	1	_1	1	- J - J	-0 _9	5 9	-0 _9	5 9	-0 _9	2	_2		-1 0	_9	-1 9
X33	_1	1 1	-1 _1	1 1		-2 9	_2 _2	-2 9	_2 _2	-2		-2 9	-2 9	_2 _2	-2	_2 _2
134 Var		1 9	_1 _2	1 9	_1	2 1	-2 _1	1	_1	1	_1	1	2 1	-2 _1	2 1	_1
135 V26	-2	2	-2	2	1	_1	-1	_1	1	_1	1	-1	_1	-1	-1	1
130 V97	0	0	0	0		0	0	0	0	0		0	0	0	0	0
131 1 1 200	1	-1	1	-1		0	0	0	0	0	0	0	0	0	0	0
100 100	1	-1	1	-1		0	0	0	0	0	0	0	Ő	0	0	0
×40	2	-2	2	-2		0	0	0	0	0	0	0	Ő	0	0	0
χ_{41}	2	-2	2	-2	0	0	0	0	0	0	0	0	0	0	0	0
×41	-1	1	-1	1		-4	4	-4	4	-4	0	0	0	0	0	0
χ ₄₃	-1	1	-1	1	-4	4	-4	4	-4	4	0	0	0	0	0	0
χ_{44}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^7:S_8(\text{continued})$

	4B				4C							4D				-
	8C	8D	4AC	4AD	4AE	4AF	4AG	4AH	4AI	8E	8MF	4AJ	4AK	8G	4AL	8H
<i>χ</i> 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	-1	-1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ3	1	1	-1	-1	-1	-1	-1	-1	1	1	1	1	1	1	1	1
χ_4	-1	-1	-1	-1	-1	-1	-1	-1	1	1	1	1	1	1	1	1
χ_5	-2	-2	2	2	2	2	2	2	0	0	0	0	0	0	0	0
χ_6	2	2	2	2	2	2	2	2	0	0	0	0	0	0	0	0
χ7	-2	-2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X8	2	2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X9	1	1	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{10}	-1	-1	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1
X11	2	2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{12}	-2	-2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{13}	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{14}	1	1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{15}	0	0	2	2	2	2	2	2	-2	-2	-2	-2	-2	-2	-2	-2
χ_{16}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{17}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{18}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X19	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{20}	0	0	-2	-2	-2	-2	-2	-2	0	0	0	0	0	0	0	0
χ_{21}	0	0	-2	-2	-2	-2	-2	-2	0	0	0	0	0	0	0	0
X22	0	0	2	2	2	2	2	2	2	2	2	2	2	2	2	2
χ_{23}	-1	1	1	1	-1	-1	1	-1	1	-1	1	-1	1	-1	1	-1
χ_{24}	1	-1	1	1	-1	-1	1	-1	1	-1	1	-1	1	-1	1	-1
χ_{25}	-1	1	-1	-1	1	1	-1	1		-1	1	-1	1	-1	1	-1
χ_{26}	1	-1	-1	-1	1	1	-1	1	1	-1	1	-1	1	-1	1	-1
χ_{27}	2	-2	2	2	-2	-2	2	-2	0	0	0	0	0	0	0	0
χ_{28}	-2	2	2	2	-2	-2	2	-2		0	0	0	0	0	0	0
χ_{29}	2	-2	0	0	0	0	0	0		0	0	0	0	0	0	0
X30	-2	2	0	0	0	0	0	0		1	0	0	0	1	0	1
X31	-1	1	1	1	-1	-1	1	-1	-1	1	-1	1	-1	1	-1	1
X32	1	-1		1	-1	-1	1	-1	-1	1	-1	1	-1	1	-1	1
X33	-2	2	0	0	0	0	0	0		0	0	0	0	0	0	0
X34	1	-2	1	1	1	1	1	1		1	1	1	1	1	1	1
X35	1	-1	-1	-1 1	1	1	-1	1	-1	1	-1 1	1	-1	1	-1	1
X36	-1	0	-1	-1	1	1 9	-1	1	-1	1 2	-1	1	-1	1 2	-1	1 2
X37	0	0		∠ 0	-2 0	-2	∠ 0	-2 0		∠ 0	-2 0	2 0	-2 0	∠ 0	-2	2 0
X38	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0
X39	0	0		0	0	0	0	0		0	0	0	0	0	0	0
X40	0	0		0	0	0	0	0		0	0	0	0	0	0	0
X41	0	0	_0	_9	0 9	0 0	_9	0		0	0	0	0	0	0	0
X42	0	0	- <u>-</u> 2	-2 _2	∠ ?	∠ ?	-∠ _?	∠ ?		0	0	0	0	0	0	0
×43	0	0	2	-2		_2 _2	-2	?	2	_2	0 9	_2	0 9	_2	2	_2
X44		U	4	4	-2	-2	4	-2	4	-2	4	-2	2	-2	2	-2

The character table of $2^7:S_8(\text{continued})$

		5A					6A							6B		
	5A	10A	10B	10C	6K	12A	6L	12B	6M	12C	12D	6N	6O	6P	12E	12F
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1	1	1	1	1
χ_3	2	2	2	2	-2	-2	-2	-2	-2	-2	-2	-2	0	0	0	0
χ_4	2	2	2	2	2	2	2	2	2	2	2	2	0	0	0	0
χ_5	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_6	-1	-1	-1	-1	1	1	1	1	1	1	1	1	-1	-1	-1	-1
χ_7	0	0	0	0	-1	-1	-1	-1	-1	-1	-1	-1	1	1	1	1
χ_8	0	0	0	0	1	1	1	1	1	1	1	1	1	1	1	1
χ_9	1	1	1	1	0	0	0	0	0	0	0	0	-2	-2	-2	-2
χ_{10}	1	1	1	1	0	0	0	0	0	0	0	0	-2	-2	-2	-2
χ_{11}	-2	-2	-2	-2	-1	-1	-1	-1	-1	-1	-1	-1		1	1	1
χ_{12}	-2	-2	-2	-2	1	1	1	1	1	1	1	1	1	1	1	1
χ_{13}	0	0	0	0	1	1	1	1	1	1	1	1		1	1	1
χ_{14}	0	0	0	0	-1	-1	-1	-1	-1	-1	-1	-1		1	1	1
χ_{15}		2	2	1	0	0	0	0	0	0	0	0		2	2	2
χ_{16}	1	1	1	1	2	2	2	2	2	2	⊿ 0	2	0	0	0	0
χ_{17}	1	1	1	1	-2	-2	-2 2	-2	-2	-2	-2	-2		0	0	0
X18	-1	-1	-1	-1	2	2	2	2	2	2	2	2		0	0	0
X19 X00	-1	-1	-1	-1	-2	-2	-2	-2	-2	-2	-2	-2	-1	-1	-1	-1
X20	0	0	0	0	1	-1	1	-1	1	-1	1	-1	-1	-1	-1	-1
X21 X22	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ ₂₂	1	-1	-1	1	1	-1	-1	1	1	-1	1	-1	1	-1	-1	1
χ ₂₄	1	-1	-1	1	-1	1	1	-1	-1	1	-1	1	1	-1	-1	1
χ ₂₅	2	-2	-2	2	-2	2	2	-2	-2	2	-2	2	0	0	0	0
X26	2	-2	-2	2	2	-2	-2	2	2	-2	2	-2	0	0	0	0
χ ₂₇	-1	1	1	-1	-1	1	1	-1	-1	1	-1	1	-1	1	1	-1
X28	-1	1	1	-1	1	-1	-1	1	1	-1	1	-1	-1	1	1	-1
χ_{29}	0	0	0	0	-1	1	1	-1	-1	1	-1	1	1	-1	-1	1
χ_{30}	0	0	0	0	1	-1	-1	1	1	-1	1	-1	1	-1	-1	1
χ_{31}	1	-1	-1	1	0	0	0	0	0	0	0	0	-2	2	2	-2
χ_{32}	1	-1	-1	1	0	0	0	0	0	0	0	0	-2	2	2	-2
χ_{33}	-2	2	2	-2	-1	1	1	-1	-1	1	-1	1	1	-1	-1	1
χ_{34}	-2	2	2	-2	1	-1	-1	1	1	-1	1	-1	1	-1	-1	1
χ_{35}	0	0	0	0	1	-1	-1	1	1	-1	1	-1	1	-1	-1	1
χ_{36}	0	0	0	0	-1	1	1	-1	-1	1	-1	1	1	-1	-1	1
χ_{37}	2	-2	-2	2	0	0	0	0	0	0	0	0	2	-2	-2	2
χ_{38}	1	-1	-1	1	2	-2	-2	2	2	-2	2	-2	0	0	0	0
χ_{39}	1	-1	-1	1	-2	2	2	-2	-2	2	-2	2	0	0	0	0
χ_{40}	-1	1	1	-1	2	-2	-2	2	2	-2	2	-2	0	0	0	0
χ_{41}	-1	1	1	-1	-2	2	2	-2	-2	2	-2	2	0	0	0	0
χ_{42}	0	0	0	0	-1	1	1	-1	-1	1	-1	1	-1	1	1	-1
χ_{43}	0	0	0	0	1	-1	-1	1	1	-1	1	-1	-1	1	1	-1
χ_{44}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^7:S_8(\text{continued})$

	6B			6C					6D					6E		
	12G	12H	6Q	12I	12J	6R	6S	6T	6U	6V	6W	6X	6Y	12K	6Z	12L
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	1	1	1	1
<i>χ</i> з	0	0	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_4	0	0	-1	-1	-1	-1	1	1	1	1	1	1	-1	-1	-1	-1
χ_5	-1	-1	2	2	2	2	0	0	0	0	0	0	0	0	0	0
χ_6	-1	-1	-2	-2	-2	-2	0	0	0	0	0	0	0	0	0	0
χ_7	1	1	-1	-1	-1	-1	1	1	1	1	1	1	1	1	1	1
χ_8	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1	1	1	1	1
χ_9	-2	-2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{10}	-2	-2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{11}	1	1	-1	-1	-1	-1	1	1	1	1	1	1	-1	-1	-1	-1
χ_{12}	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{13}	1	1	-2	-2	-2	-2	0	0	0	0	0	0	0	0	0	0
χ_{14}	1	1	2	2	2	2	0	0	0	0	0	0	0	0	0	0
χ_{15}	2	2	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{16}	0	0	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{17}	0	0	1	1	1	1		1	1	1	1	1	-1	-1	-1	-1
χ_{18}	0	0	2	2	2	2	0	0	0	0	0	0	0	0	0	0
χ_{19}	0	0	-2	-2	-2	-2	0	0	0	0	0	0	0	0	0	0
χ_{20}	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1		1	1	1
χ_{21}	-1	-1	1	1	1	1		1	1	1	1	1		1	1	1
χ_{22}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{23}	-1	1	1	1	-1	-1	1	-1	-1	1	-1	1		1	-1	-1
χ_{24}	-1	1	-1	-1	1	1	-1	1	1	-1	1	-1		1	-1	-1
χ_{25}	0	0	1	1	-1	-1	-1	1	1	-1	1	-1	-1	-1	1	1
χ_{26}	1	1	-1	-1	1	1		-1	-1	1	-1	1	-1	-1	1	1
X27	1	-1	2	2	-2	-2		0	0	0	0	0		0	0	0
X28	1	-1	-2	-2	2 1	4	1	1	1	1	1	1		1	1	1
X29	-1	1	-1	-1	1	1	1	-1	-1	1	-1	1	1	1	-1	-1
X30	-1 9	1 _9		1	-1	-1	0	0	1	-1	1	-1		1	-1 0	-1
X31	2	-2 _9		0	0	0	0	0	0	0	0	0		0	0	0
X32	-1	-2	_1	_1	1	1	1	-1	-1	1	-1	1	_1	_1	1	1
λ33 V24	-1	1	1	1	_1	_1	1	1	-1	-1	1	-1	_1	-1	1	1 1
134 Var	_1	1		_9	- <u>-</u> 9	-1 9	0	0	0	0	0	0	0	0	0	0
A35 V26	-1	1	2	-2	-2	-2	0	0	0	0	0	0		0	0	0
A30 V27	-2	2		0	0	0	0	0	0	0	0	0		0	0	0
137 V20	0	0	_1	-1	1	1	_1	1	1	-1	1	-1	_1	-1	1	1
138 V20	0	0	1	1	-1	_1	1	-1	-1	1	-1	1	_1	-1	1	1
X39	0	0	2	1 9	-1 _9	-1 _9		0	0	0	0	0	0	0	0	1
λ40 γ ₄₁	0	0	-2	-2	2	-2	0	0	0	0	0	0		0	0	0
X41	1	_1	_1	_1	1	- 1	_1	1	1	_1	1	_1	1	1	_1	_1
λ42 γ ₄₂	1	-1	1	1	_1	_1	1	_1	_1	-1	_1	1		1	-1	-1
λ43 <i>χ</i> 44	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^7:S_8(\text{continued})$

	7A			8A				10A				12A			15A	
	7A	14A	8I	8J	8K	8L	10D	20A	20B	10E	12M	24A	24B	12SN	15A	30A
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	1	1
<i>χ</i> 3	0	0	1	1	1	1	0	0	0	0	0	0	0	0	-1	-1
<i>χ</i> 4	0	0	-1	-1	-1	-1	0	0	0	0	0	0	0	0	-1	-1
χ_5	0	0	0	0	0	0	1	1	1	1	-1	-1	-1	-1	-1	-1
χ_6	0	0	0	0	0	0	-1	-1	-1	-1	1	1	1	1	-1	-1
χ7	-1	-1	0	0	0	0	0	0	0	0	1	1	1	1	0	0
X8	-1	-1	0	0	0	0	0	0	0	0	-1	-1	-1	-1	0	0
X9	0	0	-1	-1	-1	-1	1	1	1	1	0	0	0	0	1	1
χ_{10}	0	0	1	1	1	1	-1	-1	-1	-1	0	0	0	0	1	1
X11	0	0	0	0	0	0	0	0	0	0	-1	-1	-1	-1	1	1
χ_{12}	0	0	0	0	0	0	0	0	0	0	1	1	1	1	1	1
χ_{13}	0	0	1	1	1	1	0	0	0	0	-1	-1	-1	-1	0	0
χ_{14}	0	0	-1	-1	-1	-1	0	0	0	0	1	1	1	1	0	0
χ_{15}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	-1	-1
χ_{16}	0	0	0	0	0	0	1	1	1	1	0	0	0	0	1	1
χ_{17}	0	0	0	0	0	0	-1	-1	-1	-1	0	0	0	0	1	1
χ_{18}	1	1	0	0	0	0	-1	-1	-1	-1	0	0	0	0	-1	-1
χ_{19}	1	1	0	0	0	0	1	1	1	1	0	0	0	0	-1	-1
χ_{20}	0	0	0	0	0	0	0	0	0	0	1	1	1	1	0	0
χ_{21}	0	0	0	0	0	0	0	0	0	0	-1	-1	-1	-1	0	0
X22	-1	-1	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X23	1	-1	1	1	-1	-1	1	1	-1	-1	1	1	-1	-1	1	-1
χ_{24}	1	-1	-1	-1	1	1	-1	-1	1	1	-1	-1	1	1	1	-1
χ_{25}	0	0	1	1	-1	-1	0	0	0	0	0	0	0	0	-1	1
χ_{26}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0	-1	1
χ_{27}	0	0	0	0	0	0	1	1	-1	-1	-1	-1	1	1	-1	1
χ_{28}	0	0	0	0	0	0	-1	-1	1	1	1	1	-1	-1	-1	1
χ_{29}	-1	1	0	0	0	0	0	0	0	0	1	1	-1	-1	0	0
X30	-1	1	0	0	0	0	0	0	0	0	-1	-1	1	1	0	0
χ_{31}		0	-1	-1	1	1	1	1	-1	-1	0	0	0	0	1	-1
X32		0	1	1	-1	-1	-1	-1	1	1	0	0	0	0	1	-1
X33		0	0	0	0	0	0	0	0	0	-1	-1 1	1	1	1	-1
X34		0	1	1	1	1	0	0	0	0	1	1	-1	-1	1	-1
X35		0	1	1	-1	-1	0	0	0	0	-1	-1	1	1	0	0
X36		0	-1	-1	1	1		0	0	0		1	-1	-1	1	1
X37		0		0	0	0	1	1	1	1		0	U	0	-1	1
X38		0		0	0	0		1	-1 1	-1 1		0	0	0		-1 1
X39		1		0	0	0	-1	-1 1	1	1		0	0	0	1	-1 1
X40		-1 1		0	0	0	-1 1	-1 1	1	1	0	0	0	0	-1	1
X41		-1		0	0	0		1	-1	-1	1	1	1	1	-1	1
X42		0		0	0	0		0	0	0	1	1	-1 1	-1		0
X43		1		0	0	0		0	0	0	-1	-1	1	1		0
χ_{44}	-1	1	U	U	U	U	U	U	U	U	U	U	U	U	U	U

The character table of $2^7:S_8(\text{continued})$

				1A						2A				
	1A	2A	2B	2C	2D	2E	2F	4A	4B	4C	4D	4E	4F	4G
χ_{45}	28	-28	4	-4	-4	4	16	-14	4	-6	-4	2	2	0
χ_{46}	28	-28	4	-4	-4	4	-16	14	-4	6	4	-2	-2	0
χ_{47}	28	-28	4	-4	-4	4	14	-16	6	-4	-2	4	0	-2
χ_{48}	28	-28	4	-4	-4	4	-14	16	-6	4	2	-4	0	2
χ_{49}	140	-140	20	-20	-20	20	-40	50	-20	10	4	-14	2	8
χ_{50}	140	-140	20	-20	-20	20	50	-40	10	-20	-14	4	8	2
χ_{51}	140	-140	20	-20	-20	20	-50	40	-10	20	14	-4	-8	-2
χ_{52}	140	-140	20	-20	-20	20	40	-50	20	-10	-4	14	-2	-8
χ_{53}	140	-140	20	-20	-20	20	-20	10	0	10	8	2	-6	-4
χ_{54}	140	-140	20	-20	-20	20	10	-20	10	0	2	8	-4	-6
χ_{55}	140	-140	20	-20	-20	20	-10	20	-10	0	-2	-8	4	6
χ_{56}	140	-140	20	-20	-20	20	20	-10	0	-10	-8	-2	6	4
χ_{57}	252	-252	36	-36	-36	36	-36	54	-24	6	0	-18	6	12
χ_{58}	252	-252	36	-36	-36	36	54	-36	6	-24	-18	0	12	6
χ_{59}	252	-252	36	-36	-36	36	-54	36	-6	24	18	0	-12	-6
χ_{60}	252	-252	36	-36	-36	36	36	-54	24	-6	0	18	-6	-12
χ_{61}	280	-280	40	-40	-40	40	-40	20	0	20	16	4	-12	-8
χ_{62}	280	-280	40	-40	-40	40	20	-40	20	0	4	16	-8	-12
χ_{63}	280	-280	40	-40	-40	40	-20	40	-20	0	-4	-16	8	12
χ_{64}	280	-280	40	-40	-40	40	40	-20	0	-20	-16	-4	12	8
χ_{65}	448	-448	64	-64	-64	64	16	16	-16	-16	-16	-16	16	16
χ_{66}	448	-448	64	-64	-64	64	-16	-16	16	16	16	16	-16	-16
χ_{67}	28	28	4	4	-4	-4	16	14	4	6	-4	-2	-2	0
χ_{68}	28	28	4	4	-4	-4	-16	-14	-4	-6	4	2	2	0
χ_{69}	28	28	4	4	-4	-4	14	16	6	4	-2	-4	0	-2
χ_{70}	28	28	4	4	-4	-4	-14	-16	-6	-4	2	4	0	2
χ_{71}	140	140	20	20	-20	-20	-40	-50	-20	-10	4	14	-2	8
χ_{72}	140	140	20	20	-20	-20	50	40	10	20	-14	-4	-8	2
χ_{73}	140	140	20	20	-20	-20	-50	-40	-10	-20	14	4	8	-2
χ_{74}	140	140	20	20	-20	-20	40	50	20	10	-4	-14	2	-8
χ_{75}	140	140	20	20	-20	-20	-20	-10	0	-10	8	-2	6	-4
χ_{76}	140	140	20	20	-20	-20	10	20	10	0	2	-8	4	-6
χ_{77}	140	140	20	20	-20	-20	-10	-20	-10	0	-2	8	-4	6
χ_{78}	140	140	20	20	-20	-20	20	10	0	10	-8	2	-6	4
χ_{79}	252	252	36	36	-36	-36	-36	-54	-24	-6	0	18	-6	12
χ_{80}	252	252	36	36	-36	-36	54	36	6	24	-18	0	-12	6
χ_{81}	252	252	36	36	-36	-36	-54	-36	-6	-24	18	0	12	-6
χ_{82}	252	252	36	36	-36	-36	36	54	24	6	0	-18	6	-12
χ_{83}	280	280	40	40	-40	-40	-40	-20	0	-20	16	-4	12	-8
χ_{84}	280	280	40	40	-40	-40	20	40	20	0	4	-16	8	-12
χ_{85}	280	280	40	40	-40	-40	-20	-40	-20	0	-4	16	-8	12
χ_{86}	280	280	40	40	-40	-40	40	20	0	20	-16	4	-12	8
χ_{87}	448	448	64	64	-64	-64	16	-16	-16	16	-16	16	-16	16
χ_{88}	448	448	64	64	-64	-64	-16	16	16	-16	16	-16	16	-16

The character table of $2^7:S_8(\text{continued})$

				2B									2C					
	2J	2K	2L	2M	2N	20	4E	4F	2P	4G	4H	2Q	4I	2R	4J	2S	4K	4L
χ_{45}	4	4	4	4	-4	-4	0	0	8	-4	4	-8	4	0	-4	0	0	0
χ_{46}	4	4	4	4	-4	-4	0	0	8	-4	4	-8	4	0	-4	0	0	0
χ_{47}	-4	-4	-4	-4	4	4	0	0	4	-8	8	-4	0	-4	0	4	0	0
χ_{48}	-4	-4	-4	-4	4	4	0	0	4	-8	8	-4	0	-4	0	4	0	0
χ_{49}	4	4	4	4	-4	-4	0	0	0	-12	12	0	-4	-8	4	8	0	0
χ_{50}	-4	-4	-4	-4	4	4	0	0	12	0	0	-12	8	4	-8	-4	0	0
χ_{51}	-4	-4	-4	-4	4	4	0	0	12	0	0	-12	8	4	-8	-4	0	0
χ_{52}	4	4	4	4	-4	-4	0	0	0	-12	12	0	-4	-8	4	8	0	0
χ_{53}	-12	-12	-12	-12	12	12	0	0	8	-4	4	-8	4	0	-4	0	0	0
χ_{54}	12	12	12	12	-12	-12	0	0	4	-8	8	-4	0	-4	0	4	0	0
χ_{55}	12	12	12	12	-12	-12	0	0	4	-8	8	-4	0	-4	0	4	0	0
χ_{56}	-12	-12	-12	-12	12	12	0	0	8	-4	4	-8	4	0	-4	0	0	0
χ_{57}	-12	-12	-12	-12	12	12	0	0	0	-12	12	0	-4	-8	4	8	0	0
χ_{58}	12	12	12	12	-12	-12	0	0	12	0	0	-12	8	4	-8	-4	0	0
χ_{59}	12	12	12	12	-12	-12	0	0	12	0	0	-12	8	4	-8	-4	0	0
χ_{60}	-12	-12	-12	-12	12	12	0	0	0	-12	12	0	-4	-8	4	8	0	0
χ_{61}	-8	-8	-8	-8	8	8	0	0	-8	16	-16	8	0	8	0	-8	0	0
χ_{62}	8	8	8	8	-8	-8	0	0	-16	8	-8	16	-8	0	8	0	0	0
χ_{63}	8	8	8	8	-8	-8	0	0	-16	8	-8	16	-8	0	8	0	0	0
χ_{64}	-8	-8	-8	-8	8	8	0	0	-8	16	-16	8	0	8	0	-8	0	0
χ_{65}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{66}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{67}	4	-4	-4	4	4	-4	0	0	8	4	-4	-8	-4	0	4	0	0	0
χ_{68}	4	-4	-4	4	4	-4	0	0	8	4	-4	-8	-4	0	4	0	0	0
χ_{69}	-4	4	4	-4	-4	4	0	0		8	-8	-4	0	-4	0	4	0	0
χ70	-4	4	4	-4	-4	4	0	0	4	8	-8	-4	0	-4	0	4	0	0
χ71		-4	-4	4	4	-4	0	0		12	-12	0	4	-8	-4	8	0	0
χ72	-4	4	4	-4	-4	4	0	0	12	0	0	-12	-8	4	8	-4	0	0
χ73	-4	4	4	-4	-4	4	0	0		10	10	-12	-8	4	8	-4	0	0
X74	4	-4 19	-4 19	4	4 19	-4 19	0	0		12	-12	0	4	-8	-4	8	0	0
χ75 χ75	12	12	12	-12 19	-12 19	12	0	0		4	-4	-0	-4	4	4	4	0	0
X76	12	-12	-12	12	12	-12	0	0		8	-8	-4 -4	0	-4 -4	0	4	0	0
X77	_12	-12 19	-12 19	-12	-12	-12 12	0	0	8	4	-0	-4	-4	-4	4	- -	0	0
X 18	_12	12	12	-12	-12	12	0	0			_12	0	4	-8	-4	8	0	0
× 19	12	-12	-12	12	12	-12	0	0	12	0	0	-12	-8	4	8	-4	0	0
1 ×00	12	-12	-12	12	12	-12	0	0	12	0	0	-12	-8	4	8	-4	0	0
X82	-12	12	12	-12	-12	12	0	0	0	12	-12	0	4	-4	-4	8	0	0
X83	-8	8	8	-8	-8	8	õ	0	-8	-16	16	8	0	8	0	-8	õ	Õ
X84	8	-8	-8	8	8	-8	0	0	-16	-8	8	16	8	0	-8	0	0	0
X85	8	-8	-8	8	8	-8	0	0	-16	-8	8	16	8	0	-8	0	0	0
X86	-8	8	8	-8	-8	8	0	0	-8	-16	16	8	0	8	0	-8	0	0
χ_{87}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{88}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^7:S_8(\text{continued})$

				2D							3A				3B	
	2T	4M	4N	4O	4P	4Q	2U	4R	3A	6A	6B	6C	6D	6E	3B	6F
χ_{45}	4	2	2	4	-2	0	0	-2	10	-10	-2	2	2	-2	1	-1
χ_{46}	-4	-2	-2	-4	2	0	0	2	10	-10	-2	2	2	-2	1	-1
χ_{47}	-2	-4	-4	-2	0	2	2	0	10	-10	-2	2	2	-2	1	-1
χ_{48}	2	4	4	2	0	-2	-2	0	10	-10	-2	2	2	-2	1	-1
χ_{49}	4	2	2	4	-2	0	0	-2	20	-20	-4	4	4	-4	-1	1
χ_{50}	2	4	4	2	0	-2	-2	0	20	-20	-4	4	4	-4	-1	1
χ_{51}	-2	-4	-4	-2	0	2	2	0	20	-20	-4	4	4	-4	-1	1
χ_{52}	-4	-2	-2	-4	2	0	0	2	20	-20	-4	4	4	-4	-1	1
χ_{53}	0	-6	-6	0	-2	4	4	-2	-10	10	2	-2	-2	2	2	-2
χ_{54}	-6	0	0	-6	4	-2	-2	4	-10	10	2	-2	-2	2	2	-2
χ_{55}	6	0	0	6	-4	2	2	-4	-10	10	2	-2	-2	2	2	-2
χ_{56}	0	6	6	0	2	-4	-4	2	-10	10	2	-2	-2	2	2	-2
χ_{57}	0	6	6	0	2	-4	-4	2	0	0	0	0	0	0		0
χ_{58}	6	0	0	6	-4	2	2	-4	0	0	0	0	0	0		0
χ_{59}	-6	0	0	-6	4	-2	-2	4	0	0	0	0	0	0		0
χ_{60}	0	-6	-6	0	-2	4	4	-2	0	0	0	0	0	0		0
χ_{61}	8	4	4	8	-4	0	0	-4	10	-10	-2	2	2	-2		-1
χ_{62}	4	8	8	4	0	-4	-4	0	10	-10	-2	2	2	-2		-1
χ_{63}	-4	-8	-8	-4	0	4	4	0	10	-10	-2	2	2	-2		-1
X64	-8	-4	-4	-8	4	0	0	4	10	-10	-2	2	2	-2		-1
χ_{65}	0	0	0	0	0	0	0	0	-20	20	4	-4	-4	4	-2	2
X66	4	0	4	0 0	0	0	0 0	0	-20	10	4 9	-4	-4 0	4 0	-2	1
X67	4	-2 2	-4	2 9	0	0	-2 2	∠ ົ	10	10	2	2	-2	-2		1
X68	-4 _2	4	4 2	-2	-2	2	0	-2	10	10	2	2	-2	-2		1
X69	2	-4	-2	-4	-2 2	-2	0	0	10	10	2	2	-2	-2	1	1
X70	4	-4	-2	2	0	-2	-2	2	20	20	4	4	-2	-2	-1	-1
	2	-4	-2	4	2	-2	0	0	20	20	4	4	-4	-4	-1	-1
X12 V72	-2	4	2	-4	-2	2	0	0	20	20	4	4	-4	-4	-1	-1
×74	-4	2	4	-2	0	0	2	-2	20	20	4	4	-4	-4	-1	-1
χ_{75}	0	6	0	-6	-4	4	-2	2	-10	-10	-2	-2	2	2	2	2
χ ₇₆	-6	0	6	0	2	-2	4	-4	-10	-10	-2	-2	2	2	2	2
χ ₇₇	6	0	-6	0	-2	2	-4	4	-10	-10	-2	-2	2	2	2	2
χ_{78}	0	-6	0	6	4	-4	2	-2	-10	-10	-2	-2	2	2	2	2
χ_{79}	0	-6	0	6	4	-4	2	-2	0	0	0	0	0	0	0	0
χ_{80}	6	0	-6	0	-2	2	-4	4	0	0	0	0	0	0	0	0
χ_{81}	-6	0	6	0	2	-2	4	-4	0	0	0	0	0	0	0	0
χ_{82}	0	6	0	-6	-4	4	-2	2	0	0	0	0	0	0	0	0
χ_{83}	8	-4	-8	4	0	0	-4	4	10	10	2	2	-2	-2	1	1
χ_{84}	4	-8	-4	8	4	-4	0	0	10	10	2	2	-2	-2	1	1
χ_{85}	-4	8	4	-8	-4	4	0	0	10	10	2	2	-2	-2	1	1
χ_{86}	-8	4	8	-4	0	0	4	-4	10	10	2	2	-2	-2	1	1
χ_{87}	0	0	0	0	0	0	0	0	-20	-20	-4	-4	4	4	-2	-2
χ_{88}	0	0	0	0	0	0	0	0	-20	-20	-4	-4	4	4	-2	-2

The character table of $2^7:S_8(\text{continued})$

		3B					4A						4B			
	6G	6H	6I	6J	4AA	4AB	8A	8B	8C	8D	4AC	4AD	8E	8F	8G	8H
χ_{45}	-1	1	1	-1	6	-6	-2	2	0	0	2	-2	-2	2	2	-2
χ_{46}	-1	1	1	-1	-6	6	2	-2	0	0	-2	2	2	-2	-2	2
χ_{47}	-1	1	1	-1	6	-6	-2	2	0	0	-2	2	2	-2	-2	2
χ_{48}	-1	1	1	-1	-6	6	2	-2	0	0	2	-2	-2	2	2	-2
χ_{49}	1	-1	-1	1	-6	6	2	-2	0	0	-2	2	2	-2	-2	2
χ_{50}	1	-1	-1	1	6	-6	-2	2	0	0	-2	2	2	-2	-2	2
χ_{51}	1	-1	-1	1	-6	6	2	-2	0	0	2	-2	-2	2	2	-2
χ_{52}	1	-1	-1	1	6	-6	-2	2	0	0	2	-2	-2	2	2	-2
χ_{53}	-2	2	2	-2	6	-6	-2	2	0	0	2	-2	-2	2	2	-2
χ_{54}	-2	2	2	-2	-6	6	2	-2	0	0	2	-2	-2	2	2	-2
χ_{55}	-2	2	2	-2	6	-6	-2	2	0	0	-2	2	2	-2	-2	2
χ_{56}	-2	2	2	-2	-6	6	2	-2	0	0	-2	2	2	-2	-2	2
χ_{57}	0	0	0	0	6	-6	-2	2	0	0	2	-2	-2	2	2	-2
χ_{58}	0	0	0	0	-6	6	2	-2	0	0	2	-2	-2	2	2	-2
χ_{59}	0	0	0	0	6	-6	-2	2	0	0	-2	2	2	-2	-2	2
χ_{60}	0	0	0	0	-6	6	2	-2	0	0	-2	2	2	-2	-2	2
χ_{61}	-1	1	1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{62}	-1	1	1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{63}	-1	1	1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{64}	-1	1	1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{65}	2	-2	-2	2	0	0	0	0	0	0	0	0	0	0	0	0
χ_{66}	2	-2	-2	2	0	0	0	0	0	0	0	0	0	0	0	0
χ_{67}	1	1	-1	-1	6	6	-2	-2	0	0	2	2	2	2	-2	-2
χ_{68}	1	1	-1	-1	-6	-6	2	2	0	0	-2	-2	-2	-2	2	2
χ_{69}	1	1	-1	-1	6	6	-2	-2	0	0	-2	-2	-2	-2	2	2
χ_{70}		1	-1	-1	-6	-6	2	2	0	0	2	2	2	2	-2	-2
χ71	-1	-1	1	1	-6	-6	2	2	0	0	-2	-2	-2	-2	2	2
χ72	-1	-1	1	1	6	6	-2	-2	0	0	-2	-2	-2	-2	2	2
χ73	-1	-1	1	1	-6	-6	2	2	0	0	2	2	2	2	-2	-2
χ_{74}	-1	-1	1	1	6	6	-2	-2	0	0	2	2	2	2	-2	-2
χ_{75}		2	-2	-2	6	6	-2	-2	0	0	2	2	2	2	-2	-2
χ_{76}		2	-2	-2	-6	-6	2	2	0	0	2	2	2	2	-2	-2
χ77		2	-2	-2	0 6	0 6	-2	-2	0	0	-2	-2	-2	-2	2	2
χ78		2	-2	-2	-0 C	-0 C	2	2	0	0	-2	-2	-2	-2	2	2
X79		0	0	0	0	0 6	-2	-2	0	0	2	2	2	2	-2	-2
X80		0	0	0	-0 C	-0 C	2	2	0	0	2	2	2	2	-2	-2
χ ₈₁		0	0	0	0	0	-2	-2	0	0	-2	-2	-2	-2	2	2
X82		1	1	1	-0	-0 0	2	2	0	0	-2	-2	-2	-2	2	2
X83		1	-1	-1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{84}		1	-1 1	-1 1		0	0	0	0	0		0	0	0	0	0
X85		1	-1 1	-1 1	0	0	0	0	0	0		0	0	0	0	0
χ_{86}		1	-1	-1		0	0	0	0	0		0	0	0	0	0
X87	-2	-2	2	2	0	0	0	0	0	0		0	0	0	0	0
χ_{88}	-2	-2	2	2	U	U	U	U	U	U	0	0	U	U	0	U

The character table of $2^7:S_8(\text{continued})$

	4B				4C							4D				
	8 <i>I</i>	8J	4AE	4AF	4AG	4AH	8K	8L	4AL	8L	8M	8N	8O	8P	8Q	8R
χ_{45}	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2	0
χ_{46}	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2	0
χ_{47}	0	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2
χ_{48}	0	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2
χ_{49}	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2	0
χ_{50}	0	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2
χ_{51}	0	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2
χ_{52}	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2	0
χ_{53}	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2	0
χ_{54}	0	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2
χ_{55}	0	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2
χ_{56}	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2	2
χ_{57}	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2	0
χ_{58}	0	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2
χ_{59}	0	0	0	0	0	0	0	0	0	-2	0	2	0	-2	0	2
χ_{60}	0	0	0	0	0	0	0	0	2	0	-2	0	2	0	-2	0
χ_{61}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{62}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{63}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{64}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{65}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{66}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{67}	0	0	0	0	0	0	0	0	2	0	2	0	-2	0	-2	-2
χ_{68}	0	0	0	0	0	0	0	0	2	0	2	0	-2	0	-2	0
χ_{69}		0	0	0	0	0	0	0	0	-2	0	-2	0	2	0	2
X70		0	0	0	0	0	0	0	0	-2	0	-2	0	2	0	2
X71		0	0	0	0	0	0	0	-2	0	-2	0	2	0 9	2	0
X72		0	0	0	0	0	0	0	0	2	0	2	0	-2 -2	0	-2
X73	0	0	0	0	0	0	0	0	_2	0	_2	0	2	-2	2	-2
X74	0	0	0	0	0	0	0	0	-2	0	-2 -2	0	2	0	2	0
X15	0	0	0	0	0	0	0	0	0	2	0	2	0	-2	0	-2
×77	0	0	0	0 0	0 0	0 0	0	Õ	0	2	0	2	0	-2	0	-2
	0	0	0	0 0	0 0	0 0	0	Õ	-2	0	-2	0	2	0	2	0
×70	0	0	0	0	0	0	0	0	2	0	2	0	-2	0	-2	0
X80	0	0	0	0	0	0	0	0	0	-2	0	-2	0	2	0	2
X81	0	0	0	0	0	0	0	0	0	-2	0	-2	0	2	0	2
χ_{82}	0	0	0	0	0	0	0	0	2	0	2	0	-2	0	-2	0
χ_{83}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{84}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{85}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{86}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X87	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X88	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^7:S_8(\text{continued})$

7

A group of the form $2^6:A_8$ as an inertia factor group of $2^8:O_8^+(2)$

Prologue

The group $\overline{G} = 2^6: A_8$ is an inertia factor group of $2^8: O_8^+(2)$. This group is also a maximal subgroup of $O_8^+(2)$ of index 135 and order 1290240. As an inertia factor group, our group \overline{G} plays an essential role in the construction of the character table of $2^8: O_8^+(2)$ as there is a block of irreducible characters in the character table of $2^8: O_8^+(2)$ corresponding to \overline{G} . In this chapter we look at two ways of constructing \overline{G} . In the first method, we use combinatorics and the natural action of A_8 on 2^6 . In the second method, we use GAP and we construct \overline{G} inside $O_8^+(2)$. We then compute the Fischer-Clifford matrices of \overline{G} which can then be used together with the ordinary character tables of the inertia factors of A_8 to compute its full ordinary character table. For more reading, on the methods used, one can also go to [1, 2, 3, 26, 37, 81, 82, 99, 120, 126, 94].

7.1. Introduction

The group $\overline{G} = 2^6 \cdot A_8$ is a maximal subgroup of $O_8^+(2)$ of index 135 and order 1290240. It is also an inertia factor of $O_8^+(2)$. As an inertia factor it plays an essential role in the construction of the character table of $2^8 \cdot O_8^+(2)$ as there is a block of irreducible characters in this table that corresponds to \overline{G} . In the construction of \overline{G} , A_8 acts on the elementary abelian group 2^6 . The action on 2^6 is multiplication on the right of the six dimensional row vector space $N = 2^6$. This requires A_8 to be represented by 6×6 matrices. It then becomes necessary to reconstruct A_8 from a 8×8 representation to a 6×6 representation. Here we look at two ways to do this.

Although it is much simpler and natural to consider the embedding of $2^6:A_8$ into $O_8^+(2)$ (see Section 7.3), but it is interesting to construct this group combinatorially and this is our main reason for discussing the first method. In our first method, we first take an 8-dimensional module V on which S_8 acts naturally by permuting its basis elements. We then obtain two submodules of V, namely

 M_1 and M_2 of dimensions 1 and 7 respectively. Let $W = M_2/M_1$, then dim(W) = 6 and W is a *G*-invariant where $G = S_6$ or A_8 (see Theorem 7.2.2). Let α and β be two permutation cycles of orders 7 and 3 respectively where $A_8 = \langle \alpha, \beta \rangle$. We then, by the action of α and β on the generators of W, get a matrix representation of both α and β . These are 6×6 matrix representations. We are then able to represent A_8 by 6×6 matrices. Letting this A_8 act on W, we obtain three orbits of lengths 1, 28 and 35 respectively. These have corresponding point stabilizers which we obtain from the ATLAS [23]. We are then able to construct \overline{G} .

For the second method we use GAP. We first construct $O_8^+(2)$ from the general orthogonal group $GO_8^+(2)$. We then construct, $\overline{G} = 2^6 : A_8$, inside $O_8^+(2)$. This has only one proper normal subgroup, namely 2^6 , which we can always obtain from \overline{G} . We then obtain the 6 generators of 2^6 which are 8×8 matrices. From the generators of \overline{G} , we are able to get two, 8×8 matrix generators of A_8 namely, a and b each of order 4. We then let a and b act on the generators of 2^6 by conjugation. Since $2^6 \leq \overline{G}$ the result of these actions are elements of 2^6 . We get a 6×6 matrix representation of both. This leads us to a 6×6 representation of A_8 . We then let this A_8 , using GAP, to act on 2^6 . Using the representatives of resulting orbits, we obtain corresponding point stabilizers of A_8 . These turn out to be the same as those obtained by combinatorics above.

The two groups constructed have the same character table, and through GAP, one can confirm that they are in deed isomorphic. We compute the Fischer-Clifford matrices which together with the character tables of the inertia factor groups of A_8 we use to compute the full character table of $2^6:A_8$. Note that there might be other easier methods to achieve this but our aim is to use the Fischer-Clifford theory to compute the character table.

7.2. The Combinatorics Method

The combinatorics method can also be found in [1] and [99] and is used extensively in [94] and [126]. The group S_8 acts naturally on a module of dimension 8 by permuting the basis elements which generate the module. Let V be the 8-dimensional natural module of S_8 over GF(2), where $V = \langle e_1, e_2, e_3, e_4, e_5, e_6, e_7, e_8 \rangle$, and $e_i^2 = 1$ for $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$. We regard V as a multiplicative elementary abelian 2-group of order 2^8 .

Theorem 7.2.1. Let V be the natural module of S_8 over GF(2). Then there exist S_8 submodules M_1 and M_2 of V such that $V \supset M_2 \supset M_1 \supset 0$ and that

$$dim(M_2) = 7$$
 and $dim(M_1) = 1$.

PROOF. Let $V = \langle e_1, e_2, e_3, e_4, e_5, e_6, e_7, e_8 \rangle$, and $e_i^2 = 1$ for $i \in \{1, 2, 3, 4, 5, 6, 7, 8\}$. Then S_8 acts naturally on V and this natural action results in the following orbits :

1. $O_0 = \{1_V\}$ and $|O_0| = 1$.

2.
$$O_1 = \{e_i | 1 \le i \le 8\}$$
 and $|O_1| = 8$.
3. $O_2 = \{e_i e_j | 1 \le i, j \le 8, i \ne j\}$ and $|O_2| = \binom{8}{2} = 28$.
4. $O_3 = \{e_i e_j e_k | 1 \le i, j, k \le 8, \text{ distinct } i, j, k\}$ and $|O_3| = \binom{8}{3} = 56$.
5. $O_4 = \{e_i e_j e_k e_l | 1 \le i, j, k, l \le 8, \text{ distinct } i, j, k, l\}$ and $|O_4| = \binom{8}{4} = 70$.
6. $O_5 = \{e_i e_j e_k e_l e_m | 1 \le i, j, k, l, m \le 8, \text{ distinct } i, j, k, l, m\}$ and $|O_5| = \binom{8}{5} = 56$.
7. $O_6 = \{e_i e_j e_k e_l e_m e_n | 1 \le i, j, k, l, m, n \le 8, \text{ distinct } i, j, k, l, m, n\}$ and $|O_6| = \binom{8}{6} = 28$.
8. $O_7 = \{e_i e_j e_k e_l e_m e_n e_o | 1 \le i, j, k, l, m, n, o \le 8, \text{ distinct } i, j, k, l, m, n, o\}$ and $|O_7| = \binom{8}{7} = 8$.
9. $O_8 = \{e_i e_j e_k e_l e_m e_n e_o e_p | 1 \le i, j, k, l, m, n, o, p \le 8, \text{ distinct } i, j, k, l, m, n, o, p\}$ and $|O_8| = 1$.

9.
$$O_8 = \{e_i e_j e_k e_l e_m e_n e_o e_p | 1 \le i, j, k, l, m, n, o, p \le 8, \text{ distinct } i, j, k, l, m, n, o, p\}$$
 and $|O_8| = \begin{pmatrix} 8\\ 8 \end{pmatrix} = 1.$

Thus S_8 produces 9 orbits on V. Set $M_1 = \langle e_1 e_2 e_3 e_4 e_5 e_6 e_7 e_8 \rangle$. Then M_1 is an S_8 - invariant submodule of V with $dim(M_1) = 1$. Now set $M_2 = O_0 \cup O_2 \cup O_4 \cup O_6 \cup O_8$. Then $|M_2| = 128$, so we have $dim(M_2) = 7$. Since $M_1 = O_0 \cup O_8$, we obtain that $V \supset M_2 \supset M_1 \supset 0$. This implies that M_2 is a reducible S_8 - invariant submodule of V.

Since S_8 is 8-transitive, A_8 is 6-transitive on $\{e_1, e_2, e_3, e_4, e_5, e_6, e_7, e_8\}$. It is clear that $O_0, O_1, O_2, O_4, O_5, O_6$ are also orbits under the action of A_8 . Now since A_8 does not have a proper subgroup of index less than 8, O_7 remains as an orbit of length 8. Obviously O_8 also remains as an orbit of length 1.

Theorem 7.2.2. Let $W = M_2/M_1$, then dim(W) = 6. Also W is a G - invariant module where $G = S_8$ or A_8 .

PROOF. It is clear that dim(W) = 6, since $dim(M_1) = 1$ and $dim(M_2) = 7$. If $g \in G$ and $\alpha \in M_2$, then since M_2 is G invariant, $g(\alpha M_1) = g(\alpha)M_1 \in M_2/M_1 \quad \forall g \in G$ and $\alpha \in M_2$. So W is S_8 (A₈) invariant.

Let $W = \langle e_1 e_2 M_1, e_1 e_3 M_1, e_1 e_4 M_1, e_1 e_5 M_1, e_1 e_6 M_1, e_1 e_7 M_1 \rangle$. The set $B = \{e_1 e_2, e_1 e_3, e_1 e_4, e_1 e_5, e_1 e_6, e_1 e_7\}$ is a linearly independent set. Let

$$\gamma_1 = e_1 e_2 M_1, \gamma_2 = e_1 e_3 M_1, \gamma_3 = e_1 e_4 M_1, \gamma_4 = e_1 e_5 M_1, \gamma_5 = e_1 e_6 M_1, \gamma_6 = e_1 e_7 M_1.$$

Also, if $\alpha = (1\ 2\ 3\ 4\ 5\ 6\ 7)$ and $\beta = (6\ 7\ 8)$ then $A_8 = <\alpha,\beta>$.

We obtain

$$\alpha: \gamma_1 \to \gamma_1 \gamma_2, \ \gamma_2 \to \gamma_1 \gamma_3, \ \gamma_3 \to \gamma_1 \gamma_4, \ \gamma_4 \to \gamma_1 \gamma_5, \ \gamma_5 \to \gamma_1 \gamma_6, \text{ and } \gamma_6 \to \gamma_1$$

We give two examples for the action of α . Under the action of α we have

$$\gamma_2 = e_1 e_3 M_1 \rightarrow e_2 e_4 M_1 = e_1 e_2 e_1 e_4 M_1 = \gamma_1 \gamma_3$$

That is $\alpha(\gamma_2) = \gamma_1 \gamma_3$. Also

$$\gamma_6 = e_1 e_7 M_1 \to e_2 e_1 M_1 = \gamma_1.$$

That is $\alpha(\gamma_6) = \gamma_1$. Hence α can be represented by the following matrix

	(1	1	0	0	0	0	
		1	0	1	0	0	0	
o —		1	0	0	1	0	0	
$\alpha =$		1	0	0	0	1	0	,
		1	0	0	0	0	1	
		1	0	0	0	0	0 /	

with $o(\alpha) = 7$.

Similarly for β we have

$$\beta: \gamma_1 \to \gamma_1, \ \gamma_2 \to \gamma_2, \ \gamma_3 \to \gamma_3, \ \gamma_4 \to \gamma_4, \ \gamma_5 \to \gamma_6, \ \gamma_6 \to \gamma_1 \gamma_2 \gamma_3 \gamma_4 \gamma_5 \gamma_6$$

As an example we see that

$$\gamma_6 = e_1 e_7 M_1 \to e_1 e_8 M_1 = e_2 e_3 e_4 e_5 e_6 e_7 M_1 = \gamma_1 \gamma_2 \gamma_3 \gamma_4 \gamma_5 \gamma_6$$

That is $\beta(\gamma_6) = \gamma_1 \gamma_2 \gamma_3 \gamma_4 \gamma_5 \gamma_6$. Here we obtain

$$\beta = \begin{pmatrix} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \\ 1 & 1 & 1 & 1 & 1 & 1 \end{pmatrix},$$

with $o(\beta) = 3$. We are now able to write all the elements of A_8 as 6×6 matrices. By acting A_8 directly on W, using the orbits of A_8 on M_2 and the fact that $M_1 = \{1, e_1e_2e_3e_4e_5e_6e_7e_8\}$, we can see that A_8 has 3 orbits namely

$$\Delta_0 = \{O_0 M_1\} = \{O_8 M_1\} = \{M_1\},\$$

$$\Delta_1 = \{O_2 M_1\} = \{O_6 M_1\} = \{e_i e_j M_1 | \text{distinct } e_i, e_j\},\$$

$$\Delta_2 = \{O_4 M_1\} = \{e_i e_j e_k e_l M_1 | \text{distinct } e_i, e_j, e_k, e_l\}.$$

Clearly $|\Delta_0| = 1$, $|\Delta_1| = 28$, $|\Delta_2| = \frac{70}{2} = 35$ and $W = \Delta_0 \cup \Delta_1 \cup \Delta_2$.

Theorem 7.2.3. A_8 acts irreducibly on W.

PROOF. Let $U \leq W$ be such that $U \neq 0$ and U is A_8 invariant. Since $U \neq 0 \exists x \in U$ such that $x \neq 0$. Since $x \neq 0$ and $U \leq W$ we have two cases.

Case 1 : Suppose $x \in \Delta_1$, $x = e_i e_j M_1$ for distinct i, j. Hence $g(x) \in U \quad \forall g \in A_8$. However

$$\{g(x)|g \in A_8\} = \Delta_1 \quad \Rightarrow \Delta_1 \subseteq U \quad \Rightarrow e_i e_j M_1 \in U \quad \forall \ i, j.$$

Hence we have $\gamma_1, \gamma_2, \gamma_3, \gamma_4, \gamma_5, \gamma_6 \in U$.

Case 2: Suppose $x \in \Delta_2$, then $x = e_i e_j e_k e_l M_1$ for some distinct e_i, e_j, e_k, e_l . Hence $g(x) \in U \quad \forall \ g \in A_8$. Now

$$\{g(x)|g \in A_8\} = \Delta_2 \Rightarrow \Delta_2 \subseteq U.$$

Since $\Delta_2 \subseteq U$, $e_k e_l e_m e_r M_1$ and $e_k e_l e_m e_i M_1$ are in U for distinct k, l, m, r, i. Since U is closed we get

$$(e_k e_l e_m e_r M_1)(e_k e_l e_m e_i M_1) = e_i e_r \in U \quad \forall \quad \text{distinct} \quad k, l, m, r, i.$$

This shows that $U \subseteq W$. So similar to case 1, we have U = W.

Hence W is a unique 6-dimensional GF(2) module that A_8 acts irreducibly on.

By methods of coset analysis that can be found in chapter 2, when $G = A_8$ acts on W we obtain three orbits of lengths 1, 28 and 35 respectively. These have corresponding point stabilizers K_1, K_2 and K_3 of indices 1, 28 and 35 respectively. One can immediately see that $K_1 = G$ and K_2, K_3 must each sit in a maximal subgroup of G. However any maximal subgroup of G which contains K_i must have an order divisible by $|K_i|$ and its index in G must divide 28 and 35 respectively. From the ATLAS [23], we get that up to isomorphism and conjugacy there is only one maximal subgroup of G, in each case, that would contain K_2 and the other K_3 and these are the symmetric group S_6 and the group $2^4:(S_3 \times S_3)$ respectively. However since $|K_2| = |S_6|$ we have $K_2 \cong S_6$. Similarly we have $K_3 \cong 2^4:(S_3 \times S_3)$. For each $g \in G$, the number of fixed points $g \in G$ in N is equal to $k = |C_N(g)|$. Since the zero vector of N is fixed by every $g \in G$ we have

$$k = 1 + \chi(G|K_2)(g) + \chi(G|K_3)(g) = 1 + (\chi(G|K_2) + \chi(G|K_3))(g).$$

From this we determine, $\chi = \chi(A_8|2^6)$, the permutation character of A_8 on 2^6 . We have

$$\chi = 1a + I_{S_6}^{A_8} + I_{2^4:(S_3 \times S_3)}^{A_8} = 3 \times 1a + 7a + 14a + 2 \times 20a,$$

where $I_{S_6}^{A_8} = 1a + 7a + 20a$ and $I_{2^4:(S_3 \times S_3)}^{A_8} = 1a + 14a + 20a$, are the characters of A_8 induced from the identity characters of S_6 and $2^4:(S_3 \times S_3)$ respectively. Since $C_N(g) \leq N$, we must have $k = 2^n$ where $n \in \{1, 2, 3, 4, 5, 6\}$. Hence we obtain the values of the k's in Table 7.1.

CHAPTER 7. A GROUP OF THE FORM $2^6:A_8$ AS AN INERTIA FACTOR GROUP OF $2^8:O_8^+(2)$

$[g]_{A_8}$	1a	2a	2b	3a	3b	4a	4b	5a	6a	6b	7a	7b	15a	15b
$\chi(A_8 S_6)$	28	4	8	10	1	0	2	3	1	2	0	0	0	0
$\chi(A_8 2^4:(S_3 \times S_3))$	35	11	7	5	2	3	1	0	2	1	0	0	0	0
k	64	16	16	16	4	4	4	4	4	4	1	1	1	1

7.3. The GAP Method

In this section for all our computations we use GAP [41]. We first construct $O_8^+(2)$ inside the general orthogonal group $GO_8^+(2)$. This we do by getting the maximal normal subgroup of $GO_8^+(2)$ and this is a group of 8×8 matrices of size 174182400 over GF(2). We then construct $\overline{G} = 2^6:A_8$ inside $O_8^+(2)$ by first constructing an 8-dimensional row vector space U, over GF(2). We then let $O_8^+(2)$ to act on U and we get three orbits of lengths 1,120 and 135. Using the ATLAS [23] and Programme C (see Appendix A), given below, the maximal subgroup of index 135 is $2^6:A_8$ which corresponds to the third orbit. We then get the stabilizer of a representative of this orbit in $O_8^+(2)$, which gives us a group of 8×8 matrices of size 1290240 which is our $2^6:A_8$. We are now ready to construct 2^6 and A_8 inside our \overline{G} . Note that we use ATLAS [23] for character tables of A_8 and S_6 . The character table of $2^4:(S_3 \times S_3)$ is given in the GAP Library [41], but it is also presented in the table below.

	1A	2A	2B	2C	4A	4B	$_{3A}$	6A	$_{3B}$	3C	2D	2E	4C	4D	6B	6C
X1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1
χ_3	1	1	1	-1	-1	-1	1	1	1	1	-1	1	-1	1	1	-1
χ_4	1	1	1	-1	-1	-1	1	1	1	1	1	-1	1	-1	-1	1
χ_5	2	2	2	0	0	0	-1	-1	-1	2	0	2	0	2	-1	0
χ_6	2	2	2	0	0	0	-1	-1	-1	2	0	-2	0	-2	1	0
χ_7	2	2	2	0	0	0	-1	-1	2	-1	-2	0	-2	0	0	1
χ_8	2	2	2	0	0	0	-1	-1	2	-1	2	0	2	0	0	-1
χ_9	4	4	4	0	0	0	1	1	-2	-2	0	0	0	0	0	0
χ_{10}	6	2	-2	2	-2	0	3	-1	0	0	0	0	0	0	0	0
χ_{11}	6	2	-2	2	-2	0	3	-1	0	0	0	0	0	0	0	0
χ_{12}	9	-3	1	1	1	-1	0	0	0	0	3	3	-1	-1	0	0
χ_{13}	9	-3	1	1	1	-1	0	0	0	0	-3	-3	1	1	0	0
χ_{14}	9	-3	1	-1	-1	1	0	0	0	0	3	-3	-1	1	0	0
χ_{15}	9	-3	1	-1	-1	1	0	0	0	0	-3	3	1	-1	0	0
χ_{16}	12	4	-4	0	0	0	-3	1	0	0	0	0	0	0	0	0

Character Table of $2^4:(S_3 \times S_3)$

We first pay our attention to A_8 . We first obtain the generators of \overline{G} . We get four of these and from the four we pick two, call them a and b that generate A_8 . We give a, b and their inverses below. Note that o(a) = o(b) = 4.

	Γ1	0	0	0	0	1	0	0 -	1		0	0	0	0	0	1	0	()]	
		1	0	1	0	1	0	1			0	0	1	1	1	1	1	(
	1	0	1	0	0	1	0	0			1	0	0	1	0	0	0	1	L	
	0	0	0	1	0	0	0	0		,	1	0	1	0	0	0	0]	L	
a =	1	1	0	0	1	1	0	0	,	b =	1	1	0	0	0	1	1	1	ιl	
	0	0	0	0	0	1	0	0			1	1	1	1	0	1	0]	L	
	1	0	0	0	0	1	1	0			0	1	0	0	0	0	1	(
	0	0	0	0	0	0	0	1 _			1	1	1	1	0	0	0	(
	[1	0	0	0	0	1	0	0]		[() () (0	0	1	1	1	1]
	0	1	0	1	0	1	0	1			() (0	1	1	1	1	1	0	
	1	0	1	0	0	0	0	0				1 () (0	1	1	0	1	0	
$a^{-1} - $	0	0	0	1	0	0	0	0		$b^{-1} - $.	1 (0	1 (0	1	0	1	0	
$a^{-1} =$	1	1	0	1	1	1	0	1	,	0 —		1	1 (0	0	1	1	0	0	
	0	0	0	0	0	1	0	0				1 () (0	0	0	0	0	0	
	1	0	0	0	0	0	1	0			() () [1	1	1	1	0	0	
	0	0	0	0	0	0	0	1				1 () () (0	0	1	0	1	

The group $N = 2^6$ is the only proper normal subgroup of \overline{G} and we use GAP [41] to obtain this normal subgroup. We then obtain its generators, which are given below.

[- 1	0	0	0	0	0	0	1			Γ1	0	0	0	0	0	0	1	٦
	0	1	0	0	0	0	0	0				1	0	0	0	0	0	1	
	0	0	1	0	0	0	0	0				0	1	0	0	0	0	0	
	0	0	0	1	0	0	0	0				0	0	1	0	0	0	0	
$\gamma_1 =$	0	0	0	0	1	0	0	0	,	$\gamma_2 =$		0	0	0	1	0	0	1	,
	0	0	0	0	0	1	0	0				0	0	0	0	1	0	1	
	0	1	0	0	0	0	1	0				1	0	0	1	1	1	0	
	0	0	0	0	0	0	0	1				0	0	0	0	0	0	1	
L	- 0	0	0	0	0	0	0		l		LU	0	0	0	0	0	0	T	
	[1]	0	0	0	0	0	0	0	1	Γ	1	0	0	0	0	0	0	0]	
	0	1	0	0	0	0	0	1			0	1	0	0	0	0	0	1	
	0	0	1	0	0	0	0	0			0	0	1	0	0	0	0	1	
$\gamma_3 =$	0	0	0	1	0	0	0	0			0	0	0	1	0	0	0	0	
	0	0	0	0	1	0	0	0		$, \gamma_4 = \left \right $	0	0	0	0	1	0	0	1	,
	0	0	0	0	0	1	0	1			0	0	0	0	0	1	0	0	
	1	0	0	0	1	0	1	0			1	0	0	1	0	1	1	0	
	0	0	0	0	0	0	0	1			0	0	0	0	0	0	0	1	
	-								-	-	•							_	
	1	0	0	0	0	0	0	1			[1	0	0	0	0	0	0	0]
	0	1	0	0	0	0	0	0			0	1	0	0	0	0	0	0	
	0	0	1	0	0	0	0	1			0	0	1	0	0	0	0	1	
	0	0	0	1	0	0	0	0			0	0	0	1	0	0	0	1	
$\gamma_5 =$	0	0	0	0	1	0	0	1	,	$\gamma_6 =$	0	0	0	0	1	0	0	1	.
	0	0	0	0	0	1	0	0			0	0	0	0	0	1	0	1	
	0	1	0	1	0	1	1	0			0	0	1	1	1	1	1	0	
	0	0	0	0	0	0	0	1			0	0	0	0	0	0	0	1	

Noting that the generators of 2^6 and A_8 are both 8×8 matrices. Computing the conjugate of each γ_i with respect to a, that is $a\gamma_i a^{-1}$ and noting that 2^6 is normal in $2^6 \cdot A_8$ we get that $a\gamma_i a^{-1} = \gamma_{j_1}\gamma_{j_2}\cdots\gamma_{j_k}$, where $\gamma_{j_r} = \gamma_j$ or 1 for some $j_r = 1, \cdots, 6$. We denote this as $\gamma_i \to \gamma_{j_1}\gamma_{j_2}\cdots\gamma_{j_k}$. We then get

$$\gamma_1 \to \gamma_1, \ \gamma_2 \to \gamma_2 \gamma_5, \ \gamma_3 \to \gamma_2 \gamma_4 \gamma_5 \gamma_6, \ \gamma_4 \to \gamma_4, \ \gamma_5 \to \gamma_1 \gamma_4 \gamma_5, \ \gamma_6 \to \gamma_1 \gamma_2 \gamma_3 \gamma_4.$$

Similarly with b we get

$$\begin{split} \gamma_1 &\to \gamma_1 \gamma_2 \gamma_3 \gamma_4 \gamma_6, \ \gamma_2 \to \gamma_2 \gamma_3 \gamma_4 \gamma_5 \gamma_6, \ \gamma_3 \to \gamma_1 \gamma_2 \gamma_3 \gamma_5 \gamma_6, \ \gamma_4 \to \gamma_1 \gamma_3 \gamma_4, \\ \gamma_5 &\to \gamma_4, \ \gamma_6 \to \gamma_2 \gamma_3 \gamma_4. \end{split}$$

Representing this information in matrix form, where the i-th row will correspond to the i-th conjugate, we get a 6×6 matrix representation of $G = A_8$. Hence we have $A_8 = \langle a', b' \rangle$, where

	1	0	0	0	0	0		1	1	1	1	0	1]
-	0	1	0	0	1	0		0	1	1	1	1	1	
a'	0	1	0	1	1	1	h'	1	1	1	0	1	1	
a' =	0	0	0	1	0	0	, o =	1	0	1	1	0	0	.
	1	0	0	1	1	0		0	0	0	1	0	0	
	1	1	1	1	0	0		0	1	1	1	0	0	

We now turn our attention to the inertia factor groups of $2^6:A_8$. Here we use Programme C below, to act G on Irr(N), $N = 2^6$. To be able to do this we need to rewrite N as a row vector space V of dimension 6 over GF(2), that is V:= FullRowSpace(GF(2),6). We have two procedures at our disposal. First we can act G on V from right and this action gives us the orbits of G acting as a permutation group on the conjugacy classes of N. Secondly we act G^t , that is the set consists of transpose of elements of G, on V from right. This action is equivalent to multiplying the column vectors of V on the left by G. This action gives the orbits of G acting as a permutation group on the irreducible characters of N.

PROGRAMME C for $2^6:A_8$

$$\begin{split} & gap>V:=FullRowSpace(GF(2),6);\\ & gap>m1:=(OneGF(2))^*[6\times 6 \text{ matrix group generators}];\\ & gap>m2:=(OneGF(2))^*[6\times 6 \text{ matrix group generators}];\\ & gap>m:=Group(m1,m2);\\ & gap>k:=OrbitLengths(m,V);\\ & gap>l:=OrbitLengths(Group(List(m,TransposedMat)),N); \end{split}$$

From the above, the action of G on Irr(N) produces three orbits of lengths 1, 28 and 35 respectively. We then take representatives of the orbits of lengths 28 and 35. For each of the orbit representative we find its stabilizer in G. For the representative of the orbit of length 28, the corresponding stabilizer is a group of 6×6 matrices of size 720 isomorphic to S_6 . For the orbit of length 35 the corresponding stabilizer is a group of 6×6 matrices of size 576 isomorphic to $2^4 : (S_3 \times S_3)$. This is the same result which we got in Section 2. We use GAP to check our calculations for the number of fixed points using Programme F (see Appendix A) and we list the values of the k's in Table 7.2.

						Tal	ble 7.	2:						
$[g]_{A_8}$	1a	2a	2b	3a	3b	4a	4b	5a	6a	6b	7a	$7\mathrm{b}$	15a	15b
k	64	16	16	16	4	4	4	4	4	4	1	1	1	1

Since the two $2^6:A_8$ constructed are isomorphic we use one of them to compute the Fisher-Clifford matrices and its character table.

7.4. The Conjugacy Classes of $2^6:A_8$

We first give the representatives of the conjugacy classes of A_8 in Table 7.3.

$[a]_{C}$		6	× 6	mat	rix	C 1.0.		$\begin{bmatrix} a \end{bmatrix}_{\alpha}$		6	× 6 :	mat	rix		$\left \left[a \right]_{\alpha} \right $
[9]G	/ 1	0	0	0	0	0)	[[9]G]	[9]G	(0	0	0	1	0	0)	[[9]G]
		1	0	0	0	$\left(\begin{array}{c} 0 \\ 0 \end{array} \right)$				0	1	1	0	1	
		1	1	0	0					1	1	1	0	1	
1A		0	1	1	0		1	2A		1	0	1	0		105
		0	0	1	1					1	1	1	1	1	
		0	0	0	1	0				1	1	1	1		
		0	0	0	0	1 /				0	0	0	0	1 /	
	$\left \left(\begin{array}{c} 1 \end{array} \right) \right $	0	0	0	0	0				1	1	1	1	0	
	0	1	0	0	0	0				1	0	0	0	0	
2B	0	0	1	0	0	0	210	3A		1	0	0	0	0	112
	0	0	0	1	0	0			1	1	1	1	0	0	
	0	0	0	0	1	1			0	0	0	1	0	0	
	$\setminus 1$	0	1	0	1	0 /				0	0	1	1	1 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	1	0	0	0)			0	0	1	1	1	1	
	1	0	1	1	0	1			0	0	1	1	0	0	
чD	1	0	1	0	0	1	1 1 1 0 0	410	0	0	0	0	1	0	1.000
3B	0	1	1	1	0	0	1 1 20	4B	1	0	1	0	1	1	1 200
	0	1	0	0	0	0			0	1	0	1	0	1	
		0	1	1	1	1			$ $ $ $ $ $ $ $	1	1	0	0	0 /	
	/ 0	0	1	1	1	0 \			/ 0	1	0	1	0	1	
	0	0	1	1	0	0				0	1	0	0	1	
	0	0	0	1	0	0			0	0	0	1	0	0	
4A		0	1	1	0	1	2 520	5a		0	0	1	1	1	1 344
		1	0	1	0	1				0	0	0	0	1	
		1	1	1	1	1				1	1	1	1		
		0	0	1	0	0)				0	1	1	1	1)	
		0	1	0	0	1)				0	1	0	0		
		0	1	0	1	1				0	1	1	0	1	
6A		1	1	1	1		1 680	6B		0	1	1	1		3 360
		1	1	1	1					1	1	1	1		
		1	1	1	1					1	1	1	1		
		1	1	1	1	1 /				0	1	0	0	1)	
		0	1	1	1	0)				0	1	0	0	1	
		0	1	1	0	0				0	0	1	0		
7A		0	0	1	0		2 880	7B		1	0	0	0	0	2 880
		0	1	1	0	1				1	1	1	1	1	
	0	1	1	0	0	0			0	0	0	1	1	1	
	$\setminus 1$	0	0	0	0	0 /				1	1	0	0	0 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	0	1	0	1	1			$\int 0$	0	1	1	1	0)	
	1	0	1	0	0	1			0	0	1	1	1	1	
15A	1	1	1	1	0	0	1 344	15B	1	1	1	1	1	1	1 344
1011	1	1	1	1	1	1	1 0 4 4	100	1	0	0	0	0	0	1 044
	0	0	0	1	1	1			0	1	1	0	0	0	
	0	1	1	0	0	0/				0	1	1	0	1/	

Table 7.3: Conjugacy classes of A_8

From the methods of coset analysis, which can also be found in chapter 2 and by Programmes A and B (see Appendix A), we are able to compute the conjugacy classes of $2^6:A_8$ which are given

in Table 7.4. We give a very brief summary of coset analysis. We look at the action of \overline{G} on $N\overline{g}$, for the split extension it is suffices to look at the coset $Ng, g \in G$. First N acts on Ng and we get k orbits. Then we act $C_G(g)$ on these orbits and f_j of these orbits, fuse to form one orbit with $\sum f_j = k$, and d_j a representative of these fused orbits. For this we use Programme A (see Appendix A).

PROGRAMME A for $2^6:A_8$

```
gap>V:=FullRowSpace(GF(2), 6);
     gap>gr1:=(OneGF(2))*[6 \times 6 matrix group generators];
     gap>gr2:=(OneGF(2))*[6 \times 6 matrix group generators];
                   gap>grp:=Group(gr1,gr2);
                gap>Ccl:=ConjugacyClasses(grp);
                 gap>O:=Union(Orbits(grp,V));
                      gap>for i in [1..14] do
                 >Print(Representative(Ccl[i]));
                 > w := One(GF(q))^*[0,0,\cdots,0];
                             >e:=[];
                >while Difference(O,e) <> [] do
                            >d:=[];
                         >for x in O do;
            >y:=[x+w+(x^{*}(Representative((Ccl)[i]))];
                        >d:=Union(d,y);
                              > \text{od};
                           >Print(d);
                        >e:=Union(d,e);
                 > if Difference(O,e) <> [] then
              >w:=Representative(Difference(O,e));
                               >fi;
                              > \text{od};
                             >r:=[];
                 >u:=One(GF(2))^*[0, 0, \cdots, 0];
                >while Difference(O,e) <> [] do
                            >m:=[];
       >for g in Centralizer(grp,Representative(Ccl[i])) do
                           >l:=[u^*g];
                        >m:=Union(m,l);
                              >od;
>Print("A block for the vectors under the action of a centralizer");
                           >Print(m);
                        >r:=Union(m,r);
                 >if Difference(O,r) <> [ ] then
              >u:=Representative(Difference(O,r));
                               >fi;
                              > od;
        >Print("**********
                              > \text{od};
```

Let o(dg) = k and o(g) = m. If $w = (dg)^m$, then if w = (0, 0, 0, 0, 0, 0), k = m. On the other hand if $w \neq (0, 0, 0, 0, 0, 0)$, then k = 2m. To get w we use Programme B (see Appendix A).

PROGRAMME B for $2^6:A_8$

$$\begin{split} & \text{gap>V:=FullRowSpace}(GF(2),6);\\ & \text{gap>m1:=}(\text{OneGF}(2))^*[6\times 6 \text{ matrix group generators}];\\ & \text{gap>m2:=}(\text{OneGF}(q))^*[6\times 6 \text{ matrix group generators}];\\ & \text{gap>m:=Group}(m1,m2);\\ & \text{gap>c:=ConjugacyClasses}(m);\\ & \text{gap>g:=Representative}(c[i]);\\ & \text{gap>d:=One}(\text{GF}(2))^*[\alpha_1,\alpha_2,\cdots,\alpha_6];\\ & \text{gap>w:=}d+d*g+d*g^2+\cdots+d*g^{k-1}; \end{split}$$

gap>Print(w);

We obtain that $2^6: A_8$ has altogether 41 conjugacy classes which are given in Table 7.4 below.

$g \in A_8$	k	f_j	d_{j}	w	$[x]_{2^6:A_8}$	$ C_{2^6:A_8}(x) $
1A	2^{6}	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	1A	1 290 240
		28	(0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 1)	2A	46 080
		35	(0, 0, 0, 1, 0, 1)	(0, 0, 0, 1, 0, 1)	2B	36 864
2A	2^{4}	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	2C	3 072
		1	(0, 0, 1, 0, 0, 1)	(0, 0, 0, 0, 0, 0)	2D	$3\ 072$
		1	(0, 0, 1, 1, 1, 1)	(0, 0, 0, 0, 0, 0)	$2\mathrm{E}$	3 072
		1	(0, 0, 1, 1, 1, 1)	(0, 0, 0, 0, 0, 0, 0)	$2\mathrm{F}$	3 072
		12	(0, 0, 0, 0, 0, 1)	(1, 1, 1, 0, 0, 1)	4A	256
2B	2^{4}	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	2G	1 536
		1	(0, 0, 1, 1, 1, 1)	(0, 0, 1, 0, 0, 0)	$4\mathrm{B}$	1 536
		3	(1, 0, 1, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0)	$2\mathrm{H}$	512
		3	(0, 0, 1, 1, 0, 1)	(0, 0, 1, 0, 1, 0)	$4\mathrm{C}$	512
		8	(0, 0, 0, 0, 0, 1)	(0, 0, 0, 1, 1, 0)	4D	192
3A	2^{4}	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	3A	2 880
		5	(1, 1, 0, 0, 1, 0)	(0, 0, 1, 1, 1, 1)	6A	576
		10	(0, 1, 0, 0, 1, 1)	(0, 0, 1, 1, 0, 0)	6B	288
3B	2^{2}	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	3B	72
		1	(1, 1, 0, 1, 0, 0)	(1, 0, 0, 0, 0, 1)	6B	72
		1	(1, 0, 1, 0, 1, 0)	(1, 0, 1, 0, 1, 0)	6C	72
		1	$\left(0,0,0,1,0,0 ight)$	(1, 1, 1, 1, 1, 1)	6D	72
4A	2^{2}	1	(0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	4E	64
		1	(1, 1, 0, 1, 0, 0)	(0, 0, 0, 0, 0, 0, 0)	4F	64
		1	(1, 0, 1, 0, 1, 0)	(0, 0, 0, 0, 0, 0)	4G	64
		1	$\left(0,0,0,1,0,0 ight)$	(0, 0, 0, 0, 0, 0)	4H	64
4B	2^{2}	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	4I	32
		1	(1, 1, 0, 1, 0, 0)	(0, 0, 1, 1, 1, 1)	8A	32
		1	$\left(1,0,1,0,1,0 ight)$	(0, 0, 0, 0, 0, 0)	4J	32
		1	$\left(0,0,0,1,0,0 ight)$	(1, 1, 1, 0, 1, 0)	8B	32
5A	2^{2}	1	$\left(0,0,0,0,0,0 ight)$	(0, 0, 0, 0, 0, 0)	5A	60
		3	$\left(0,1,0,1,0,0 ight)$	(0, 0, 0, 1, 0, 1)	10A	20
6A	2^{2}	1	(0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	6F	48
		1	(1, 0, 1, 0, 1, 0)	(0, 1, 1, 0, 1, 0)	12A	48
		2	(1, 0, 1, 0, 1, 0)	(1, 0, 1, 0, 1, 0)	12B	24
6B	2^{2}	1	(0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	6G	24
		1	$\left(0,0,0,0,0,0\right)$	(0, 0, 0, 0, 0, 1)	6H	24
					continued	on next page

Table 7.4: Conjugacy Classes of $2^6:A_8$

		-	· · · · · · · · · · · · · · · · · · ·	1	1 8)	
$g \in AS_8$	k	f_j	d_{j}	w	$[x]_{2^7:A_8}(x) $	
		1	$\left(0,0,0,0,0,0 ight)$	(0, 0, 0, 0, 0, 0)	6I	24
		1	$\left(0,0,0,0,0,0 ight)$	(0, 0, 0, 0, 0, 0)	6J	24
7A	1	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	7A	7
7B	1	1	$\left(0,0,0,0,0,0\right)$	(0, 0, 0, 0, 0, 0)	7B	7
15A	1	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	15A	15
15B	1	1	(0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0)	15B	15

Table 7.4 (continued from previous page)

7.5. The Fischer-Clifford matrices of $2^6:A_8$

The inertia factor groups are $H_1 = A_8$, $H_2 = S_6$ and $H_3 = 2^4: (S_3 \times S_3)$, as we discuss in section 7.3. We construct H_2 and H_3 inside A_8 in terms of 6×6 matrices. Their conjugacy classes are given in Table 7.5 and Table 7.6 respectively. The fusions of the inertia factor groups into A_8 , which can also be done using Programme D (see Appendix A), are given in Table 7.7.

PROGRAMME D for $2^6:A_8$

```
gap>g:=Group(H_1);
gap>T1:=CharacterTable(g);
gap>h:=Group(H_2);
gap>T2:=CharacterTable(h);
gap>T3:=Group(H_3);
gap>T3:=CharacterTable(k);
gap>FusionConjugacyClasses(h,g);
gap>FusionConjugacyClasses(k,g);
```

Using these fusions, and properties of Fischer-Clifford matrices which can be found in chapter 5.2.1, we are now able to obtain the Fischer-Clifford matrices of $2^6:A_8$. For each class representative $g \in A_8$, we construct a Fischer-Clifford matrix M(g) which are given in the Table 7.8.

$[g]_{S_6}$		6 :	× 6 1	mat	rix		$ [g]_{S_6} $	$[g]_{S_6}$		6 >	< 6 1	mati	rix		$ [g]_{S_6} $
	$\begin{pmatrix} 1 \end{pmatrix}$	0	0	0	0	0)			$\begin{pmatrix} 1 \end{pmatrix}$	0	1	0	1	0)	
	0	1	0	0	0	0			0	1	0	0	0	0	
1Δ	0	0	1	0	0	0	1	24	0	0	1	0	0	1	45
111	0	0	0	1	0	0	1	211	0	0	0	1	0	1	-10
	0	0	0	0	1	0			0	0	0	0	1	1	
	10	0	0	0	0	1 /			\ 0	0	0	0	0	1 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	0	0	0	0	0)			$\begin{pmatrix} 1 \end{pmatrix}$	0	0	1	1	0)	
	1	1	0	1	1	0			0	1	1	1	0	0	
2B	0	0	1	1	1	0	15	2C	0	0	1	0	0	0	15
20	0	0	0	1	0	0	10	20	0	0	0	1	0	0	10
	0	0	0	0	1	0			0	0	0	0	1	0	
	(0	0	0	0	0	1 /			\ 0	0	0	0	0	1 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	0	1	1	0	0)			$\begin{pmatrix} 1 \end{pmatrix}$	0	0	1	1	0)	
	0	0	1	1	0	0			0	0	1	1	0	0	
3A	0	0	1	0	0	0	40	3B	1	1	1	1	1	0	40
	0	1	0	1	0	0	-		1	0	0	0	1	0	
	0	0	0	0	1	0			0	0	0	0	1	0	
		0	0	0	0	1 /			(0	0	0	0	0	1 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	0	1	1	0	0			$\begin{pmatrix} 1 \end{pmatrix}$	0	1	1	0	0	
	1	1	0	1	1	0			0	0	1	1	0	0	
4A	0	0	1	1	1	1	90	4B	0	0	1	1	1	0	90
	0	0	0	1	0	1			0	1	0	1	0	1	
	0	0	0	0	1	1			0	0	0	1	0	0	
		0	0	0	0	1 /				0	0	0	0	1 /	
		0	1	1	0	0)			$\begin{pmatrix} 1 \\ 0 \end{pmatrix}$	0	0	1	1	0)	
		1	0	1	1	0				0	1	1	0	0	
6A		1	1	1	1	0	120	6B		0	1	0	0		120
		1	0	1	0					1	0	1	0		
		0	0	0	1					0	0	0	1		
		0	1	1	0	1 /				0	0	0	0	1 /	
		1	1	1	0	1									
		1	1	1	1										
5A		1	1	1	1		144								
		T T	0	1	0										
		0	0	1	0										
		0	0	0	0	1 /									

Table 7.5: Conjugacy classes of S_6

7.6. The Character Table of $2^6:A_8$

We use the Fischer-Clifford matrices (Table 7.8) and the character tables of the inertia factors groups A_8 , S_6 , $2^4:(S_3 \times S_3)$ together with the fusions of these inertia factors into A_8 which are given in Table 7.7 to obtain the full character table of $2^6:A_8$. The Fischer-Clifford matrix M(g)will be partitioned row-wise into blocks, where each block corresponds to an inertia group \overline{H}_i . Then given the character table of the inertia factor group H_i of \overline{H}_i , we therefore take the columns of this character table which correspond to the classes of H_i which fuse to class [g] in A_8 and multiply these columns by the rows of the Fischer-Clifford matrix M(g) which correspond to \overline{H}_i

CHAPTER 7. A GROUP OF THE FORM $2^6:A_8$ AS AN INERTIA FACTOR GROUP OF $2^8:O_8^+(2)$

$[g]_{2^4:(S_3 \times S_3)}$	6×6 matrix	$ [g]_{2^4:(S_3 \times S_3)} $	$[g]_{2^4:(S_3 \times S_3)}$	6×6 matrix	$ [g]_{2^4:(S_3 \times S_3)} $
1A	$\left(\begin{array}{ccccccccccccc} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	1	2A	$\left(\begin{array}{ccccccccccc} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 1 & 1 & 1 \end{array}\right)$	6
2B	$\left(\begin{array}{cccccccccccc} 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 0 & 1 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 0 & 1 & 1 \end{array}\right)$	9	2C	$\left(\begin{array}{cccccccccc} 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 1 & 1 & 1 & 1 & 0 \\ 0 & 0 & 1 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 0 & 1 & 1 \end{array}\right)$	36
2D	$\left(\begin{array}{cccccccccccc} 1 & 0 & 0 & 0 & 0 & 0 \\ 1 & 1 & 0 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 0 & 1 & 1 \end{array}\right)$	12	$2\mathrm{E}$	$\left(\begin{array}{ccccccccccc} 1 & 0 & 0 & 1 & 1 & 0 \\ 1 & 1 & 1 & 1 & 0 & 0 \\ 0 & 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	12
3A	$\left(\begin{array}{ccccccccc} 0 & 1 & 0 & 1 & 1 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 1 & 1 & 0 & 1 & 0 & 0 \\ 1 & 1 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 0 & 1 & 1 \end{array}\right)$	16	3B	$\left(\begin{array}{ccccccccc} 1 & 1 & 1 & 0 & 1 & 0 \\ 0 & 1 & 1 & 1 & 0 & 0 \\ 1 & 1 & 1 & 1 & 1 & 0 \\ 1 & 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	32
3C	$\left(\begin{array}{ccccccccccc} 1 & 1 & 0 & 0 & 0 & 0 \\ 1 & 1 & 0 & 1 & 1 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 0 & 1 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	32	4A	$\left(\begin{array}{cccccccccc} 0 & 1 & 0 & 1 & 1 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 \\ 1 & 1 & 0 & 1 & 0 & 0 \\ 1 & 1 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	36
4B	$\left(\begin{array}{cccccccccccc} 1 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 1 \\ 0 & 1 & 0 & 1 & 0 & 1 \\ 0 & 0 & 0 & 0 & 1 & 1 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	72	4C	$\left(\begin{array}{cccccccccc} 1 & 0 & 0 & 1 & 0 & 0 \\ 0 & 1 & 1 & 1 & 1 & 0 \\ 0 & 0 & 1 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 0 & 1 & 1 & 0 & 1 \end{array}\right)$	36
4D	$\left(\begin{array}{cccccccccccc} 1 & 0 & 0 & 1 & 0 & 0 \\ 1 & 1 & 1 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 1 & 1 & 1 & 0 & 1 & 1 \end{array}\right)$	36	6A	$\left(\begin{array}{cccccccccc} 1 & 0 & 0 & 1 & 1 & 0 \\ 0 & 0 & 1 & 1 & 0 & 0 \\ 1 & 1 & 1 & 1 & 1 & 0 \\ 1 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	48
6B	$\left(\begin{array}{cccccccccccc} 1 & 0 & 0 & 1 & 1 & 0 \\ 0 & 1 & 1 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array}\right)$	96	6C	$\left \left(\begin{array}{ccccccc} 0 & 1 & 0 & 1 & 1 & 0 \\ 1 & 1 & 0 & 1 & 1 & 0 \\ 0 & 1 & 0 & 0 & 1 & 0 \\ 1 & 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 0 & 0 & 1 \end{array} \right) \right $	96
$[x]_{S_6}$	\longrightarrow	$[g_1]_{A_8}$	$[x]_{2^4:(S_3 \times S_3)}$	\longrightarrow	$[g_1]_{A_8}$
-------------	-------------------	---------------	------------------------------	-------------------	---------------
1A		1A	1A		1A
2A		2B	2A		2B
2B		2B	2B		2A
2C		2A	2C		2B
3A		3A	2D		2A
3B		3B	2E		2A
4A		4B	3A		3A
4B		4B	3B		3B
5A		5A	3C		3B
6A		6A	4A		4A
6B		6B	4B		4B
			4C		4A
			4 <i>D</i>		4A
			6A		6A
			6B		6B
			6C		6B

Table 7.7: Fusion of S_6 and $2^4:(S_3 \times S_3)$ into A_8

and then fill the portion of the character table of $2^6:A_8$ which is in the block corresponding to \overline{H}_i for the classes of $2^6:A_8$ which come from the coset Ng. The set of irreducibles characters of $2^6:A_8$ will be partitioned into three blocks B_1 , B_2 , and B_3 corresponding to the inertia factors A_8 , S_6 and $2^4:(S_3 \times S_3)$ respectively. In fact $B_1 = \{\chi_i | 1 \le i \le 14\}$, $B_2 = \{\chi_i | 15 \le i \le 25\}$ and $B_3 = \{\chi_i | 25 \le i \le 41\}$. Note that the centralizers of elements of $2^6:A_8$ were listed in the last column of Table 7.4. We use Fischer-Clifford matrices and partial character tables of inertia factor groups and computed the character table of \overline{G} . This character table is given in Table 7.10. We then convert the character table to the GAP format and used Programme E (see Appendix A) to test its validity and to compute the possible power maps. We list the power maps of $2^6:A_8$ in Table 7.9.

Table 7.9: Power maps of elements of $2^6:A_8$

$[g]_{A_8}$	$[x]_{2^6:A_8}$	2	3	5	7	$[g]_{A_8}$	$[x]_{2^6:A_8}$	2	3	5	7
1A	1A					2A	2F	1A			
	2A	1A					2C	1A			
	2B	1A					2D	1A			
							$2\mathrm{E}$	1A			
							$2\mathrm{F}$	1A			
							4A	2A			
2B	2G	1A				3B	3A	1A			
	4B	2B					6C	3B	2B		
	2H	1A					6D	3B	2B		
	$4\mathrm{C}$	2B					6E	3B	2A		
	$4\mathrm{D}$	2B									
3A	3B	1A				4A	$4\mathrm{E}$	2C			
	6A	3B	2B			4F		2D			
								contin	ued on	next	page

		Tubi	0 110	(00110	maca	nom p	remous pe	(9e			
$[g]_{A_8}$	$[x]_{2^6:A_8}$	2	3	5	7	$[g]_{A_8}$	$[x]_{2^6:A_8}$	2	3	5	7
	6B	3B	2A				$4\mathrm{G}$	$2\mathrm{E}$			
							$4\mathrm{H}$	2F			
4B	4I	2G				5A	5A			1A	
	8A	4B					10A	5A		2A	
	4J	2H									
	8B	$4\mathrm{C}$									
6B	6H	3A	2C			6A	6G	3A	2G		
	6I	3B	2D				12A	6A	4B		
	6J	3B	$2\mathrm{E}$				12B	6A	$4\mathrm{C}$		
	6K	3B	2F								
7A	7A				1A	7B	7B				1A
15A	15A		5A	3B		15B	15B		5A	3B	

Table 7.9 (continued from previous page)

M(1A) =	$\begin{bmatrix} 1 & 1 & 1 \\ 28 & 4 & -4 \\ 35 & -5 & 3 \end{bmatrix}$	$M(4A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 1 & -1 & 1 & -1 \\ 1 & 1 & -1 & -$
M(2A) =	$\begin{bmatrix} 1 & 1 & 1 & 1 & 1 \\ 4 & -4 & -4 & 4 & 0 \\ 3 & 3 & 3 & 3 & -1 \\ 4 & -4 & 4 & -4 & 0 \\ 4 & 4 & -4 & -4 & 0 \end{bmatrix}$	$M(4B) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 1 & -1 & 1 & -1 \\ 1 & 1 & -1 & -$
M(2B) =	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$M(6B) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 1 & -1 & 1 & -1 \\ 1 & 1 & -1 & -$
M(6A) =	$\begin{bmatrix} 1 & 1 & 1 \\ 2 & -2 & 0 \\ 1 & 1 & -1 \end{bmatrix}$	$M(3B) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 1 & -1 & 1 & -1 \\ 1 & 1 & -1 & -$
M(3A) =	$\begin{bmatrix} 1 & 1 & 1 \\ 10 & 2 & -2 \\ 5 & -3 & 1 \end{bmatrix}$	$M(5A) = \left[\begin{array}{rrr} 1 & 1\\ 3 & -1 \end{array}\right]$
M(7A) = $M(7B) =$	1	$M(15A) = \begin{bmatrix} 1 \\ M(15B) = \begin{bmatrix} 1 \end{bmatrix}$

Table 7.8: Fischer-Clifford matrices of $2^6:A_8$

		1A				2A					2B		
	1A	2A	2B	2C	2D	2E	2F	4A	2G	4B	2H	4C	4D
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	7	7	7	-1	-1	-1	-1	-1	3	3	3	3	3
χ_3	14	14	14	6	6	6	6	6	2	2	2	2	2
χ_4	20	20	20	4	4	4	4	4	4	4	4	4	4
χ_5	21	21	21	-3	-3	-3	-3	-3	1	1	1	1	1
χ_6	21	21	21	-3	-3	-3	-3	-3	1	1	1	1	1
χ_7	21	21	21	-3	-3	-3	-3	-3	1	1	1	1	1
χ_8	28	28	28	-4	-4	-4	-4	-4	4	4	4	4	4
χ_9	35	35	35	3	3	3	3	3	-5	-5	-5	-5	-5
χ_{10}	45	45	45	-3	-3	-3	-3	-3	-3	-3	-3	-3	-3
χ_{11}	45	45	45	-3	-3	-3	-3	-3	-3	-3	-3	-3	-3
χ_{12}	56	56	56	8	8	8	8	8	0	0	0	0	0
χ_{13}	64	64	64	0	0	0	0	0	0	0	0	0	0
χ_{14}	70	70	70	-2	-2	-2	-2	-2	2	2	2	2	2
χ_{15}	28	4	-4	4	-4	-4	4	0	8	4	0	-4	0
χ_{16}	28	4	-4	-4	4	4	-4	0	4	8	-4	0	0
χ_{17}	140	20	-20	-4	4	4	-4	0	12	0	4	-8	0
χ_{18}	140	20	-20	4	-4	-4	4	0	0	12	-8	4	0
χ_{19}	140	20	-20	12	-12	-12	12	0	4	8	-4	0	0
χ_{20}	140	20	-20	-12	12	12	-12	0	8	4	0	-4	0
χ_{21}	252	36	-36	-12	12	12	-12	0	0	12	-8	4	0
χ_{22}	252	36	-36	12	-12	-12	12	0	12	0	4	-8	0
χ_{23}	280	40	-40	8	-8	-8	8	0	-16	-8	0	8	0
χ_{24}	280	40	-40	-8	8	8	-8	0	-8	-16	8	0	0
χ_{25}	448	64	-64	0	0	0	0	0	0	0	0	0	0
χ_{26}	35	-5	3	11	3	3	-5	-1	7	-5	-1	3	-1
χ_{27}	35	-5	3	3	11	-5	3	-1	-5	7	3	-1	-1
χ_{28}	35	-5	3	-5	3	3	11	-1	7	-5	-1	3	-1
χ_{29}	35	-5	3	3	-5	11	3	-1	-5	7	3	-1	-1
χ_{30}	70	-10	6	14	-2	14	-2	-2	2	2	2	2	-2
χ_{31}	70	-10	6	-2	14	-2	14	-2	2	2	2	2	-2
χ_{32}	70	-10	6	14	14	-2	-2	-2	2	2	2	2	-2
χ_{33}	70	-10	6	-2	-2	14	14	-2	2	2	2	2	-2
χ_{34}	140	-20	12	12	12	12	12	-4	4	4	4	4	-4
χ_{35}	210	-30	18	-6	-6	-6	-6	2	-10	14	6	-2	-2
χ_{36}	210	-30	18	-6	-6	-6	-6	2	14	-10	-2	6	-2
χ_{37}	315	-45	27	3	27	-21	3	-1	-9	3	-1	-5	3
χ_{38}	315	-45	27	-21	3	3	27	-1	3	-9	-5	-1	3
χ_{39}	315	-45	27	3	-21	27	3	-1	-9	3	-1	-5	3
χ_{40}	315	-45	27	27	3	3	-21	-1	3	-9	-5	-1	3
χ_{41}	420	-60	36	-12	-12	-12	-12	4	4	4	4	4	-4

Table 7.10: Character table of $2^6:A_8$

	3A				3B				4A				4B		
	3A	6A	6B	3B	6C	6D	6E	4E	4F	4G	4H	4I	8A	4J	8B
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	4	4	4	1	1	1	1	-1	-1	-1	-1	1	1	1	1
χ_3	-1	-1	-1	2	2	2	2	2	2	2	2	0	0	0	0
χ_4	5	5	5	-1	-1	-1	-1	0	0	0	0	0	0	0	0
χ_5	6	6	6	0	0	0	0	1	1	1	1	-1	-1	-1	-1
χ_6	-3	-3	-3	0	0	0	0	1	1	1	1	-1	-1	-1	-1
χ_7	-3	-3	-3	0	0	0	0	1	1	1	1	-1	-1	-1	-1
χ_8	1	1	1	1	1	1	1	0	0	0	0	0	0	0	0
χ_9	5	5	5	2	2	2	2	-1	-1	-1	-1	-1	-1	-1	-1
χ_{10}	0	0	0	0	0	0	0	1	1	1	1	1	1	1	1
χ_{11}	0	0	0	0	0	0	0	1	1	1	1	1	1	1	1
χ_{12}	-4	-4	-4	-1	-1	-1	-1	0	0	0	0	0	0	0	0
χ_{13}	4	4	4	-2	-2	-2	-2	0	0	0	0	0	0	0	0
X14	-5	-5	-5	1	1	1	1	-2	-2	-2	-2	0	0	0	0
χ_{15}	10	2	-2	1	-1	-1	1	0	0	0	0	2	-2	0	0
χ_{16}	10	2	-2	1	-1	-1	1	0	0	0	0	0	0	2	-2
χ_{17}	20	4	-4	-1	1	1	-1	0	0	0	0	0	0	-2	2
χ_{18}	20	4	-4	-1	1	1	-1	0	0	0	0	-2	2	0	0
χ_{19}	-10	-2	2	2	-2	-2	2	0	0	0	0	0	0	-2	2
χ_{20}	-10	-2	2	2	-2	-2	2	0	0	0	0	-2	2	0	0
χ_{21}	0	0	0	0	0	0	0	0	0	0	0	2	-2	0	0
χ_{22}	0	0	0	0	0	0	0	0	0	0	0	0	0	2	-2
χ_{23}	10	2	-2		-1	-1	1	0	0	0	0	0	0	0	0
χ_{24}	10	2	-2		-1	-1	1	0	0	0	0	0	0	0	0
χ_{25}	-20	-4	4	-2	2	2	-2	0	0	0	0	0	0	0	0
χ_{26}		-3	1	2	0	0	-2	3	-1	-1	-1	1	1	-1	-1
χ_{27}	5	-3	1		0	0	-2	-1	3	-1	-1	-1	-1	1	1
χ_{28}		-3	1		0	0	-2	-1	-1	3	-1	1	1	-1	-1
χ_{29}		-3	1		0	0	-2	-1	-1	-1	3	-1	-1	1	1
X30		ა ი	-1		ა ი	-3 9	-1	2	-2	-2	2	0	0	0	0
X31	-0 E	ა ე	-1	1	ა ე	-ə 9	-1	-2	2	2	-2	0	0	0	0
X32	-0	ა ე	-1 1	1	-ə 9	ა ე	-1	2	2	-2	-2	0	0	0	0
X33	-0	ა ა	-1		-3	0	-1	-2	-2	2	2	0	0	0	0
X34	15		3	-4	0	0	4	2	0 9	0 9	0 9	0	0	0	0
X35	15	-9 _0	ა ვ		0	0	0	_2	-2 2	_2 _2	-2 2	0	0	0	0
X36		-9	0		0	0	0	_1	1	-2 _1	∠ 3	1	1	-1	_1
X37		0	0		0	0	0	-1 2	-1 _1	-1 _1	ວ _1	1	1 _1	-1 1	-1 1
X38		0	0		0	0	0	_1	-1	-1	-1	1	-1	-1	_1
X39		0	0		0	0	0	_1	_1	-1	-1 _1	1	_1	-1	-1
χ ₄₁	-15	9	-3		0	0	0	0	0	0	0	0	0	0	0

Character table of $2^6: A_8$ (continued)

	5A			6A				6B		7A	7B	15A	15B
	5A	10A	6F	12A	12B	6G	6H	6I	6J	7A	7B	15A	15B
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	2	2	0	0	0	-1	-1	-1	-1	0	0	-1	-1
χ_3	-1	-1	-1	-1	-1	0	0	0	0	0	0	-1	-1
χ_4	0	0	1	1	1	1	1	1	1	-1	-1	0	0
χ_5	1	1	-2	-2	-2	0	0	0	0	0	0	1	1
χ_6	1	1	1	1	1	0	0	0	0	0	0	A	/A
χ_7	1	1	1	1	1	0	0	0	0	0	0	/A	А
χ_8	-2	-2	1	1	1	-1	-1	-1	-1	0	0	1	1
χ_9	0	0	1	1	1	0	0	0	0	0	0	0	0
χ_{10}	0	0	0	0	0	0	0	0	0	В	/B	0	0
χ_{11}	0	0	0	0	0	0	0	0	0	/B	В	0	0
χ_{12}	1	1	0	0	0	-1	-1	-1	-1	0	0	1	1
χ_{13}	-1	-1	0	0	0	0	0	0	0	1	1	-1	-1
χ_{14}	0	0	-1	-1	-1	1	1	1	1	0	0	0	0
χ_{15}	3	-1	2	-2	0	1	-1	-1	1	0	0	0	0
χ_{16}	3	-1	-2	2	0	-1	1	1	-1	0	0	0	0
χ_{17}	0	0	0	0	0	-1	1	1	-1	0	0	0	0
χ_{18}	0	0	0	0	0	1	-1	-1	1	0	0	0	0
χ_{19}	0	0	-2	2	0	0	0	0	0	0	0	0	0
χ_{20}	0	0	2	-2	0	0	0	0	0	0	0	0	0
χ_{21}	-3	1	0	0	0	0	0	0	0	0	0	0	0
χ_{22}	-3	1	0	0	0	0	0	0	0	0	0	0	0
χ_{23}	0	0	2	-2	0	-1	1	1	-1	0	0	0	0
χ_{24}	0	0	-2	2	0	1	-1	-1	1	0	0	0	0
χ_{25}	3	-1	0	0	0	0	0	0	0	0	0	0	0
χ_{26}	0	0	1	1	-1	2	0	0	-2	0	0	0	0
χ_{27}	0	0	1	1	-1	0	2	-2	0	0	0	0	0
χ_{28}	0	0	1	1	-1	-2	0	0	2	0	0	0	0
χ_{29}	0	0	1	1	-1	0	-2	2	0	0	0	0	0
χ_{30}	0	0	-1	-1	1	-1	1	-1	1	0	0	0	0
χ_{31}	0	0	-1	-1	1	1	-1	1	-1	0	0	0	0
χ_{32}	0	0	-1	-1	1	-1	-1	1	1	0	0	0	0
χ_{33}	0	0	-1	-1	1	1	1	-1	-1	0	0	0	0
χ_{34}	0	0	1	1	-1	0	0	0	0	0	0	0	0
χ_{35}	0	0	-1	-1	1	0	0	0	0	0	0	0	0
χ_{36}	0	0	-1	-1	1	0	0	0	0	0	0	0	0
χ_{37}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{38}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{39}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{40}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{41}	0	0	1	1	-1	0	0	0	0	0	0	0	0

Character table of $2^6: A_8$ (continued)

$$A = -E(15)^7 - E(15)^{11} - E(15)^{13} - E(15)^{14}$$
$$B = E(7) + E(7)^2 + E(7)^4$$

8

A Group of the Form $2^8:O_8^+(2)$ as a maximal subgroup of $O_{10}^+(2)$

Prologue

The group $\overline{G} = 2^8: O_8^+(2)$ is a group of order 44590694400. It is also a maximal subgroup of index 527 of $O_{10}^+(2)$. In turn $2^{10+16} O_{10}^+(2)$ is a maximal subgroup of the monster $M = F_1$. The group \overline{G} has three inertia factor groups namely, $O_8^+(2)$, SP(6,2) and $2^6:A_8$ of index 1, 120, and 135 respectively in $O_8^+(2)$. We first give a detailed definition of $O_8^+(2)$ and we then compute the Fischer-Clifford matrices of \overline{G} which together with the partial character tables of inertia factor groups are used to compute the full character table of \overline{G} .

8.1. Bilinear Forms

Definition 8.1.1. A symmetric bilinear form f is a function $f: V \times V \to \mathbb{F}_q$ which first satisfies *linearity* in x that is

$$f(\lambda_1 x_1 + \lambda_2 x_2, y) = \lambda_1 f(x_1, y) + \lambda_2 f(x_2, y).$$

If linearity is also satisfied in y, then we say f is bilinear. We say f is symmetric if f(x, y) = f(y, x). If f is bilinear and symmetric we say f is in symmetric bilinear form.

Definition 8.1.2. We define the quadratic form $Q: V \to \mathbb{F}_q$ to be a function satisfying

$$Q(\lambda x + \mu y) = \lambda^2 Q(x) + \lambda \mu f(x, y) + \mu^2 Q(y),$$

for some symmetric bilinear form f, called the *associated bilinear form*.

Definition 8.1.3. The *kernel* of f is the subspace of all x such that $f(x, y) = 0 \quad \forall y$. Also the kernel of the quadratic form Q, is the set of all $x \in \text{ker}(f)$ such that Q(x) = 0.

We can now define nullity, rank, singular and isotropic subspaces.

Definition 8.1.4. The *nullity* and *rank* of f is the dimension and codimension of its kernel respectively. We say f is *non-singular* if the nullity of f is zero.

A subspace W of V is said to be *totally isotropic* for f if $f(x, y) = 0 \forall x, y \in W$.

We also define the *Witt index* of a quadratic form Q as the greatest dimension of any totally isotropic subspace for Q.

Note that if any two non-singular quadratic forms Q_1 and Q_2 over \mathbb{F}_q have the same Witt index, then they are equivalent to a scalar multiple of each other. We now define the *general orthogonal* group.

8.2. Orthogonal Groups

Definition 8.2.1. The general orthogonal group GO(V, f), where f is a bilinear form, is defined as the group of linear maps g satisfying $f(u^g, v^g) = f(u, v) \quad \forall u, v \in V$. We write it as GO(n, q) or $GO_n(q)$, where V is an n-dimensional vector space over GF(q).

The elements of a general orthogonal group GO(V, f) have determinants ± 1 . This is since for if M is the matrix of the form and $g \in G(V, f)$, then $gMG^t = M$ and so $\det(g) = \det(M(G^t)^{-1}M^{-1}) = \det(g)^{-1}$. Hence we have $[\det(g)]^2 = 1$ thus $\det(g) = \pm 1$. The elements of the group that have determinant 1 is a subgroup of index 2 called the *special orthogonal group* SO(n,q). The *projective special orthogonal group* PSO(n,q) is the group obtained from SO(n,q) by factoring it by the group of scalar matrices they contain. When n = 2m + 1, that is when n is odd, all non-singular quadratic forms on a space of dimension n over \mathbb{F}_q have Witt index m and are equivalent up to scalar forms. If n = 2m, then n is even and we, up to equivalence, get two types of quadratic form namely, the *plus* type with Witt index m and the *minus* type with Witt index m - 1. Hence if n is odd we get GO(n,q) and when n is even $GO^{\epsilon}(n,q)$ with $\epsilon = +$ or $\epsilon = -$. We are more interested in n = 2m and in particular $\epsilon = +$.

For an orthogonal group we define a *reflection* r_v , for each vector $v \in V$ for which $f(v, v) \neq 0$ as elements of GO(n, q) defined by

$$r_v: x \to x - 2 \frac{f(x,v)}{f(v,v)} v.$$

In characteristic 2 if we let $\frac{1}{2}f(v,v) = Q(v)$ in the equation above, then for each vector v of norm 1 we define the *orthogonal transvection* t_v by

$$t_v: w \to w + f(w, v)v.$$

This is a linear map and preserves the quadratic form since

$$\begin{aligned} Q(w+f(w,v)v) &= \frac{1}{2}f(w+f(w,v)v,w+f(w,v)v) \\ &= \frac{1}{2}[f(w,w)+f(w,f(w,v)v)+f(f(w,v)v,w)+f(f(w,v)v,f(w,v)v)] \\ &= \frac{1}{2}[f(w,w)+f(w,f(w,v)v)+f(w,f(w,v)v)+f(f(w,v)v,f(w,v)v)] \\ &= \frac{1}{2}[f(w,w)+2f(w,f(w,v)v)+f(f(w,v)v,f(w,v)v)] \\ &= \frac{1}{2}f(w,w)+0+\frac{1}{2}f(f(w,v)v,f(w,v)v) \\ &= Q(w)+Q(f(w,v)v). \end{aligned}$$

Orthogonal groups of dimension ≥ 6 can be generated by these transvections. The quasideterminant of an element x is defined to be +1 or -1 depending on whether x can be written as a product of even or odd number of of orthogonal transvections.

Thus the quasi-determinant of an element of $GO_{2m}^+(q)$ is the sign of the permutation describing its action on this set. The kernel of the quasi-determinant map is a subgroup of index 2 in $GO_{2m}^+(q)$ which we denote $\Omega_{2m}^+(q)$. These are simple for all $m \geq 3$ and all q. We define $\Omega_{2m}^{\epsilon}(q)$ as a subgroup of index 1 or 2 in $SO_{2m}^{\epsilon}(q)$. The image of $P\Omega_{2m}^{\epsilon}(q)$ in $PSO_{2m}^{\epsilon}(q)$ is denoted $O_{2m}^{\epsilon}(q)$ and is the commutator subgroup of $SO_{2m}^{\epsilon}(q)$. Also the group $Sp_{2m-2}(q)$ is a maximal subgroup of both groups $O_{2m}^{\epsilon}(q)$. Looking at the orders of orthogonal groups in particular those of even dimension and $\epsilon = +$, we have that

$$\begin{aligned} |GO_{2m}^{+}(q)| &= \Pi_{i=1}^{m}(q^{i-1})(q^{i-1}+1)q^{2i-2} \\ &= 2q^{m(m-1)}(q^m-1)\Pi_{i=1}^{m-1}(q^{2i}-1) \end{aligned}$$

For further reading one can also go to [19, 23, 64, 52, 60, 74, 103, 106, 125].

In our case we have $O_8^+(2)$ is of index 2 in $GO_8^+(2)$ and hence $P\Omega_8^+(2) = O_8^+(2) \cong PSO_8^+(2)$. Also for the order $O_8^+(2)$ since it is of index 2 in $GO_8^+(2)$ we have

$$|O_8^+(2)| = 2^{12}(2^4 - 1)\Pi_{i=1}^3(2^{2i} - 1).$$

We are interested in the group $2^{8}:O_{8}^{+}(2)$ which is a maximal subgroup of $O_{10}^{+}(2)$. The group $O_{10}^{+}(2)$ has nine conjugacy classes of maximal subgroups. It has exactly four conjugacy classes of involutions presented in the ATLAS [23] by 2A, 2B, 2C and 2D respectively. In $O_{10}^{+}(2)$, we have $N_{O_{10}^{+}(2)}(2^{8}) = 2^{8}:O_{8}^{+}(2)$ and using the list of maximal subgroups of $O_{10}^{+}(2)$ given in the ATLAS [23], we can see that $2^{8}:O_{8}^{+}(2)$ is a maximal subgroup of $O_{10}^{+}(2)$. There are three non-equivalent 8-dimensional 2-modular representations of the group $O_{8}^{+}(2)$, with corresponding vector spaces V_{1}, V_{2} and V_{3} . Here we have

$$\dim_{GF(2)}(V_1) = 8$$
, $\dim_{GF(4)}(V_2) = \dim_{GF(4)}(V_3) = 8$.

We also have that $V_1 = 2^8$ is irreducible over GF(2) and here we are concerned with the group $V_1:O_8^+(2)$. We also note that V_2 and V_3 are irreducible over GF(4). Hence $O_{10}^+(2)$ has only one class for the maximal subgroups representatives of type $2^8:O_8^+(2)$

Let $\overline{G} = N:G$, where $N = 2^8$ is the vector space of dimension 8 over GF(2) and $G = O_8^+(2)$ acts irreducibly on N. We use the method of coset analysis, which was discussed in chapter 2, to determine the conjugacy classes of \overline{G} . We then construct the complete character table using Fischer-Clifford matrices and partial character tables of inertia factor groups. The complete fusion of $2^8:O_8^+(2)$ into $O_{10}^+(2)$ will also be fully determined. Our computations were done using GAP [41]

8.3. The action of $O_8^+(2)$ on 2^8

We generate $O_8^+(2)$ as a matrix group as the only proper normal subgroup of $GO_8^+(2)$ by three 8×8 matrices α , β and γ of orders 15, 15 and 4 as follows.

$$\alpha = \begin{pmatrix} 1 & 1 & 1 & 1 & 0 & 0 & 1 & 1 & 1 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 1 & 0 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 1 & 1 & 0 & 0 & 0 \\ 0 & 1 & 1 & 0 & 0 & 1 & 0 & 1 \\ 0 & 1 & 0 & 0 & 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 1 & 0 & 0 & 1 & 0 \end{pmatrix},$$
$$\beta = \begin{pmatrix} 1 & 0 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 0 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 & 1 \\ 0 & 0 & 0 & 1 & 0 & 0 & 1 & 0 \end{pmatrix},$$

	$\begin{pmatrix} 1 \end{pmatrix}$	0	0	0	1	0	0	0)
	0	1	0	0	0	0	1	0
	0	0	1	0	0	0	0	0
a. —	0	0	0	1	0	0	0	0
$\gamma \equiv$	0	0	0	0	1	0	0	0
	0	1	0	0	0	1	0	0
	0	0	0	0	0	0	1	0
	$\setminus 1$	0	0	0	1	0	0	1 /

We give the class representatives for each $g \in O_8^+(2)$ in terms of 8×8 matrices over GF(2) in Table 8.1 where $[g]_G$ is the class containing g and M is the matrix that represents that particular class. This is written in GAP format [41].

$[g]_G$					1	M				$[g]_G$	$[g]_G$					Λ	Λ					$ [g]_G $
	1	1	0	0	0	0	0	0	0 \			1	1	0	0	1	1	1	1	1)		
)	1	0	0	0	0	0	0				0	1	1	1	1	0	1	1		
)	0	1	0	0	0	0	0				0	0	1	0	0	0	1	0		
1)	0	0	1	0	0	0	0	1	20		0	0	0	1	1	0	1	0		2720
14)	0	0	0	1	0	0	0	1	Za		1	1	0	1	1	1	0	1		3780
)	0	0	0	0	1	0	0				0	1	1	1	0	0	1	1		
)	0	0	0	0	0	1	0				0	1	0	0	1	0	0	0		
)	0	0	0	0	0	0	1 /				1	1	1	1	1	0	0	1 /		
	()	1	0	1	0	1	1	0)			1	1	0	1	0	0	1	1	0)		
)	0	1	0	0	1	0	0				0	0	0	1	0	1	1	0		
)	0	0	0	0	1	1	0				0	1	0	1	1	1	1	1		
$^{2}\mathrm{h}$)	0	1	0	0	1	0	1	1575	20		1	0	1	1	1	1	1	0		3780
20)	1	0	1	1	0	0	1	1070	20		0	1	0	1	1	0	0	1		5100
)	0	0	0	0	1	0	0				0	0	1	1	1	1	1	0		
)	1	0	0	0	0	0	0				1	0	0	0	0	0	1	0		
		1	0	1	0	0	0	0	0 /				0	0	0	0	1	0	0	1 /		
	()	0	1	0	1	0	0	0)			1	0	0	1	0	0	0	0	0)		
)	1	0	1	1	0	0	0				1	1	0	1	1	1	0	0		
)	1	0	0	0	0	0	1				1	1	0	0	1	1	0	1		
24)	0	0	0	0	0	1	0	3780	20		1	0	1	0	0	0	0	0		56700
2u		1	0	1	0	1	0	1	0	5780	20		1	0	0	0	0	1	0	0		30700
)	1	1	1	1	1	0	0				1	0	0	0	1	0	0	0		
)	0	1	1	1	1	0	0				1	0	1	0	0	0	1	0		
		1	0	1	0	0	0	1	0 /				0	0	0	0	0	0	0	1 /		
	1	1	0	0	0	0	1	0	1			(1	1	1	1	1	0	0	0)		
)	0	1	0	1	0	0	0				1	0	0	0	1	0	0	1		
		1	0	0	0	0	0	0	0				1	1	0	1	1	1	0	1		
39		1	1	0	0	0	0	1	1	89600	3b		1	1	1	0	1	1	0	1		2240
Ja		1	0	0	0	0	0	0	1		50		0	1	0	0	0	1	0	0		2240
		1	0	1	1	1	1	1	0				1	1	1	1	0	0	1	0		
		1	0	1	0	0	0	0	1				0	1	0	0	0	1	0	1		
)	0	0	1	0	1	0	0 /				1	0	0	0	1	1	1	1 /		
																		с	onti	nued	on	next page

|--|

$[g]_G$	M								$ [g]_G $	$[g]_G$	M								$ [g]_G $
	(0	0	0	1	0	0	0	0)			(1	0	0	1	1	1	1	1	
	0	0	1	1	0	1	1	1			1	1	1	0	1	1	1	0	
	1	1	0	0	0	0	1	1			0	0	0	1	1	1	1	1	
20	1	0	0	0	0	0	0	0	268800	24	0	1	0	0	0	0	0	1	2240
30	1	0	0	0	1	0	0	1	208800	50	1	0	0	0	0	1	1	0	2240
	0	0	0	0	0	1	0	0			1	0	0	0	1	0	1	0	
	1	0	0	1	0	1	1	0			0	0	1	1	1	1	0	1	
	$\setminus 1$	0	0	1	0	0	0	1/			$\begin{pmatrix} 1 \end{pmatrix}$	0	0	0	1	1	1	1 /	
	(1	1	1	1	1	0	0	1			(0	0	0	0	0	0	0	1	
	0	1	0	0	0	0	0	1			0	0	0	1	1	0	1	0	
	1	1	0	1	1	0	1	1			0	0	0	1	1	0	0	0	
20	0	0	1	0	0	1	0	1	2240	10	0	0	0	1	0	1	0	1	27800
Je	1	1	0	0	1	1	1	0	2240	4a	0	0	0	1	0	0	0	0	37800
	0	1	1	0	0	0	0	0			0	1	1	1	1	1	0	0	
	1	0	1	1	0	0	1	1			0	1	0	0	0	0	0	0	
	<u> </u>	1	0	0	0	1	0	0 /			$\begin{pmatrix} 1 \end{pmatrix}$	0	0	1	0	0	0	0 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	1	1	1	0	0	0 \			$\begin{pmatrix} 1 \end{pmatrix}$	0	0	0	0	1	1	0 \	
	1	1	1	0	0	0	1	1			0	0	0	1	0	0	0	1	
	0	1	1	0	0	0	1	0			0	1	1	1	0	1	1	1	
46	0	1	1	0	0	0	0	1	007200	10	1	0	1	0	0	1	1	0	240200
40	1	1	1	0	0	1	1	1	907200	40	0	0	1	0	0	1	1	0	540200
	1	1	0	0	1	0	1	1			1	1	1	0	1	1	1	0	
	1	1	0	0	0	0	1	1			0	0	1	0	0	0	1	0	
	(0	1	1	1	0	1	1	1 /			(0	0	0	0	1	0	0	1 /	
	(0	0	1	0	0	1	0	1			$\int 0$	1	0	0	1	0	0	0)	
	0	0	0	0	1	0	0	0			0	0	0	1	0	1	0	0	
	0	1	0	0	1	0	1	0			0	1	0	0	0	0	0	0	
1d	1	0	0	0	0	0	0	0	907200	10	1	0	1	1	1	1	1	0	907200
44	0	0	1	0	1	0	1	0	301200	40	0	0	1	0	1	0	0	0	301200
	1	0	1	0	0	0	0	1			0	0	0	1	0	0	0	1	
	0	0	1	0	0	0	0	0			0	1	0	0	0	0	0	1	
	$\setminus 1$	0	1	1	0	0	1	1 /			(0	0	1	0	1	0	1	0 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	0	1	1	1	1	0)			$\begin{pmatrix} 1 \end{pmatrix}$	0	0	1	0	1	0	1	
	1	0	1	0	1	0	0	0			1	1	1	1	0	1	1	0	
	1	0	1	0	0	0	1	0			0	0	0	1	0	1	0	1	
4f	1	0	1	0	1	0	0	1	2721600	5a	0	1	0	1	0	1	1	0	580608
	0	1	1	0	0	1	0	1		04	1	0	1	0	1	0	0	1	000000
	0	0	0	1	0	1	0	1			0	0	0	0	0	1	0	0	
	1	1	1	0	1	1	0	1			1	0	0	0	0	0	0	1	
	(0	0	1	1	1	1	0	1 /			$\setminus 1$	0	1	0	0	1	1	0 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	0	0	1	0	1	0	1			$\begin{pmatrix} 0 \end{pmatrix}$	1	1	0	1	0	0	1	
	1	0	0	1	1	0	0	0			0	0	1	0	0	1	0	0	
	0	1	0	0	0	1	1	0			0	0	0	0	0	1	1	0	
5b	0	0	0	0	1	0	1	0	580608	50	0	0	0	1	1	0	1	0	580608
	1	1	1	1	1	0	1	0	000000		1	0	0	0	0	0	1	0	
	1	1	0	1	0	1	1	1			0	1	1	0	0	1	1	0	
	0	1	1	0	0	0	0	0			0	0	1	0	0	1	1	0	
	$\setminus 1$	1	0	0	0	1	1	1 /			$\setminus 1$	1	1	1	1	0	0	1 /	
																с	onti	nued of	n next page

Table 8.1 (continued from previous page)

$[g]_G$	M									[g] _G	M	-	0	/					$ [g]_G $
	/ 0	0	0	0	0	1	0	1 \	1101-1	1011	/ 1	0	0	1	1	0	1	0 \	10111
	0	0	1	1	0	0	1	1			0	0	1	0	1	1	1	1	
		0	1	0	0	1	0	1			0	1	1	0	1	0	1	0	
		0	0	1	0	1	0	1			1	0	0	0	1	1	1	1	
6a		1	1	1	1	0	1	1	2419200	6b	1	0	1	0	1	0	0	1	806400
		0	0	0	0	1	0	0			1	1	1	1	0	0	1	0	
		1	1	1	0	1	0	0			1	0	1	0	0	1	1	0	
		0	0	0	0	1	0	ů)				1	1	1	1	0	0	ů)	
		0	1	1	1	1	0	1)				0	1	0	0	0	0	1)	
		0	1	1	1	1	1	$\begin{pmatrix} 1 \\ 0 \end{pmatrix}$				0	1	0	0	0	1	$\begin{pmatrix} 1 \\ 0 \end{pmatrix}$	
		0	1	1	1	1	1					0	1	0	1	0	1		
		0	1	1	1	1	1					1	1	1	1	0	1	1	
6c		1	1	1	1	0	1		2419200	6d		1	1	1	1	0	1		806400
		1	1	1	1	1	0					0	1	1	1	1	1		
		1	1	1	1	1	0					1	1	1	1	1	1		
		1	0	1	1	1	0					1	1	0	0	0	0		
		1	0	1	0	0	0	0 /				0	1	0	1	0	0	0 /	
		1	1	1	0	1	0					1	0	0	0	1	0		
		0	0	0	0	1	0	0			0	0	0	1	0	1	1	0	
	0	0	0	0	0	1	0					0	0	0	0	1	1	0	
6e	1	0	1	1	0	0	1	1	806400	6f	0	1	0	0	1	1	1	1	100800
	0	1	0	1	1	1	1	1	806400		1	0	0	0	0	0	1	0	
	1	0	0	1	0	1	0	1			1	1	1	1	1	0	0	0	
		0	0	1	0	0	0	1			1	1	1	0	0	1	1	1	
	$\setminus 1$	1	1	1	1	0	1	0 /			$\setminus 1$	0	0	1	0	0	1	0 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	1	1	1	0	0	0			$\int 0$	1	0	0	0	1	1	0)	
	0	0	0	1	1	0	0	0			1	0	0	0	0	1	1	0	
	1	0	0	1	0	0	0	1			0	0	1	0	0	0	0	0	
6g	0	1	1	1	1	1	1	0	100800	6h	0	0	0	1	1	0	0	1	806400
~8	1	0	0	1	0	0	0	0			1	1	0	0	0	1	1	1	
	0	1	0	1	1	0	1	0			1	1	1	0	1	1	0	0	
	0	1	1	1	1	1	0	0			1	1	1	0	0	0	1	1	
	$\setminus 1$	0	0	1	1	0	0	1 /			$\setminus 1$	1	0	0	1	1	1	0 /	
	$\int 1$	0	0	1	0	0	0	0)			$\int 0$	0	0	1	0	1	1	0)	
	1	1	0	0	1	0	1	1			0	0	1	0	0	1	0	0	
	1	1	1	1	0	1	0	1			0	1	0	0	0	0	1	0	
61	0	0	0	0	1	0	0	0	604800	61	0	0	0	1	1	1	1	1	100800
	1	1	1	0	0	0	1	1	004000		0	0	0	0	0	1	1	0	100000
	0	0	0	1	1	0	0	0			1	0	1	0	0	0	0	1	
	1	0	0	0	0	0	1	0			0	1	0	0	0	1	1	0	
	$\setminus 1$	0	0	0	1	0	0	1 /			$\setminus 1$	0	0	1	1	1	1	1 /	
	(0	1	0	0	1	0	0	1			/ 1	0	1	1	1	1	0	1	
	0	0	1	1	0	0	1	1			1	1	0	0	1	1	0	0	
	1	0	1	0	0	0	1	0			1	0	0	0	0	1	1	0	
£1.	1	0	0	1	0	0	1	0	9410900	61	0	1	0	1	0	1	1	0	60.4900
OK	0	0	1	1	0	1	1	1	2419200	01	1	0	1	0	0	1	0	0	004800
	0	1	0	0	1	1	1	1			1	0	1	0	1	0	0	1	
	1	1	1	1	0	1	0	0			1	0	1	1	1	0	1	1	
	$ $ $ $ $ $ $ $	0	0	0	0	1	0	0 /			0	1	1	0	1	0	1	0 /	
									I		1					с	onti	nued or	n next page

Table 8.1 (continued from previous page)

$[g]_G$	M								$ [g]_G $	$[g]_G$	M								$ [g]_G $
	(0	0	0	0	1	1	1	1			(1	1	0	0	1	0	1	1)	
	0	0	0	0	0	0	1	0			0	0	1	1	1	1	0	0	
	0	0	0	1	0	1	1	0			1	0	0	0	1	1	1	1	
Green	1	1	0	0	1	1	0	0	604800	Gm	0	0	0	1	1	0	0	0	7957600
om	1	0	0	0	0	1	1	0	004800	011	0	1	0	0	0	0	1	0	7257600
	1	0	1	0	1	0	1	0			0	1	0	0	0	0	0	1	
	1	1	1	0	1	1	0	0			1	1	0	1	1	0	1	1	
	$\begin{pmatrix} 1 \end{pmatrix}$	0	0	1	0	1	1	0/			\ 0	1	1	0	0	1	0	0/	
	(0	0	0	1	0	1	1	0 \			(1	0	1	0	1	0	0	1	
	1	1	0	1	1	1	1	0			0	0	0	1	0	1	1	0	
	0	0	1	0	1	0	0	0			1	0	0	1	0	0	1	0	
80	1	1	1	1	1	0	1	0	5442200	8h	1	0	0	0	1	1	1	1	5442200
oa	0	1	0	1	1	1	1	1	0440200	00	0	0	1	1	0	0	1	1	3443200
	1	0	0	1	1	1	1	1			1	1	0	0	1	1	0	1	
	1	1	1	0	0	0	1	1			1	0	1	1	0	0	1	1	
	<u> </u>	0	1	0	0	0	0	1 /			$\begin{pmatrix} 1 \end{pmatrix}$	1	0	0	1	0	1	1 /	
	(0	0	0	1	0	0	0	0 \			(0	0	0	1	1	0	0	0 \	
	1	1	1	0	1	1	0	1			0	0	0	1	1	1	1	1	
	1	0	0	0	1	0	0	1			1	1	0	0	0	0	1	1	
0-	1	0	0	0	0	1	0	1	80000	01	0	0	0	0	1	0	1	0	6451900
9a	1	0	0	0	1	1	1	1	89600	90	1	0	0	0	1	0	1	0	6451200
	1	1	0	0	1	1	0	0			1	1	1	0	0	1	1	1	
	1	1	0	0	1	1	0	1			1	0	0	1	1	0	1	0	
	<u> </u>	0	0	1	1	1	1	1 /			$\begin{pmatrix} 1 \end{pmatrix}$	0	1	1	1	1	0	1 /	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	0	0	1	1	0	0 \			(0	1	1	0	0	0	0	1	
	0	0	1	0	1	0	0	0			1	0	1	0	0	1	0	1	
	0	0	0	0	0	0	0	1			1	1	1	1	0	1	0	0	
0.0	1	0	1	0	1	0	1	0	6451200	105	0	0	1	0	0	1	0	1	8700120
90	1	0	0	0	1	0	0	0	0451200	10a	1	0	0	0	1	1	1	1	8709120
	1	0	0	1	0	1	0	0			0	0	0	0	0	1	0	0	
	1	0	1	0	1	0	0	0			0	0	1	1	0	0	1	1	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	0	1	1	1	0	1 /			(0	0	0	0	0	1	0	1 /	
	$\begin{pmatrix} 0 \end{pmatrix}$	1	0	0	0	1	0	1			$\begin{pmatrix} 1 \end{pmatrix}$	0	1	0	0	0	1	0)	
	0	1	1	0	1	0	0	0			1	1	0	1	1	1	0	1	
	0	1	0	0	1	0	1	0			0	1	0	0	1	1	1	1	
10b	0	1	1	1	1	1	1	0	8709120	10c	1	0	1	0	0	0	0	0	8709120
100	0	0	0	1	1	1	1	1	0100120	100	0	0	0	0	1	0	1	0	0100120
	1	1	1	0	1	1	0	0			1	0	0	0	0	1	0	0	
	1	0	0	0	0	1	1	0			1	1	1	1	1	0	0	0	
	(0	1	0	1	1	1	1	1 /			$\setminus 1$	0	0	0	0	0	1	0 /	
	$\begin{pmatrix} 0 \end{pmatrix}$	1	0	1	0	1	1	0)			$\begin{pmatrix} 0 \end{pmatrix}$	0	0	1	0	1	0	0)	
	1	0	1	1	0	1	1	1			1	1	1	1	0	1	1	0	
	1	0	1	0	1	1	1	1			1	1	1	0	1	1	0	0	
12a	0	0	0	1	1	1	1	1	4838400	12b	1	0	0	1	0	1	1	0	907200
	0	1	0	1	1	0	0	1			1	1	1	1	1	0	0	1	
	0	0	1	1	1	1	1	0			1	1	0	1	0	1	1	1	
	1	1	0	1	0	0	1	1			0	1	0	0	0	1	0	1	
	\ 0	0	0	0	1	0	0	1 /			\ 0	1	1	0	1	0	0	1 /	
																с	onti	nued or	next nage

Table 8.1 (continued from previous page)

$[q]_G$	М									$[q]_G$	M	-	<u> </u>	/					$[[q]_G]$
1011	/ 0	1	1	0	0	1	1	0 \	1001-1		/ 1	1	0	0	1	1	0	0 \	1.01-1
	0	0	0	0	1	0	0	0			0	0	0	0	0	1	0	1	
	0	1	0	1	0	0	0	1			1	0	1	0	0	1	0	0	
	1	0	0	0	1	0	0	0			1	1	1	1	1	1	1	1	
12c	0	0	0	1	1	0	0	0	1209600	12d		0	1	0	0	1	0	1	7257600
	0	0	1	1	0	0	1	1			1	0	0	1	1	0	1	0	
	1	0	0	1	0	0	1	0			1	1	1	1	0	0	0	0	
		0	0	1	0	0	0	1)				0	1	0	0	0	0	1	
	$\begin{pmatrix} 0 \\ \end{pmatrix}$	1	1	1	1	0	1	1)			(1	0	0	1	1	0	0	0 \	
		0	0	0	0	0	1	$\left[\begin{array}{c} - \\ 0 \end{array} \right]$				0	0	0	0	1	0	1	
	0	0	1	0	0	0	1	0			1	0	0	1	0	0	0	0	
	1	1	1	1	1	0	1	0				1	0	1	1	1	0	0	
12e	1	0	1	0	0	0	0	0	7257600	12f		0	0	0	0	0	0	1	1209600
	0	1	1	0	0	0	1	0			0	1	1	0	0	0	1	0	
	0	0	1	0	0	1	0	0				0	0	1	0	0	0	1	
		0	0	0	1	0	0	ů)				1	1	1	1	0	0		
	(0	0	1	1	0	1	1	1)			(0	1	1	0	0	1	0	0 \	
		0	1	0	0	0	0					0	0	1	1	0	1	0	
	0	0	0	0	1	0	0	1			1	0	0	0	1	0	0	0	
	0	1	1	1	1	1	0	1				1	0	0	1	0	1	1	
12g	0	1	0	1	1	0	0	0	7257600	15a		1	1	0	1	1	0	1	11612160
	1	0	1	0	0	1	1	0			1	0	0	0	1	0	1	0	
	0	0	1	0	1	1	1	1			0	1	0	0	0	1	0	0	
		1	1	1	1	0	0	1)				0	0	1	1	0	0	0	
	, / 0	1	0	1	1	1	1	1			, / 0	0	0	1	1	1	1	1	
	0	1	1	0	0	1	0	1			0	1	1	0	0	1	0	1	
	0	1	1	1	1	1	0	1			0	0	0	0	1	1	1	1	
	1	1	0	0	1	1	0	1			0	1	1	0	0	0	0	1	
15b	0	0	1	1	0	1	1	1	11612160	15c	1	1	0	0	0	0	1	1	11612160
	0	1	0	1	1	0	0	1			0	1	0	0	0	0	0	0	
	0	0	1	1	1	1	0	1			0	1	0	0	0	1	0	1	
	$\begin{pmatrix} 1 \end{pmatrix}$	1	0	1	1	0	1	1				1	0	1	0	1	1	0	
	/ 1	0	1	1	1	0	1	1										,	
	1	1	0	0	1	1	0	1											
	1	0	0	1	0	0	1	0											
_	1	0	1	1	1	1	1	0	2 40002000										
7a	1	0	0	1	0	1	0	0	24883200										
	0	0	1	0	0	0	0	1											
	1	1	0	1	1	0	1	1											
	<u> </u>	0	1	1	1	0	1	1 /											

Table 8.1 (continued from previous page)

We obtain that $O_8^+(2)$ has 53 conjugacy classes. The action of $O_8^+(2)$ on 2^8 gives rise to three orbits of lengths 1, 120 and 135 with corresponding point stabilizers that we get from the ATLAS [23] namely, $O_8^+(2)$, $Sp_6(2)$ and 2^6 : A_8 . Let ρ_1 and ρ_2 be permutation characters of $O_8^+(2)$ of degrees 120 and 135. Then from ATLAS [23], we deduce that $\chi_{\rho_1} = 1a + 35a + 84a$ and $\chi_{\rho_2} = 1a + 50a + 84a$. Suppose $\chi = \chi(O_8^+(2)|2^8)$ is the permutation character of $O_8^+(2)$ on 2^8 . Then we get

$$\chi = 1a + I_{Sp_6(2)}^{O_8^+(2)} + I_{2^6:A_8}^{O_8^+(2)}$$

 $= 3 \times 1a + 35a + 50a + 2 \times 84a,$

where $I_{Sp_6(2)}^{O_8^+(2)}$ and $I_{2^6:A_8}^{O_8^+(2)}$ are the characters of $O_8^+(2)$ induced from the identity characters of $Sp_6(2)$ and $2^6:A_8$ respectively.

For each class representative $g \in O_8^+(2)$, we calculate the values of $\chi(Sp_6(2)|2^8)$, $\chi(2^6:A_8|2^8)$ on g and $k = \chi(g)$ on 2^8 , the number of fixed points of g on 2^8 . These are given in Table 8.2.

	$[g]_{O_8^+(2)}$	1	a	2a	2b	2c	2d	2e	3a	3b	3c	3d	3e	4a	4b	4c
χ	$(Sp_6(2) 2^8)$	1:	20	32	24	8	0	0	36	0	0	3	6	12	4	8
>	$\chi(2^6:A_8 2^8)$	1:	35	31	39	7	15	15	27	0	0	0	9	3	11	7
	k	2	56	64	64	16	16	16	64	1	1	4	16	16	16	16
	$[g]_{O_8^+(2)}$	4	4d	4e	4f	5a	5b	5c	$_{6a}$	6b	6c	6d	6e	6f	6g	6h
2	$\chi(Sp_6(2) 2^8)$		0	0	1	10	0	0	12	0	0	8	0	0	0	6
	$\chi(2^6:A_8 2^8)$		3	3	2	5	0	0	3	0	0	7	0	0	3	9
	k		4	4	4	16	1	1	16	1	1	16	1	1	4	16
	$[g]_{O_8^+}$	2)	6i	6j	6k	6l	6m	6n	7a	8a	8b	9a	9b	o 90	c 10)a
	$\chi(Sp_6(2) 2^8)$	3)	0	0	2	0	0	2	1	2	2	3	0	0	2	2
	$\chi(2^6:A_8 2^8)$	3)	3	3	1	3	3	1	2	1	1	0	0	0) 1	L
		k	4	4	4	4	4	4	4	4	4	4	1	1	4	1
	$[g]_{O_8^+(2)}$	10)b	10c	12a	12	b 1	2c	12d	12e	12f	12	g 1	5a	15b	15c
χ	$(Sp_6(2) 2^8)$	()	0	3	0		0	3	2	0	0		1	0	0
λ	$(2^6:A_8 2^8)$	()	0	0	0		0	0	1	0	0		2	0	0
	k	1	L	1	4	1		1	4	4	1	1		4	1	1

Table 8.2:

We can also check our calculations for the values of k using Programme F (see appendix). Let $k_1 = \chi(Sp_6(2)|2^8)(g), k_2 = \chi(2^6:A_8|2^8)(g)$ then $k = k_1 + k_2 + 1$, where $k_1 := cut[1] + cut[3] + cut[7], k_2 := cut[1] + cut[6] + cut[7]$, then k := 3 * cut[1] + cut[3] + cut[6] + 2 * cut[7]. Here cut[i] is the i-th row of the character table of $O_8^+(2)$ as shown in the ATLAS [23].

Having obtained the values of the k's for the various classes of G, then we need to calculate the $f'_j s$ corresponding to the various k's. For this purpose we use Programme A (see appendix) For a class representative $dg \in \overline{G}$, where $d \in 2^8$ and $g \in O_8^+(2)$ and o(g) = m, by Theorem 3.3.10 [99] we have

$$o(dg) = \begin{cases} m & \text{if } w = 1_N \\ 2m & \text{otherwise} \end{cases}$$

To calculate the orders of the class representatives $dg \in \overline{G}$, we use Programme B (see appendix)

Here if o(g) = m and $w = 1_N$ then o(dg) = m otherwise if $w \neq 1_N$, then o(dg) = 2m. Table 8.3 gives detailed information about the conjugacy classes of \overline{G}

$g \in O_8^+(2)$	k	f_j	d_{j}	w	$[x]_{2^8:O_8^+(2)}$	$ C_{2^8:O_8^+(2)}(x) $
1A	2^{8}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	1A	44 590 694 400
		120	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(1, 0, 1, 0, 1, 0, 1, 0)	2A	371 589 120
		135	$\left(0,0,0,0,0,0,0,1\right)$	$\left(1,0,0,1,0,0,0,1\right)$	2B	330 301 440
2A	2^{6}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2C	7 077 888
		6	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2D	$1\ 179\ 648$
		9	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	$2\mathrm{E}$	786 432
		48	$\left(0,0,0,1,0,0,0,0 ight)$	(1, 0, 0, 0, 0, 1, 1, 1)	4A	$147 \ 456$
2B	2^{6}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2F	2 949 120
		6	(0, 0, 0, 0, 0, 0, 1, 0)	(0, 1, 0, 1, 0, 0, 1, 0)	4B	491 520
		10	$\left(0,0,0,0,0,0,0,1\right)$	(0, 1, 0, 0, 0, 0, 0, 0)	4C	294 912
		15	(0, 0, 0, 1, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2G	196 608
		32	(0, 0, 1, 0, 0, 0, 0, 0)	(1, 0, 0, 0, 0, 1, 1, 1)	4D	92 160
2C	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2H	737 280
		15	(0, 0, 0, 0, 0, 0, 1, 0)	(1, 1, 1, 1, 1, 1, 0, 0)	4E	49 152
2D	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2I	737 280
		15	(0, 0, 0, 0, 0, 0, 1, 0)	(1, 1, 0, 1, 0, 0, 1, 0)	4F	49 152
2E	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	2J	49 152
		8	(0, 0, 0, 1, 0, 0, 0, 0)	(0, 0, 0, 1, 0, 0, 0, 0)	4G	6 144
		1	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(1, 1, 0, 1, 0, 0, 1, 0)	4H	49 152
		6	(0, 0, 0, 0, 0, 0, 0, 1)	(1, 0, 1, 0, 1, 0, 1, 0)	4I	8 192
3A	2^{6}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	3A	4 976 640
		36	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(1, 0, 1, 0, 0, 1, 0, 1)	6A	138 240
		27	(0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 1, 0, 1, 1, 1)	6B	184 320
3B	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	3B	77 760
3C	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	3C	77 760
3D	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	3D	7 776
	. 1	3	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 1, 0, 0, 0, 0)	6 <i>C</i>	2 592
3E	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	3E	10 368
		6	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(1, 1, 0, 1, 0, 0, 1, 0)	6D	1 728
	- 24	9	(0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 1, 0, 0, 0, 0)	6 <i>E</i>	1 152
4A	24	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	4J	73 728
		12	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	4K	6 144
4.0	04	3	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0)	4L	24 576
4B	24		(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	4 <i>M</i>	8 192
		4	(0, 0, 0, 1, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	41N	2 048
		1	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	40 4D	8 192 4 006
		2	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	41	4 090
4C	9 4	1	(0, 0, 0, 1, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	±~; 4R	3 072
40	<u> </u>	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	40	3 072
		2	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	45	1 024
		3	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	41	1 024
		4	(0,0,0,1,0,0,0,0) (0,0,0,1,0,0,0,1)	(0,0,0,0,0,0,0,0,0,0)	8A	768
		4	(0, 0, 1, 1, 0, 0, 0, 0)	(1, 1, 1, 1, 1, 1, 0, 0, 0)	8B	768
4D	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	4V	768
	-	3	(0, 0, 0, 0, 0, 0, 0, 0, 0, 0)	(1, 1, 0, 1, 0, 0, 1, 0)	8C	256
4E	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	4W	768
		3	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(1,0,0,1,0,0,1,0)	8D	256
4F	2^{2}	1	(0,0,0,0,0,0,0,0,0,0)	(0,0,0,0,0,0,0,0,0,0)	4X	256
		1	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 1, 0, 0, 0, 0)	8E	256
	1				continu	ied on next page

Table 8.3: Conjugacy Classes of $2^8:O_8^+(2)$

a±()			,			
$g \in O_8^+(2)$	k	f_j	d_j	w	$[x]_{2^8:O_8^+(2)}$	$ C_{2^8:O_8^+(2)}(x) $
		2	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 1, 1, 1)	8F	128
5A	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	5A	4 800
		5	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 1, 1, 1)	10A	960
		10	(0,0,0,0,0,0,0,0,1)	(0, 1, 0, 1, 0, 0, 1, 1)	10B	480
5B	1	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	5B	300
5C	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	5C	300
6A	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6F	27 648
		12	(0,0,0,0,0,0,1,0)	(1, 0, 1, 1, 1, 0, 0, 0)	12A	2 304
		3	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6G	9 216
6 <i>B</i>	1	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6 <i>H</i>	1 728
6C	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6 <i>I</i>	1 728
6D	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6J	4 608
		1	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 1, 1, 1)	12B	4 608
		3	(0, 0, 0, 1, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6K	1 536
		3	(0, 0, 0, 1, 0, 0, 0, 1)	(0, 0, 0, 1, 1, 0, 1, 1)	12C	1 536
		8	(0, 0, 1, 1, 0, 0, 0, 0)	(, 0, 1, 0, 0, 1, 0, 1)	12D	576
6E	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6L	288
6F	1	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6M	288
6G	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6N	864
		3	(0, 0, 0, 0, 0, 0, 0, 1, 0)	0, 0, 0, 0, 0, 0, 0, 0, 0)	6O	288
6H	2^{4}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6P	3 456
		6	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6Q	576
		9	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6R	384
6I	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6S	864
		3	(0, 0, 0, 0, 0, 0, 1, 0)	(1, 0, 1, 0, 0, 1, 0, 0)	12E	288
6J	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6T	864
		3	(0, 0, 0, 0, 0, 0, 1, 0)	(1, 0, 1, 1, 1, 0, 0, 0)	12F	288
6K	2^{2}	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6U	288
		1	(0,0,0,0,0,0,1,0)	(1, 0, 1, 0, 1, 0, 1, 0)	12G	288
		2	(0, 0, 0, 0, 0, 0, 0, 1)	(0, 1, 0, 1, 0, 1, 0, 1)	12H	144
6L	2^{2}	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6V	288
		3	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 1, 1, 0, 1, 1)	121	96
6M	2^{2}	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6W	288
		3	(0,0,0,0,0,0,1,0)	(,0,1,0,0,1,0,1)	12J	96
6N	22	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	6X	96
		1	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 1, 1, 0, 0, 0)	12K	96
	- 2	2	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(1,0,1,1,1,0,0,0)	12L	48
7A	22		(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	7A	28
			(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 1, 0, 0, 0, 1)	14A	28
			(0, 0, 0, 0, 0, 0, 0, 0, 1)	(1, 0, 1, 0, 1, 0, 1, 0)	14B	28
0.4	02	1	(0,0,0,1,0,0,0,0)	(0, 0, 0, 0, 0, 0, 1, 1, 1)	14C	28
8A	22		(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	8G	128
			(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	8H	128
0 D	02	2	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	01	199
85	2-		(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	8J	128
		1	(0, 0, 0, 0, 0, 0, 0, 1, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	or.	64
0.4	<u>9</u> 2	1		(0, 0, 0, 0, 0, 0, 0, 0, 0)		108
3/1	4	1 2	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)		36
0 <i>B</i>	1	1		(1, 0, 1, 0, 0, 1, 0, 0)	0R	27
9D	1	1		(0, 0, 0, 0, 0, 0, 0, 0, 0)	9D 0C	21
10.4	1 92	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	10C	80
10/4	4-	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	100	lod on mont in a
					contini	ueu on next page

Table 8.3 (continued from previous page)

$g \in O_8^+(2)$	k	f_j	d_j	w	$[x]_{2^8:O_8^+(2)}$	$ C_{2^8:O_8^+(2)}(x) $
		1	(0,0,0,0,0,0,1,0)	(1, 1, 1, 1, 1, 1, 0, 0)	20A	80
		2	(0,0,0,0,0,0,0,0,1)	(0, 0, 0, 1, 1, 1, 0, 0)	20B	40
10B	1	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	10D	20
10C	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	10E	20
12A	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12M	576
		3	(0, 0, 1, 0, 0, 1, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12N	192
12B	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	120	144
12C	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12P	144
12D	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12Q	144
		3	(0,0,1,0,0,1,0,0)	(0,0,0,0,0,0,0,0,0)	12R	48
12E	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12S	96
		1	(0,0,0,0,0,0,1,0)	(1, 0, 1, 1, 0, 0, 0, 1)	24A	96
		1	(0, 0, 0, 0, 0, 0, 1, 0, 0)	(1, 0, 1, 0, 1, 0, 1, 0)	24B	96
		1	(0,0,0,0,1,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12T	96
12F	1	1	(0,0,0,0,0,0,0,0,0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12U	24
12G	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	12V	24
15A	2^{2}	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	15A	60
		1	(0, 0, 0, 0, 0, 0, 0, 0, 1)	(1, 0, 1, 0, 1, 1, 1, 1)	30A	60
		1	(0,0,0,0,0,0,1,0)	(1,0,1,0,1,0,1,0)	30B	60
		1	(0,0,0,1,0,0,0,0)	(0, 0, 0, 0, 1, 1, 1, 1)	30C	60
15B	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	15B	15
15C	1	1	(0, 0, 0, 0, 0, 0, 0, 0, 0)	(0, 0, 0, 0, 0, 0, 0, 0, 0)	15C	15

Table 8.3 (continued from previous page)

8.4. The Character Table of $2^8:O_8^+(2)$

8.4.1 Inertia factor groups Sp(6,2) and $2^6:A_8$ and their fusion into $O_8^+(2)$

When $O_8^+(2)$ acts on 2^8 we get three orbits of conjugacy classes so that by Brauer [14] when $O_8^+(2)$ acts on $Irr(2^8)$ we also get three orbits of irreducible characters. In this case the orbit lengths of the irreducible characters are also 1,120 and 135. These have corresponding point stabilizers H_1 , H_2 and H_3 of indices 1,120 and 135 respectively. From the ATLAS [23] the corresponding inertia factor groups are $H_1 = O_8^+(2)$, $H_2 = Sp_6(2)$ and $H_3 = 2^6:A_8$. We have seen from Table 8.3 that $2^8:O_8^+(2)$ has 124 conjugacy classes and hence it has 124 irreducible characters.

We give the fusion of the conjugacy classes of $2^6:A_8$ and $Sp_6(2)$ into $O_8^+(2)$ respectively in two tables, namely Tables 8.5 and 8.6.

The work required for the fusions of $2^6:A_8$ into $O_8^+(2)$ are listed in Table 8.4. We use programme D (see appendix) to compute the fusion of SP(6,2) into $O_8^+(2)$.

							Table	e 8.4:									_
Class of	$O_8^+(2)$	1a	2a	2b	2c	2d	2e	3a	3e		4a	4b	4c	4d	4e	4f	_
Class of	$2^6:A_8$																_
1a	,	135															
2a				1													
2b			3														
2c			36														
2d					15												
2e						15											
2f	·						1										
2g	;			30													
2h							6										
3a	,							27									
3b	,								9								
4a	,											1					
4b	,										3						
4c												2					
4d													1				
4e												8					
4f																1	
4g	;													3			
4h															3		
4i													6				
4j																2	
$\chi(O_8^+(2))$	$ 2^6:A_8)$	135	39	31	15	15	7	27	9		3	11	7	3	3	3	_
Class	$of O_{1}^{+}(2)$) 5a	5b	5c	6a 6	ih 6c	6d	бе	6f	ნთ	6h	6i	6i	6k	61	6m	
Class	$rac{1}{2} of \frac{2}{2} of \frac{2}{2}$) 00	00	00	04 0	00	ou	00	01	05	011	01	oj	OK	01	0111	
	59.2.118	5															
	6a				3												
	6h				0		1										
	60 60												3				
	6d											3					
	60 60											0		1			
	6f						6							1			
	6g						0				Q						
	6b										5						
	61														3		
	6i														0		1
$\gamma(Q_2^+)$	$\frac{0}{(2) 2^6:A_8}$) 5	0	0	3	0 0	7	0	0	0	9	3	3	1	3	0	1
<u></u>	(-) 0	/ -	-							·							
	Class o	$\frac{\text{of } O_8^+(2)}{c_8^-(2)}$	7a	8a	8b	10a	12a	12b	12c	12d	1 12	e 1	2f 1	2g 1	15a	15b	
	Class o	of $2^{\circ}:A_8$															
	1	7a 	1														
	7	7b	1														
	8	3a		1													
	8	3b			1												
	1	0a				1											
	1	2a					3				_	-					
	1	2b									3			r			
	1	5a												ļ	1		
	1	5b													1		
	$\chi(O_8^+)$	$2) 2^6:A_8)$	2	1	1	1	3	0	0	0	3	()	0	2	0	

$[x]_{2^6:A_8}$	\longrightarrow	$[g_1]_{O_8^+(2)}$	$[x]_{2^6:A_8}$	\longrightarrow	$[g_1]_{O_8^+(2)}$
1A		1A	2A		2B
2B		2A	2C		2A
2D		2C	2E		2D
2F		2E	2G		2B
2H		2E	3A		3A
3B		3E	4A		4B
4B		4A	4C		4B
4D		4C	4E		4B
4F		4F	4G		4D
4H		4E	4I		4C
4J		4F	5A		5A
6A		6A	6B		6D
6C		6J	6D		6I
6E		6K	6F		6D
6G		6H	6H		6M
6I		6L	6J		6N
7A		7A	7B		7A
8A		8A	8B		8B
10A		10A	12A		12A
12B		12E	15A		15A
15B		15A			

Table 8.5: The fusion of $2^6: A_8$ into $O_8^+(2)$

Table 8.6: The fusion of SP(6,2) into $O_8^+(2)$

$[x]_{SP(6,2)}$	$\longrightarrow [g_1]_{O_8^+(2)}$	$[x]_{SP(6,2)}$	$\longrightarrow [g_1]_{O_8^+(2)}$
1 <i>A</i>	1A	2A	2B
2B	2A	2C	2B
2D	2E	3A	3A
3B	3D	3C	3E
4A	4A	4B	4C
4C	4C	4D	4B
4E	4C	6A	6D
5A	5A	6B	6A
6C	6G	6D	6D
6E	6K	6F	6H
6G	6N	7A	7A
8A	8A	8B	8B
9A	9A	10A	10A
12A	12E	12B	12E
12C	12D	15A	15A

8.4.2 The Fischer-Clifford Matrices of \overline{G}

By using the fusions of the inertia factor groups and the properties of the Fischer-Clifford matrix from chapter 5, we computed the Fischer-Clifford matrices. These are given in Table 8.7.

M(s	g)	M(g)	
$M(1A) = \begin{bmatrix} 1 & 1\\ 120 & 8\\ 135 & -9 \end{bmatrix}$	$\begin{bmatrix} 1\\ -8\\ 7 \end{bmatrix}$	$M(2C) = \begin{bmatrix} 1 & 1\\ 15 & -1 \end{bmatrix}$	
$M(2A) = \begin{bmatrix} 1 & 1 \\ 24 & 8 \\ 3 & 3 \\ 36 & -12 \end{bmatrix}$	$ \begin{array}{ccc} 1 & 1 \\ -8 & 0 \\ 3 & -1 \\ 4 & 0 \end{array} $	$M(2D) = \left[\begin{array}{rrr} 1 & 1\\ 15 & -1 \end{array} \right]$	
$M(2B) = \begin{bmatrix} 1 & 1 \\ 2 & -2 \\ 30 & 10 \\ 1 & 1 \\ 30 & -10 \end{bmatrix}$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$M(3D) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(2E) = \begin{bmatrix} 1 & 1 \\ 8 & 0 & -1 \\ 1 & -1 \\ 6 & 0 \end{bmatrix}$	$ \begin{array}{ccc} 1 & 1 \\ -8 & 0 \\ 1 & 1 \\ 6 & -2 \end{array} $	$M(4D) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(3A) = \begin{bmatrix} 1 & 1\\ 36 & -4\\ 27 & 3 \end{bmatrix}$	$\begin{bmatrix} 1\\ 4\\ -5 \end{bmatrix}$	$M(4E) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(3E) = \begin{bmatrix} 1 & 1 \\ 6 & 2 \\ 9 & -3 \end{bmatrix}$	1 -2 1	$M(6G) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(4A) = \begin{bmatrix} 1 & 1\\ 12 & 0\\ 3 & -1 \end{bmatrix}$	$\begin{bmatrix} 1 \\ -4 \\ 3 \end{bmatrix}$	$M(6I) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(4B) = \begin{bmatrix} 1 & 1 \\ 4 & 0 \\ 1 & 1 \\ 2 & -2 \\ 8 & 0 & - \end{bmatrix}$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$M(6J) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(4C) = \begin{bmatrix} 1 & 1 \\ 1 & -1 \\ 1 & 1 \\ 6 & -6 \\ 1 & -1 \\ 6 & 6 & - \end{bmatrix}$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$M(6L) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$	
$M(4F) = \begin{bmatrix} 1 & 1 \\ 1 & 1 & -2 \\ 2 & -2 \end{bmatrix}$	$\begin{bmatrix} 1 \\ -1 \\ 0 \end{bmatrix}$	$M(7A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 \\ 1 & -1 & -$	
$M(5A) = \begin{bmatrix} 1 & 1 \\ 10 & 2 \\ 5 & -3 \end{bmatrix}$	$\begin{bmatrix} 1\\ -2\\ 1 \end{bmatrix}$	$M(6M) = \begin{bmatrix} 1 & 1\\ 3 & -1 \end{bmatrix}$ continued on next part	age

Table 8.7: The Fischer-Clifford Matrices of $2^8:O_8^+(2)$

M(g)	M(g)
$M(6A) = \begin{bmatrix} 1 & 1 & 1 \\ 12 & 0 & -4 \\ 3 & -1 & 3 \end{bmatrix}$	$M(9A) = \begin{bmatrix} 1 & 1 \\ 3 & -1 \end{bmatrix}$
$M(6D) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 \\ 2 & -2 & 2 & -2 & 0 \\ 6 & 6 & -2 & -2 & 0 \\ 1 & 1 & 1 & 1 & -1 \\ 6 & -6 & -2 & 2 & 0 \end{bmatrix}$	$M(12A) = \begin{bmatrix} 1 & 1\\ 3 & -1 \end{bmatrix}$
$M(6H) = \begin{bmatrix} 1 & 1 & 1 \\ 6 & 2 & -2 \\ 9 & -3 & 1 \end{bmatrix}$	$M(12D) = \begin{bmatrix} 1 & 1\\ 3 & -1 \end{bmatrix}$
$M(6K) = \begin{bmatrix} 1 & 1 & 1 \\ 2 & -2 & 0 \\ 1 & 1 & -1 \end{bmatrix}$	$M(3B) = M(3C) = \begin{bmatrix} 1 \end{bmatrix}$
$M(6N) = \begin{bmatrix} 1 & 1 & 1 \\ 2 & -2 & 0 \\ 1 & 1 & -1 \end{bmatrix}$	$M(5A) = M(5B) = \begin{bmatrix} 1 \end{bmatrix}$
$M(8A) = \begin{bmatrix} 1 & 1 & 1 \\ 2 & -2 & 0 \\ 1 & 1 & -1 \end{bmatrix}$	$M(6B) = M(6C) = M(6E) = M(6F) = \begin{bmatrix} 1 \end{bmatrix}$
$M(8B) = \begin{bmatrix} 1 & 1 & 1 \\ 2 & -2 & 0 \\ 1 & 1 & -1 \end{bmatrix}$	$M(9B) = M(9C) = \begin{bmatrix} 1 \end{bmatrix}$
$M(10A) = \begin{bmatrix} 1 & 1 & 1 \\ 2 & -2 & 0 \\ 1 & 1 & -1 \end{bmatrix}$	$M(10B) = M(10C) = \begin{bmatrix} 1 \end{bmatrix}$
$M(12E) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 \\ 1 & -1 & -$	$M(12B) = M(12C) = M(12F) = M(12G) = \begin{bmatrix} 1 \end{bmatrix}$
$M(15A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 1 & 1 & -1 & -$	$M(15B) = M(15C) = \begin{bmatrix} 1 \end{bmatrix}$

Table 8.7 (continued from previous page)

We use programme D (see appendix) to give the fusion of $2^8:O_8^+(2)$ into $O_{10}^+(2)$ in Table 8.8.

$[g]_{2^8:O_8^+(2)}$	\longrightarrow	$[y]_{O_{10}^+(2)}$	$[g]_{2^8:O_8^+(2)}$	\longrightarrow	$[y]_{O_{10}^+(2)}$
1A		1A	2A		2B
2B		2A	2C		2A
2D		2D	2E		2C
2F		2B	2G		2D
2H		2C	2I		2C
2J		2D	3A		3A
3B		3C	3C		3C
			conti	nued or	n next page

Table 8.8: The fusion of $2^8:O_8^+(2)$ into $O_{10}^+(2)$

	Sie 8.8 (continued	from previous	page)
$[g]_{2^8:O_8^+(2)}$	\longrightarrow $[y]_{O_{10}^+(2)}$	$[g]_{2^8:O_8^+(2)}$ -	\rightarrow $[y]_{O_{10}^+(2)}$
3D	3D	3E	3B
4A	4B	4B	4A
4C	4B	4D	4C
4E	4D	4F	4D
4G	4F	4H	4D
41	4E	4J	4A
4K	4G	4L	4E
4M	4B	4N	4I
40	4E	4P	4D
4Q	4H	4R	4C
45	4G	4T	4I
4U	4F	4V	4H
4W	4H	4X	4I
5A	5A	5B	5B
5C	5B	6A	6B
68	6A	6C	6G
6 <i>D</i>	61	6F	6 <i>A</i>
6E	6F	6G	6D
6 <i>H</i>	6 <i>H</i>	6 <i>I</i>	6 <i>H</i>
6.1	6B	6K	6D
6L	6L	6M	6L
6N	6E	60	6M
6P	6C	60	6 <i>K</i>
68	6 I	65	61
6T	6 <i>F</i>	6U	61
6V	61 61	6W	61
6 X	6 <i>K</i>	74	7 A
84	8B	8B	84
80	8C	80	8C
86	8C	8E	80 80
	8.4	84	80
8G	8D	8H	84
81	8 <i>E</i>	81	8 <i>B</i>
81 8 <i>K</i>	8E 8D	85 97	8D 8 F
	8D 0 4	08	08
	9A 0 <i>P</i>	3D 10.4	3D 10 A
90 10 <i>B</i>	9D 10B	10A 10C	10A 10B
10D	100	100	10D
10D	10C	10 <i>E</i>	10C
127	12D 19A	120	12D 19D
	12A 19E	195	12D 12E
	12E 19E	12F 19F	12E 12C
120	1 <i>4E</i> 197	197	120
121 19 <i>K</i>	121	197	1 <i>21</i> 19 <i>1</i> 2
12Λ	121	12L 19 M	12A 19E
12/1/	12A 1911	1211	12F 19U
120	12H	12F	12 <i>H</i> 19 <i>I</i>
12Q	120	12K	12J
125	12D	121	12F
	12L	127	12L
	14A	148	148
140	14C	15A	15C
15B	15E	15C	15 <i>E</i>
		continue	ed on next page

Table 8.8 (continued from previous page)

240		(comunada	menn previe	up pug	,e)
$[g]_{2^8:O_8^+(2)}$	\longrightarrow	$[y]_{O_{10}^+(2)}$	$[g]_{2^8:O_8^+(2)}$	\longrightarrow	$[y]_{O_{10}^+(2)}$
18A		18A	20A		20A
20B		20B	24A		24A
24B		24B	30A		30C
30B		30F	30C		30D

Table 8.8 (continued from previous page)

We use programme E (see appendix) to give the power maps of elements of $2^8:O_8^+(2)$ in Table 8.9.

$[g]_{S_8}$	$[x]_{2^8:O_8^+(2)}$	2	3	5	7	$[g]_{S_8}$	$[x]_{2^8:O_8^+(2)}$	2	3	5	7
1A	1A	1A				2A	2C	1A			
	2A	1A					2D	1A			
	$2\mathrm{B}$	1A					2D	1A			
							$2\mathrm{E}$	1A			
							4A	2A			
2B	$2\mathrm{F}$	1A				2E	2J	1A			
	4B	2B					$4\mathrm{G}$	2A			
	$4\mathrm{C}$	2B					$4\mathrm{H}$	2B			
	2G	1A					4I	2B			
	4D	2A									
2C	2H	1A				2D	2I	1A			
	$4\mathrm{E}$	2B					$4\mathrm{F}$	2B			
3A	3A	3A	1A			3D	3D	3D	1A		
	6A	3A	2A				$6\mathrm{C}$	3D	2B		
	6B	3A	2B								
3B	3B	3B	1A			3C	3C	3C	1A		
3E	3E	3E	1A			4A	4J	2C			
	6D	3E	2A				4K	2D			
	$6\mathrm{E}$	3E	2B				4L	2C			
4B	$4\mathrm{M}$	2C				4C	4R	2F			
	4N	2D					4S	2G			
	40	2C					$4\mathrm{T}$	2G			
	4P	2C					$4\mathrm{U}$	2F			
	4Q	$2\mathrm{E}$					8A	$4\mathrm{C}$			
							8B	4B			
4D	4V	2H				4E	4W	2I			
	8C	$4\mathrm{E}$					8D	$4\mathrm{F}$			
4F	4X	$2\mathrm{E}$				5A	5A	5A	5A	1A	
	8E	$4\mathrm{H}$					10A	5A	5A	2B	
	8F	4G					10B	5A	5A	2A	
5B	5B	5B	5B	1A		5C	$5\mathrm{C}$	$5\mathrm{C}$	$5\mathrm{C}$	1A	
6A	6F	3A	2C			6D	6 J	3A	2F		
	12A	6B	4A				12B	6A	$4\mathrm{C}$		
	6G	3A	2D				6K	3A	2G		
							12C	6A	4B		
							12D	6B	4D		
								contin	ued on	next pa	age

Table 8.9: The Power Maps of elements of $2^8:O_8^+(2)$

[g]ss [x]_{2^8.O_h^+(2)} 2 3 5 7 [g]ss [x]_{2^8.O_h^+(2)} 2 3 5 7 6B 6H 3B 2C 6C 6I 3C 2C 6E 6L 3B 2C 6C 6I 3C 2C 6H 6P 3E 2C 6G 6N 3D 2D 6I 6P 3E 2C 6J 6T 3E 2C 6I 6R 3E 2 6J 6T 3E 2C 6I 6S 3E 2 6J 6T 3E 2C 6I 6U 3E 2 6L 6L 3E 2I 6I 6L 3E 2 6L 6L 3E 2I						nueu	nom pr	evious page)		1		1
6B 6H 3B 2C 6C 6I 3C 2C 6E 6L 3B 2H 6F 6M 3C 2I 6H 6P 3E 2C 6G 6M 3D 2C 6R 3E 2C 6G 6M 3D 2D 60 3D 2C 6I 6S 3E 2C 6G 6T 3E 2C 6G 3D 2D 6I 6S 3E 2C 6J 6T 3E 2C 6D 4A 2D	$[g]_{S_8}$	$[x]_{2^8:O_8^+(2)}$	2	3	5	7	$[g]_{S_8}$	$[x]_{2^8:O_8^+(2)}$	2	3	5	7
6E 6L 3B 2H 6F 6M 3C 2I 6H 6P 3E 2C 6G 6N 3D 2C 6Q 3E 2D 6G 6N 3D 2C 6R 3E 2D 61 6N 3E 2C 12E 6E 4A 12F 6E 4A 12G 6E 4A 12I 6E 4A 12G 6E 4C 12I 6E 4E 12I 6N 6X 3E 2J 6M 6W 3E 2I 6N 6X 3E 2J 6M 6W 3E 2I 12H 6D 4G 4D 12J 6E 4F 12J 12K 6D 4G 2J 6M 6W 3E 2I 12K 6D 4G 4H 2A 12J 6E 4F 12K 6D 4G 14A 2A 18A 9A 3D I <tr< td=""><td>6B</td><td>6H</td><td>3B</td><td>2C</td><td></td><td></td><td>6C</td><td>6I</td><td>3C</td><td>2C</td><td></td><td></td></tr<>	6B	6H	3B	2C			6C	6I	3C	2C		
6H 6P 3E 2C 6G 6N 3D 2C 6Q 3E 2D 6O 6O 3D 2D 6R 3E 2E 60 6T 3E 2C 12E 6E 4A 12F 6E 4A 12E 6E 4C 12F 6E 4A 12G 6E 4C 12I 6E 4E 12H 6D 4D 12I 6E 4F 12H 6D 4D 12I 6E 4F 12L 6E 4H 2A 12I 6B 3D 14A 7A 7A 7A 7A 4A 2A 18A 9A 14B 7A 14A 2A 18A 9A	6E	6L	3B	2H			6F	6M	3C	2I		
6Q 3E 2D 60 3D 2D 61 68 3E 2E 61 6T 3E 2C 61 68 3E 2F 61 6T 3E 2C 6K 6U 3E 2F 6L 6V 3E 2H 12G 6E 4C 12I 6E 4E 12I 6E 4E 12H 6D 4D - 12I 6E 4F 12I 6E 4F 12K 6D 4G - 6M 6W 3E 2I - 12K 6D 4G - 6M 6W 3E 2I - 12K 6D 4G - 6M 6W 3E 2I - 12K 6D 4G - 6M 6W 3E 2I - 12K 6D 4G - 12K 6M 3D - - 7A 7A 7A 1A 14K 2A 18K<	6H	6P	3E	2C			6G	6N	3D	2C		
6R $3E$ $2E$ $6I$ $6T$ $3E$ $2C$ $6I$ $6S$ $3E$ $2C$ $6E$ $4A$ $12F$ $6E$ $4A$ $6K$ $6U$ $3E$ $2F$ C $12F$ $6E$ $4A$ $12G$ $6E$ $4C$ C $12I$ $6E$ $4E$ C $12H$ $6D$ $4D$ C $12I$ $6E$ $4F$ C $12K$ $6D$ $4D$ C $12I$ $6E$ $4F$ C $12K$ $6D$ $4G$ C $12I$ $6E$ $4F$ C $12K$ $6D$ $4G$ C $12I$ $6E$ $4F$ C $12L$ $6E$ $4H$ C $12K$ $6E$ $4F$ C $14K$ $7A$ $7A$ $7A$ $7A$ $7A$ $4A$ $2A$ AB $9A$ $3D$ C $9B$ $9B$ $9B$ $3D$ C BE AD AD		6Q	3E	2D				6O	3D	2D		
6I 6S 3E 2C 6J 6T 3E 2C 12E 6E 4A 12F 6E 4A 6K 6U 3E 2F 12I 6E 4A 12G 6E 4C 12I 6E 4C 12H 6D 4D 12I 6E 4F 12H 6D 4G 12I 6E 4F 12K 6E 4H 12I 6E 4F 12K 6E 4H 12I 6E 4F 12L 6E 4H 12I 6M 3D 14A 7A 7A 7A 1A 9A 9A 14B 7A 14E 2B 18A 9A 9C 9B 9B 9B 9B 9B 9B 9B 9B 9A 4G 12I 12I 12I 3D		6R	3E	$2\mathrm{E}$								
12E 6E 4A 12F 6E 4A 6K 6U 3E 2F 6L 6V 3E 2H 12G 6E 4C 12I 6E 4E 12I 6E 4E 12H 6D 4D	6I	6S	3E	2C			6J	6T	3E	2C		
6K 6U 3E 2F Image: Constraint of the sector o		12E	6E	4A				12F	6E	4A		
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	6K	$6\mathrm{U}$	3E	$2\mathrm{F}$			6L	6V	3E	2H		
12H 6D 4D $4D$ $6N$ $6X$ $3E$ $2J$ $6M$ $6W$ $3E$ $2I$ 12K $6D$ $4G$ $12J$ $6E$ $4F$ $12J$ $6E$ $4F$ 12L $6E$ $4H$ $12J$ $6E$ $4F$ $12J$ $6E$ $4F$ 7A $7A$ $7A$ $7A$ $7A$ $7A$ $9A$ $9A$ $9D$ $3D$ 14A $7A$ $14A$ $14A$ $2A$ $18A$ $9A$ $6C$ C 9B $9B$ $3D$ C BB BJ C		12G	6E	$4\mathrm{C}$				12I	6E	$4\mathrm{E}$		
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		12H	6D	4D								
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	6N	6X	3E	2J			6M	6W	3E	2I		
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$		12K	6D	$4\mathrm{G}$				12J	6E	$4\mathrm{F}$		
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		12L	6E	$4\mathrm{H}$								
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	7A	7A	7A	7A	7A	1A	9A	9A	9A	3D		
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		14A	7A	14A	14A	2A		18A	9A	6C		
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$		14B	7A	14B	14B	2B						
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$		14C	7A	14C	14C	2B						
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	9B	9B	9B	3D			9C	9C	9C	3D		
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	8A	8G	4J				8B	8J	4M			
$\begin{array}{ c c c c c c c c c c } 81 & 4L & & & & & & & & & & & & & & & & & $		$8\mathrm{H}$	4K					8K	4N			
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		81	4L					8L	4P			
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	10A	10C	5A	10C	2F		12A	12M	6F	4J		
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$		20A	10A	20A	4B			12N	6G	4L		
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		20B	10B	20B	4D							
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	10B	10D	5B	10D	2H		10C	10E	$5\mathrm{C}$	10E	2I	
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	12B	12O	6H	4J			12C	12P	6I	4J		
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	12D	12Q	6N	4J			12E	12S	6J	4R		
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		12R	6N	4L				24A	12B	8A		
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$								24B	12C	8B		
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$								12T	6K	4S		
$ \begin{array}{ccccccccccccccccccccccccc$	12F	12U	6L	4V			12G	12V	6M	4W		
30A 15A 10B 6B 30B 15A 10A 6A 15C 15C 5C 3C 30C 15A 10A 6A 15C 15C 15C 3C	15A	15A	15A	5A	3A		15B	15B	15B	5B	3B	
30B 15A 10A 6A 15C 15C 15C 5C 3C 30C 15A 10A 6A 15C		30A	15A	10B	6B							
30C 15A 10A 6A		30B	15A	10A	6A		15C	15C	15C	$5\mathrm{C}$	3C	
		30C	15A	10A	6A							

Table 8.9 (continued from previous page)

To compute the character table of $2^8:O_8^+(2)$, as an example consider the following. Let $C_1(2B), C_2(2B), C_3(2B)$ be the partial character tables of the inertia factors for the classes that fuse to $2B \in O_8^+(2)$. The portions of the character table of $\overline{G} = 2^8:O_8^+(2)$ corresponding to the coset 2B are :

		1						г ,	1	1		1
	1								1	1	1	1
	4							4	4	4	4	4
	11							11	11	11	11	11
	-5							-5	-5	-5	-5	-5
	-5							-5	-5	-5	-5	-5
	10							10	10	10	10	10
	20							20	20	20	20	20
	4							4	4	4	4	4
	4							4	4	4	4	4
	15							15	15	15	15	15
	26							26	26	26	26	26
	10							10	10	10	10	10
	10							10	10	10	10	10
	20							20	20	20	20	20
	-10							-10	-10	-10	-10	-10
	э 20							5	о 20	о 20	о 20	о 20
	39							39	39	39	39	39
	-9							-9	-9	-9	-9	-9
	-9							-9	-9	-9	-9	-9
	20							20	20	20	20	20
	00							00	00	00	00	00
	-20							-20	-20	-20	-20	-20
	-20							-20	-20	-20	-20	-20
	24							24	24	24	24	24
	-40							-40	-40	-40	-40	-40
C(2R)M(2R) =	-40	l	-		-	-	1_	-40	-40	-40	-40	-40
$C_1(2B)M_1(2B) =$	36		1	1	1	1] =	36	36	36	36	36
	20							30	50 20	50 20	50 20	50 20
	-30							-30	-30	-30	-30	-30
	-30							-30	-30	-30	-30	-30
	04							04	04	04	04	04
	0							0	0	0	0	0
	40							40	40	40	40	40
	15							15	40	40	40	40
	15							15	15	15	15	15
	15							15	15	15	15	15
	-60							-60	-60	-60	-60	-60
	20							20	20	20	20	20
	20							20	20	20	20	20
	64							64	<u>-</u> 0 64	<u>-</u> 0 64	<u>-</u> 0 64	6 4
	0							0	0	0	0	0
	0							Ő	0	0	Ő	0
	12							12	12	12	12	12
	-36							-36	-36	-36	-36	-36
	-36							-36	-36	-36	-36	-36
	51							51	51	51	51	51
	-45							-45	-45	-45	-45	-45
	-45							-45	-45	-45	-45	-45
	0							0	0	0	0	0
	Ő							ő	Ő	Ő	ő	Ő
	40							40	40	40	40	40
	-45							-45	-45	-45	-45	-45

$C_2(2B)M_2(2B) =$	$\left[\begin{array}{c} 1\\ -5\\ -5\\ 9\\ -11\\ 15\\ -5\\ 15\\ -24\\ -10\\ 4\\ -35\\ 25\\ 5\\ 40\\ 40\\ 21\\ -51\\ -39\\ 50\\ 10\\ -24\\ 40\\ -45\\ -16\\ -30\\ 45\\ 20\\ 0\\ \end{array}\right]$	$1 \\ 3 \\ 3 \\ 1 \\ 5 \\ 7 \\ -5 \\ 7 \\ 8 \\ 6 \\ 4 \\ 5 \\ 9 \\ -3 \\ 8 \\ 8 \\ -11 \\ 12 \\ 10 \\ 8 \\ -8 \\ 8 \\ 3 \\ -16 \\ 2 \\ -3 \\ -12 \\ 0 \\ 0 \\ 1 \\ 2 \\ -12 \\ 0 \\ 0 \\ 1 \\ 1 \\ 2 \\ -12 \\ 0 \\ 0 \\ 1 \\ 1 \\ 1 \\ 1 \\ 2 \\ -12 \\ 0 \\ 0 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 2 \\ -12 \\ 0 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1 \\ 1$		$-2 \\ 10$	-2 - 6	2 -2	$\begin{bmatrix} 0\\0 \end{bmatrix} =$	$\left[\begin{array}{cccc} 32\\ 80\\ 80\\ 48\\ 128\\ 240\\ -160\\ 240\\ 192\\ 160\\ 128\\ 80\\ 320\\ -288\\ 320\\ -288\\ 320\\ -288\\ 288\\ -48\\ 160\\ 320\\ 192\\ -320\\ 320\\ 0\\ -512\\ 0\\ 0\\ -512\\ 0\\ 0\\ -320\\ 0\\ 0\\ 0\\ -320\\ 0\\ 0\\ 0\\ -320\\ 0\\ 0\\ 0\\ 0\\ -320\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0\\ 0$	$\begin{array}{c} 8\\ 40\\ 40\\ -8\\ 72\\ 40\\ -40\\ 40\\ 128\\ 80\\ 32\\ 120\\ 40\\ -40\\ 0\\ 0\\ -152\\ 232\\ 88\\ -80\\ 80\\ 128\\ 80\\ 128\\ 80\\ -120\\ -128\\ 80\\ -120\\ -160\\ 0\\ 0\end{array}$	$\begin{array}{c} -8\\ -8\\ -8\\ -8\\ -24\\ -8\\ -72\\ 40\\ -72\\ 0\\ -72\\ 40\\ -72\\ -82\\ 40\\ -104\\ 8\\ -128\\ -128\\ -128\\ 24\\ 72\\ -112\\ -80\\ 0\\ 128\\ -128\\ -82\\ 128\\ 48\\ -72\\ 32\\ 0\\ \end{array}$	$egin{array}{c} 0 \\ -16 \\ -16 \\ 16 \\ 0 \\ 16 \\ -32 \\ 16 \\ 0 \\ 0 \\ -80 \\ 32 \\ 16 \\ 64 \\ 64 \\ 64 \\ -128 \\ -80 \\ 96 \\ 0 \\ -64 \\ -64 \\ 64 \\ -96 \\ 0 \\ -64 \\ 96 \\ 64 \\ 0 \\ 0 \\ \end{array}$	0 0 0	
$C_3(2B)M_3(2B) =$	$\begin{array}{c} 1\\ 7\\ 14\\ 20\\ 21\\ 21\\ 21\\ 21\\ 28\\ 35\\ 45\\ 56\\ 64\\ 70\\ 4\\ 4\\ -5\\ -5\\ -5\\ -5\\ -5\\ -10\\ -10\\ -10\\ -10\\ -10\\ -10\\ 20\\ 20\\ -20\\ 20\\ 20\\ -20\\ 20\\ -30\\ 36\\ 36\\ 40\\ 40\\ -45\\ -45\\ -45\\ -45\\ -60\\ 64\\ \end{array}$	$\begin{array}{c}1&&&\\&3&&\\&&&\\&&&\\&&&\\&&&\\&&&\\&&&\\&&&\\&&$	$\begin{bmatrix} 1\\ 30 \end{bmatrix}$	1 -10	1 6	1 -2	$\begin{bmatrix} -1\\ 0 \end{bmatrix} =$	$\left[\begin{array}{c} 31\\ 97\\ 74\\ 140\\ 51\\ 51\\ 148\\ -115\\ -45\\ -45\\ -45\\ -45\\ -45\\ 205\\ 205\\ 205\\ -155\\ -155\\ -155\\ 50\\ 50\\ 50\\ 50\\ 20\\ 380\\ 100\\ 260\\ 140\\ 390\\ -300\\ 36\\ 396\\ -440\\ -200\\ 45\\ 45\\ -315\\ -315\\ 60\\ 64\\ \end{array}\right.$	$\begin{array}{c} -9\\ -23\\ -6\\ -20\\ 11\\ 11\\ 11\\ -12\\ 85\\ 75\\ 55\\ 66\\ 4\\ 50\\ -76\\ 64\\ 50\\ -76\\ -36\\ -36\\ -75\\ 45\\ 45\\ -30\\ -30\\ -30\\ -30\\ -30\\ -30\\ -30\\ -30$	$\begin{array}{c} 7\\ 25\\ 26\\ 44\\ 27\\ 27\\ 52\\ 5\\ 27\\ 27\\ 56\\ 64\\ 82\\ 52\\ 28\\ 37\\ -35\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\ 2\\$	$\begin{array}{c} -1 \\ 1 \\ 10 \\ 12 \\ 19 \\ 19 \\ 19 \\ 20 \\ 45 \\ 51 \\ 51 \\ 56 \\ 64 \\ 66 \\ -12 \\ -4 \\ -19 \\ -19 \\ 5 \\ 5 \\ -14 \\ -14 \\ -14 \\ -14 \\ 20 \\ -4 \\ -28 \\ 4 \\ 12 \\ -58 \\ -10 \\ 36 \\ 12 \\ 72 \\ 56 \\ -51 \\ -27 \\ -68 \\ 64 \end{array}$	$\begin{array}{c} -1\\ -7\\ -14\\ -20\\ -21\\ -21\\ -21\\ -21\\ -21\\ -21\\ -21\\ -21$	

We used the Fischer-Clifford matrices and partial character tables of inertia factor groups and computed the character table of \overline{G} . This character table is given in Table 8.10. We converted this character table to the GAP format and used Programme E (see appendix) to test its validity and to compute the power maps.

8.4.3 The Character Table of \overline{G}

		1A			2A					2B			2C		2D	
	1A	2A	2B	2C	2D	2E	4A	2F	4B	4C	2G	4D	2H	4E	2I	4F
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	28	28	28	-4	-4	-4	-4	4	4	4	4	4	4	4	4	4
χ_3	35	35	35	3	3	3	3	11	11	11	11	11	-5	-5	-5	-5
χ_4	35	35	35	3	3	3	3	-5	-5	-5	-5	-5	11	11	-5	-5
χ_5	35	35	35	3	3	3	3	-5	-5	-5	-5	-5	-5	-5	11	11
χ_6	50	50	50	18	18	18	18	10	10	10	10	10	10	10	10	10
χ_7	84	84	84	20	20	20	20	20	20	20	20	20	4	4	4	4
χ_8	84	84	84	20	20	20	20	4	4	4	4	4	20	20	4	4
χ_9	84	84	84	20	20	20	20	4	4	4	4	4	4	4	20	20
χ_{10}	175	175	175	-17	-17	-17	-17	15	15	15	15	15	15	15	15	15
χ_{11}	210	210	210	-14	-14	-14	-14	26	26	26	26	26	10	10	10	10
χ_{12}	210	210	210	-14	-14	-14	-14	10	10	10	10	10	26	26	10	10
χ_{13}	210	210	210	-14	-14	-14	-14	10	10	10	10	10	10	10	26	26
χ_{14}	300	300	300	12	12	12	12	20	20	20	20	20	20	20	20	20
χ_{15}	350	350	350	-2	-2	-2	-2	-10	-10	-10	-10	-10	-10	-10	-10	-10
χ_{16}	525	525	525	45	45	45	45	5	5	5	5	5	5	5	5	5
χ_{17}	567	567	567	-9	-9	-9	-9	39	39	39	39	39	-9	-9	-9	-9
χ_{18}	567	567	567	-9	-9	-9	-9	-9	-9	-9	-9	-9	39	39	-9	-9
χ_{19}	567	567	567	-9	-9	-9	-9	-9	-9	-9	-9	-9	-9	-9	39	39
χ_{20}	700	700	700	92	92	92	92	20	20	20	20	20	20	20	20	20
χ_{21}	700	700	700	-4	-4	-4	-4	60	60	60	60	60	-20	-20	-20	-20
χ_{22}	700	700	700	-4	-4	-4	-4	-20	-20	-20	-20	-20	60	60	-20	-20
χ_{23}	700	700	700	-4	-4	-4	-4	-20	-20	-20	-20	-20	-20	-20	60	60
χ_{24}	840	840	840	8	8	8	8	24	24	24	24	24	-40	-40	-40	-40
χ_{25}	840	840	840	8	8	8	8	-40	-40	-40	-40	-40	24	24	-40	-40
χ_{26}	840	840	840	8	8	8	8	-40	-40	-40	-40	-40	-40	-40	24	24
χ_{27}	972	972	972	108	108	108	108	36	36	36	36	36	36	36	36	36
χ_{28}	1050	1050	1050	58	58	58	58	50	50	50	50	50	-30	-30	-30	-30
χ_{29}	1050	1050	1050	58	58	58	58	-30	-30	-30	-30	-30	50	50	-30	-30
χ_{30}	1050	1050	1050	58	58	58	58	-30	-30	-30	-30	-30	-30	-30	50	50
χ_{31}	1344	1344	1344	64	64	64	64	64	64	64	64	64	0	0	0	0
χ_{32}	1344	1344	1344	64	64	64	64	0	0	0	0	0	64	64	0	0
χ_{33}	1344	1344	1344	64	64	64	64	0	0	0	0	0	0	0	64	64
χ_{34}	1400	1400	1400	-72	-72	-72	-72	40	40	40	40	40	40	40	40	40
χ_{35}	1575	1575	1575	-57	-57	-57	-57	15	15	15	15	15	15	15	15	15
χ_{36}	1575	1575	1575	-57	-57	-57	-57	15	15	15	15	15	15	15	15	15
χ_{37}	1575	1575	1575	-57	-57	-57	-57	15	15	15	15	15	15	15	15	15
χ_{38}	2100	2100	2100	52	52	52	52	-60	-60	-60	-60	-60	20	20	20	20
χ_{39}	2100	2100	2100	52	52	52	52	20	20	20	20	20	-60	-60	20	20
χ_{40}	2100	2100	2100	52	52	52	52	20	20	20	20	20	20	20	-60	-60
χ_{41}	2240	2240	2240	-64	-64	-64	-64	64	64	64	64	64	0	0		0
χ_{42}	2240	2240	2240	-64	-64	-64	-64		0	0	0	0	64	64		0
χ_{43}	2240	2240	2240	-64	-64	-64	-64	0	0	0	0	0	0	0	64	64
χ_{44}	2268	2268	2268	-36	-36	-36	-36	12	12	12	12	12	-36	-36	-36	-36

Table 8.10: The Character Table of $2^8:O_8^+(2)$

		1A			2A					2B			2C		2D	
	1A	2A	2B	2C	2D	2E	4A	2F	4B	4C	2G	4D	2H	4E	2I	4F
χ_{45}	2268	2268	2268	-36	-36	-36	-36	-36	-36	-36	-36	-36	12	12	-36	-36
χ_{46}	2268	2268	2268	-36	-36	-36	-36	-36	-36	-36	-36	-36	-36	-36	12	12
χ_{47}	2835	2835	2835	-45	-45	-45	-45	51	51	51	51	51	-45	-45	-45	-45
χ_{48}	2835	2835	2835	-45	-45	-45	-45	-45	-45	-45	-45	-45	51	51	-45	-45
χ_{49}	2835	2835	2835	-45	-45	-45	-45	-45	-45	-45	-45	-45	-45	-45	51	51
χ_{50}	3200	3200	3200	128	128	128	128	0	0	0	0	0	0	0	0	0
χ_{51}	4096	4096	4096	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{52}	4200	4200	4200	-24	-24	-24	-24	40	40	40	40	40	40	40	40	40
χ_{53}	6075	6075	6075	27	27	27	27	-45	-45	-45	-45	-45	-45	-45	-45	-45
χ_{54}	120	8	-8	24	8	-8	0	32	8	-8	0	0	0	0	0	0
χ_{55}	840	56	-56	-24	-8	8	0	80	40	-8	-16	0	0	0	0	0
χ_{56}	1800	120	-120	168	56	-56	0	80	40	-8	-16	0	0	0	0	0
χ_{57}	2520	168	-168	-72	-24	24	0	48	-8	-24	16	0	0	0	0	0
χ_{58}	2520	168	-168	120	40	-40	0	128	72	-8	-32	0	0	0	0	0
χ_{59}	3240	216	-216	72	24	-24	0	240	40	-72	16	0	0	0	0	0
χ_{60}	4200	280	-280	72	24	-24	0	-160	-40	40	0	0	0	0	0	0
χ_{61}	4200	280	-280	264	88	-88	0	240	40	-72	16	0	0	0	0	0
χ_{62}	6720	448	-448	-192	-64	64	0	192	128	0	-64	0	0	0	0	0
χ_{63}	8400	560	-560	-240	-80	80	0	160	80	-16	-32	0	0	0	0	0
χ_{64}	10080	672	-672	480	160	-160	0	128	32	-32	0	0	0	0	0	0
χ_{65}	12600	840	-840	24	8	-8	0	80	120	40	-80	0	0	0	0	0
χ_{66}	12600	840	-840	-168	-56	56	0	320	40	-104	32	0	0	0	0	0
χ_{67}	12600	840	-840	408	136	-136	0	-80	-40	8	16	0	0	0	0	0
χ_{68}	14400	960	-960	-192	-64	64	0	320	0	-128	64	0	0	0	0	0
χ_{69}	20160	1344	-1344	192	64	-64	0	320	0	-128	64	0	0	0	0	0
χ_{70}	22680	1512	-1512	-72	-24	24	0	-288	-152	24	64	0	0	0	0	0
χ_{71}	22680	1512	-1512	-72	-24	24	0	288	232	24	-128	0	0	0	0	0
χ_{72}	22680	1512	-1512	504	168	-168	0	-48	88	72	-80	0	0	0	0	0
χ_{73}	25200	1680	-1680	48	16	-16	0	160	-80	-112	96	0	0	0	0	0
χ_{74}	25200	1680	-1680	-336	-112	112	0	320	80	-80	0	0	0	0	0	0
χ_{75}	25920	1728	-1728	576	192	-192	0	192	128	0	-64	0	0	0	0	0
χ_{76}	33600	2240	-2240	-192	-64	64	0	-320	0	128	-64	0	0	0	0	0
χ_{77}	33600	2240	-2240	576	192	-192	0	320	0	-128	64	0	0	0	0	0
χ_{78}	37800	2520	-2520	-504	-168	168	0	0	120	72	-96	0	0	0	0	0
χ_{79}	40320	2688	-2688	384	128	-128	0	-512	-128	128	0	0	0	0	0	0
χ_{80}	45360	3024	-3024	-144	-48	48	0	0	80	48	-64	0	0	0	0	0
χ_{81}	48600	3240	-3240	-648	-216	216	0		-120	-72	96	0		0		0
χ_{82}	50400	3360	-3360	96	32	-32	0	-320	-160	32	64	0		0		0
X83	61440	4096	-4096		0	0	0	0	0	0	0	0	0	0		0
χ_{84}	135	-9	7	39	-9	7	-1		-9	7	-1	-1	15	-1	15	-1
χ_{85}	945	-63	49	-15	33	17	-7		-23	25	10	-7	-15	1	-15	1
χ_{86}	1890	-126	98	258	-30	66	-14	140	-6	26	10	-14	90	-6	90	-6
χ_{87}	2700	-180	140	204	12	76	-20	140	-20	44	12	-20	60	-4	60	-4
χ_{88}	2835	-189	147	-45	99	51	-21	51	11	27	19	-21	-45	3	-45	3

The character table of $2^8:O_8^+(2)$ (continued)

		1A			2A					2B			2C		2D	
	1A	2A	2B	2C	2D	2E	4A	2F	4B	4C	2G	4D	2H	4E	2I	4F
χ_{89}	2835	-189	147	-45	99	51	-21	51	11	27	19	-21	-45	3	-45	3
X90	2835	-189	147	-45	99	51	-21	51	11	27	19	-21	-45	3	-45	3
χ_{91}	3780	-252	196	-60	132	68	-28	148	-12	52	20	-28	-60	4	-60	4
χ_{92}	4725	-315	245	213	69	117	-35	-115	85	5	45	-35	45	-3	45	-3
X93	6075	-405	315	27	171	123	-45	-45	75	27	51	-45	-45	3	-45	3
χ_{94}	6075	-405	315	27	171	123	-45	-45	75	27	51	-45	-45	3	-45	3
χ_{95}	7560	-504	392	456	72	200	-56	56	56	56	56	-56	120	-8	120	-8
χ_{96}	8640	-576	448	192	192	192	-64	64	64	64	64	-64	0	0	0	0
χ_{97}	9450	-630	490	138	234	202	-70	130	50	82	66	-70	-30	2	-30	2
χ_{98}	3780	-252	196	132	-60	4	4	244	-76	52	-12	-4	-60	4	-60	4
χ_{99}	3780	-252	196	-156	36	-28	4	124	-36	28	-4	-4	60	-4	60	-4
χ_{100}	4725	-315	245	405	-123	53	-3	205	-75	37	-19	5	45	-3	45	-3
χ_{101}	4725	-315	245	-171	69	-11	-3	205	-75	37	-19	5	45	-3	45	-3
χ_{102}	4725	-315	245	117	-27	21	-3	-155	45	-35	5	5	-75	5	165	-11
χ_{103}	4725	-315	245	117	-27	21	-3	-155	45	-35	5	5	165	-11	-75	5
χ_{104}	9450	-630	490	-54	42	10	-6	50	-30	2	-14	10	-30	2	210	-14
χ_{105}	9450	-630	490	-54	42	10	-6	50	-30	2	-14	10	210	-14	-30	2
χ_{106}	9450	-630	490	522	-150	74	-6	50	-30	2	-14	10	-30	2	210	-14
χ_{107}	9450	-630	490	522	-150	74	-6	50	-30	2	-14	10	210	-14	-30	2
χ_{108}	18900	-1260	980	84	-108	-44	20	20	20	20	20	-20	-60	4	-60	4
χ_{109}	18900	-1260	980	-204	-12	-76	20	380	-100	92	-4	-20	60	-4	60	-4
χ_{110}	18900	-1260	980	468	-108	84	-12	100	-60	4	-28	20	180	-12	180	-12
X111	18900	-1260	980	-492	84	-108	20	260	-60	68	4	-20	180	-12	180	-12
X112	18900	-1260	980	372	-204	-12	20	140	-20	44	12	-20	-180	12	-180	12
χ_{113}	28350	-1890	1470	-162	126	30	-18	390	-170	54	-58	30	-90	6	-90	6
X114	28350	-1890	1470	-162	126	30	-18	-330	70	-90	-10	30	-90	6	-90	6
χ_{115}	34020	-2268	1764	-540	36	-156	36	36	36	36	36	-36	180	-12	180	-12
χ_{116}	34020	-2268	1764	324	-252	-60	36	396	-84	108	12	-36	-180	12	-180	12
χ_{117}	37800	-2520	1960	168	-216	-88	40	-440	200	-56	72	-40	-120	8	-120	8
χ_{118}	37800	-2520	1960	-408	-24	-152	40	-200	120	-8	50	-40	120	-8	120	-8
X119	42525	-2835	2205	-075	333	-3	-27	45	-75	-27	-51	45	45	-3	45	-3
X120	42525	-2833	2205	1003	-243	189	-21	40	-() 45	-21	-01 07	40	40	-3 97	40 915	-3 01
X121	42525	-2835	2205	189	45	93	-27	-315	45	-99	-27	45 45	405	-21	-315	21
χ ₁₂₂	42525	-2835	2205	189	45	93	-27	-315	45	-99 26	-21	45	-315	21 19	405	-27
X123	60490	-3/80	2940	102	202	100	-30	64	-100	-30	-08	60 64	-180	12	-180	12
X124	60480	-4032	3136	-192	-192	-192	64	64	64	64	64	-64	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

		2E				3A		3B	3C	3D			3E			4A	
	2J	4G	4H	4I	3A	6A	6B	3B	3C	3D	6C	3E	6D	6E	4J	4K	4L
χ1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	-4	-4	-4	-4	10	10	10	10	10	1	1	1	1	1	8	8	8
χ3	3	3	3	3	14	14	14	5	5	-1	-1	2	2	2	7	7	7
χ_4	3	3	3	3	5	5	5	14	5	-1	-1	2	2	2	7	7	7
χ_5	3	3	3	3	5	5	5	5	14	-1	-1	2	2	2	7	7	7
χ_6	2	2	2	2	5	5	5	5	5	-4	-4	5	5	5	-2	-2	-2
χ7	4	4	4	4	21	21	21	-6	-6	3	3	3	3	3	4	4	4
χ_8	4	4	4	4	-6	-6	-6	21	-6	3	3	3	3	3	4	4	4
X 9	4	4	4	4	-6	-6	-6	-6	21	3	3	3	3	3	4	4	4
χ_{10}	-1	-1	-1	-1	-5	-5	-5	-5	-5	13	13	4	4	4	-1	-1	-1
χ_{11}	2	2	2	2	39	39	39	-15	-15	-6	-6	3	3	3	6	6	6
χ_{12}	2	2	2	2	-15	-15	-15	39	-15	-6	-6	3	3	3	6	6	6
χ_{13}	2	2	2	2	-15	-15	-15	-15	39	-6	-6	3	3	3	6	6	6
χ_{14}	12	12	12	12	30	30	30	30	30	3	3	-6	-6	-6	8	8	8
χ_{15}	-2	-2	-2	-2	35	35	35	35	35	-1	-1	-1	-1	-1	26	26	26
χ_{16}	-19	-19	-19	-19	30	30	30	30	30	12	12	3	3	3	-7	-7	-7
χ_{17}	-9	-9	-9	-9	81	81	81	0	0	0	0	0	0	0	15	15	15
χ_{18}	-9	-9	-9	-9	0	0	0	81	0	0	0	0	0	0	15	15	15
χ_{19}	-9	-9	-9	-9	0	0	0	0	81	0	0	0	0	0	15	15	15
χ_{20}	-4	-4	-4	-4	-20	-20	-20	-20	-20	-2	-2	7	7	7	0	0	0
χ_{21}	12	12	12	12	55	55	55	10	10	7	7	4	4	4	-4	-4	-4
χ_{22}	12	12	12	12	10	10	10	55	10	7	7	4	4	4	-4	-4	-4
χ_{23}	12	12	12	12	10	10	10	10	55	7	7	4	4	4	-4	-4	-4
χ_{24}	8	8	8	8	-24	-24	-24	30	30	3	3	3	3	3	16	16	16
χ_{25}	8	8	8	8	30	30	30	-24	30	3	3	3	3	3	16	16	16
χ_{26}	10	10	10	10	30	30	30	30	-24	3	3		3	3	16	16	16
χ_{27}	12	12	12	12		15	15		15	0	0		0	0		10	10
X28	-0 6	-0 6	-0 6	-0 6	15	15	15	15	15	-3	-3 9	6	0 6	0 6	-10	-10	-10
X29	-0 6	-0 6	-0 6	-0 6	15	15	10	10	10		-ə 9	G	6	6	-10	-10	-10
X30	-0	-0	-0	-0	10	10	10	10	10	-3	-3 6	G	6	6	-10	-10	-10
χ31 χ31	0	0	0	0	04 94	04 94	04 94	-24	-24	-0	-0 6	-0	-0	-0		0	0
X32	0	0	0	0	24	-24 24	-24	24	-24	-0	-0	-0	-0	-0		0	0
X33	-8	-8	-8	-8	50	-24 50	-24 50	50	50	-0	-0	-0	-0	-0	-16	-16	-16
X34	-0	-0	-0	-0	90	90	90	-45	-45	-4	-4	0	0	0	11	-10	11
X35	-9	-9	-9	-9	-45	-45	-45	90	-45	9	9	0	0	0		11	11
1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	_9	-9	_9	-9	-45	-45	-45	-45	90	g	9	0	0	0	11	11	11
X31 Y20	4	-5	-5	-0	75	75	75	-60	-60	-6	-6		3	3	12	12	12
100 120	4	4	4	4	-60	-60	-60	75	-60	-6	-6	3	3	3	12	12	12
×40	4	4	4	4	-60	-60	-60	-60	75	-6	-6	3	3	3	12	12	12
χ ₄₁	0	- 0	0	- 0	-4	-4	-4	-40	-40	-10	-10		2	2		0	0
×41 ×42	0	0	0	0	-40	-40	-40	-4	-40	-10	-10		2	2		0	Ő
χ ₄₂	0	0	0	0	-40	-40	-40	-40	-4	-10	-10	2	2	2		0	õ
χ ₄₄	12	12	12	12	81	81	81	0	0	0	0	0	0	0	-12	-12	-12

The character table of $2^8:O_8^+(2)$ (continued)

		2E				3A		3B	3C	3D			3E			4A	
	2J	4G	4H	4I	3A	6A	6B	3B	3C	3D	6C	3E	6D	6E	4J	4K	4L
χ_{45}	12	12	12	12	0	0	0	81	0	0	0	0	0	0	-12	-12	-12
χ_{46}	12	12	12	12	0	0	0	0	81	0	0	0	0	0	-12	-12	-12
χ_{47}	3	3	3	3	-81	-81	-81	0	0	0	0	0	0	0	3	3	3
χ_{48}	3	3	3	3	0	0	0	-81	0	0	0	0	0	0	3	3	3
χ_{49}	3	3	3	3	0	0	0	0	-81	0	0	0	0	0	3	3	3
χ_{50}	0	0	0	0	-40	-40	-40	-40	-40	14	14	-4	-4	-4	0	0	0
χ_{51}	0	0	0	0	64	64	64	64	64	-8	-8	-8	-8	-8	0	0	0
χ_{52}	8	8	8	8	-30	-30	-30	-30	-30	15	15	-3	-3	-3	-8	-8	-8
χ_{53}	-21	-21	-21	-21	0	0	0	0	0	0	0	0	0	0	-9	-9	-9
χ_{54}	8	0	-8	0	36	-4	4	0	0	3	-1	6	2	-2	12	0	-4
χ_{55}	-8	0	8	0	144	-16	16	0	0	-6	2	6	2	-2	36	0	-12
χ_{56}	-8	0	8	0	0	0	0	0	0	-9	3	18	6	-6	-12	0	4
χ_{57}	-24	0	24	0	216	-24	24	0	0	9	-3	0	0	0	60	0	-20
χ_{58}	-24	0	24	0	216	-24	24	0	0	9	-3	0	0	0	12	0	-4
χ_{59}	24	0	-24	0	324	-36	36	0	0	0	0	0	0	0	36	0	-12
χ_{60}	24	0	-24	0	180	-20	20	0	0	-3	1	12	4	-4	84	0	-28
χ_{61}	24	0	-24	0	180	-20	20	0	0	-3	1	12	4	-4	-12	0	4
χ_{62}	0	0	0	0	396	-44	44	0	0	6	-2	12	4	-4	0	0	0
χ_{63}	-16	0	16	0	-180	20	-20	0	0	21	-7	6	2	-2	24	0	-8
χ_{64}	32	0	-32	0	-216	24	-24	0	0	9	-3	18	6	-6	48	0	-16
χ_{65}	8	0	-8	0	540	-60	60	0	0	-9	3	-18	-6	6	60	0	-20
χ_{66}	8	0	-8	0	0	0	0	0	0	18	-6	18	6	-6	-36	0	12
χ_{67}	-56	0	56	0	0	0	0	0	0	18	-6	18	6	-6	-36	0	12
χ_{68}	0	0	0	0	540	-60	60	0	0	-18	6	0	0	0	0	0	0
χ_{69}	64	0	-64	0	216	-24	24	0	0	18	-6	-18	-6	6	0	0	0
χ_{70}	-24	0	24	0	324	-36	36	0	0	0	0		0	0	108	0	-36
χ_{71}	-24	0	24	0	324	-30	30	0	0	0	0		0	0	-30	0	12
<i>χ</i> 72	-24	0	24	0	54	-30	30 60	0	0	0	0		0	0	-30	0	12
<i>χ</i> 73	-48	0	48	0	540	-60	60	0	0	10	-3	10	0	0	-24	0	8 24
X74	10	0	-10	0	-540	26	-00	0	0	-18	0	10	0	-0	12	0	-24
X75	64	0	64	0	-524	30 40	-30	0	0	30	10	6	0 2	0 9	0	0	0
X76	04	0	-04	0	180	-40	20	0	0	24	-10	19	4	-2	0	0	0
X77	24	0	-24	0	-180	20	-20	0	0	-24 -27	0	-12	-4	4	-60	0	20
X78	0	0	-24	0	216	-24	24	0	0	-18	6		0	0	-00	0	20
X19 X00	-48	0	48	0	-324	-24	-36	0	0	0	0	0	0	0	72	0	-24
180 Vo1	-24	0	24	0		0	0	0	0	0	0		0	0	-36	0	12
181 Veo	32	0	-32	0		0	0	0	0	_9	3	18	6	-6	-48	0	16
X82	0	0	0	0	-576	64	-64	0	0	24	-8	-24	-8	8	0	0	0
100 Yo1	7	-1	7	-1	27	3	-5	0	0	0	0	9	-3	1	3	-1	3
λ04 X85	17	1	17	-7	108	12	-20	0	0	0	0	9	-3	1	9	-3	9
X86	18	-6	18	2	-27	-3	5	0	Ő	0	õ	18	-6	2	6	-2	6
X87	28	-4	28	-4	135	15	-25	0	0	0	0 0	-9	3	-1	12	-4	12
X88	3	3	3	-5	162	18	-30	0	0	0	0	0	0	0	3	-1	3

The character table of $2^8:O_8^+(2)$ (continued)

		2E				3A		3B	3C	3D			3E			4A	
	2J	4G	4H	4I	3A	6A	6B	3B	3C	3D	6C	3E	6D	6E	4J	4K	4L
χ_{89}	3	3	3	-5	-81	-9	15	0	0	0	0	0	0	0	3	-1	3
χ_{90}	3	3	3	-5	-81	-9	15	0	0	0	0	0	0	0	3	-1	3
χ_{91}	20	4	20	-12	27	3	-5	0	0	0	0	9	-3	1	12	-4	12
χ_{92}	-27	-3	-27	13	135	15	-25	0	0	0	0	18	-6	2	-15	5	-15
χ_{93}	-21	3	-21	3	0	0	0	0	0	0	0	0	0	0	-9	3	-9
χ_{94}	-21	3	-21	3	0	0	0	0	0	0	0	0	0	0	-9	3	-9
χ_{95}	8	-8	8	8	-108	-12	20	0	0	0	0	-9	3	-1	0	0	0
χ_{96}	0	0	0	0	108	12	-20	0	0	0	0	-18	6	-2	0	0	0
χ_{97}	10	2	10	-6	-135	-15	25	0	0	0	0	9	-3	1	6	-2	6
χ_{98}	4	-4	4	4	270	30	-50	0	0	0	0	9	-3	1	12	-4	12
χ_{99}	-28	4	-28	4	270	30	-50	0	0	0	0	9	-3	1	24	-8	24
χ_{100}	-11	5	-11	-3	135	15	-25	0	0	0	0	18	-6	2	-15	5	-15
χ_{101}	5	-11	5	13	135	15	-25	0	0	0	0	18	-6	2	-15	5	-15
χ_{102}	21	-3	21	-3	135	15	-25	0	0	0	0	18	-6	2	21	-7	21
χ_{103}	21	-3	21	-3	135	15	-25	0	0	0	0	18	-6	2	21	-7	21
χ_{104}	26	-14	26	10	-135	-15	25	0	0	0	0	9	-3	1	6	-2	6
χ_{105}	10	2	10	-6	-135	-15	25	0	0	0	0	9	-3	1	6	-2	6
χ_{106}	10	2	10	-6	-135	-15	25	0	0	0	0	9	-3	1	6	-2	6
χ_{107}	-44	-4	-44	20	540	60	-100	0		0	0	-9	3	-1	36	-12	36
χ_{108}	20	4	20	-12	540	60	-100	0		0	0	-9	3	-1	10	0	10
χ_{109}	30	-12	30	4	135	15	-25	0		0	0	-30	12	-4	12	-4	12
X110	-12	12	-12 19	-12	-270	-30	50	0		0	0	18	-0 6	2	12	-4	12
X111	-12	-12	-12 18	20	-270	-30	50 75	0		0	0	10	-0	2	24	-0 10	24 30
X112 X112	30	6	-10	-18	405	45	-75	0	0	0	0	0	0	0	42	-14	-30 42
X113 X114	-60	12	-60	4	0	0	0	0	0	0	0 0	0	0	0	36	-12	36
χ115 X115	36	-12	36	4	0	0	0	0	0	0	0	0	0	0	0	0	0
X116	8	-8	8	8	270	30	-50	0	0	0	0	9	-3	1	-24	8	-24
χ_{117}	40	8	40	-24	270	30	-50	0	0	0	0	9	-3	1	-48	16	-48
χ_{118}	-3	-27	-3	37	0	0	0	0	0	0	0	0	0	0	-27	9	-27
χ_{119}	-51	21	-51	-11	0	0	0	0	0	0	0	0	0	0	-27	9	-27
χ_{120}	-27	21	-27	-19	0	0	0	0	0	0	0	0	0	0	-3	1	
χ_{121}	-3	-3	-3	5	0	0	0	0	0	0	0	0	0	0	9	-3	9
χ_{122}	-3	-3	-3	5	0	0	0	0	0	0	0	0	0	0	9	-3	9
χ_{123}	12	12	12	-20	-405	-45	75	0	0	0	0	0	0	0	12	-4	12
χ_{124}	0	0	0	0	-540	-60	100	0	0	0	0	-18	6	-2	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

			4B					4C				4D		4E			$4\mathrm{F}$	
	4M	4N	4O	4P	4Q	4R	4S	4T	4U	8A	8B	4V	8C	4W	8D	4X	8E	8F
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X3	-1	-1	-1	-1	-1	3	3	3	3	3	3	-1	-1	-1	-1	-1	-1	-1
χ_4	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	3	3	-1	-1	-1	-1	-1
χ_5	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	3	3	-1	-1	-1
χ_6	6	6	6	6	6	2	2	2	2	2	2	2	2	2	2	2	2	2
χ7	4	4	4	4	4	4	4	4	4	4	4	0	0	0	0	0	0	0
χ_8	4	4	4	4	4	0	0	0	0	0	0	4	4	0	0	0	0	0
X9	4	4	4	4	4	0	0	0	0	0	0	0	0	4	4	0	0	0
X10	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	3	3	3
<i>χ</i> 11	-2	-2	-2	-2	-2		2	2	2	2	2	-2	-2	-2	-2	2	2	2
X12	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	2	2	-2	-2		2	2
X13	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2		2		2	2
X14	0	0	0	0	0		0	0	0	0	0	0	0		0		0	0
X15	1	2	2 1	1	1	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2		1	2 1
X16	1	1	1	1	1	-5	-5	-5	-5	-5	-5	-0	-J 2	-0	-0 2	1	1	1
χ_{17}	-1	-1	-1	-1	-1	-1	-1 3	-1 3	-1 3	-1	-1 3	-1	-1	3	े २	1	-1	-1
X18 X10	-1	-1	-1	-1	-1	3	3	3	3	3	3	3	3	-1	-1	-1	-1	-1
×19 ×20	8	8	8	8	8	0	0	0	0	0	0	0	0	0	0	0	0	0
X21	-4	-4	-4	-4	-4	4	4	4	4	4	4	0	0	0	0	0	0	0
χ_{22}	-4	-4	-4	-4	-4	0	0	0	0	0	0	4	4	0	0	0	0	0
χ ₂₃	-4	-4	-4	-4	-4	0	0	0	0	0	0	0	0	4	4	0	0	0
χ_{24}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{25}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{26}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{27}	8	8	8	8	8	0	0	0	0	0	0	0	0	0	0	0	0	0
X28	-2	-2	-2	-2	-2	2	2	2	2	2	2	-2	-2	-2	-2	-2	-2	-2
χ_{29}	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	2	2	-2	-2	-2	-2	-2
X30	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	2	2	-2	-2	-2
χ_{31}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X32	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X33	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X34	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{35}	3	3	3	3	3	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_{36}	3	3	3	3	3	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
X37	3	3	3	3	3	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
X38	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4		0		0		0	0
X39	-4	-4	-4	-4	-4		0	0	0	0	0	-4	-4		0		0	0
χ_{40}	-4	-4	-4	-4	-4		0	0	0	0	0		0	-4	-4		0	0
X41	0	0	0	0	0		0	0	0	0	0	0	0		0		0	0
χ_{42}	0	0	0	0	0		0	0	0	0	0	0	0		0		0	0
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	4	4	4	4	4	-4	-4	-4	-4	-4	-4	0	0	0	0		0	0

The character table of $2^8:O_8^+(2)$ (continued)
			4B					4C				4D		$4\mathrm{E}$			4F	
	4M	4N	4O	4P	4Q	4R	4S	4T	4U	8A	8B	4V	8C	4W	8D	4X	8E	8F
χ_{45}	4	4	4	4	4	0	0	0	0	0	0	-4	-4	0	0	0	0	0
χ_{46}	4	4	4	4	4	0	0	0	0	0	0	0	0	-4	-4	0	0	0
χ_{47}	3	3	3	3	3	-5	-5	-5	-5	-5	-5	3	3	3	3	-1	-1	-1
χ_{48}	3	3	3	3	3	3	3	3	3	3	3	-5	-5	3	3	-1	-1	-1
χ_{49}	3	3	3	3	3	3	3	3	3	3	3	3	3	-5	-5	-1	-1	-1
χ_{50}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{51}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{52}	-8	-8	-8	-8	-8	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{53}	-1	-1	-1	-1	-1	3	3	3	3	3	3	3	3	3	3	3	3	3
χ_{54}	4	0	4	-4	0	8	-6 10	0	2	0	-2	0	0	0	0	0	0	0
χ_{55}	-4	0	-4 19	4 10	0	4	-10	-4	-2	4	2	0	0	0	0	0	0	0
χ56 χ		0	12	-12	0	4	-2 10	-4	0 2	-4	2	0	0	0	0	0	0	0
X57	4	0	4	-4 _1	0	-4	-6	-8	2	-4	-2	0	0	0	0	0	0	0
X58	-4	0	-4	-4	0	12	-0	-0	6	-4	-6	0	0	0	0	0	0	0
X 59	-4	0	-4	4	0	-8	6	0	-2	0	2	0	0	0	0	0	0	0
X61	12	0	12	-12	Õ	12	-10	4	-2	4	-6	0	0	0	0	0	0	0
χ_{62}	0	0	0	0	0	0	-8	0	-8	8	0	0	0	0	0	0	0	0
χ ₆₃	8	0	8	-8	0	-8	12	8	-4	0	-4	0	0	0	0	0	0	0
χ_{64}	16	0	16	-16	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{65}	4	0	4	-4	0	-12	2	-4	-6	4	6	0	0	0	0	0	0	0
χ_{66}	-12	0	-12	12	0	0	-6	-8	2	0	6	0	0	0	0	0	0	0
χ_{67}	4	0	4	-4	0	-4	2	4	-6	4	-2	0	0	0	0	0	0	0
χ_{68}	0	0	0	0	0	0	8	0	8	-8	0	0	0	0	0	0	0	0
χ_{69}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X70	4	0	4	-4	0	8	-6	0	2	0	-2	0	0	0	0	0	0	0
<i>χ</i> 71	-12	0	-12	12	0	8	-6	0	2	0	-2	0	0	0	0	0	0	0
<i>χ</i> 72	4	0	4	-4	0	-12	10	-4	2	-4	6	0	0	0	0	0	0	0
<i>χ</i> 73	-8	0	-8	8	0	-8	12	8	-4	0	-4	0	0	0	0	0	0	0
χ_{74}	-8	0	-8	8	0	-16	12	0	-4	0	4	0	0	0	0	0	0	0
χ_{75}	0	0	0	0	0	0	8	0	8	-8	0	0	0	0	0	0	0	0
χ_{76}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ77	10	0	10	10	0	0	-8	0	-8	8	0	0	0	0	0	0	0	0
X78	12	0	12	-12	0	0	6	8	-2	0	-6	0	0	0	0	0	0	0
<i>χ</i> 79		0	0	0	0	16	10	0	0	0	0	0	0	0	0	0	0	0
X80	-0	0	-0 20	0 20	0	10	-12	0	4	0	-4	0	0	0	0	0	0	0
X81	20	0	20 16	-20 16	0		-0	-0	2	0	0	0	0	0	0	0	0	0
X82	0	0	-10 0	10	0		0	0	0	0	0	0	0	0	0		0	0
183 Vo4	11	-1	-5	3	1	7	5	-1	-3	1	1	3	-1	3	1	3	1	
Λ84 Υοπ	-7	5	-0	1	-3	9	3	1	-5	-3	3	-3	1	-3	1	1	-3	1
100 X86	30	-10	-2	14	-2		-2	2	-2	-2	2	6	-2	6	-2	2	2	-2
χ ₈₇	12	-4	- 12	12	-4	4	-4	4	-4	-4	4	0	0	0	0	0	0	0
χ_{88}	3	7	-13	-5	-1	-5	-7	3	1	-1	1	3	-1	3	-1	-1	3	-1

The character table of $2^8:O_8^+(2)$ (continued)

			4B					4C				4D		$4\mathrm{E}$			$4\mathrm{F}$	
	4M	4N	4O	4P	4Q	4R	4S	4T	4U	8A	8B	4V	8C	4W	8D	4X	8E	8F
χ_{89}	3	7	-13	-5	-1	-5	-7	3	1	-1	1	3	-1	3	-1	-1	3	-1
χ_{90}	3	7	-13	-5	-1	-5	-7	3	1	-1	1	3	-1	3	-1	-1	3	-1
X91	-4	12	-4	-4	-4	4	-4	4	-4	-4	4	0	0	0	0	0	0	0
χ_{92}	-7	-11	9	1	5	-11	-1	-3	7	5	-5	-3	1	-3	1	-3	1	1
χ_{93}	-1	3	-17	-9	3	3	9	-5	1	3	-3	3	-1	3	-1	3	-1	-1
χ_{94}	-1	3	-17	-9	3	3	9	-5	1	3	-3	3	-1	3	-1	3	-1	-1
χ_{95}	16	-16	16	16	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{96}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{97}	-18	6	14	-2	-2	2	-2	2	-2	-2	2	-6	2	-6	2	-2	-2	2
χ_{98}	-4	-4	-4	-4	4	12	12	-4	-4	0	0	0	0	0	0	-4	4	0
χ_{99}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{100}	25	5	-23	1	-3	5	7	-3	-1	1	-1	-3	1	-3	1	1	-3	1
χ_{101}	-7	5	9	1	-3	5	7	-3	-1	1	-1	-3	1	-3	1	5	1	-3
χ_{102}	-11	1	5	-3	1	-7	-5	1	3	1	-1	-3	1	9	-3	-3	1	1
χ_{103}	-11	1	5	-3	1	-7	-5	1	3	1	-1	9	-3	-3	1	-3	1	1
χ_{104}	-18	6	14	-2	-2	-2	2	-2	2	2	-2	-6	2	6	-2	2	2	-2
χ_{105}	-18	6	14	-2	-2	-2	2	-2	2	2	-2	6	-2	-6	2		2	-2
χ_{106}	14	6	-18	-2	-2	-2	2	-2	2	2	-2	-6	2	6	-2	-2	-2	2
χ_{107}		6	-18	-2	-2	-2	10	-2	2	2	-2	6	-2	-6	2	-2	-2	2
χ_{108}	4	4	4	4	-4	-12	-12	4	4	0	0	0	0		0		-4	0
X109	-8	-ð	-8	-8	8	0	0	0	0	0	0	0	0		0		0	0
X110	-4	12	-4	-4	-4	-4 10	4	-4	4	4	-4	0	0		0		4	0
X111	-4	-4	-4	-4	4	-12	-12	4	4	0	0	0	0		0		-4	0
X112	-6	2	26	10	-6	-2	0 2	_2	2	2	_2	6	_2	6	-2	_2	_2	2
X113	18	-6	-14	2	-0	-2	2	-2 -2	2	2	-2 -2	-6	-2	-6	-2	2	-2	-2
X114 V115	4	4	4	4	-4	12	12	-4	-4	0	0	0	0	0	0	-4	4	0
×115	-8	-8	-8	-8	8	0	0	0	0	0	0	0	Õ	0	0	0	0	0
X110 X117	8	8	8	8	-8	0	0	0	0	0	0	0	0	0	0	0	0	0
X117	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X119	21	1	-27	-3	1	-3	-9	5	-1	-3	3	-3	1	-3	1	-3	1	1
χ_{120}	-11	1	5	-3	1	-3	-9	5	-1	-3	3	-3	1	-3	1	1	5	-3
X121	-15	-3	1	-7	5	9	3	1	-5	-3	3	-3	1	9	-3	1	-3	1
χ_{122}	-15	-3	1	-7	5	9	3	1	-5	-3	3	9	-3	-3	1	1	-3	1
χ123	12	-4	12	12	-4	-4	4	-4	4	4	-4	0	0	0	0	0	0	0
χ_{124}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

		5A		5B	5C		6A		6B	6C		6D				6E	6F
	5A	10A	10B	5B	5C	6F	12A	6G	12B	12C	6H	6I	6J	12D	6K	12E	12F
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	3	3	3	3	3	2	2	2	2	2	-2	-2	-2	-2	-2	-2	-2
<i>χ</i> 3	5	5	5	0	0	6	6	6	-3	-3	2	2	2	2	2	1	1
χ_4	0	0	0	5	0	-3	-3	-3	6	-3	1	1	1	1	1	2	1
χ_5	0	0	0	0	5	-3	-3	-3	-3	6	1	1	1	1	1	1	2
χ_6	0	0	0	0	0	-3	-3	-3	-3	-3	1	1	1	1	1	1	1
X7	4	4	4	-1	-1	5	5	5	2	2	5	5	5	5	5	-2	-2
χ_8	-1	-1	-1	4	-1	2	2	2	5	2	-2	-2	-2	-2	-2	5	-2
X9	-1	-1	-1	-1	4	2	2	2	2	5	-2	-2	-2	-2	-2	-2	5
X10	0	0	0	0	0	-5	-5	-5	-5	-5	3	3	3	3	3	3	3
<i>χ</i> 11	5	5	5	0	0	7	7	7	1	1	-1	-1	-1	-1	-1	1	1
X12	0	0	0	5	0	1	1	1	7	1	1	1	1	1	1	-1	1
χ_{13}	0	0	0	0	5	1	1	1	1	7	1	1	1	1	1	1	-1
χ_{14}	0	0	0	0	0	6	6	6	6	6	2	2	2	2	2	2	2
χ_{15}	0	0	0	0	0	-5	-5	-5	-5	-5	-1	-1	-1	-1	-1	-1	-1
χ_{16}	0	0	0	0	0	6	6	6	6	6	2	2	2	2	2	2	2
<i>χ</i> 17	7	7	7	-3	-3	9	9	9	0	0	-3	-3	-3	-3	-3	0	0
X18	-3	-3	-3	7	-3	0	0	0	9	0	0	0	0	0	0	-3	0
X19	-3	-3	-3	-3	7	0	0	0	0	9	0	0	0	0	0	0	-3
χ_{20}	0	0	0	0	0	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4	-4
χ_{21}	0	0	0	0	0	-1	-1	-1	2	2	3	3	3	3	3	-2	-2
X22	0	0	0	0	0	2	2	2	-1	2	-2	-2	-2	-2	-2	3	-2
X23	0	0	0	0	0	2	2	2	2	-1	-2	-2	-2	-2	-2	-2	3
X24	-5	-5	-5		0	8	8	8	-10	-10	0	0	0	0	0		2
χ_{25}	0	0	0	-5	0	-10	-10	-10	8	-10	2	2	2	2	2	0	2
χ_{26}		0	0		-5	-10	-10	-10	-10	8	2	2	2	2	2	2	0
χ_{27}	-3	-3	-3	-3	-3	17	17	17	0	0	0	1	1	0	1		0
X28		0	0			-17	-17	-17	17	7	-1	-1	-1	-1	-1	1	3
X29		0	0		0		(7	7	-17	17	3 3	ა ი	ა ი	ა ე	ა ი	-1	ວ 1
X30		1	1				1	1	0	-17	3	ა ⊿	3 4	3 4	ა ⊿	3	-1
χ31 χ22	-1	-1	-1	4	4	4	4	4	-0	-0	4	4	4	4	4		0
X32	4	4	4	-1	1	-0	-0	-0	4 Q	-0	0	0	0	0	0	4	4
X33		4	4		-1	-6	-0 -6	-6	-6	-6	-2	_2	-2	-2	-2	_2	-9
X34		0	0		0	-6	-0 -6	-0 -6	-0	-0	-6	-2	-6	-2	-6	3	-2
X35	0	0	0		0	3	3	3	-6	3	3	3	3	-0	-0	-6	3
130 V97		0	0		0	3	3	3	3	-6	3	3	3	3	3	3	-6
×30		0	0		0	-5	-5	-5	4	4	3	3	3	3	3	-4	-4
100 100		Ő	Ő		0	4	4	4	-5	4	-4	-4	-4	-4	-4	3	-4
×40		Ő	Ő		0	4	4	4	4	-5	-4	-4	-4	-4	-4	-4	3
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	-5	-5	-5		0	-4	-4	-4	8	8	4	4	4	4	4		0
×41		Ő	Ő	-5	0	8	8	8	-4	8	0	0	0	0	0	4	0
χ ₄₃		0 0	0 0		-5	8	8	8	8	-4	0	Ũ	0	Ũ	0	0	4
χ_{44}	-2	-2	-2	3	3	9	9	9	0	0	-3	-3	-3	-3	-3	0	0

The character table of $2^8:O_8^+(2)$ (continued)

		5A		5B	5C		6A		6B	6C		6D				6E	6F
	5A	10A	10B	5B	5C	6F	12A	6G	12B	12C	6H	6I	6J	12D	6K	12E	12F
χ_{45}	3	3	3	-2	3	0	0	0	9	0	0	0	0	0	0	-3	0
χ_{46}	3	3	3	3	-2	0	0	0	0	9	0	0	0	0	0	0	-3
χ_{47}	5	5	5	0	0	-9	-9	-9	0	0	3	3	3	3	3	0	0
χ_{48}	0	0	0	5	0	0	0	0	-9	0	0	0	0	0	0	3	0
χ_{49}	0	0	0	0	5	0	0	0	0	-9	0	0	0	0	0	0	3
χ_{50}	0	0	0	0	0	8	8	8	8	8	0	0	0	0	0	0	0
χ_{51}	-4	-4	-4	-4	-4	0	0	0	0	0	0	0	0	0	0	0	0
χ_{52}	0	0	0	0	0	-6	-6	-6	-6	-6	-2	-2	-2	-2	-2	-2	-2
χ_{53}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{54}	10	2	-2	0	0	12	0	-4	0	0	8	4	0	-4	0	0	0
χ_{55}	20	4	-4	0	0	24	0	-8	0	0	-4	4	-4	4	0	0	0
χ_{56}	0	0	0	0	0	-24	0	8	0	0	-4	4	-4	4	0	0	0
χ_{57}	10	2	-2	0	0	0	0	0	0	0	-12	-12	4	4	0	0	0
χ_{58}	10	2	-2	0	0	24	0	-8	0	0	8	16	-8	0	0	0	0
χ_{59}	20	4	-4	0	0	36	0	-12	0	0	12	0	4	-8	0	0	0
χ_{60}	0	0	0	0	0	-36	0	12	0	0	8	4	0	-4	0	0	0
χ_{61}	0	0	0	0	0	-12	0	4	0	0	12	0	4	-8	0	0	0
χ_{62}	10	2	-2	0	0	12	0	-4	0	0	-12	0	-4	8	0	0	0
χ_{63}	0	0	0	0	0	-12	0	4	0	0	16	20	-8	-4	0	0	0
χ_{64}	-10	-2	2	0	0	24	0	-8	0	0	-16	-8	0	8	0	0	0
χ_{65}	0	0	0	0	0	12	0	-4	0	0	-4	-8	4	0	0	0	0
χ_{66}	0	0	0		0	-48	0	16	0	0	8	-8	8	-8	0	0	0
χ_{67}		0	0		0	24	0	-8	0	0	4	-4	4	-4	0	0	0
χ_{68}		0	0		0	12	0	-4		0	-4	-8	4	0	0		0
χ_{69}	-20	-4	4		0	24	0	-8	0	0	8	16	-8	0	0		0
<i>χ</i> 70	-10	-2	2		0	-36	0	12		0	0	12	-8	4	0		
X71	-10	-2	2		0	-30	0	12		0	10	12	-8	4	0		
X72	-10	-2	2		0	30 19	0	-12		0	12	0	4	-8	0		
X73		0	0		0	-12	0	4		0	-0	-4	0	4	0		
X74	10	0 0	0		0	26	0	-4 19		0	10	4	4	-4	0		0
X75		0	-2		0	-30	0	12		0	-12	-16	-4	0	0		
X76		0	0		0	-36	0	12		0	-4	-10	4	0	0		0
X11 V79	0	0	0	0	0	-00	0	0	0	0	0	0	0	0	0	0	0
×70	10	2	-2	0	0	-24	0	8	0	0	16	8	0	-8	0	0	0
X19 X80	-20	-4	4	0	0	36	0	-12	0	0	0	-12	8	-4	0	0	0
×81	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X82	0	Õ	õ	0	0	48	õ	-16		0	-8	8	-8	8	0 0		0
X83	20	4	-4	0	Õ	0	Ũ	0	0	0	0	õ	0	õ	Ũ	0	0
X84	5	-3	1	0	0	3	-1	3	0	0	7	-5	-1	3	-1	0	0
χ ₈₅	10	-6	2	0	0	12	-4	12	0	0	4	4	4	4	-4	0	0
X86	-5	3	-1	0	0	-3	1	-3	0	0	-7	5	1	-3	1	0	0
χ ₈₇	0	0	0	0	0	15	-5	15	0	0	11	-1	3	7	-5	0	0
X88	5	-3	1	0	0	18	-6	18	0	0	-6	18	10	2	-6	0	0

The character table of $2^8:O_8^+(2)$ (continued)

		5A		5B	5C		6A		6B	6C		6D				6E	6F
	5A	10A	10B	5B	5C	6F	12A	6G	6H	6I	6J	12B	6K	12C	12D	6L	6M
χ_{89}	5	-3	1	0	0	-9	3	-9	0	0	3	-9	-5	-1	3	0	0
χ_{90}	5	-3	1	0	0	-9	3	-9	0	0	3	-9	-5	-1	3	0	0
χ_{91}	-10	6	-2	0	0	3	-1	3	0	0	7	-5	-1	3	-1	0	0
χ_{92}	0	0	0	0	0	15	-5	15	0	0	11	-1	3	7	-5	0	0
χ_{93}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{94}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{95}	5	-3	1	0	0	-12	4	-12	0	0	-4	-4	-4	-4	4	0	0
χ_{96}	-5	3	-1	0	0	12	-4	12	0	0	4	4	4	4	-4	0	0
χ_{97}	0	0	0	0	0	-15	5	-15	0	0	-11	1	-3	-7	5	0	0
χ_{98}	15	-9	3	0	0	6	-2	6	0	0	10	-14	-6	2	2	0	0
χ_{99}	15	-9	3	0	0	6	-2	6	0	0	-14	10	2	-6	2	0	0
χ_{100}	0	0	0	0	0	-9	3	-9	0	0	7	-5	-1	3	-1	0	0
χ_{101}	0	0	0	0	0	-9	3	-9	0	0	7	-5	-1	3	-1	0	0
χ_{102}	0	0	0	0	0	-9	3	-9	0	0	7	-5	-1	3	-1	0	0
χ_{103}	0	0	0	0	0	-9	3	-9	0	0	7	-5	-1	3	-1	0	0
χ_{104}	0	0	0	0	0	9	-3	9	0	0	-7	5	1	-3	1	0	0
χ_{105}	0	0	0	0	0	9	-3	9	0	0	-7	5	1	-3	1	0	0
χ_{106}	0	0	0	0	0	9	-3	9	0	0	-7	5	1	-3	1	0	0
χ_{107}	0	0	0	0	0	9	-3	9	0	0	-7	5	1	-3	1	0	0
χ_{108}	0	0	0	0	0	12	-4	12	0	0	-4	-4	-4	-4	4	0	0
χ_{109}	0	0	0	0	0	12	-4	12		0	-4	-4	-4	-4	4	0	0
X110	0	0	0	0	0	-9	3	-9		0		-5	-1	3	-1		0
χ_{111}	0	0	0	0	0	-6	2	-6	0	0	14	-10	-2	6	-2	0	0
χ_{112}		0	0	0	0	-6	2	-6		0	-10	14	6	-2	-2		
χ_{113}		0	0		0	-27	9	-27		0	-3	9	5 -	1	-3		
X114	15	0	0		0	-27	9	-27		0	-3	9	5	1	-3		
X115	-10	9	-ə 9		0	0	0	0		0		0	0	0	0		
X116	-15	9	-3		0	6	0	6		0		14	0 6	0	0		
X117		0	0		0	0 6	-2	6		0	10	-14	-0	2 C	2		
X118		0	0		0	0	-2	0		0	-14	10	2	-0	2		
X119		0	0		0		0	0				0	0	0	0		
X120		0	0		0	0	0	0				0	0	0	0		
X121		0	0			0	0	0		0		0	0	0	0		
X122		0	0		0	97	0	0 97		0	2	0	5	1	บ จ		
X123	15	_0	2 2		0	-12	-9 1	41 _19		0		-9 1	-0	-1 /	ن _1		
X124	10	-9	ა	0	0	-12	4	-14		U	4	4	4	4	-4		

The character table of $2^8:O_8^+(2)$ (continued)

	6G			6H		6I		6J			6K		6L		6M	
	6N	6O	6P	6Q	6R	6S	12E	6T	12F	6U	12G	12H	6V	12I	6W	12J
<i>χ</i> 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	5	5	-1	-1	-1	-1	-1	-1	-1	1	1	1	1	1	1	1
χ3	3	3	0	0	0	0	0	0	0	2	2	2	-2	-2	-2	-2
χ_4	3	3	0	0	0	0	0	0	0	-2	-2	-2	2	2	-2	-2
χ_5	3	3	0	0	0	0	0	0	0	-2	-2	-2	-2	-2	2	2
χ_6	0	0	3	3	3	3	3	3	3	1	1	1	1	1	1	1
χ_7	-1	-1	5	5	5	-1	-1	-1	-1	-1	-1	-1	1	1	1	1
χ_8	-1	-1	-1	-1	-1	5	5	-1	-1	1	1	1	-1	-1	1	1
χ_9	-1	-1	-1	-1	-1	-1	-1	5	5	1	1	1	1	1	-1	-1
χ_{10}	1	1	-2	-2	-2	-2	-2	-2	-2	0	0	0	0	0	0	0
X11	-2	-2	-5	-5	-5	1	1		1	-1	-1	-1		1		1
X12	-2	-2		1	1	-5	-5		1	1	1	1	-1	-1	1	1
χ_{13}	-2	-2		1	1	1	1	-5	-5	1	1	1		1	-1	-1
χ_{14}	3	3		0	0	0	0	0	0	2	2	2		2	2	2
χ_{15}		7		1	1	1	1		1	-1	-1	-1	-1	-1	-1	-1 1
χ_{16}	0	0		3	3	3	3		ა ი	-1	-1	-1	-1	-1	-1	-1
X17	0	0		0	0	0	0		0	0	0	0		0		0
X18	0	0		0	0	0	0		0	0	0	0		0		0
X19	2	0 2	-1	-1	_1	-1	_1	1	_1	-1	_1	-1	-1	-1	-1	-1
X20	_1	_1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	2	-1	-1	-1
X21	-1	-1	2	2	2	-4	-4	2	2	-2	-2	-2	0	-2	-2	-2
λ22 	-1	-1	2	2	2	2	2	-4	-4	-2	-2	-2	-2	-2		0
×24	-1	-1	-1	-1	-1	-1	-1	-1	-1	- 3	-	- 3	-1	-1	-1	-1
X24	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	3	3	-1	-1
X26	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1	3	3
X27	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X28	1	1	4	4	4	-2	-2	-2	-2	2	2	2	0	0	0	0
χ_{29}	1	1	-2	-2	-2	4	4	-2	-2	0	0	0	2	2	0	0
X30	1	1	-2	-2	-2	-2	-2	4	4	0	0	0	0	0	2	2
χ_{31}	-2	-2	4	4	4	-2	-2	-2	-2	-2	-2	-2	0	0	0	0
X32	-2	-2	-2	-2	-2	4	4	-2	-2	0	0	0	-2	-2	0	0
X33	-2	-2	-2	-2	-2	-2	-2	4	4	0	0	0	0	0	-2	-2
χ_{34}	0	0	-3	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1	1
χ_{35}	-3	-3	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{36}	-3	-3	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{37}	-3	-3	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X38	-2	-2	1	1	1	1	1	1	1	3	3	3	-1	-1	-1	-1
X39	-2	-2	1	1	1	1	1	1	1	-1	-1	-1	3	3	-1	-1
χ_{40}	-2	-2	1	1	1	1	1	1	1	-1	-1	-1	-1	-1	3	3
χ_{41}	2	2	-4	-4	-4	2	2	2	2	-2	-2	-2	0	0	0	0
χ_{42}	2	2	2	2	2	-4	-4	2	2	0	0	0	-2	-2	0	0
χ_{43}	2	2	2	2	2	2	2	-4	-4	0	0	0	0	0	-2	-2
χ_{44}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	6G			6H		6I		6J			6K		6L		6M	
	6N	6O	6P	6Q	6R	6S	12E	6T	12F	6U	12G	12H	6V	12I	6W	12J
χ_{45}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{46}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{47}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{48}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{49}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{50}	2	2	-4	-4	-4	-4	-4	-4	-4	0	0	0	0	0	0	0
χ_{51}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{52}	3	3	3	3	3	3	3	3	3	1	1	1	1	1	1	1
χ_{53}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{54}	3	-1	6	2	-2	0	0	0	0	2	-2	0	0	0	0	0
χ_{55}	6	-2	-6	-2	2	0	0	0	0	2	-2	0	0	0	0	0
χ_{56}	3	-1	6	2	-2	0	0	0	0	2	-2	0	0	0	0	0
χ_{57}	9	-3	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{58}	-3	1	12	4	-4	0	0	0	0	-4	4	0	0	0		0
χ_{59}		0	0	0	0	0	0	0	0	0	0	0	0	0		0
χ_{60}	9	-3	10	0	0		0	0	0	-4	4	0	0	0		0
χ ₆₁	-3	1	12	4	-4		0	0	0		0	0	0	0		0
X62	-0	2 1	-12	-4 0	4 9		0	0	0	0	0	0	0	0		0
X63	-3	1	-0	-2	2		0	0	0	-2	_2 _2	0	0	0		0
X64	-5 3	-1	6	-2	_2		0	0	0	2	-2 _2	0	0	0		0
X65	6	-1	-6	-2	-2		0	0	0	2	-2 _2	0	0	0		0
X60	6	-2 -2	-6	-2 -2	2	0	0	0	0	-2	-2	0	0	0		0
101 V69	-6	2	-12	-4	4	0	0	0	0	-4	4	0	0	0	0	0
X69	6	-2	-6	-2	2	0	0	0	0	2	-2	0	0	0		0
X70	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X71	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
X72	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ73	-3	1	12	4	-4	0	0	0	0	4	-4	0	0	0	0	0
χ_{74}	-6	2	6	2	-2	0	0	0	0	2	-2	0	0	0	0	0
χ_{75}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{76}	-6	2	6	2	-2	0	0	0	0	-2	2	0	0	0	0	0
X77	0	0	0	0	0	0	0	0	0	-4	4	0	0	0	0	0
χ_{78}	9	-3	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{79}	-6	2	-12	-4	4	0	0	0	0	4	-4	0	0	0	0	0
χ_{80}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{81}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{82}	3	-1	6	2	-2	0	0	0	0	-2	2	0	0	0	0	0
X83	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{84}	0	0	9	-3	1	3	-1	3	-1	1	1	-1	3	-1	3	-1
χ_{85}	0	0	-9	3	-1	3	-1	3	-1	1	1	-1	-3	1	-3	1
χ_{86}		0	0	0	0	6	-2	6	-2	2	2	-2	0	0		0
X87		0	9	-3	1	-3	1	-3	1	-1	-1	1	3	-1	3	-1
χ_{88}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	6G			6H		6I		6J			6K		6L		6M	
	6N	6O	6P	6Q	6R	6S	12E	6T	12F	6U	12G	12H	6V	12I	6W	12J
χ_{89}	0	0	0	0	0	63	21	63	-21	0	0	0	63	-21	63	-21
χ_{90}	0	0	9	-3	1	84	-28	84	-28	1	1	-1	84	-28	84	-28
χ_{92}	0	0	18	-6	2	105	-35	105	-35	2	2	-2	105	-35	105	-35
χ_{93}	0	0	0	0	0	135	-45	135	-45	0	0	0	135	-45	135	-45
χ_{94}	0	0	0	0	0	135	-45	135	-45	0	0	0	135	-45	135	-45
χ_{95}	0	0	-9	3	-1	168	-56	168	-56	-1	-1	1	168	-56	168	-56
χ_{96}	0	0	-18	6	-2	192	-64	192	-64	-2	-2	2	192	-64	192	-64
χ_{97}	0	0	9	-3	1	210	-70	210	-70	1	1	-1	210	-70	210	-70
χ_{98}	0	0	9	-3	1	84	-28	84	-28	-1	-1	1	84	-28	84	-28
χ_{99}	0	0	9	-3	1	84	-28	84	-28	-1	-1	1	84	-28	84	-28
χ_{100}	0	0	-18	6	-2	105	-35	105	-35	0	0	0	105	-35	105	-35
χ_{101}	0	0	-18	6	-2	105	-35	105	-35	0	0	0	105	-35	105	-35
χ_{102}	0	0	-18	6	-2	105	-35	105	-35	0	0	0	105	-35	105	-35
χ_{103}	0	0	-18	6	-2	105	-35	105	-35	0	0	0	105	-35	105	-35
χ_{104}	0	0	-9	3	-1	210	-70	210	-70	3	3	-3	210	-70	210	-70
χ_{105}	0	0	-9	3	-1	210	-70	210	-70	3	3	-3	210	-70	210	-70
χ_{106}	0	0	-9	3	-1	210	-70	210	-70	3	3	-3	210	-70	210	-70
χ_{107}	0	0	-9	3	-1	210	-70	210	-70	3	3	-3	210	-70	210	-70
χ_{108}	0	0	-9	3	-1	420	-140	420	-140	1	1	-1	420	-140	420	-140
χ_{109}	0	0	-9	3	-1	420	-140	420	-140	1	1	-1	420	-140	420	-140
χ_{110}	0	0	-9	3	-1	420	-140	420	-140	1	1	-1	420	-140	420	-140
χ_{111}	0	0	-9	3	-1	420	-140	420	-140	1	1	-1	420	-140	420	-140
χ_{112}	0	0	-9	3	-1	420	-140	420	-140	1	1	-1	420	-140	420	-140
χ_{113}	0	0	0	0	0	630	-210	630	-210	0	0	0	630	-210	630	-210
χ_{114}	0	0	0	0	0	630	-210	630	-210	0	0	0	630	-210	630	-210
χ_{115}	0	0	0	0	0	756	-252	756	-252	0	0	0	756	-252	756	-252
χ_{116}	0	0	0	0	0	756	-252	756	-252	0	0	0	756	-252	756	-252
χ_{117}	0	0	9	-3	1	840	-280	840	-280	-1	-1	1	840	-280	840	-280
χ_{118}	0	0	9	-3	1	840	-280	840	-280	-1	-1	1	840	-280	840	-280
χ_{119}	0	0	945	-315	945	-315	0	0	0	945	-315	945	-315	-1	-1	1
χ_{120}	0	0	945	-315	945	-315	0	0	0	945	-315	945	-315	-1	-1	1
χ_{121}	0	0	945	-315	945	-315	0	0	0	945	-315	945	-315	-1	-1	1
$\chi_{122}0$	0	945	-315	945	-315	0	0	0	945	-315	945	-315	-1	-1	1	
$\chi_{123}0$	0	0	0	0	1260	-420	1260	-420	0	0	0	1260	-420	1260	-420	
$\chi_{124}0$	0	-18	6	-2	1344	-448	1344	-448	2	2	-2	1344	-448	1344	-448	

The character table of $2^8:O_8^+(2)$ (continued)

		6N			7A				8A			8B	
	6X	12K	12L	7A	14A	14B	14C	8G	8H	8I	8J	8K	8L
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	-1	-1	-1	0	0	0	0	-2	-2	-2	2	2	2
<i>χ</i> 3	0	0	0	0	0	0	0	1	1	1	1	1	1
χ_4	0	0	0	0	0	0	0	1	1	1	1	1	1
χ_5	0	0	0	0	0	0	0	1	1	1	1	1	1
χ_6	-1	-1	-1	1	1	1	1	0	0	0	0	0	0
χ_7	1	1	1	0	0	0	0	0	0	0	0	0	0
χ_8	1	1	1	0	0	0	0	0	0	0	0	0	0
χ_9	1	1	1	0	0	0	0	0	0	0	0	0	0
χ_{10}	2	2	2		0	0	0	-1	-1	-1	-1	-1	-1
χ_{11}	-1	-1	-1		0	0	0		0	0		0	0
χ_{12}	-1	-1	-1		0	0	0		0	0		0	0
X13	-1	-1	-1		1	1	1		0	0		0	0
X14	1	1	1	-1	-1	-1	-1		2	2	-2	-2	-2
χ_{15}	_1	_1	_1		0	0	0	-1	-1	-1	_1	-1	-1
χ_{17}	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
×17 ×18	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
χ_{19}	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
χ_{20}	-1	-1	-1	0	0	0	0	2	2	2	-2	-2	-2
χ_{21}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{22}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{23}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{24}	-1	-1	-1	0	0	0	0	0	0	0	0	0	0
χ_{25}	-1	-1	-1	0	0	0	0	0	0	0	0	0	0
χ_{26}	-1	-1	-1	0	0	0	0	0	0	0	0	0	0
χ_{27}	0	0	0	-1	-1	-1	-1	-2	-2	-2	2	2	2
χ_{28}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{29}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{30}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{31}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{32}	0	0	0		0	0	0		0	0		0	0
χ_{33}	0	0	0		0	0	0		0	0		0	0
χ_{34}	1	1	1		0	0	0		0	0		0	0
χ_{35}	0	0	0		0	0	0		1	1		1	1
X36	0	0	0		0	0	0		1	1		1	1
X37	1	1	1		0	0	0		1	1		1	1
X38		1	1		0	0	0		0	0		0	0
X39	1	1	1		0	0	0		0	0		0	0
χ_{41}	0	1 0	0		0	0	0		0	0		0	0
χ ₄₁ γ ₄₂	0	0	0		0	0	0		0	0		0	0
χ ₄₃	0	0	0		0	0	0		0	0		0	0
χ_{44}	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

		6N			7A				8A			8B	
	6X	12K	12L	7A	14A	14B	14C	8G	8H	8I	8J	8K	8L
χ_{45}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{46}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{47}	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
χ_{48}	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
χ_{49}	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
χ_{50}	0	0	0	1	1	1	1	0	0	0	0	0	0
χ_{51}	0	0	0	1	1	1	1	0	0	0	0	0	0
χ_{52}	-1	-1	-1	0	0	0	0	0	0	0	0	0	0
χ_{53}	0	0	0	-1	-1	-1	-1	1	1	1	1	1	1
χ_{54}	2	-2	0	1	-1	1	-1	2	-2	0	2	-2	-2
χ_{55}	-2	2	0	0	0	0	0	-2	2	0	2	-2	-2
χ_{56}	-2	2	0		-1	1	-1	2	-2	0	-2	2	2
χ_{57}	0	0	0		0	0	0	-2	2	0		-2	-2
χ_{58}	0	0	0		0	0	0	-2	2	0	-2	2	2
χ_{59}	0	0	0	-1	1	-1	1		-2	0	-2	2	2
χ_{60}	0	0	0		0	0	0		-2	0		-2	-2
X61	0	0	0		0	0	0	-2	2	0		-2	-2
χ_{62}	0	0	0		0	0	0		0	0		0	0
χ_{63}	2	-2	0		0	0	0		0	0		0	0
X64	2	-2 _2	0		0	0	0	2	_2	0	_2	2	2
X65 Xcc	2	_2	0		0	0	0	_2	2	0	_2	2	2
X60 X67	-2	2	0	0	0	0	0	2	-2	0	-2	2	2
X68	0	0	0	1	-1	1	-1	0	0	0 0		0	0
X69	-2	2	0	0	0	0	0	0	0	0	0	0	0
X70	0	0	0	0	0	0	0	-2	2	0	-2	2	2
χ ₇₁	0	0	0	0	0	0	0	2	-2	0	2	-2	-2
χ_{72}	0	0	0	0	0	0	0	-2	2	0	2	-2	-2
χ_{73}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{74}	-2	2	0	0	0	0	0	0	0	0	0	0	0
χ_{75}	0	0	0	-1	1	-1	1	0	0	0	0	0	0
χ_{76}	-2	2	0	0	0	0	0	0	0	0	0	0	0
χ_{77}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{78}	0	0	0	0	0	0	0	-2	2	0	-2	2	2
χ_{79}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{80}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{81}	0	0	0	-1	1	-1	1	2	-2	0	2	-2	-2
χ_{82}	2	-2	0	0	0	0	0	0	0	0	0	0	0
χ_{83}	0	0	0	1	-1	1	-1	0	0	0	0	0	0
χ_{84}	1	1	-1	2	0	-2	0	1	1	-1	1	1	1
χ_{85}	-1	-1	1	0	0	0	0	1	1	-1	1	1	1
χ_{86}	0	0	0	0	0	0	0		0	0		0	0
χ_{87}	1	1	-1	-2	0	2	0		0	0		0	0
χ_{88}	0	0	0	0	0	0	0	-1	-1	1	-1	-1	-1

The character table of $2^8:O_8^+(2)$ (continued)

		6N			7A				8A			8B	
	6X	12K	12L	7A	14A	14B	14C	8G	8H	8I	8J	8K	8L
χ_{89}	0	0	0	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{90}	0	0	0	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{91}	-1	-1	1	0	0	0	0	0	0	0	0	0	0
χ_{92}	0	0	0	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{93}	0	0	0	-1	$-A + \overline{A}$	1	$A - \overline{A}$	1	1	-1	1	1	1
χ_{94}	0	0	0	-1	$A - \overline{A}$	1	$-A + \overline{A}$	1	1	-1	1	1	1
χ_{95}	-1	-1	1	0	0	0	0	0	0	0	0	0	0
χ_{96}	0	0	0	2	0	-2	0	0	0	0	0	0	0
χ_{97}	1	1	-1	0	0	0	0	0	0	0	0	0	0
χ_{98}	1	1	-1	0	0	0	0	0	0	0	0	0	0
χ_{99}	-1	-1	1	0	0	0	0	-2	-2	2	2	2	2
χ_{100}	-2	-2	2	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{101}	2	2	-2	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{102}	0	0	0	0	0	0	0	1	1	-1	1	1	1
χ_{103}	0	0	0	0	0	0	0	1	1	-1	1	1	1
χ_{104}	-1	-1	1	0	0	0	0	0	0	0	0	0	0
χ_{105}	-1	-1	1	0	0	0	0	0	0	0	0	0	0
χ_{106}	1	1	-1	0	0	0	0	0	0	0	0	0	0
χ_{107}	1	1	-1	0	0	0	0	0	0	0	0	0	0
χ_{108}	1	1	-1	0	0	0	0	0	0	0	0	0	0
χ_{109}	-1	-1	1	0	0	0	0	2	2	-2	-2	-2	-2
χ_{110}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{111}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{112}	0	0	0	0	0	0	0	2	2	-2	-2	-2	-2
χ_{113}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{114}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{115}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{116}	0	0	0	0	0	0	0	-2	-2	2	2	2	2
χ_{117}	-1	-1	1	0	0	0	0	0	0	0	0	0	0
χ_{118}	1	1	-1	0	0	0	0	0	0	0	0	0	0
χ_{119}	0	0	0	0	0	0	0	1	1	-1	1	1	1
χ_{120}	0	0	0	0	0	0	0	1	1	-1	1	1	1
χ_{121}	0	0	0	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{122}	0	0	0	0	0	0	0	-1	-1	1	-1	-1	-1
χ_{123}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{124}	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	9A		9B	9C		10A		10B	10C	12A		12B	12C
	9A	18A	9B	9C	10C	20A	20B	10D	10E	12M	12N	120	12P
χ_1	9450	3780	3780	4725	4725	4725	4725	9450	9450	9450	9450	18900	18900
χ_2	-630	-252	-252	-315	-315	-315	-315	-630	-630	-630	-630	-1260	-1260
<i>χ</i> 3	490	196	196	245	245	245	245	490	490	490	490	980	980
χ_4	138	132	-156	405	-171	117	117	-54	-54	522	522	84	-204
χ_5	234	-60	36	-123	69	-27	-27	42	42	-150	-150	-108	-12
χ_6	202	4	-28	53	-11	21	21	10	10	74	74	-44	-108
χ7	-70	4	4	-3	-3	-3	-3	-6	-6	-6	-6	20	20
χ_8	130	244	124	205	205	-155	-155	50	50	50	50	20	380
χ_9	50	-76	-36	-75	-75	45	45	-30	-30	-30	-30	20	-100
χ_{10}	82	52	28	37	37	-35	-35	2	2	2	2	20	92
χ_{11}	66	-12	-4	-19	-19	5	5	-14	-14	-14	-14	20	-4
χ_{12}	-70	-4	-4	5	5	5	5	10	10	10	10	-20	-20
χ_{13}	-30	-60	60	45	45	-75	165	-30	210	-30	210	-60	60
χ_{14}	2	4	-4	-3	-3	5	-11	2	-14	2	-14	4	-4
χ_{15}	-30	-60	60	45	45	165	-75	210	-30	210	-30	-60	60
χ_{16}	2	4	-4	-3	-3	-11	5	-14	2	-14	2	4	-4
χ_{17}	10	4	-28	-11	5	21	21	26	26	10	10	-44	20
χ_{18}	2	-4	4	5	-11	-3	-3	-14	-14	2	2	-4	4
χ_{19}	10	4	-28	-11	5	21	21	26	26	10	10	-44	20
χ_{20}	-6	4	4	-3	13	-3	-3	10	10	-6	-6	20	-12
χ_{21}	-135	270	270	135	135	135	135	-135	-135	-135	-135	540	540
χ_{22}	-15	30	30	15	15	15	15	-15	-15	-15	-15	60	60
χ_{23}	25	-50	-50	-25	-25	-25	-25	25	25	25	25	-100	-100
χ_{24}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{25}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{26}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{27}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{28}	9	9	9	18	18	18	18	9	9	9	9	-9	-9
χ_{29}	-3	-3	-3	-6	-6	-6	-6	-3	-3	-3	-3	3	3
X30		10				2	2					-1	-1
χ_{31}	0	12	24	-15	-15	21	21	0	0	0	0	30	0
X32	-2	-4 10	-0	15	15	- (91	- (-2	-2	-2	-2	-12	0
<i>χ</i> 33	18	12		-15	-15	21 11	21 11	18	18	14	14	30	8
X34	-10	-4		5	5	-11	-11	-10	-10	6	6	4	-0
X35	14	-4		_23	9	5	5	14	14	-18	-18	4	-0
X36	-2	-4		-25		_3	_3	-2	-2	-10	-10	4	-8
X37	-2	-4		-3	-3	-5	-5	-2	-2	-2	-2 -2	-4	-0
138 V20	2	- 12		5	5	-7	-7	-2	-2	-2	-2 -2	-12	0
740	-2	12		7	7	-5	-5	2	2	2	2	-12	0
Λ40 Υ ₄₁	2	-4	0	-3	-3	1	1	-2	-2	-2	-2	4	0
χ ₄₀	-2	-4	0	-1	-1	3	3	2	2	2	2	4	Ő
X42 X43	-2	0	0	1	1	1	1	2	2	2	2	0	Ő
χ_{44}	2	0	0	-1	-1	-1	-1	-2	-2	-2	-2	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	9A		9B	9C		10A		10B	10C	12A		12B	12C
	9A	18A	9B	9C	10C	20A	20B	10D	10E	12M	12N	120	12P
χ_{45}	-6	0	0	-3	-3	-3	9	-6	6	-6	6	0	0
χ_{46}	2	0	0	1	1	1	-3	2	-2	2	-2	0	0
χ_{47}	-6	0	0	-3	-3	9	-3	6	-6	6	-6	0	0
χ_{48}	2	0	0	1	1	-3	1	-2	2	-2	2	0	0
χ_{49}	-2	-4	0	1	5	-3	-3	2	2	-2	-2	4	0
χ_{50}	-2	4	0	-3	1	1	1	2	2	-2	-2	-4	0
χ_{51}	2	0	0	1	-3	1	1	-2	-2	2	2	0	0
χ_{52}	0	15	15	0	0	0	0	0	0	0	0	0	0
χ_{53}	0	-9	-9	0	0	0	0	0	0	0	0	0	0
χ_{54}	0	3				0	0			0	0	0	0
χ_{55}	0	0				0	0			0	0	0	0
χ_{56}	15	0				0	0				0	10	10
<i>χ</i> 57	-15	0	0	-9	-9	-9	-9 2	9	9	9	9	12	
X58	-15	-2			0 0	ა ი	ა ი				-ə 0	-4 19	-4 19
X59	-10	0		-9	-9	-9	-9		0		0	0	
X60	0	0	0	0	0	0	0	0	0	0	0	0	0
X62	-11	10	-14	7	7	7	7	-7	-7	-7	-7	-4	-4
X63	1	-14	10	-5	-5	-5	-5	5	5	5	5	-4	-4
χ_{64}	-3	-6	2	-1	-1	-1	-1	1	1	1	1	-4	-4
χ_{65}	-7	2	-6	3	3	3	3	-3	-3	-3	-3	-4	-4
χ_{66}	5	2	2	-1	-1	-1	-1	1	1	1	1	4	4
χ_{67}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{68}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{69}	0	0	0	0	0	0	0	0	0	0	0	0	0
X70	0	0	0	0	0	0	0	0	0	0	0	0	0
χ71	9	9	-9	18	-18	0	0	9	9	-9	-9	9	-9
<i>χ</i> 72	-3	-3	3	-6	6	0	0	-3	-3	3	3	-3	3
χ73	1	1	-1	2	-2	0	0	1	1	-1	-1	1	-1
χ_{74}	3	-3	-3	0	0	0	0	9	-9	-9	9	3	3
χ_{75}	-1	1		0	0	0	0	-3	3	3	-3	-1	-1
χ_{76}	3	-3	-3	0	0	0	0	-9	9	9	-9	3	3
χ77	-1	1				0	0		-3	-3	3	-1	-1
χ78		1		-2	-2	-2	-2	-1	-1	-1	-1	-1	-1
X79		1		-2	-2	-2	-2	-1	-1	-1	-1	-1	-1
χ_{80}	-1	-1	-1 -1			2	2	1 9	1 		1	1 9	1 9
X81	1 1	-3 1	3 1			-0 -0	0 0	Ĵ 1	-3 1	1 1	-3 1	-3 1	Ĵ 1
χ_{82}	-1 2	_2 _2	-1 2			∠ 6	-2 -6	-1	2	-1	3 T	3	-1 2
X83		-5		0	0		-0 0	-5	1	-5		-5	1
X84	1	1	1	-2	2	-2	2 0	1		1	-1	1	
180 Yee	1	1	-1	-2		0	0	-1	-1	1	1	1	-1
X87	-1	-1			-2	0	0			-1	-1	-1	
X88	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	9A		9B	9C		10A		10B	10C	12A		12B	12C
	9A	18A	9B	9C	10C	20A	20B	10D	10E	12M	12N	12O	12P
χ_{89}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{90}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{91}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{92}	0	0	-2	-1	-1	1	1	0	0	0	0	0	2
χ_{93}	0	0	-2	-1	-1	1	1	0	0	0	0	0	2
χ_{94}	0	0	2	1	1	-1	-1	0	0	0	0	0	-2
χ_{95}	0	0	2	-1	-1	1	1	0	0	0	0	0	-2
χ_{96}	0	0	2	-1	-1	1	1	0	0	0	0	0	-2
χ_{97}	0	0	-2	1	1	-1	-1	0	0	0	0	0	2
χ_{98}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{99}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{100}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{101}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{102}	0	-1	-1	0	0	0	0	0	0	0	0	0	0
χ_{103}	0	-1	-1	0	0	0	0	0	0	0	0	0	0
χ_{104}	0	1	1	0	0	0	0	0	0	0	0	0	0
χ_{105}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{106}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{107}	-3	-6	6	3	3	3	3	-3	-3	-3	-3	0	0
χ_{108}	1	2	-2	-1	-1	-1	-1	1	1	1	1	0	0
χ_{109}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{110}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{111}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{112}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{113}	-1	0	0	-1	-1	-1	-1	1	1	1	1	0	0
χ_{114}	-1	0	0	-1	-1	-1	-1	1	1	1	1	0	0
χ_{115}	1	0	0	1	1	1	1	-1	-1	-1	-1	0	0
χ_{116}	1	0	0	1	1	1	1	-1	-1	-1	-1	0	0
χ_{117}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{118}	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{119}		0	0		0	0	0	0	0	0	0	0	0
χ_{120}		0	0		0	0	0	0	0	0	0	0	0
χ_{121}		0	0		0	0	0	0	0	0	0	0	0
χ_{122}		0	0		0	0	0	0	0	0	0	0	0
χ_{123}		0	0		0	0	0	0	0	0	0	0	0
χ_{124}	0	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	12D			12E			12F	12G		15A			15B	15C
	12Q	12R	12S	24A	24B	12T	12U	12V	15A	30A	30B	30C	15B	15C
<i>χ</i> 1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	-1	-1	0	0	0	0	0	0	0	0	0	0	0	0
X3	1	1	0	0	0	0	-1	-1	-1	-1	-1	-1	0	0
χ_4	1	1	-1	-1	-1	-1	0	-1	0	0	0	0	-1	0
χ_5	1	1	-1	-1	-1	-1	-1	0	0	0	0	0	0	-1
χ_6	-2	-2	-1	-1	-1	-1	-1	-1	0	0	0	0	0	0
χ_7	1	1	1	1	1	1	0	0	1	1	1	1	-1	-1
X8	1	1	0	0	0	0	1	0	-1	-1	-1	-1	1	-1
X9	1	1	0	0	0	0	0	1	-1	-1	-1	-1	-1	1
χ_{10}	-1	-1	-1	-1	-1	-1	-1	-1	0	0	0	0	0	0
X11	0	0	-1	-1	-1	-1		1	-1	-1	-1	-1	0	0
X12	0	0		1	1	1	-1	1		0	0	0	-1	0
<i>χ</i> 13	0	0		1	1	1		-1		0	0	0	0	-1
χ_{14}	-1	-1		0	0	0	0	0		0	0	0	0	0
X15	-1	-1		1	1	1		1		0	0	0	0	0
X16	2	2		1	1	1		0		1	1	1	0	0
X17	0	0		-1	-1	-1		0		1	1	1	1	0
X18	0	0		0	0	0	-1	1		0	0	0	1	1
X29	0	0		0	0	0		-1		0	0	0	0	1
X20	-1	-1	1	1	1	1		0		0	0	0	0	0
X21	-1	-1		0	0	0	1	0		0	0	0	0	0
X22	-1	-1		0	0	0		1		0	0	0	0	0
X23	1	1	0	0	0	0	0	0	1	1	1	1	0	0
χ ₂₅	1	1	0	0	0	0	0	0		0	0	0	1	0
×26	1	- 1	0	0	0	0	0	0	0	0	0	0	0	1
χ_{27}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ ₂₈	-1	-1	-1	-1	-1	-1	1	1	0	0	0	0	0	0
χ ₂₉	-1	-1	1	1	1	1	-1	1	0	0	0	0	0	0
χ_{30}	-1	-1	1	1	1	1	1	-1	0	0	0	0	0	0
χ_{31}	0	0	0	0	0	0	0	0	-1	-1	-1	-1	1	1
X32	0	0	0	0	0	0	0	0	1	1	1	1	-1	1
X33	0	0	0	0	0	0	0	0	1	1	1	1	1	-1
χ_{34}	2	2	0	0	0	0	0	0	0	0	0	0	0	0
χ_{35}	-1	-1	2	2	2	2	-1	-1	0	0	0	0	0	0
χ_{36}	-1	-1	-1	-1	-1	-1	2	-1	0	0	0	0	0	0
X37	-1	-1	-1	-1	-1	-1	-1	2	0	0	0	0	0	0
X38	0	0	-1	-1	-1	-1	0	0	0	0	0	0	0	0
χ_{39}	0	0	0	0	0	0	-1	0	0	0	0	0	0	0
χ_{40}	0	0	0	0	0	0	0	-1	0	0	0	0	0	0
χ_{41}	0	0	0	0	0	0	0	0	1	1	1	1	0	0
χ_{42}	0	0	0	0	0	0	0	0	0	0	0	0	1	0
χ_{43}	0	0	0	0	0	0	0	0	0	0	0	0	0	1
χ_{44}	0	0	-1	-1	-1	-1	0	0	1	1	1	1	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	12D			12E			12F	12G		15A			15B	15C
	12Q	12R	12S	24A	24B	12T	12U	12V	15A	30A	30B	30C	15B	15C
χ_{45}	0	0	0	0	0	0	-1	0	0	0	0	0	1	0
χ_{46}	0	0	0	0	0	0	0	-1	0	0	0	0	0	1
χ_{47}	0	0	1	1	1	1	0	0	-1	-1	-1	-1	0	0
χ_{48}	0	0	0	0	0	0	1	0	0	0	0	0	-1	0
χ_{49}	0	0	0	0	0	0	0	1	0	0	0	0	0	-1
χ_{50}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{51}	0	0	0	0	0	0	0	0	-1	-1	-1	-1	-1	-1
χ_{52}	1	1	0	0	0	0	0	0	0	0	0	0	0	0
χ_{53}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{54}	3	-1	2	-2	0	0	0	0	1	-1	1	-1	0	0
χ_{55}	0	0	-2	2	-2	2	0	0	-1	1	-1	1	0	0
χ_{56}	-3	1	-2	2	2	-2	0	0	0	0	0	0	0	0
χ_{57}	-3	1	2	-2	2	-2	0	0	1	-1	1	-1	0	0
χ_{58}	3	-1	0	0	0	0	0	0	1	-1	1	-1	0	0
χ_{59}	0	0	0	0	2	-2	0	0	-1	1	-1	1	0	0
χ_{60}	3	-1	-2	2	0	0	0	0	0	0	0	0	0	0
χ_{61}	-3	1	0	0	-2	2	0	0	0	0	0	0	0	0
χ_{62}	0	0	0	0	2	-2	0	0	1	-1	1	-1	0	0
χ_{63}	-3	1	-2	2	0	0	0	0	0	0	0	0	0	0
χ_{64}	3	-1	0	0	0	0	0	0	-1	1	-1	1	0	0
χ_{65}	-3	1	0	0	-2	2	0	0	0	0	0	0	0	0
χ_{66}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{67}	0	0	2	-2	-2	2	0	0	0	0	0	0	0	0
χ_{68}	0	0	0	0	-2	2	0	0	0	0	0	0	0	0
χ_{69}	0	0	0	0	0	0	0	0		-1	1	-1	0	0
X70	0	0	2	-2	0	0	0	0	-1	1	-1	1	0	0
χ71	0	0		-2	0	0	0	0	-1	1	-1	1	0	0
χ72	0	0		0	2	-2	0	0	-1	1	-1	1	0	0
χ73	3	-1	-2	2	0	0	0	0		0	0	0	0	0
χ74	0	0		-2	0	0		0		0	0	0	0	0
<i>χ</i> 75	0	0		0	-2	2		0		-1	1	-1	0	0
X76	0	0		0	0	0		0		0	0	0	0	0
χ77	0	1		0	2	-2		0		0	0	0	0	0
X78	3	-1		0	0	0		0		1	1	1	0	0
X79		0	0 9	บ จ	0	0		0		-1 1	1	-1 1		
X80	0	0	-2	2	0	0		0		-1	1	-1	0	0
X81	0 9	1		0	0	0		0		0	0	0		0
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$		1		0	0	0		0		1	_1	1	0	0
X83	0	0	1	1	1	1	0	0	-1		-1		0	0
X84	0	0		1	-1	-1		0		0	0	-2 9	0	0
185 Vac	0	0	_1	_1	1	1		0		0	0	2 9	0	0
180 Vo7	0	0	1	-1	-1	-1		0		0	0	0	0	0
χ_{88}	0	0	-2	-2	2	2	0	0		0	0	-2	0	0

The character table of $2^8:O_8^+(2)$ (continued)

	12D			12E			12F	12G		15A			15B	15C
	12Q	12R	12S	24A	24B	12T	12U	12V	15A	30A	30B	30C	15B	15C
χ_{89}	0	0	1	1	-1	-1	0	0	-1	$B - \overline{B}$	$-B + \overline{B}$	1	0	0
χ_{90}	0	0	1	1	-1	-1	0	0	-1	$-B + \overline{B}$	$B - \overline{B}$	1	0	0
χ_{91}	0	0	1	1	-1	-1	0	0	2	0	0	-2	0	0
χ_{92}	0	0	1	1	-1	-1	0	0	0	0	0	0	0	0
X93	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{94}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{95}	0	0	0	0	0	0	0	0	2	0	0	-2	0	0
χ_{96}	0	0	0	0	0	0	0	0	-2	0	0	2	0	0
χ_{97}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0
χ_{98}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{99}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{100}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0
χ_{101}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0
χ_{102}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0
χ_{103}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0
χ_{104}	0	0	1	1	-1	-1	0	0	0	0	0	0	0	0
χ_{105}	0	0	1	1	-1	-1	0	0	0	0	0	0	0	0
χ_{106}	0	0	1	1	-1	-1	0	0	0	0	0	0	0	0
χ_{107}	0	0	1	1	-1	-1	0	0	0	0	0	0	0	0
χ_{108}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{109}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{110}	0	0	-1	-1	1	1	0	0	0	0	0	0	0	0
χ_{111}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{112}	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{113}	0	0		1	-1	-1	0	0	0	0	0	0	0	0
χ_{114}	0	0		1	-1	-1	0	0	0	0	0	0		0
χ_{115}	0	0		0	0	0	0	0	0	0	0	0		0
χ_{116}		0		0	0	0	0	0		0	0	0		
χ_{117}		U		0	0	0				0	0	0		
X118		0		0	0	0				0	0	0		
χ_{119}		0		0	0	0				0	0	0		
χ_{120}		U		0	0	0				0	0	0		
χ_{121}		U		0	0	0				0	0	0		
X122		0		1	1	1				0	0	U		
<i>χ</i> 123		U	-1	-1	1	1				0	0	0		
χ_{124}	0	U	0	0	0	0	0	0	0	0	0	0	0	0

The character table of $2^8:O_8^+(2)$ (continued)

$$A = E(7) + E(7)^2 + E(7)^4$$

$$B = -E(15)^7 - E(15)^{11} - E(15)^{13} - E(15)^{14}$$

9

$2^{4} \cdot S_6$ and $2^{5} \cdot S_6$ as maximal subgroups of HS and HS:2 respectively

Prologue

The group HS:2 is a full automorphism group of the Higman-Sims group HS. The groups $2^4 \cdot S_6$ and $2^5 \cdot S_6$ are maximal subgroups of HS and HS:2 respectively. The group $2^4 \cdot S_6$ is of order 11520 and $2^5 \cdot S_6$ is of order 23040 and each of them is of index 3850 in HS and HS:2 respectively. The aim of this chapter is to compute $\overline{G} = 2^5 \cdot S_6$ as a group of the form $2^4 \cdot S_6 \cdot 2$, that is $\overline{G} = \overline{G_1} \cdot 2$. We then compute the Fischer-Clifford matrices of $\overline{G_1}$ and \overline{G} respectively. These together with the partial character tables of the inertia factors of $\overline{G_1}$ and \overline{G} are used to compute the full character tables of $\overline{G_1}$ and \overline{G} respectively. We then fuse $\overline{G_1}$ into \overline{G} .

9.1. Introduction

The Higman-Sims group, HS is a sporadic simple group of order $2^{9}.3^{2}.5^{3}.7.11 = 44352000$. This is a group that was discovered in 1967 by Higman and Sims [49]. This is a simple group of index two in the group of automorphisms of the Higman-Sims graph. Higman and Sims were attending a presentation by Marshall Hall on the Hall-Janko group J_2 , which is a permutation group on a hundred points with the stabilizer of a point a subgroup with the other two orbits of length 36 and 63. They then thought of a group of permutations on a 100 points containing the Mathieu group M_{22} , which has a permutation representation on 22 and 77 points . From these two ideas they found HS, with one-point stabilizer isomorphic to M_{22} . Higman, in 1969 [50], independently discovered this group as a doubly transitive group acting on a certain "geometry" on 176 points. In his classical paper Conway [22] showed that HS is a subgroup of each of the Conway groups Co_1, Co_2 and Co_3 . This group is also one of the seven sporadic groups found in Co_1 but not in the Mathieu groups and this set of groups is also known as *second generation* of sporadic groups. The group HS:2 is of order $88704000 = 2^{10}.3^2.5^3.7.11$ and it is the full automorphism group of HS. The aim of this chapter is to compute the Fischer-Clifford matrices of $\overline{G_1}$ and \overline{G} . We use these matrices and the partial character tables of each inertia factor group to compute the full character table of each group. The notation and method used is taken from F. Ali [1]. For more information on Fischer-Clifford theory and projective characters of non-split extensions one is encouraged to read [1, 2, 55, 66, 107, 108, 109].

We follow the work of Conway leading up to the computations of the groups HS and HS:2 in the subsections below.

9.1.1 The Conway Groups

Leech created a lattice that gives the tightest lattice packing of spheres in 24 dimension [72]. Conway analyzed the symmetry of this lattice in detail in [22] and discovered three previously unknown sporadic groups namely the Co_1, Co_2 and Co_3 . Let us give the definition of a Leech lattice which is given as Theorem 5.1 in [125].

Definition 9.1.1. A *Leech lattice* Λ is a 24 dimensional even integral lattice containing no vectors of norm 2, 196560 vectors of norm 4, 16773120 vectors of norm 6 and 398034000 vectors of norm 8.

We first construct the biggest Conway group $Aut(\Lambda) = .O = 2.Co_1$ as a group of 24×24 matrices and simultaneously we construct a Leech lattice. All the vectors of norm 8 in the Leech lattice fall into congruence classes of 48 pairs of mutually perpendicular vectors called the *crosses* and we get 8292375 such crosses. When .O acts on crosses, the stabilizer of a cross is a group $2^{12}:M_{24}$, which is a maximal in .O. So .O is a group of order $8292375.2^{12}.|M_{24}|$. The group .O is a perfect group with Z(.O) = 2. The quotient of this group by the center is a group denoted by $.1 = Co_1$ and is of order

$$|Co_1| = 4157776806543360000 = 2^{21} \cdot 3^9 \cdot 5^4 \cdot 7^2 \cdot 11 \cdot 13 \cdot 23.$$

Note that the action of O on crosses is transitive and Co_1 is a simple group.

Also .*O* acts transitively on vectors of norm 4 having the products ± 4 or 0. These three orbits of $2^{12}:M_{24}$ on vectors of norm 4 are fused into a single orbit under $2.Co_1$. The stabilizer of a vector of norm 4 is denoted by Co_2 , where

$$|Co_2| = 42305421312000 = 2^{18} \cdot 3^6 \cdot 5^3 \cdot 7 \cdot 11 \cdot 23.$$

Lastly O is transitive on vectors of norm 6. The stabilizer of a vector of norm 6 is denoted by Co_3 and is of order

$$|Co_3| = 423054213122000 = 2^{10}.3^7.5^3.7.11.23.$$

From the ATLAS [23] we see that $Co_3 \leq Co_2 \leq Co_1$ with Co_2 and Co_3 both maximal subgroup of Co_1 and Co_3 a maximal subgroup of Co_2 .

9.1.2 The Higman-Sims Group

We get the Higman-Sims group HS by showing that Co_3 acts transitively on the set S of 11178 vectors of norm 4 which have inner product -2 with vector v, when $v = (-2^{12}, 0^{12})$. The monomial group $2 \times M_{12}$ fixes v and has six orbits on S. When $u = (-5, -1^{23})$, the group M_{23} fixes u and has five orbits on S. The only way for both these sets of orbits to fuse into orbits for Co_3 is a single orbit of length 11178. Thus the stabilizer in Co_3 of such a vector in S, is a subgroup of index 11178. This is the Higman-Sims group HS of order

$$|HS| = 44352000 = 2^9 \cdot 3^2 \cdot 5^3 \cdot 7 \cdot 11$$

Moreover if we let $w = (5, 1, 1^{22})$ and $x = (-1, -5, -1^{22})$ the stabilizer of the set $\{w, x\}$ is the monomial group M_{22} :2 and we get an involution of the group which interchanges the two vectors. This results in HS extending to HS:2 which is a full automorphism group of HS. A complete list of maximal subgroups of the Conway groups is provided in Table 5.3 of [125]. For further reading one can also go into [22, 72, 89, 125].

We use [124] to find two 20×20 matrices a and b with a from class 2A, b from class 5A and $HS = \langle a, b \rangle$. Again using [124] we find two 20×20 matrices c, d from classes 2C and 5C of HS:2 respectively, with $HS:2 = \langle c, d \rangle$. From the HS computed, HS:2 is an automorphism group of an isomorphic copy of it.

9.1.3 The Groups $2^{4}S_{6}$ and $2^{5}S_{6}$

The group $2^{4} \cdot S_6$, is a group of order 11520, and a maximal subgroup of HS.. There is a group $2^{4} \cdot S_6$ which is maximal in $\overline{M_{22}}$ and hence sits inside HS:2. Our group $2^{4} \cdot S_6$ is a subgroup of HS and hence is also a subgroup of HS:2. For further reading on $2^4 \cdot S_6$ as a maximal subgroup of $\overline{M_{22}}$ one can read [81] and [120]. The group $2^{5} \cdot S_6$ is a group of order 23040 and is a maximal subgroup of the automorphism group of the Higman-Sims HS:2. The groups $2^{4} \cdot S_6$ and $2^{5} \cdot S_6$ are unique maximal subgroups of their form in HS and HS:2 respectively. Using generators a and b of HS and Programme G (see Appendix A) we obtain elements a'_1 and b'_1 with $o(a'_1) = 2$, $o(b'_1) = 5$ and $\overline{G'_1} = \langle a'_1, b'_1 \rangle = 2^{4} \cdot S_6$. Similarly using generators c and d from HS:2 and Programme H (see Appendix A) we obtain two elements c' and d' with o(c') = 2, o(d') = 5 and $\overline{G'} = \langle c', d' \rangle = 2^{5} \cdot S_6$. Both Programme G and Programme H are obtained from [124]. Our aim is to compute $\overline{G} = 2^{5} \cdot S_6$ as $\overline{G_1} \cdot 2$, where $\overline{G_1} = 2^4 \cdot S_6$ and we compute these inside HS:2. Since $\overline{G'_1}$ is in HS we seek for its isomorphic copy $\overline{G_1}$ in HS:2. The extension of $\overline{G_1}$ is $\overline{G_1} \cdot 2 = \overline{G}$ and $\overline{G} \cong \overline{G'}$.

Having obtained $\overline{G'}$, using GAP [41], we get three of its subgroups of order 11520. By methods of coset analysis (see Chapter 2), we determine that each of these three subgroups is of the form $2^{4} S_6$. From these three subgroups only one, $\overline{G_1}$, is isomorphic to $\overline{G'_1}$ in HS. The group $\overline{G_1}$ has seven generators of which five are of order two, one of order 5 and one of order 6. To this list of seven generators we add one of the generator of HS:2 of order two namely c. The group generated by these eight elements is $\overline{G} = 2^4 \cdot S_6 \cdot 2 = 2^5 \cdot S_6$.

The groups $2^{4} S_{6}$ and $2^{5} S_{6}$ will be discussed fully in Sections 9.2 and 9.3 respectively.

9.2. The Group $\overline{G_1} = 2^{4} \cdot S_6$

9.2.1 Construction of $G_1 \cong S_6$

From [124] we get two 20×20 matrices a and b over GF(2) with o(a) = 2, o(b) = 5, o(ab) = 11 and $HS = \langle a, b \rangle$. Again from [124] we get Programme G (see Appendix A).

Let $a'_1 = output[1]$ and $b'_1 = output[2]$. Then we have $o(a'_1) = 2$, $o(b'_1) = 5$, $o(a'_1b'_1) = 6$ and $\overline{G'_1} = \langle a'_1, b'_1 \rangle = 2^{4} \cdot S_6$. Up to isomorphism, there is only one group of the type $2^{4} \cdot S_6$ that is a maximal subgroup of HS and this has 21 conjugacy classes of which two are classes of involutions. Going back to [124], we get two 20×20 matrices c and d with o(c) = 2, o(d) = 5 and $HS: 2 = \langle c, d \rangle$. Again from [124] we get Programme H (see Appendix A).

Let $c'_1 = output[1]$ and $d'_1 = output[2]$. We get that $o(c'_1) = 2$, $o(d'_1) = 10$, $o(c'_1d'_1) = 6$ and $\overline{G'} = \langle c'_1, d'_1 \rangle = 2^5 S_6$. Using GAP [41], we get eight normal subgroups of $\overline{G'}$. Three of these groups (we call them S1, S2, S3) are of order 11520 and for each group the conjugacy class 2A has 15 elements and when S_6 acts on 2^4 we get two orbits of length 1 and 15 hence all these groups are of the form $2^{4} S_{6}$. One of them $(S2 = 2^{4} : S_{6})$, however has 24 conjugacy classes and is thus not a maximal subgroup of HS. The other one (S3, a split extension of 2^5 by A_6) has five classes of involutions and again is not a maximal subgroup of HS, this group from [81] and [120] is actually a maximal subgroup of $\overline{M_{22}}$. This leaves us with the group $S1 = \overline{G_1} \cong \overline{G'_1}$. See Remark 9.2.1 for more details on groups S1, S2 and S3. The group $\overline{G_1}$ has seven generators $a_1, a_2, a_3, a_4, a_5, a_6$ and a_7 with a_1 of order 2, a_2 of order 5 and a_3 of order 6 and the rest of order 2. We use GAP to compute normal subgroups of $\overline{G_1}$ and it has only one proper normal subgroup, the elementary abelian group $N_1 = 2^4$. Our aim is to act $\overline{G_1}$ on N_1 and to do this we use Programme C (see Appendix A) and this requires us to consider N_1 as a full row space V_1 of dimension four over GF(2). The action of $\overline{G_1}$ on V_1 is multiplication of V_1 from the right. This then requires us to rewrite $\overline{G_1}$, from a 20 × 20 representation to a 4 × 4 one. To do this we act $\overline{G_1}$ on N_1 by acting the seven generators a_i , $i = 1, \dots, 7$ of $\overline{G_1}$ on the four generators λ_i , $i = 1, \dots, 4$ of N_1 .

Writing this action as maps we get.

$$\begin{split} a_1 &: \lambda_1 \to \lambda_2, \ \lambda_2 \to \lambda_1, \ \lambda_3 \to \lambda_1 \lambda_3 \lambda_4, \ \lambda_4 \to \lambda_1 \lambda_2 \lambda_4; \\ a_2 &: \lambda_1 \to \lambda_2, \ \lambda_2 \to \lambda_4, \ \lambda_3 \to \lambda_1 \lambda_2, \ \lambda_4 \to \lambda_2 \lambda_3 \lambda_4; \\ a_3 &: \lambda_1 \to \lambda_2 \lambda_3 \lambda_4, \ \lambda_2 \to \lambda_4, \ \lambda_3 \to \lambda_1 \lambda_2 \lambda_3, \ \lambda_4 \to \lambda_2; \end{split}$$

For the rest that is a_4 to a_7 we get

$$\lambda_i: \lambda_1 \to \lambda_1, \ \lambda_2 \to \lambda_2, \ \lambda_3 \to \lambda_3, \ \lambda_4 \to \lambda_4.$$

Writing this in matrix form we get :

$$\alpha_1 := \begin{pmatrix} 0 & 1 & 0 & 0 \\ 1 & 0 & 0 & 0 \\ 1 & 0 & 1 & 1 \\ 1 & 1 & 0 & 1 \end{pmatrix}; \quad \alpha_2 := \begin{pmatrix} 0 & 1 & 0 & 0 \\ 0 & 0 & 0 & 1 \\ 1 & 1 & 0 & 0 \\ 0 & 1 & 1 & 1 \end{pmatrix}; \\ \alpha_3 := \begin{pmatrix} 0 & 1 & 1 & 1 \\ 0 & 0 & 0 & 1 \\ 1 & 1 & 1 & 0 \\ 0 & 1 & 0 & 0 \end{pmatrix}.$$

For the rest α_4 to α_7 we get

$$\alpha_i := \left(\begin{array}{rrrr} 1 & 0 & 0 & 0 \\ 0 & 1 & 0 & 0 \\ 0 & 0 & 1 & 0 \\ 0 & 0 & 0 & 1 \end{array} \right).$$

Let $G_1 = \langle \alpha_1, \alpha_2, \alpha_3 \rangle \cong S_6$, that is the action of $\overline{G_1}$ on N_1 is isomorphic to S_6 .

Remark 9.2.1. Note that $N_1 = 2^4$ is generated by 4 commuting involutions from the class 2A of HS. From ATLAS we can see that $S1 = \overline{G_1} = N_{HS}(N_1)$, $S2 = N_{\overline{M}_{22}}(N_1)$, and $N_{HS:2}(N_1) = 2^{5} \cdot S_6 = \overline{G}$. As observed, S1, S2, S3 are non-isomorphic maximal subgroups of \overline{G} and that S2 and S3 do not sit inside HS. Our computations show that

$$S1 = 2^{4} \cdot S_{6} = \overline{G} \cap HS \leq_{max} HS \leq_{max} HS:2$$

$$S2 = 2^{4} \cdot S_{6} = \overline{G} \cap \overline{M}_{22} \leq_{max} \overline{M}_{22} \leq_{max} HS:2;$$

$$S3 = 2^{4} \cdot (A_{6} \times 2) \cong 2^{5} \cdot A_{6} \leq_{max} \overline{G} \leq_{max} HS:2;$$

$$S1 \cap S2 = S2 \cap S3 = S1 \cap S3 = N_{M_{22}}(N_{1}) = 2^{4} \cdot A_{6} \leq_{max} M_{22} \leq_{max} HS.$$

In addition to the character table of $S1 = \overline{G_1}$ we also give the character tables of S2 and S3, in Table 1 and Table 2 of Appendix B respectively. It is also interesting to note that except for the conjugacy classes, the character tables of S1 and S2 are the same. A pictorial view of Remark 9.2.1 is given in Figure 9.1, where $A = 2^4 \cdot A_6$.

We compute the permutation characters of HS:2 when acting on S1, S2 and S3. For interest sake we also include $\chi(HS|S1)$ and $\chi(2^5.S_6|Si)$, i = 1, 2, 3. We use GAP[41] for our computations.

$$\begin{split} \chi(HS|S1) &= 1a + 22a + 77aa + 154a + 175a + 693a + 770a + 825a + 1056a = \chi(HS:2|2^5.S_6) \\ \chi(2^5.S_6|S1) &= 1a + 1b, \\ \chi(2^5.S_6|S2) &= 1a + 1c, \\ \chi(2^5.S_6|S3) &= 1a + 1d. \end{split}$$



Figure 9.1: S1,S2 and S3

 $\chi(HS:2|S1) = 1a + 1b + 22a + 22b + 77aa + 77bb + 154a + 154b + 175a + 175b + 693a + 693b + 770a + 770b + 825a + 825b + 1056a + 1056b,$

 $\chi(HS:2|S2) = 1a + 22aa + 77aaa + 154a + 175a + 231a + 693a + 770aa + 825aa + 1056a + 1925a,$ $\chi(HS:2|S3) = 1a + 22a + 22b + 77aa + 77b + 154a + 175a + 231a + 693a + 770a + 770b + 825a + 825b + 1056a + 1925b.$

Lemma 9.2.2. $\overline{G} = S1 \cup S2 \cup S3$.

Proof: First we see that $\overline{G} \supseteq S1 \cup S2 \cup S3$. But we also have $S1 \cup S2 \cup S3 = (S1 - A) \cup (S2 - A) \cup (S3 - A) \cup A$. Hence

$$\begin{split} |S1 \cup S2 \cup S3| &= |S1 - A| + |S2 - A| + |S3 - A| + |A| \\ &= (16 \times 6! - 16 \times \frac{6!}{2}) + (16 \times 6! - 16 \times \frac{6!}{2}) + (16 \times 6! - 16 \times \frac{6!}{2}) + (16 \times \frac{6!}{2}) \\ &= 3(16 \times 6! - 16 \times \frac{6!}{2}) + (16 \times \frac{6!}{2}) \\ &= 3 \times 16 \times 6! - 2 \times 16 \times \frac{6!}{2} \\ &= 2 \times 16 \times 6! \\ &= |2^5 . S_6|. \end{split}$$

Thus $2^5 \cdot S_6 = S1 \cup S2 \cup S3.\square$

Theorem 9.2.3. *HS*:2 has only three conjugacy classes of subgroups of type 2^4 . A_6 .2. In particular S1 and S2 are of type 2^4 . A_6 .2₁ and S3 is of type 2^4 . $(A_6 \times 2)$.

Proof: From the ATLAS we can see that if $H \leq HS:2$ is of type $2^4.A_6.2$, then H must sit in one of the maximal subgroups of HS:2 of type HS, \overline{M}_{22} or $2^5.S_6$. Also since $N_{HS:2}(S1) \supseteq N_{\overline{G}}(S1) = \overline{G}$ and \overline{G} is maximal but not normal in HS:2, we have $N_{HS:2}(S1) = \overline{G}$. Hence $[HS:2:N_{HS:2}(S1)] =$ $[HS:2:\overline{G}] = 3850$. Similarly since $N_{HS:2}(S2) = N_{HS:2}(S3) = \overline{G}$, we have $[HS:2:N_{HS:2}(S2)] =$ $[HS:2:N_{HS:2}(S3)] = 3850$. Hence we have 3 conjugacy classes for the subgroups of type $2^4.A_6.2$ in HS:2. Thus the total number of subgroups of type $2^4.A_6.2$ in HS:2 is $3 \times 3850 = 11550.\Box$

9.2.2 Conjugacy Classes and Inertia Factors of $\overline{G_1}$

Using GAP [41], we compute the conjugacy classes of $2^{4} \cdot S_6$. The action of $\overline{G_1}$ on N_1 is viewed as the action of of G_1 on V_1 . If G_1 acts on N_1 , we get two orbits of length 1 and 15. From the ATLAS [23], by checking on the indices of maximal subgroups of S_6 , we can see that there are two inertia factor groups namely S_6 and $S_4 \times 2$. The full inertia groups are of the form $\overline{H_i} = 2^4 \cdot H_i$ of indices 1 and 15 in $2^4 \cdot S_6$ respectively. We note that $H_1 \cong S_6$ and $H_2 \cong S_4 \times 2$. The character tables of H_1 and H_2 are easy to compute. The fusion of $S_4 \times 2$ into S_6 is given in Table 9.1.

Table 9.1: The fusion of $S_4 \times 2$ into S_6

$[x]_{S_4 \times 2}$	\longrightarrow	$[g_1]_{S_6}$
1A		1A
2A		2C
2B		2B
2C		2B
2D		2A
2E		2A
3A		3A
4A		4A
4B		4B
6A		6A

We computed the conjugacy classes of $2^{4} \cdot S_6$ by using GAP [41] and then fused them into HS. Having the length of each coset, we use the fusion map to convert the conjugacy classes of $2^{4} \cdot S_6$ into the form that is required for the computation of Fischer-Clifford matrices (that is into a form normally obtained by coset-analysis). We give the conjugacy classes of $2^{4} \cdot S_6$ in Table 9.2.

9.2.3 Fischer-Clifford Matrices of $\overline{G_1}$

Again from the inertia factors and fusions we compute the Fischer-Clifford matrices.

$[g]_{S_6}$	$[x]_{2^{4}\cdot S_{6}}$	$C_{2^{4}\cdot S_{6}}(x)$	\longrightarrow HS
1A	1A	11520	1A
	2A	768	2A
	2B	96	2B
2A	4A	384	4A
	4B	128	4B
	2C	64	2A
2B	$4\mathrm{C}$	64	4B
	4D	32	4B
2C	2D	192	2A
	$4\mathrm{E}$	64	4B
3A	3A	192	3A
	6A	24	6B
3B	3B	18	3A
4A	$4\mathrm{F}$	16	4B
	8A	16	8A
4B	4G	16	4C
	8B	16	8A
5A	5A	5	5C
6A	6B	12	6A
	12A	12	12A
6B	6C	6	6B

Table 9.2: Conjugacy Classes Of $2^4 S_6$

From the fusions we get $M(1A) = \begin{bmatrix} 1 & 1 \\ 15 & -1 \end{bmatrix}$.

Let $Irr(HS) = \{\psi_i : 1 \le i \le 24\}$ as given in the ATLAS. We get :

$[x]_{HS}$	1A	2A
ψ_2	22	6
ψ_3	77	13

Let r and s be the rows of the Fischer-Clifford matrix M(1A). Since $\langle (\psi_2)_N, 1_N \rangle = 7$ we get the following decomposition, 22 = 7 + 15s. Thus s = 1 and this shows that the partial character table of H_2 comes from the ordinary table of H_2 . Hence we use the ordinary character table of $S_4 \times 2$. We use the properties of the Fischer - Clifford matrices and the fusion of $S_4 \times 2$ into S_6 , the centralizer orders of $2^{4}S_6$, the fusion of \overline{G} into HS, together with restriction of HS to \overline{G} that forces the signs of the Fischer-Clifford matrices. We give these in Table 9.3

Note the change of sign in M(2A). For example we calculate the partial character table corresponding to coset $2A \in S_6$. From M(2A) we get $M_1(2A) = \begin{bmatrix} 1 & 1 & 1 \\ 0 & -6 & 2 \end{bmatrix}$

Let $C_1(2A), C_2(2A)$ be the partial character tables of the inertia factors for the classes that fuse to $2A \in S_6$. Then the portions of the character table of $\overline{G} = 2^4 \cdot S_6$ corresponding to the coset 2A are :

M(1A) =	$\left[\begin{array}{rrr}1&1\\15&-1\end{array}\right]$	$M(2A) = \begin{bmatrix} 1 & 1 & 1 \\ -1 & 1 & 1 \\ 0 & -6 & 2 \end{bmatrix}$					
M(2B) =	$\begin{bmatrix} 1 & 1 & 1 \\ 1 & 1 & -1 \\ 2 & -2 & 0 \end{bmatrix}$	$M(2C) = \left[\begin{array}{rrr} 1 & 1 \\ 3 & -1 \end{array} \right]$					
M(3A) =	$\begin{bmatrix} 1 & 1 \\ 3 & -1 \end{bmatrix}$	$M(3B) = \left[\begin{array}{c} 1 \end{array}\right]$					
M(4A) =	$\begin{bmatrix} 1 & 1 \\ 1 & -1 \end{bmatrix}$	$M(4B) = \begin{bmatrix} 1 & 1\\ 1 & -1 \end{bmatrix}$					
M(6A) =	1 1 1 -1	$M(5A) = M(6B) = \begin{bmatrix} 1 \end{bmatrix}$					

Table 9.3: The Fischer-Clifford matrices of $2^{4} \cdot S_6$

We get the character table of $2^4 S_6$ in Table 9.4 which can be compared to the one in GAP.

•

	1A			2A			2B		2C		3A		3B	4A	
	1a	2a	2b	4a	4b	2c	4c	4d	2d	4e	3a	6a	3b	4f	8a
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	-1	-1	-1	1	1	1	-1	-1	1	1	1	-1	-1
χ_3	5	5	-3	-3	-3	1	1	1	1	1	2	2	-1	-1	-1
χ_4	5	5	3	3	3	1	1	1	-1	-1	2	2	-1	1	1
χ_5	5	5	-1	-1	-1	1	1	1	3	3	-1	-1	2	1	1
χ_6	5	5	1	1	1	1	1	1	-3	-3	-1	-1	2	-1	-1
χ_7	9	9	-3	-3	-3	1	1	1	-3	-3	0	0	0	1	1
χ_8	9	9	3	3	3	1	1	1	3	3	0	0	0	-1	-1
χ_9	10	10	-2	-2	-2	-2	-2	-2	2	2	1	1	1	0	0
χ_{10}	10	10	2	2	2	-2	-2	-2	-2	-2	1	1	1	0	0
χ_{11}	16	16	0	0	0	0	0	0	0	0	-2	-2	-2	0	0
χ_{12}	15	-1	-1	-5	3	3	-1	-1	3	-1	3	-1	0	1	-1
χ_{13}	15	-1	1	5	-3	3	-1	-1	-3	1	3	-1	0	-1	1
χ_{14}	15	-1	-1	7	-1	-1	3	-1	3	-1	3	-1	0	-1	1
χ_{15}	15	-1	1	-7	1	-1	3	-1	-3	1	3	-1	0	1	-1
χ_{16}	30	-2	2	-2	-2	2	2	-2	-6	2	-3	1	0	0	0
χ_{17}	30	-2	-2	2	2	2	2	-2	6	-2	-3	1	0	0	0
χ_{18}	45	-3	3	3	-5	1	-3	1	3	-1	0	0	0	1	-1
χ_{19}	45	-3	-3	9	1	-3	1	1	-3	1	0	0	0	1	-1
χ_{20}	45	-3	3	-9	-1	-3	1	1	3	-1	0	0	0	-1	1
χ_{21}	45	-3	-3	-3	5	1	-3	1	-3	1	0	0	0	-1	1

Table 9.4: The character table of $2^{4} \cdot S_6$

The character	table	of	$2^{4} \cdot S_{6}$	(continued $)$
---------------	-------	----	---------------------	----------------

	4B		5a	6A		6B
	4g	8b	5a	6b	12a	6c
χ_1	1	1	1	1	1	1
χ_2	1	1	1	-1	-1	-1
χ_3	-1	-1	0	0	0	1
χ_4	-1	-1	0	0	0	-1
χ_5	-1	-1	0	-1	-1	0
χ_6	-1	-1	0	1	1	0
χ_7	1	1	-1	0	0	0
χ_8	1	1	-1	0	0	0
χ_9	0	0	0	1	1	-1
χ_{10}	0	0	0	-1	-1	1
χ_{11}	0	0	1	0	0	0
χ_{12}	1	-1	0	1	-1	0
χ_{13}	1	-1	0	-1	1	0
χ_{14}	-1	1	0	1	-1	0
χ_{15}	-1	1	0	-1	1	0
χ_{16}	0	0	0	1	-1	0
χ_{17}	0	0	0	-1	1	0
χ_{18}	-1	1	0	0	0	0
χ_{19}	1	-1	0	0	0	0
χ_{20}	1	-1	0	0	0	0
χ_{21}	-1	1	0	0	0	0

9.3. The Group $\overline{G} = 2^{5} S_6$

Having completed the computation of the full character table of $2^{4} \cdot S_6$, we now turn our attention to $2^{5} \cdot S_6$. We compute $2^{5} \cdot S_6 = 2^{4} \cdot S_6 \cdot 2$, this we do by adding the generator c of HS:2, that is from $\overline{G_1}$ we get $\overline{G} = \langle a_1, a_2, a_3, a_4, a_5, a_6, a_7, c \rangle$. Since $2^{5} \cdot S_6$ is the only group of its type that is a maximal subgroup of HS:2, we have $\overline{G} \cong \overline{G'}$, where $\overline{G'}$ was computed using Programme H. Our aim is to compute the full character table of $2^{5} \cdot S_6$. We first want to let \overline{G} act on the elementary abelian group $N = 2^5$. We use GAP [41] to compute $N = 2^5$ as a normal subgroup of \overline{G} .

9.3.1 Construction of $G \cong S_6$

For the action of \overline{G} we use Programme C (see Appendix A). We consider N as a full row vector space V of dimension 5 over GF(2). For us to be able to act on a five dimensional vector space V it becomes necessary to rewrite \overline{G} from 20×20 to a 5×5 representation. To do this we first take the eight generators of \overline{G} namely a_1 to a_7 and c. We let these act on generators γ_i , $1 = 1, \dots, 5$ of our elementary abelian group $N = 2^5$.

Writing these as maps we get :

$$\begin{aligned} a_1: \gamma_1 \to \gamma_1, \ \gamma_2 \to \gamma_3 \gamma_4, \ \gamma_3 \to \gamma_1 \gamma_3, \ \gamma_4 \to \gamma_1 \gamma_2 \gamma_3, \ \gamma_5 \to \gamma_1 \gamma_2 \gamma_3 \gamma_4 \gamma_5; \\ a_2: \gamma_1 \to \gamma_2 \gamma_3 \gamma_4, \ \gamma_2 \to \gamma_3, \ \gamma_3 \to \gamma_1 \gamma_3, \ \gamma_4 \to \gamma_2, \ \gamma_5 \to \gamma_2 \gamma_5; \\ a_3: \gamma_1 \to \gamma_1 \gamma_2, \ \gamma_2 \to \gamma_1 \gamma_2 \gamma_3 \gamma_4, \ \gamma_3 \to \gamma_4, \ \gamma_4 \to \gamma_1 \gamma_4, \ \gamma_5 \to \gamma_1 \gamma_2 \gamma_3 \gamma_4 \gamma_5; \\ c: \gamma_1 \to \gamma_3, \ \gamma_2 \to \gamma_2, \ \gamma_3 \to \gamma_1, \ \gamma_4 \to \gamma_4, \ \gamma_5 \to \gamma_5; \end{aligned}$$

For the rest a_4 to a_7 we get

$$a_i: \gamma_1 \to \gamma_1, \ \gamma_2 \to \gamma_2, \ \gamma_3 \to \gamma_3, \ \gamma_4 \to \gamma_4, \ \gamma_5 \to \gamma_5.$$

Writing this in matrix form we get :

$$\beta_1 = \begin{pmatrix} 1 & 0 & 0 & 0 & 0 \\ 0 & 0 & 1 & 1 & 0 \\ 1 & 0 & 1 & 0 & 0 \\ 1 & 1 & 1 & 0 & 0 \\ 1 & 1 & 1 & 1 & 1 \end{pmatrix}, \quad \beta_2 = \begin{pmatrix} 0 & 1 & 1 & 1 & 0 \\ 0 & 0 & 1 & 0 & 0 \\ 1 & 0 & 1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 1 \end{pmatrix},$$

	$\left(1 \right)$	1	0	0	0)		0	0	1	0	0)	١
	1	1	1	1	0		0	1	0	0	0	
$\beta_3 =$	0	0	0	1	0	$, \beta_4 =$	1	0	0	0	0	.
	1	0	0	1	0		0	0	0	1	0	
	1	1	1	1	1)		0	0	0	0	1))

For the rest β_5 to β_8 we get that $\beta_i = I_5$.

Let $G = \langle \beta_1, \beta_2, \beta_3, \beta_4 \rangle$, then $G \cong S_6$ which means that the action of \overline{G} on N is isomorphic to S_6 .

9.3.2 Conjugacy Classes and Inertia Factors of \overline{G}

The action of \overline{G} on N is reflected by the action of G on V. We use Programme C (see Appendix A). When G acts on V we get four orbits of conjugacy classes of lengths 1, 6, 10 and 15. Let G^t be the set of all transpose of elements of G. The group G^t can also be generated by transpose matrices of each generator of G. When G^t acts on V, which is the equivalent of G acting on Irr(N), by Brauer's Theorem [14] we get four orbits but these are of lengths 1, 1, 15 and 15. These have corresponding point stabilizers H_1, H_2, H_3 and H_4 . Let the full inertia groups be $\overline{H_i} = 2^5.H_i$, i = 1, 2, 3, 4. From the ATLAS [23], the corresponding inertia factor groups are S_6 , S_6 , $S_4 \times 2$ and $S_4 \times 2$. Where we get $H_1 \cong H_2 \cong S_6$ and $H_3 \cong H_4 \cong S_4 \times 2$. The character tables of S_6 and that of HS:2 are obtained from the ATLAS [23]. We give also, the fusion of $S_4 \times 2$ into S_6 in Table 9.5.

Table 9.5 :	The	fusion	of	S_4	\times	2	into	S_6
---------------	-----	--------	----	-------	----------	----------	------	-------

$[x]_{S_4 \times 2}$	$\longrightarrow [g_1]_{S_6}$
1A	1A
2A	2C
2B	2B
2C	2B
2D	2A
2E	2A
3A	3A
4A	4A
4B	4B
6A	6A

We computed the conjugacy classes of $2^{5} S_6$ by using GAP [41] and then fused them into HS:2. Having the length of each coset, we use the fusion map to convert the conjugacy classes of $2^{5} S_6$ into the form that is required for the computation of Fischer-Clifford matrices (that is into a form normally obtained by coset-analysis). We give the conjugacy classes of $2^{5} S_6$ in Table 9.6

$[g]_{S_6}$	$[x]_{2^{5} \cdot S_{6}}$	$C_{2^{5}\cdot S_{6}}(x)$	\longrightarrow $HS:2$		
	1A	23040	1A		
1A	2A	3840	$2\mathrm{D}$		
	2B	2304	2C		
	2C	1536	2A		
	2D	768	2C		
	4A	768	4A		
2A	$2\mathrm{E}$	256	2D		
	4B	256	4B		
	4C	192	4A		
	$2\mathrm{F}$	192	$2\mathrm{B}$		
	2G	128	2A		
	2H	128	2D		
2B	4D	128	4D		
	$4\mathrm{E}$	128	4B		
	$4\mathrm{F}$	64	$4\mathrm{C}$		
	4G	64	4A		
	2I	384	2A		
2C	2J	384	2C		
	$4\mathrm{H}$	64	4B		
	4I	64	4D		
	3A	144	3A		
3A	6A	144	6C		
	6B	48	6E		
	6C	48	6B		
3B	$_{3\mathrm{B}}$	36	3A		
	6D	36	6A		
	4J	32	4A		
4A	8A	32	$8\mathrm{C}$		
	4K	32	4B		
	8B	32	8A		
	4L	32	4C		
$\parallel 4B$	8C	32	8A		
	4M	32	4D		
<u> </u>	8D	32	8D		
$\parallel 5A$	5A	10	$5\mathrm{C}$		
	10A	5	10D		
	6E	24	6D		
6A	6F	24	6A		
	12A	24	12A		
	12B	24	12B		
$\ 6B$	6G	12	$6\mathrm{E}$		
	6H	12	6A		

Table 9.6: Conjugacy Classes Of $2^{5.}S_6$

9.3.3 Fischer Clifford Matrices of \overline{G}

From the fusions and orbit lengths and centralizer orders, we compute the Fischer-Clifford matrix M(1A).

$$M(1A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 15 & 5 & -3 & -1 \\ 15 & -5 & 3 & -1 \end{bmatrix}$$

Having computed M(1A) we want to determine the type of partial character tables we are going to use for our computations. We follow the methods used by F.Ali [1]. We use the character table of $HS:2 = \langle a, b \rangle$. Let $Irr(HS:2) = \{\Psi_i : 1 \le i \le 39\}$, the notation is the same at the one used in the ATLAS [23].

$C_{\overline{G}}(x)$	23040	3840	2304	1536
$[x]_{HS:2}$	1A	2A	2B	2C
Ψ_2	1	-1	-1	1
Ψ_3	22	0	8	6
Ψ_4	22	0	-8	6
Ψ_5	77	5	21	13
Ψ_6	77	-5	-21	13

Let $\gamma_1, \gamma_2, \gamma_3, \gamma_4$ be the rows of the Fischer - Cliford matrix M(1A). First we get

$$<(\Psi_2)_N, 1_N> = \frac{1}{32}(1-6-10+15) = 0,$$

$$<(\Psi_3)_N, 1_N> = \frac{1}{32}(22\times1+6\times0+10\times8+15\times6) = \frac{1}{32}(22+80+90) = 6,$$

$$<(\Psi_4)_N, 1_N> = \frac{1}{32}(22\times1+6\times0+10\times(-8)+15\times6) = \frac{1}{32}(22-80+90) = 1,$$

$$<(\Psi_5)_N, 1_N> = \frac{1}{32}(77\times1+6\times5+10\times21+15\times13) = \frac{1}{32}(77+30+210+195) = 16.$$

Restricting the character Ψ_3 to N, since $\langle (\Psi_3)_N, 1_N \rangle = 6$, we get the following equations, where a, b, c represent coefficients of $\gamma_2, \gamma_3, \gamma_4$ respectively.

$$22 = 6 + a + 15b + 15c$$

$$0 = 6 - a + 5b - 5c,$$

$$8 = 6 - a - 3b + 3c,$$

$$6 = 6 + a - b - c.$$

Solving we get : a = 1, b = 0 and c = 1. So we have the following decomposition.

$$(\Psi_3)_N = 6\gamma_1 + \gamma_2 + \gamma_4.$$

Considering the coefficients of γ_2 and γ_4 we deduce that we have irreducible characters χ_2 and $\chi_4 \in Irr(\overline{G})$ with $deg(\chi_2) = 1$ and $deg(\chi_4) = 15$. Since $deg(\chi_2) = 1$, we only need to use the

$M(1A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & -1 & 1 \\ 15 & 5 & -3 & -1 \\ 15 & -5 & 3 & -1 \end{bmatrix}$	$M(2A) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 \\ -1 & 1 & -1 & 1 & -1 & 1 \\ -6 & 6 & 2 & -2 & 0 & 0 \\ 1 & 1 & 1 & 1 & -1 & -1 \\ -6 & -6 & 2 & 2 & 0 & 0 \\ 1 & -1 & 1 & -1 & -1 & 1 \end{bmatrix}$
$M(2B) = \begin{bmatrix} 1 & 1 & 1 & 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 & 1 & -1 \\ -2 & 2 & 2 & -2 & 0 & 0 \\ 1 & 1 & 1 & 1 & -1 & -1 \\ -2 & -2 & 2 & 2 & 0 & 0 \\ 1 & -1 & 1 & -1 & -1 & 1 \end{bmatrix}$	$M(2C) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 \\ 3 & 3 & -1 & -1 \\ -3 & 3 & 1 & -1 \end{bmatrix}$
$M(3A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 \\ 3 & -3 & 1 & -1 \\ 3 & 3 & -1 & -1 \end{bmatrix}$	$M(3B) = M(5A) = \begin{bmatrix} 1 & 1\\ 1 & -1 \end{bmatrix}$
$M(4A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ -1 & -1 & 1 & 1 \\ -1 & 1 & -1 & 1 \\ 1 & -1 & -$	$M(4B) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & 1 & -1 & 1 \\ -1 & 1 & -1 & 1 \\ 1 & -1 & -$
$M(6A) = \begin{bmatrix} 1 & 1 & 1 & 1 \\ 1 & -1 & 1 & -1 \\ -1 & 1 & -1 & 1 \\ 1 & 1 & -1 & -$	$M(6B) = \left[\begin{array}{rrr} 1 & 1 \\ -1 & 1 \end{array} \right]$

Table 9.7: The Fischer-Clifford matrices of $2^{5} S_6$

ordinary character table of H_2 . For $deg(\chi_4) = 15$, if $[x_1, x_2, \dots, x_t]$ is the transpose of the partial entries for the projective characters of H_4 on 1A, then $C_4(1A)M_4(1A)$ is a $t \times 4$ matrix with first set entry $15x_1 = 15$, hence $x_1 = 1$. This shows that the partial character table of H_4 that we used contains a character of degree 1. Thus the partial character table is an ordinary character table of H_4 . Similarly, one can show that $\langle (\Psi_3)_N, \gamma_2 \rangle = 6$. This gives us $(\Psi_3)_N = \gamma_1 + 6\gamma_2 + \gamma_3$. So again H_1 and H_3 have partial character tables that each contains a character of degree 1. Thus the partial character tables of S_6 and $S_4 \times 2$ respectively. Using fusions, centralizer orders of \overline{G} and properties of Fischer-Clifford matrices that can also be found in [1], [99] and [120], we complete Table 9.8 of Fischer - Clifford matrices.

To compute the character table of \overline{G} , as an example consider the following. Let $C_1(2A), C_2(2A), C_3(2A), C_4(2A)$ be the partial character tables of the inertia factors for the classes that fuse to $2A \in S_6$. Then the portions of the character table of $\overline{G} = 2^5 \cdot S_6$ corresponding to the coset 2A are :

	1		1	1	1	1	1	1
	-1		-1	-1	-1	-1	-1	-1
	-3		$^{-3}$	$^{-3}$	$^{-3}$	$^{-3}$	$^{-3}$	$^{-3}$
	3		3	3	3	3	3	3
	-1		-1	-1	-1	-1	-1	-1
$C_1(2A)M_1(2A) =$	1	$[1 \ 1 \ 1 \ 1 \ 1 \ 1] =$	1	1	1	1	1	1
	-3		$^{-3}$	$^{-3}$	$^{-3}$	-3	$^{-3}$	$^{-3}$
	3		3	3	3	3	3	3
	-2		$^{-2}$	-2	-2	-2	$^{-2}$	-2
	2		2	2	2	2	2	2
	0		0	0	0	0	0	0



The fusion of \overline{G} into HS:2 together with the restriction of characters of HS:2 to \overline{G} forces the signs of the Fischer-Clifford matrices and the order of the elements of the conjugacy classes of \overline{G} . Hence we give the character table in Table 9.8.

		1A			2A						2B					
	1a	2a	2b	2c	2d	4a	2e	4b	4c	2f	2g	2h	4d	4e	4f	4 g
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	-1	-1	-1	-1	-1	-1	1	1	1	1	1	1
χ_3	5	5	5	5	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1
χ_4	5	5	5	5	3	3	3	3	3	3	1	1	1	1	1	1
χ_5	5	5	5	5	-1	-1	-1	-1	-1	-1	1	1	1	1	1	1
χ_6	5	5	5	5	1	1	1	1	1	1	1	1	1	1	1	1
χ_7	9	9	9	9	-3	-3	-3	-3	-3	-3	1	1	1	1	1	1
χ_8	9	9	9	9	3	3	3	3	3	3	1	1	1	1	1	1
χ_9	10	10	10	10	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2	-2
χ_{10}	10	10	10	10	2	2	2	2	2	2	-2	-2	-2	-2	-2	-2
χ_{11}	16	16	16	16	0	0	0	0	0	0	0	0	0	0	0	0
χ_{12}	1	-1	-1	1	-1	1	-1	1	-1	1	1	-1	1	-1	1	-1
χ_{13}	1	-1	-1	1	1	-1	1	-1	1	-1	1	-1	1	-1	1	-1
χ_{14}	5	-5	-5	5	3	-3	3	-3	3	-3	1	-1	1	-1	1	-1
χ_{15}	5	-5	-5	5	-3	3	-3	3	-3	3	1	-1	1	-1	1	-1
χ_{16}	5	-5	-5	5	1	-1	1	-1	1	-1	1	-1	1	-1	1	-1
χ_{17}	5	-5	-5	5	-1	1	-1	1	-1	1	1	-1	1	-1	1	-1
χ_{18}	9	-9	-9	9	3	-3	3	-3	3	-3	1	-1	1	-1	1	-1
χ_{19}	9	-9	-9	9	-3	3	-3	3	-3	3	1	-1	1	-1	1	-1
χ_{20}	10	-10	-10	10	2	-2	2	-2	2	-2	-2	2	-2	2	-2	2
χ_{21}	10	-10	-10	10	-2	2	-2	2	-2	2	-2	2	-2	2	-2	2
χ_{22}	16	-16	-16	16	0	0	0	0	0	0	0	0	0	0	0	0
χ_{23}	15	5	-3	-1	-5	7	3	-1	-1	-1	-1	3	3	-1	-1	-1
χ_{24}	15	5	-3	-1	5	-7	-3	1	1	1	-1	3	3	-1	-1	-1
χ_{25}	15	5	-3	-1	7	-5	-1	3	-1	-1	3	-1	-1	3	-1	-1
χ_{26}	15	5	-3	-1	-7	5	1	-3	1	1	3	-1	-1	3	-1	-1
χ_{27}	30	10	-6	-2	-2	-2	-2	-2	2	2	2	2	2	2	-2	-2
χ_{28}	30	10	-6	-2	2	2	2	2	-2	-2	2	2	2	2	-2	-2
χ_{29}	45	15	-9	-3	3	-9	-5	-1	3	3	-3	1	1	-3	1	1
χ_{30}	45	15	-9	-3	9	-3	1	5	-3	-3	1	-3	-3	1	1	1
χ_{31}	45	15	-9	-3	-9	3	-1	-5	3	3	1	-3	-3	1	1	1
χ_{32}	45	15	-9	-3	-3	9	5	1	-3	-3	-3	1	1	-3	1	1
χ_{33}	15	-5	3	-1	-5	-7	3	1	-1	1	-1	-3	3	1	1	1
χ_{34}	15	-5	3	-1	5	7	-3	-1	1	-1	1	3	3	1	-1	1
χ_{35}	15	-5	3	-1	7	5	-1	-3	-1	1	3	1	-1	-3	-1	1
χ_{36}	15	-5	3	-1	-7	-5	1	3	1	-1	3	1	-1	-3	-1	1
χ_{37}	30	-10	6	-2	-2	2	-2	2	2	-2	2	-2	2	-2	-2	0
χ_{38}	30	-10	6	-2	2	-2	2	-2	-2	2	2	-2	2	-2	-2	2
χ_{39}	45	-15	9	-3	3	9	-5	1	3	-3	-3	-1	1	3	1	-1
χ_{40}	45	-15	9	-3	9	3	1	-5	-3	3	1	3	-3	-1	1	-1
χ_{41}	45	-15	9	-3	-9	-3	-1	5	3	-3	1	3	-3	-1	1	-1
χ_{42}	45	-15	9	-3	-3	-9	5	-1	-3	3	-3	-1	1	3	1	-1

Table 9.8: The character table of $2^5 S_6$

	2C				3A				3B		4A			
	2i	2j	4h	4i	3a	6a	6b	6c	3b	6d	4j	8a	4k	8b
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	-1	-1	-1	-1	1	1	1	1	1	1	-1	-1	-1	-1
<i>χ</i> 3	1	1	1	1	2	2	2	2	-1	-1	-1	-1	-1	-1
χ_4	-1	-1	-1	-1	2	2	2	2	-1	-1	1	1	1	1
χ_5	3	3	3	3	-1	-1	-1	-1	2	2	1	1	1	1
χ_6	-3	-3	-3	-3	-1	-1	-1	-1	2	2	-1	-1	-1	-1
χ7	-3	-3	-3	-3	0	0	0	0	0	0	1	1	1	1
X8	3	3	3	3	0	0	0	0	0	0	-1	-1	-1	-1
X9	2	2	2	2	1	1	1	1	1	1	0	0	0	0
χ_{10}	-2	-2	-2	-2	1	1	1	1	1	1	0	0	0	0
X11	0	0	0	0	-2	-2	-2	-2	-2	-2	0	0	0	0
χ_{12}	1	-1	1	-1	1	-1	1	-1	1	-1	-1	-1	1	1
χ_{13}	-1	1	-1	1	1	-1	1	-1	1	-1	1	1	-1	-1
χ_{14}	1	-1	1	-1	2	-2	2	-2	-1	1	1	1	-1	-1
χ_{15}	-1	1	-1	1	2	2	2	-2	-1	1	-1	-1	1	1
χ_{16}	3	-3	3	-3	-1	1	-1	1	2	-2	-1	-1	1	1
χ_{17}	-3	3	-3	3	-1	1	-1	1	2	-2	1	1	-1	-1
χ_{18}	-3	3	-3	3	0	0	0	0	0	0	-1	-1	1	1
χ_{19}	3	-3	3	-3	0	0	0	0	0	0	1	-1	-1	1
χ_{20}	2	-2	2	-2	1	-1	1	-1	1	-1	0	0	0	0
χ_{21}	-2	2	-2	2	1	-1	1	-1	1	-1	0	0	0	0
χ_{22}	0	0	0	0	-2	2	-2	2	-2	2	0	0	0	0
χ_{23}	3	3	-1	-1	3	-3	-1	1	0	0	-1	1	-1	1
χ_{24}	-3	-3	1	1	3	-3	-1	1	0	0	1	-1	1	-1
χ_{25}	3	3	-1	-1	3	-3	-1	1	0	0	1	-1	1	-1
χ_{26}	-3	-3	1	1	3	-3	-1	1	0	0	-1	1	-1	1
χ_{27}	-6	-6	2	2	-3	3	1	-1	0	0	0	0	0	0
χ_{28}	6	6	-2	-2	-3	3	1	-1	0	0	0	0	0	0
X29	3	3	-1	-1		0	0	0	0	0	-1	1	-1	1
X30	-3	-3	1	1		0	0	0	0	0	-1	1	-1	1
χ_{31}	3	ა ე	-1	-1		0	0	0	0	0		-1	1	-1
X32	-3	-3 9	1	1		0	1	1	0	0	1	-1	1	-1
X33	-3	ა ე	1	-1	3	ა ი	-1	-1	0	0	1	-1	-1	1
X34	່ ວ າ	-ə -	-1	1))	ა ი	-1	-1	0	0	-1	1	1	-1
X35	-0 9	ა ე	1	-1))	ა ი	-1	-1	0	0	-1	1	1	-1
X36	3 6	-3 6	-1 0	1	3 9	ა ი	-1 1	-1 1		0		-1	-1	1
X37		-0 6	-2 2	2 0	-0 9	-ə 9	1	1		0	0	0	0	0
X38	0- 9	0 9	2 1	-2 1	-3 0	-3 0	1	1		0	1	1	1	1
X39	-3 9	ა ი	1	-1 1		0	0	0		0	1	-1 1	-1 1	1
X40) _2	-0 2	-1 1	_1		0	0	0		0	_1	-1 1	-1 1	_1
X41	2	່ງ _າ	_1	-1 1		0	0	0		0	-1	1	1	-1
X42	5	-0	-1	T		0	0	0	0	0	-1	T	T	-1

The character table of $2^5:S_6$ (continued)

	4B				5A		6A			6B		
	41	8c	4m	8d	5a	10a	6e	6f	12a	12b	6g	6h
χ_1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	1	1	1	1	1	1	-1	-1	-1	-1	-1	-1
<i>χ</i> ₃	-1	-1	-1	-1	0	0	0	0	0	0	1	1
χ_4	-1	-1	-1	-1	0	0	0	0	0	0	-1	-1
χ_5	-1	-1	-1	-1	0	0	-1	-1	-1	-1	0	0
χ_6	-1	-1	-1	-1	0	0	1	1	1	1	0	0
χ7	1	1	1	1	-1	-1	0	0	0	0	0	0
X8	1	1	1	1	-1	-1	0	0	0	0	0	0
X9	0	0	0	0	0	0	1	1	1	1	-1	-1
X10	0	0	0	0	0	0	-1	-1	-1	-1	1	1
X11	0	0	0	0	1	1	0	0	0	0	0	0
X12	1	1	-1	-1	1	-1	1	-1	1	-1	-1	1
X13	1	1	-1	-1	1	-1	-1	1	-1	1	1	-1
X14	-1	-1	1	1	0	0	0	0	0	0	-1	1
χ_{15}	-1	-1	1	1	0	0	0	0	0	0	1	-1
χ_{16}	-1	-1	1	1	0	0	-1	1	-1	1	0	0
<i>χ</i> 17	-1	-1	1	1	0	0	1	-1	1	-1	0	0
χ_{18}	1	1	-1	-1	-1	1	0	0	0	0	0	0
X19	1	1	-1	-1	-1	1	0	0	0	0	0	0
X20	0	0	0	0	0	0	1	-1	1	-1	1	-1
χ_{21}	0	0	0	0	0	0	-1	1	-1	1	-1	1
X22	0	0	0	0	1	-1	0	0	0	0	0	0
X23	-1	1	-1	1	0	0	-1	1	1	-1	0	0
χ_{24}	-1	1	-1	1	0	0	1	-1	-1	1	0	0
χ_{25}	1	-1	1	-1	0	0	-1	1	1	-1	0	0
χ_{26}	1	-1	1	-1	0	0	1	-1	-1	1	0	0
χ27	0	0	0	0	0	0	-1	1	1	-1	0	0
χ_{28}	0	0	1	1	0	0		-1	-1	1		0
X29		-1	1	-1	0	0	0	0	0	0		0
X30	-1	1	-1	1	0	0	0	0	0	0		0
X31	-1	1	-1 1	1	0	0	0	0	0	0		0
X32	1	-1	1	-1	0	0	1	1	1	1		0
X33	1	-1	-1	1	0	0	1	1	-1	-1 1		0
X34	1	-1	-1	1	0	0	-1	-1	1	1		0
X35	1	1	1	-1	0	0	1	1	-1	-1		0
X36	0	1 0	1 0	-1	0	0	1	-1 1	_1	_1		0
×20	0	0	0	0	0	0	_1	_1	-1	1		0
X38	_1	1	1	_1	0	0	0	0	0	0		0
X39	1	_1	_1	1	0	0	0	0	0	0		0
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	1	-1	-1	1	0	0	0	0	0	0		0
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	-1	1	1	-1	0	0	0	0	0	0	0	0

The character table of $2^5:S_6(\text{continued})$
9.4. Fusion of $2^{4} \cdot S_6$ into $2^5 \cdot S_6$

We give the fusion of $\overline{G_1}$ into \overline{G} in Table 9.9

$[x]_{2^4 \cdot S_6}$	\longrightarrow	$[g_1]_{2^5 \cdot S_6}$	$[x]_{2^{4} \cdot S_{6}}$	\longrightarrow	$[g_1]_{2^5 \cdot S_6}$
1A		1A	4E		4H
2A		2C	4F		4K
2B		2F	4G		4L
2C		2G	5A		5A
2D		2I	6A		6B
3A		3A	6B		6F
3B		3B	6C		6G
4A		4A	8A		8C
4B		4B	8B		8B
4C		4E	12A		12A
4D		4F			

Table 9.9: The fusion of $2^4 \cdot S_6$ into $2^5 \cdot S_6$

10

A group of the form $5^{3} L(3,5)$ as a maximal subgroup of the Lyons Group Ly

Prologue

The group $\overline{G} = 5^3 \cdot L(3,5)$ is a subgroup of order 46500000 and of index 1113229656 in Ly. The group \overline{G} in turn has L(3,5) and $5^2:2.A_5$ as inertia factors. The group $5^2:2.A_5$ is of order 3000 and is of index 124 in L(3,5). The aim of this chapter is to compute the Fischer-Clifford matrices of \overline{G} , which together with associated partial character tables of the inertia factor groups, are used to compute a full character table of \overline{G} . We will obtain that the partial projective character table corresponding to $5^2:2A_5$ is required, hence we have to compute the Schur multiplier and projective character table of $5^2:2A_5$.

10.1. Introduction

The Lyons group was discovered in 1970 by Richard Lyons [78], using the concept of classifying simple groups with an involution centralizer $2.A_n$. The smallest value of n for which $2.A_n$ has noncentral involutions is n = 8, for which the McLaughlin group M^cL , has an involution centralizer $2.A_8$. The only other case that arises is n = 11 which is in the Lyons group Ly that is the Lyons group has an involution centralizer $2A_{11}$. Moreover, a 3-cycle in $2.A_{11}$ centralizes $2.A_8$ and the full centralizer of this 3-cycle in Ly is the triple cover $3.M^cL$ of the McLaughlin group. The Lyons group Ly, is a sporadic simple group of order $2^8.3^7.5^6.7.11.31.37.67 = 51765179004000000$.

The existence of this group and its uniqueness up to isomorphism was shown by C C Sims [118, 119], using *coset enumeration* and it is often referred to as the "Lyons-Sims" group. The group Lyhas elements of order 37 and 67 which cannot be found in the monster and is one of the six sporadic simple groups called the "pariahs" which are not subgroups of the monster. The other five "pariahs" being $J_1, J_3, J_4, O'N$ and Ru, the last to be determined in [122] being J_1 . The group $5^{3} L(3,5)$ is also maximal in the baby monster B. The group Ly has nine conjugacy classes of maximal subgroups. One of the maximal subgroups of the form, $\overline{G} = N.G$ is a group of order $46500000 = 2^{6}.3.5^{6}.31$, where $N \cong 5^{3}$ and $G \cong L(3,5)$. The aim of this chapter is to compute the Fischer-Clifford matrices which together with partial character tables will be used to compute a character table for \overline{G} . The notation used is consistent with that of the ATLAS [23] and method used is taken from [1, 2]. One can read more on Fischer-Clifford theory and projective characters from [99, 120] and [55, 66, 107, 108, 109] respectively. For the theory of characters one can also read Character Theory of Finite Groups [60].

10.2. Construction of $\overline{G} \cong 5^{3} \cdot L(3,5)$

From the ATLAS of group representation [124] we get two 111 × 111 matrices a, b, with o(a) = 2, o(b) = 5, o(ab) = 14 and $Ly = \langle a, b \rangle$. Again from [124] we get Programme I (see Appendix A), where if we use a = input[1] and b = input[2] and we get $\bar{x} = output[1]$ and $\bar{y} = output[2]$, where $o(\bar{x}) = 2$, $o(\bar{y}) = 3$, $o(\bar{x}\bar{y}) = 31$ and $\overline{G} = \langle \bar{x}, \bar{y} \rangle$. From [124] we see that $o(\bar{x}\bar{y}\bar{x}\bar{y}^2) = 25$ and if we let $gen[1] = (\bar{x}\bar{y}\bar{x}\bar{y}^2)^5$, then o(gen[1]) = 5, we also get that $gen[2] = \bar{y}gen[1]\bar{y}^{-1}$, $gen[3] = \bar{x}gen[2]\bar{x}^{-1}$ and $N = 5^3 = \langle gen[1], gen[2], gen[3] \rangle$. Let $\lambda_i = gen[i]$, i = 1, 2, 3. We use GAP to compute the conjugacy classes of $5^3 \cdot L(3, 5)$ and also the fusion of $5^3 \cdot L(3, 5)$ into Ly and these are given in Table 10.1.

$[g]_{L(3,5)}$	$[x]_{5^3 \cdot L(3,5)}$	$C_{5^3 \cdot L(3,5)}(x)$	\longrightarrow	Ly
1A	1A	46500000		1A
	5A	375000		5A
2A	2A	2400		2A
	10A	600		10A
3A	3A	120		3A
	15A	30		15B
4A	4A	480		4A
4B	4B	480		4A
4C	4C	80		4A
	20A	20		20A
	5B	2500		5A
5A	$5\mathrm{C}$	1250		5B
	5D	1250		5B
5B	25A	25		25A
6A	6A	120		6B
	30A	30		30B
8A	8A	24		8B
8B	8B	24		8B
	10B	100		10A
10A	10C	50		10B
	10D	50		10B
12A	12A	24		12B
12B	12B	24		12B
continued	on next page			

Table 10.1: Conjugacy Classes Of $5^{3} L(3,5)$

Table	10.1 (contin	led from previo	us page)
$[g]_{L(3,5)}$	$[x]_{5^3 \cdot L(3,5)}$	$C_{5^3.L(3,.5)}(x)$	\longrightarrow Ly
20A	20B	20	20A
20B	20C	20	20A
24A	24A	24	24C
24B	24B	24	24B
24C	24C	24	24B
24D	24D	24	24C
31A	31A	1	31B
31B	31B	31	31A
31C	31C	31	31E
31D	31D	31	31D
31E	31E	31	31C
31F	31F	31	31B
31G	31G	31	31A
31H	31H	31	31E
31I	31I	31	31D
31J	31J	31	31C

Table 10.1 (continued from previous page)

10.3. Construction of $G \cong L(3,5)$

Our aim in this section is to let \overline{G} act on the elementary abelian group N. We use the method discussed in chapters 8 and 9. In this method N is considered as a vector space V, of dimension 3 over GF(5). For us to be able to act on a three dimensional vector space V it becomes necessary to rewrite \overline{G} from 111×111 to a 3×3 representation. To do this we have to act \overline{G} on N by letting the two generators of \overline{G} , \overline{x} and \overline{y} to act, on the generators of N, λ_i , i = 1, 2, 3 by conjugation, using GAP [41]. Writing these as maps we get :

$$\begin{split} \bar{x} &: \lambda_1 \to \lambda_1^4 \\ \lambda_2 \to \lambda_3 \\ \lambda_3 \to \lambda_2; \\ \bar{y} &: \lambda_1 \to \lambda_2 \\ \lambda_2 \to \lambda_1 \lambda_2 \lambda_3^4 \\ \lambda_3 \to \lambda_2^2 \lambda_3^4. \end{split}$$

Writing these in 3×3 matrix form, over GF(5) we get

$$x = \left(\begin{array}{rrr} 4 & 0 & 0 \\ 0 & 0 & 1 \\ 0 & 1 & 0 \end{array}\right), y = \left(\begin{array}{rrr} 0 & 1 & 0 \\ 1 & 1 & 4 \\ 0 & 2 & 4 \end{array}\right)$$

and $G = \langle x, y \rangle$. Then $G \cong L(3,5)$ which means that the action of \overline{G} on N is isomorphic to

L(3, 5).

10.4. Inertia Factors of \bar{G}

We use GAP [41] to compute the permutation character of Ly acting on $5^{3} L(3,5)$. That is

$$\begin{split} \chi(Ly|5^{3}\cdot L(3,5)) &= 1a + 45694a + 381766a + 1534500aa + 3028266a + 4226695aa + 11834746a + 18395586abc + 19212250a + 21312500ab + 22609664abc + 27252720aabbcd + 28787220aa + 29586865a + 33813560aa + 38734375a + 43110144abcde + 45648306b + 45694000ab + 56022921a + 64906250a + 71008476a \end{split}$$

We then use Programme C (see Appendix A) to compute the orbit lengths of the action on the orbits on N and on Irr(N). We let G act on a full row vector space V of dimension 3 over GF(5). We get two orbits of conjugacy classes of lengths 1 and 124. By Brauer's Theorem [14] when G acts on Irr(N), we get two inertia groups of index 1 and 124. Using the ATLAS [23], we see that $5^2:GL(2,5)$ is of index 31 in L(3,5) and |GL(2,5)| = 480. Since we are looking for the maximal subgroup of index 124 in L(3,5), we have to get a group of index 4 in GL(2,5) and the group should be of order 120. So with the help of the ATLAS [23] this can only be the group $2 \cdot A_5$. Thus the group of index 4 in $5^2:GL(2,5)$ is $5^2:2.A_5$. The full inertia groups are $\bar{H}_i = 5^3.H_i$, i = 1, 2, where $H_1 = L(3,5)$ and $H_2 = 5^2:2.A_5$. We used GAP [41] to calculate the character table of H_2 .

$[x]_{5^2:2.A_5}$	$\longrightarrow [g_1]_{L(3,5)}$
1A	1A
2A	2A
3A	3A
4 <i>A</i>	4C
5A	5A
5B	5B
5C	5B
5D	5A
5E	5B
5F	5B
5G	5A
6A	6A
10A	10A
10B	10A

Table 10.2: The fusion of $5^2:2.A_5$ into L(3,5)

10.5. Projective Character Table of $5^2:2.A_5$

From the fusions and orbit lengths and centralizer orders, we compute the Fischer-Clifford matrix M(1A) of \bar{G} .

$$M(1A) = \left[\begin{array}{rrr} 1 & 1 \\ 124 & -1 \end{array} \right]$$

Having computed M(1A) we want to determine the type of partial character tables we are going to use for our computations. We will show that the partial projective character table of H_2 is required. We follow the methods used in [1, 2] and we use the character table of $Ly = \langle a, b \rangle$. Let $Irr(Ly) = \{\Psi_i : 1 \le i \le 53\}$, where the notation is the same at the one used in the ATLAS [23]. From the list we take the values of Ψ_2 , Ψ_3 , Ψ_4 on 1A and 5A.

$C_{\overline{G}}(x)$	46500000	375000
$[x]_{Ly}$	1A	5A
Ψ_2	2480	-20
Ψ_3	2480	-20
Ψ_4	45694	69

Let γ_1, γ_2 be the rows of the Fischer-Cliford matrix M(1A). Then

$$<(\Psi_2)_N, 1_N>=\frac{1}{125}(2480-20.124)=0.$$

Since $\langle (\Psi_2)_N, 1_N \rangle = 0$, we get that 2480 = 0 + 20.124, so that $(\Psi_2)_N = 0.\gamma_1 + 20.\gamma_2$. Let $[x_1, \dots, x_t]$ be the transpose of the partial entries for the ordinary characters of $H_2 = 5^2:2.A_5$ on $1A \in L(3,5)$. Then $C_2(1A)M(1A)$ is a $t \times 2$ matrix with entries on the first column $124x_1 = 2480$. Hence $x_1 = 20$. From the ordinary character table of $H_2 = 5^2:2.A_5$, there is no character of degree 20. Similarly

$$<(\Psi_4)_N, 1_N>=\frac{1}{125}(45694-69.124)=434,$$

which gives us $x_1 = 365$ and this is a very large character degree which is not possible for $H_2 = 5^2$:2. A_5 , and this holds for the remaining characters. Hence we have to use the projective character table of H_2 . There are three primes dividing the order of H_2 namely 2, 3 and 5. Using MAGMA Programme K, or GAP Programme K' (see Appendix A), we determine the Schur multiplier of H_2 . The p - Sylow subgroups corresponding to p = 2 and 3 are cyclic, using methods from [1, 2] the Schur multipliers of both p-Sylow subgroups are trivial. Hence the Schur multiplier of H_2 is the cyclic group of order 5. The projective characters of H_2 with factor set α^{-1} where $\alpha^5 \sim 1$ is given in Table 10.3. Note that from the table we can see that 5a, 5b, 5c, 5e, 5f are all not α regular classes and we have a total of 9 α regular classes.

Let $A = -E(5) - E(5)^4$, and $A^* = 1 - A = -E(5)^2 - E(5)^3$. Then $A + A^* = 1$, $A^*A = A(A^*) = -1$, $A^2 + (A^*)^2 = 3$, $A^3 + (A^*)^3 = 4$. In fact we get a Fibonacci sequence, with $f_{i+1} = f_i + f_{i-1}$, $i \ge 2$, where $f_i = A^i + (A^*)^i$. This helps us to compute the Fischer-Clifford matrices and character table of $\overline{G} = 5^{3} \cdot L(3, 5)$.

	1a	5a	2a	4a	3a	6a	5b	5c	5d	10a	5e	5f	5g	10b
χ_1	5	0	1	1	1	1	0	0	1	1	0	0	1	1
χ_2	15	0	3	-1	0	0	0	0	Α	Α	0	0	A^*	A^*
χ_3	15	0	3	-1	0	0	0	0	A^*	A^*	0	0	Α	Α
χ_4	20	0	4	0	1	1	0	0	-1	-1	0	0	-1	-1
χ_5	20	0	4	0	1	-1	0	0	-1	1	0	0	-1	1
χ_6	25	0	5	1	-1	-1	0	0	0	0	0	0	0	0
χ_7	10	0	2	0	-1	1	0	0	-A	Α	0	0	$-A^*$	A^*
χ_8	10	0	2	0	-1	1	0	0	$-A^*$	A^*	0	0	-A	Α
χ_9	30	0	6	0	0	0	0	0	1	-1	0	0	1	-1

Table 10.3: The projective character table of $5^2:2.A_5$ with factor set α^{-1}

10.6. Fischer-Clifford Matrices of \overline{G}

Having computed the projective character table of H_2 , Table 10.3, we get the α -regular conjugacy classes. These together with the fusions of $5^2:2.A_5$ into L(3,5) in Table 10.2 help us to compute the sizes of the Fischer-Clifford matrices of \overline{G} . We use these the projective characters, the fusions, the centralizer orders of \overline{G} and properties of Fisher-Clifford matrices discussed in section 5.12.1, to compute Table 10.4 of Fischer - Clifford matrices.

To compute the character table of $5^{3} \cdot L(3,5)$, as an example consider the following. Let $C_1(5A)$ and $C_2(5A)$ be the partial character tables of the inertia factors for the classes that fuse to $5A \in L(3,5)$. The the portions of the character table of $\overline{G} = 5^{3} \cdot L(3,5)$ corresponding to the coset 5A are :

$M(1A) = \begin{bmatrix} 1 & 1\\ 124 & -1 \end{bmatrix}$	$M(2A) = \begin{bmatrix} 1 & 1 \\ 4 & -1 \end{bmatrix}$
$M(3A) = \begin{bmatrix} 1 & 1\\ -4 & 1 \end{bmatrix}$	$M(4C) = \begin{bmatrix} 1 & 1\\ 4 & -1 \end{bmatrix}$
$M(5A) = \begin{bmatrix} 1 & 1 & 1 \\ 10 & -5A^* & -5A \\ 10 & -5A & -5A^* \end{bmatrix}$	$M(10A) = \begin{bmatrix} 1 & 1 & 1 \\ 2 & -A & -A^* \\ 2 & -A^* & -A \end{bmatrix}$
$M(6A) = \begin{bmatrix} 1 & 1 \\ -4 & 1 \end{bmatrix}$	All Others = $\begin{bmatrix} 1 \end{bmatrix}$

Table 10.4: The Fischer-Clifford matrices of $5^{3} \cdot L(3,5)$



The fusion of \overline{G} to Ly together with the restriction of characters of Ly to \overline{G} forces the signs of the Fischer-Clifford matrices and the orders of the elements of the conjugacy classes of \overline{G} .

10.7. PowerMaps of \overline{G}

We then give the power maps of elements of $5^{3} L(3,5)$ in Table 10.5.

						*			())	-	
$[g]_{L(3,5)}$	$[x]_{5^3 \cdot L(3,5)}$	2	3	5	31	$[g]_{L(3,5)}$	$[x]_{5^3 \cdot L(3,5)}$	2	3	5	31
1A	1A	1A	1A	1A	1A	2A	2A	1A	2A	2A	2A
	5A	5A	5A	1A	5A		10A	5A	10A	2A	10A
3A	3A	3A	$1\mathrm{A}$	3A	3A	4A	4A	2A	4A	4A	4A
	15A	15A	5A	1A	15A	4B	4B	2A	4A	4A	4A
						4C	$4\mathrm{C}$	2A	$4\mathrm{C}$	$4\mathrm{C}$	4C
5A	5B	5B	5B	1A	5B	5B	25A	25A	25A	5A	25A
	$5\mathrm{C}$	5C	5C	1A	$5\mathrm{C}$						
	$5\mathrm{D}$	5D	5D	1A	5D						
6A	6A	3A	2A	6A	6A	10A	10B	5A	10B	2A	10B
	30A	15A	5A	6A	30A		10C	5A	10C	2A	10C
							10D	5A	10D	2A	10D
8A	8A	4A	8A	8A	8A	8B	8B	4B	8B	8B	8B
12A	12A	6A	4A	12A	12A	12B	12B	6A	4B	12B	12B
20A	20A	10A	20A	4A	20A	20B	20B	10A	20B	4B	20B
24A	24A	12A	8A	24A	24A	24B	24B	12B	8B	24B	24B
24C	$24\mathrm{C}$	12A	8A	$24\mathrm{C}$	$24\mathrm{C}$	24D	24D	12B	8B	24D	24D
31A	31A	31A	31A	31A	1A	31B	31B	31B	31B	31B	1A
31C	31C	31C	$31\mathrm{C}$	31C	1A	31D	31D	31D	31D	31D	1A
31E	31E	31E	31E	$31\mathrm{EA}$	1A	31F	31B	31F	31F	31F	1A
31G	31G	31G	31G	31G	1A	31H	31H	31H	31H	31H	1A
31I	31I	31I	31I	31I	1A	31J	31J	31J	31J	31J	1A

Table 10.5: The Power Maps of elements of $5^{3} L(3,5)$

The character table of $5^3 L(3,5)$ is given in Table 10.6.

10.8. Character Table of \overline{G}

	1A		2A		3A		4A	4B	4C			5A		5B
	1a	5a	2a	10a	3a	15a	4a	4b	4c	20a	5b	5c	5d	25a
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	30	30	6	6	0	0	6	6	2	2	5	5	5	0
χ_3	31	31	7	7	1	1	-5	-5	-1	-1	6	6	6	1
χ_4	31	31	-5	-5	1	1	А	/A	1	1	6	6	6	1
χ_5	31	31	-5	-5	1	1	/A	Α	1	1	6	6	6	1
χ_6	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_7	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_8	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_9	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{10}	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{11}	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{12}	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{13}	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{14}	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{15}	96	96	0	0	0	0	0	0	0	0	-4	-4	-4	1
χ_{16}	124	124	4	4	1	1	4	4	0	0	-1	-1	-1	-1
χ_{17}	124	124	4	4	1	1	4	4	0	0	-1	-1	-1	-1
χ_{18}	124	124	4	4	1	1	-4	-4	0	0	-1	-1	-1	-1
χ_{19}	124	124	4	4	1	1	-4	-4	0	0	-1	-1	-1	-1
χ_{20}	124	124	-4	-4	-2	-2	В	-B	0	0	-1	-1	-1	-1
χ_{21}	124	124	-4	-4	-2	-2	-B	В	0	0	-1	-1	-1	-1
χ_{22}	124	124	-4	-4	1	1	-B	В	0	0	-1	-1	-1	-1
χ_{23}	124	124	-4	-4	1	1	В	-B	0	0	-1	-1	-1	-1
χ_{24}	124	124	-4	-4	1	1	-B	В	0	0	-1	-1	-1	-1
χ_{25}	124	124	-4	-4	1	1	В	-B	0	0	-1	-1	-1	-1
χ_{26}	125	125	5	5	-1	-1	5	5	1	1	0	0	0	0
χ_{27}	155	155	11	11	-1	-1	-1	-1	-1	-1	5	5	5	0
χ_{28}	155	155	-1	-1	-1	-1	С	/C	1	1	5	5	5	0
χ_{29}	155	155	-1	-1	-1	-1	/C	С	1	1	5	5	5	0
χ_{30}	186	186	-6	-6	0	0	6	6	-2	-2	11	11	11	1
χ_{31}	620	-5	4	-1	-4	1	0	0	-4	1	20	-5	-5	0
χ_{32}	1860	-15	12	-3	0	0	0	0	4	-1	10	10	-15	0
χ_{33}	1860	-15	12	-3	0	0	0	0	4	-1	10	-15	10	0
χ_{34}	2480	-20	16	-4	-4	1	0	0	0	0	-20	5	5	0
χ_{35}	3100	-25	20	-5	4	-1	0	0	-4	1	0	0	0	0
χ_{36}	1240	-10	-8	2	4	-1	0	0	0	0	-10	-10	15	0
χ_{37}	1240	-10	-8	2	4	-1	0	0	0	0	-10	15	-10	0
χ_{38}	2480	-20	-16	4	-4	1	0	0	0	0	-20	5	5	0
χ_{39}	3720	-30	-24	6	0	0	0	0	0	0	20	-5	-5	0

Table 10.6: The character table of $5^{3} L(3,5)$

CHAPTER 10. A GROUP OF THE FORM $5^3 \cdot L(3,5)$ AS A MAXIMAL SUBGROUP OF THE LYONS GROUP LY

	6A		8A	8B		10A		12A	12B	20B	20C	24A	24B	24C	24D
	6a	30a	8a	8b	10b	10c	10d	12a	12b	20b	20c	24a	24b	24c	24d
χ_1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1
χ_2	0	0	0	0	1	1	1	0	0	1	1	0	0	0	0
χ_3	1	1	-1	-1	2	2	2	1	1	0	0	-1	-1	-1	-1
χ_4	1	1	D	-D	0	0	0	-1	-1	F	$/\mathrm{F}$	D	-D	D	-D
χ_5	1	1	-D	D	0	0	0	-1	-1	/F	F	-D	D	-D	D
χ_6	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_7	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_8	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_9	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{10}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{11}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{12}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{13}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{14}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{15}	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{16}	1	1	2	2	-1	-1	-1	1	1	-1	-1	-1	-1	-1	-1
χ_{17}	1	1	-2	-2	-1	-1	-1	1	1	-1	-1	1	1	1	1
χ_{18}	1	1	E	-E	-1	-1	-1	-1	-1	1	1	-D	D	-D	D
χ_{19}	1	1	-E	E	-1	-1	-1	-1	-1	1	1	D	-D	D	-D
χ_{20}	2	2	0	0	1	1	1	-E	E	-D	D	0	0	0	0
χ_{21}	2	2	0	0	1	1	1	E	-E	D	-D	0	0	0	0
χ_{22}	-1	-1	0	0	1	1	1	-D	D	D	-D	G	$/\mathrm{G}$	-G	-/G
χ_{23}	-1	-1	0	0	1	1	1	D	-D	-D	D	$/\mathrm{G}$	G	-/G	-G
χ_{24}	-1	-1	0	0	1	1	1	-D	D	D	-D	-G	-/G	G	$/\mathrm{G}$
χ_{25}	-1	-1	0	0	1	1	1	D	-D	-D	D	-/G	-G	/G	G
χ_{26}	-1	-1	-1	-1	0	0	0	-1	-1	0	0	-1	-1	-1	-1
χ_{27}	-1	-1	1	1	1	1	1	-1	-1	-1	-1	1	1	1	1
χ_{28}	-1	-1	-D	D	-1	-1	-1	1	1	D	-D	-D	D	-D	D
χ_{29}	-1	-1	D	-D	-1	-1	-1	1	1	-D	D	D	-D	D	-D
χ_{30}	0	0	0	0	-1	-1	-1	0	0	1	1	0	0	0	0
χ_{31}	4	-1	0	0	4	-1	-1	0	0	0	0	0	0	0	0
χ_{32}	0	0	0	0	2	-3	2	0	0	0	0	0	0	0	0
χ_{33}	0	0	0	0	2	2	-3	0	0	0	0	0	0	0	0
χ_{34}	4	-1		0	-4	1	1		0			0	0	0	0
χ_{35}	-4	1	0	0	0	0	0	0	0	0	0	0	0	0	0
χ_{36}		-1		0	2	-3	2		0		0	0	0	0	0
χ_{37}	4	-1		0	2	2	-3		0		0	0	0	0	0
χ_{38}	-4	1	0	0	4	-1	-1	0	0	0	0	0	0	0	0
χ_{39}	0	0	0	0	-4	1	1	0	0	0	0	0	0	0	0

The character table of $5^3L(3,5)$ (continued)

	31A	31B	31C	31D	31E	31F	31G	31H	31I	31J
	31a	31b	31c	31d	31e	31f	31g	31h	31i	31j
χ_1	1	1	1	1	1	1	1	1	1	1
χ_2	-1	-1	-1	-1	-1	-1	-1	-1	-1	-1
χ_3	0	0	0	0	0	0	0	0	0	0
χ_4	0	0	0	0	0	0	0	0	0	0
χ_5	0	0	0	0	0	0	0	0	0	0
χ_6	Н	/H	L	/L	K	$/\mathrm{K}$	J	$/\mathrm{J}$	Ι	/I
χ_7	/H	Н	/L	\mathbf{L}	K	/L	/K	/K	/I	Ι
χ_8	Ι	/I	Н	$/\mathrm{H}$	L	/L	К	/K	J	$/\mathrm{J}$
χ_9	/I	Ι	/H	Η	/L	\mathbf{L}	/K	Κ	/J	J
χ_{10}	J	/J	Ι	/I	Н	$/\mathrm{H}$	L	/L	K	$/\mathrm{K}$
χ_{11}	$/\mathrm{J}$	J	/I	Ι	$/\mathrm{H}$	Η	/L	\mathbf{L}	/K	Κ
χ_{12}	К	/K	J	$/\mathrm{J}$	Ι	/I	Н	$/\mathrm{H}$	/L	L
χ_{13}	/K	K	$/\mathrm{J}$	J	/I	Ι	$/\mathrm{H}$	Η	/L	L
χ_{14}	L	/L	К	/K	J	$/\mathrm{J}$	Ι	/I	Н	$/\mathrm{H}$
χ_{15}	/L	L	/K	Κ	$/\mathrm{J}$	J	/I	Ι	/H	Н
χ_{16}	0	0	0	0	0	0	0	0	0	0
χ_{17}	0	0	0	0	0	0	0	0	0	0
χ_{18}	0	0	0	0	0	0	0	0	0	0
χ_{19}	0	0	0	0	0	0	0	0	0	0
χ_{20}	0	0	0	0	0	0	0	0	0	0
χ_{21}	0	0	0	0	0	0	0	0	0	0
χ_{22}	0	0	0	0	0	0	0	0	0	0
χ_{23}	0	0	0	0	0	0	0	0	0	0
χ_{24}	0	0	0	0	0	0	0	0	0	0
χ_{25}	0	0	0	0	0	0	0	0	0	0
χ_{26}	1	1	1	1	1	1	1	1	1	1
χ_{27}	0	0	0	0	0	0	0	0	0	0
χ_{28}	0	0	0	0	0	0	0	0	0	0
χ_{29}	0	0	0	0	0	0	0	0	0	0
χ_{30}	0	0	0	0	0	0	0	0	0	0
χ_{31}	0	0	0	0	0	0	0	0	0	0
χ_{32}	0	0	0	0	0	0	0	0	0	0
χ_{33}	0	0	0	0	0	0	0	0	0	0
χ_{34}	0	0	0	0	0	0	0	0	0	0
χ_{35}	0	0	0	0	0	0	0	0	0	0
χ_{36}	0	0	0	0	0	0	0	0	0	0
χ_{37}	0	0	0	0	0	0	0	0	0	0
χ_{38}	0	0	0	0	0	0	0	0	0	0
χ_{39}	0	0	0	0	0	0	0	0	0	0

The character table of $5^{3} \cdot L(3,5)$ (continued)

χ_{14}	L	/L	ĸ	/K	J	/J	1	/1	н	/H
χ_{15}	/L	L	/K	Κ	/J	J	/I	Ι	$/\mathrm{H}$	Н
χ_{16}	0	0	0	0	0	0	0	0	0	0
χ_{17}	0	0	0	0	0	0	0	0	0	0
χ_{18}	0	0	0	0	0	0	0	0	0	0
χ_{19}	0	0	0	0	0	0	0	0	0	0
χ_{20}	0	0	0	0	0	0	0	0	0	0
χ_{21}	0	0	0	0	0	0	0	0	0	0
χ_{22}	0	0	0	0	0	0	0	0	0	0
χ_{23}	0	0	0	0	0	0	0	0	0	0
χ_{24}	0	0	0	0	0	0	0	0	0	0
χ_{25}	0	0	0	0	0	0	0	0	0	0
χ_{26}	1	1	1	1	1	1	1	1	1	1
χ_{27}	0	0	0	0	0	0	0	0	0	0
χ_{28}	0	0	0	0	0	0	0	0	0	0
χ_{29}	0	0	0	0	0	0	0	0	0	0
χ_{30}	0	0	0	0	0	0	0	0	0	0
χ_{31}	0	0	0	0	0	0	0	0	0	0
χ_{32}	0	0	0	0	0	0	0	0	0	0
χ_{33}	0	0	0	0	0	0	0	0	0	0
χ_{34}	0	0	0	0	0	0	0	0	0	0
χ_{35}	0	0	0	0	0	0	0	0	0	0
χ_{36}	0	0	0	0	0	0	0	0	0	0
χ_{37}	0	0	0	0	0	0	0	0	0	0
χ_{38}	0	0	0	0	0	0	0	0	0	0

D = E(4) = ER(-1) = iE = 2*E(4) = 2*ER(-1) = 2i

F = -1 + E(4) = -1 + ER(-1) = -1 + i

 $C = -5 + 6^* E(4) = -5 + 6^* ER(-1) = -5 + 6i$

 $A = -1 + 6^* E(4) = -1 + 6^* ER(-1) = -1 + 6i$

B = 4*E(4) = 4*ER(-1) = 4i

 $\mathbf{L} = E(31)^2 + E(31)^{10} + E(31)^{19}$

 $\mathbf{K} = E(31)^{11} + E(31)^{24} + E(31)^{27}$

 $J = E(31)^8 + E(31)^9 + E(31)^{14}$

 $\mathbf{I} = E(31)^3 + E(31)^{13} + E(31)^{15}$

 $H = E(31) + E(31)^5 + E(31)^{25}$

 $G = -E(24)^{11} + E(24)^{19}$

Α

Programmes

Programme A

 $\begin{aligned} &\text{V:=FullRowSpace}(GF(q), n); \\ &\text{gr}[1]:=(\text{OneGF}(q))^*[n \times n \text{ matrix group generators}]; \\ &\text{gr}[2]:=(\text{OneGF}(q))^*[n \times n \text{ matrix group generators}]; \end{aligned}$

÷

```
gr[k]:=(OneGF(q))^*[n \times n matrix group generators];
grp:=Group(gr[1], gr[2], \cdots, gr[k]);
{\it Ccl:=ConjugacyClasses(grp);}
O:=Union(Orbits(grp,V));
for i in [1..n(Ccl)] do
Print(Representative(Ccl)[i]);
w{:=}One(GF(q))^*[0,0,\cdots,0];
e:=[ ];
while Difference
(O,e) <> [ ] do
d := [ ];
for {\bf x} in O do;
y{:=}[x{+}w{+}(x^*(\operatorname{Representative}(\operatorname{Ccl})[i]))];
d:=Union(d,y);
od;
Print(d);
e:=Union(d,e);
{\rm if \; Difference}(O,\!e) <> [ \ ] \; {\rm then} \\
w:=Representative(Difference(O,e));
fi;
od;
r{:}{=}[ \ ];
u{:=}One(GF(q))^*[0,0,\cdots,0];
while Difference
(O,e) <> [ ] do
\mathrm{m}{:=}[ \ ];
for g in Centralizer
(grp,Representative(Ccl)[i]) do
l:=[u^*g];
m:=Union(m,l);
od;
Print("A block for the vectors under the action of a centralizer");
Print(m);
r{:=}Union(m,r);
```

Programme B

 $\begin{aligned} &\text{V:=FullRowSpace}(GF(q), n); \\ &\text{m[1]:=(OneGF(q))*}[n \times n \text{ matrix group generators}]; \\ &\text{m[2]:=(OneGF(q))*}[n \times n \text{ matrix group generators}]; \end{aligned}$

:

$$\begin{split} \mathbf{m}[\mathbf{k}] &:= (\operatorname{OneGF}(\mathbf{q}))^* [n \times n \text{ matrix group generators}];\\ \mathbf{m} &:= \operatorname{Group}(m[1], m[2], \cdots, m[k]);\\ \mathbf{c} &:= \operatorname{ConjugacyClasses}(\mathbf{m});\\ \mathbf{g} &:= \operatorname{Representative}(\mathbf{c}[\mathbf{i}]);\\ \mathbf{d} &:= \operatorname{One}(\operatorname{GF}(\mathbf{q}))^* [\alpha_1, \alpha_2, \cdots, \alpha_n];\\ \mathbf{w} &:= d + d * g + d * g^2 + \cdots + d * g^{m-1};\\ \operatorname{Print}(\mathbf{w}); \end{split}$$

PROGRAMME C

gap>V:=FullRowSpace(GF(q), n); $gap>m[1]:=(OneGF(2))*[n \times n matrix group generators];$ $gap>m[2]:=(OneGF(2))*[n \times n matrix group generators];$

÷

$$\begin{split} m[\mathbf{k}] &:= (\mathrm{OneGF}(\mathbf{q}))^* [n \times n \text{ matrix group generators}];\\ &gap > m := \mathrm{Group}(m[1], m[2], \cdots, m[k]);\\ &gap > \mathbf{k} := \mathrm{OrbitLengths}(\mathbf{m}, \mathbf{V});\\ &gap > \mathbf{l} := \mathrm{OrbitLengths}(\mathrm{Group}(\mathrm{List}(\mathbf{m}, \mathrm{TransposedMat})), \mathbf{N}); \end{split}$$

PROGRAMME D

gap>g:=Group(Main Group); gap>T1:=CharacterTable(g); gap>h:=Group(Inertia Group 1); gap>T2:=CharacterTable(h); gap>k:=Group(Inertia Group 2); gap>T3:=CharacterTable(k); gap>FusionConjugacyClasses(h,g); gap>FusionConjugacyClasses(k,g);

PROGRAMME E

$$\begin{split} & \text{gap} > \text{ct:=} \text{fuction}() \text{local ct;ct:=} \text{rec}(); \\ & > \text{ct.SizesCentralizers:=} [n \text{ Centralizer Orders}]; \\ & > \text{ct.OrdersClassRepresentatives:=} [n \text{ Class Representatives Orders}]; \\ & > \text{ct.Irr:=} [[n \times n \text{ irreducibles}]]; \\ & > \text{ct.UnderlyingCharacteristic:=} 0; \text{ct.Id:=} \overline{G}; \\ & > \text{ConvertToLibraryCharacterTable NC(ct); return ct; \text{end}; \text{ct:=} \text{ct}(); \\ & \text{gap} > \text{SetInfoLevel}(\text{InfoCharacterTable}, 2); \\ & \text{gap} > \text{IsInternallyConsistent}(\text{ct}); \end{split}$$

gap>PossiblePowerMaps(ct, p); (p-prime divisor of \overline{G}).

PROGRAMME F

$$\begin{split} & \text{gap>ct:=CharacterTable(G);} \\ & \text{gap>SetInfoLevel(InfoCharacterTable,2);} \\ & \text{gap>ct;} \\ & \text{gap>cut:=Irr(ct)}\{[i..j]\};; \\ & \text{gap>} k := k_1 * \operatorname{cut}[i_1] + k_2 * \operatorname{cut}[i_2] + \ldots + k_r * \operatorname{cut}[i_r]; \end{split}$$

Programme G

work := [];;output := [];;work[1] := a;; $\operatorname{work}[2] := b;;$ $\operatorname{work}[3] := \operatorname{work}[1] * \operatorname{work}[2];;$ $\operatorname{work}[4] := \operatorname{work}[3] * \operatorname{work}[2];;$ $\operatorname{work}[5] := \operatorname{work}[3] * \operatorname{work}[4];;$ work[6] := work[3] * work[5];; $\operatorname{work}[7] := \operatorname{work}[6] * \operatorname{work}[3];;$ work[8] := work[7] * work[4];; $\operatorname{work}[9] := \operatorname{work}[3] * \operatorname{work}[8];;$ $\operatorname{work}[2] := \operatorname{work}[1] * \operatorname{work}[9];;$ work[4] := work[3] * work[3];;work[5] := work[3] * work[4];; $\operatorname{work}[4] := \operatorname{work}[5]^{-}1;;$ $\operatorname{work}[3] := \operatorname{work}[4] * \operatorname{work}[2];;$ $\operatorname{work}[2] := \operatorname{work}[3] * \operatorname{work}[5];;$ $\operatorname{work}[6] := \operatorname{work}[7] * \operatorname{work}[7];;$ work[5] := work[7] * work[6];;work[4] := work[6] * work[5];; $\operatorname{work}[3] := \operatorname{work}[4]^{-}1;;$ $\operatorname{work}[5] := \operatorname{work}[3] * \operatorname{work}[1];;$ $\operatorname{work}[1] := \operatorname{work}[5] * \operatorname{work}[4];;$ output[1] := work[1];; $\operatorname{output}[2] := \operatorname{work}[2];;$

Programme H

work := [];;output := [];;work[1] := c;; $\operatorname{work}[2] := d;;$ work[3]:= work[1] * work[2];;work[4] := work[3] * work[2];;work[5] := work[4] * work[3];; $\operatorname{work}[6] := \operatorname{work}[2] * \operatorname{work}[2];;$ $\operatorname{work}[7] := \operatorname{work}[4] * \operatorname{work}[6];;$ work[4] := work[5] * work[7];; $\operatorname{work}[6] := \operatorname{work}[2] * \operatorname{work}[5];;$ $\operatorname{work}[5] := \operatorname{work}[6] * \operatorname{work}[3];;$ $work[2] := work[5]^{-1};;$ $\operatorname{work}[3] := \operatorname{work}[2] * \operatorname{work}[4];;$ $\operatorname{work}[2] := \operatorname{work}[3] * \operatorname{work}[5];;$ output[1] := work[1];; $\operatorname{output}[2] := \operatorname{work}[2];;$

Programme I

work := [];; output := [];; work[1] := a;; work[2] := b;; work[3] := work[1] * work[2];; work[3] := work[3] * work[2];; work[4] := work[3] * work[2];; work[5] := work[3] * work[4];; work[6] := work[3] * work[3];; work[5] := work[6] * work[3];; work[5] := work[6] * work[3];; work[1] := work[5] * work[1];; work[2] := work[7]^3;; work[6] := work[4]^{12};; work[5] := work[6]^{-1};; $\operatorname{work}[2] := \operatorname{work}[3] * \operatorname{work}[6];;$

 $\operatorname{output}[1] := \operatorname{work}[1];;$

 $\operatorname{output}[2] := \operatorname{work}[2];;$

 $\overline{G} := \operatorname{Group}(\operatorname{output}[1], \operatorname{output}[2]);$

Programme J

```
> G:=SL(2,5);
> M:=GModule(G);
> X:=CohomologyModule(G,M);
> E:=SplitExtension(X);
> Eperm:=DegreeReduction(CosetImage(E,sub<E|>));
> pMultiplicator(Eperm,2);
[ 1 ]
> pMultiplicator(Eperm,3);
[ 1 ]
> pMultiplicator(Eperm,5);
[ 5 ]
> exit;
```

 $\mathbf{Programme}~\mathbf{J}'$

gap> gg:=PerfectGroup(3000,1);

A5 2^1 5^2

gap> AbelianInvariantsMultiplier(gg);

[5]

Β

Character Tables of S2 and S3

Table 1

	2 3	8 2	8 1	5 1	7	7 1	3 2	3 1	6	6	5	4	4	2 1	2 1	6 1	6	4	4	·	1	1 2
	5	1												•						1		
	1a	2a	4a	2b	2c	3a	6a	4b	2d	4c	4d	8a	6b	12a	2e	4e	4f	8b	5a	6c	Зb	
X.1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	
X.2	1	1	-1	-1	-1	1	1	1	1	1	-1	-1	-1	-1	-1	-1	1	1	1	-1	1	
X.3	5	5	3	3	3	2	2	1	1	1	1	1	•		-1	-1	-1	-1		-1 -	-1	
X.4	5	5	-3	-3	-3	2	2	1	1	1	-1	-1			1	1	-1	-1	. :	L -:	1	
X.5	5	5	-1	-1	-1	-1	-1	1	1	1	1	1	-1	-1	3	3	-1	-1	•	. 2		
X.6	5	5	1	1	1	-1	-1	1	1	1	-1	-1	1	1	-3	-3	-1	-1	•	. 2		
X.7	9	9	-3	-3	-3	•	•	1	1	1	1	1	•		-3	-3	1	1 -:	1.			
X.8	9	9	3	3	3			1	1	1	-1	-1		. 3	3	1 1	-1					
X.9 10 10 -2 -2 -2 1 1 -2 -2 -2 1 1 2 21 1																						
X.10 10 10 2 2 2 1 1 -2 -2 -21 -1 -2 -2 1 1																						
X.11 15 -1 -1 3 -5 3 -1 3 -1 -1 -1 1 1 -1 3 -1 -1 1																						
X.12 15 -1 1 1 -7 3 -1 -1 3 -1 -1 1 -1 1 -3 1 1 -1																						
X.13 15 -1 1 -3 5 3 -1 3 -1 -1 1 -1 1 -3 1 -1 1																						
X.14 15 -1 -1 -1 7 3 -1 -1 3 -1 1 -1 1 -1 3 -1 1 -1																						
X.15 16 162 -2 12																						
X.16 30 -2 -2 2 2 -3 1 2 2 -21 1 6 -2																						
X.17	30	-2	2 -	2 -	2 -	31	2 2	2 -:	2		. 1	-1	-6	2								
X.18 45 -3 -3 1 93 1 1 -1 13 1 -1 1																						
X.19 45 -3 3 -1 -93 1 1 1 -1 3 -1 -1 1																						
X.20 45 -3 -3 5 -3 1 -3 1 1 -13 1 1 -1																						
X.21	45	-3	3 -	53		. :	1 -:	31	-1	1		. :	3 -	11	-1							

APPENDIX B. CHARACTER TABLES OF S2 AND S3

Table 2

	2 8 8 7 7 3 3 5 6 6 6 5 4 4 4 1 1 1 3 2 1 1 2 .	
		1 h
v 4		D
X.1		
X.2		
X.3	5 5 5 5 2 2 2 2 1 1 1 1 1 1 1 -1 -1 -1	
X.4	5 5 5 5 5 -1 -1 -1 -1 1 1 1 1 1 1 -1 -1 -1 -1	
X.5	5 5 -5 -5 2 2 -2 -2 -1 -1 -1 1 1 1 1 1 -1 -1	
X.6	5 5 -5 -5 -1 -1 1 1 -1 -1 -1 1 1 1 1 1 -1 -1	
X.7	8 8-8 -8 -1 -1 1 1	
X.8	88 -8 -8 -1 -1 1 1	
X.9	8 8 8 8 -1 -1 -1 -1 A A*A *A	
X.10	8 8 8 8 -1 -1 -1 -1 *A*A A A	
X.11	9999	
X.12	9 9 -9 -91 -1 -1 1 1 1 -1 -1 1 1 -1 1 1	
X.13	10 10 10 10 1 1 1 1 -2 -2 -2 -2 -2 -2	
X.14	10 10 -10 -10 1 1 -1 -1 2 2 2 -2 -2 -2 -2	
X.15	15 -1 -5 3 3 -1 1 -3 -1 -1 3 3 -1 -1 1 -1	
X.16	15 -1 -5 33 -1 1 -3 -13 -1 -13 -1 -11 1 -11	
X.17	15 -1 5 -3 3 -1 -1 3 1 -3 1 -1 3 -1 1 -1 -1 1	
X.18	15 -1 5 -3 3 -1 -1 3 1 1 -3 3 -1 -1 -1 1 1 1 -1	
X.19	30 -2 -10 6 -3 1 -1 3 -2 2 2 2 2 -2	
X.20	30 -2 10 -6 -3 1 1 -3 2 -2 -2 2 2 -2	
X.21	45 -3 -15 9 1 -3 1 1 -3 1 -1 1 -1 1	
X.22	45 -3 -15 9 1 1 -3 -3 1 1 1 -1 1 -1	
X.23	45 -3 15 -91 -1 3 -3 1 1 -1 1 1 -1	
X.24	45 -3 15 -91 3 -1 1 -3 1 1 -1 -1 1	
	2 1 1 3 2 2	
	5	
	6d 3b	
X.1	1 1	
X.2	-1 1	
X.3	-1 -1	
X.4	2 2	
X.5	1 -1	
X.6	-2 2	

- X.7 1 -1 X.8 1 -1 X.9 -1 -1 X.10 -1 -1 X.11 . . X.12 1 1 X.13 X.14 -1 1 X.15 X.16 X.17 X.18 . . X.19 . . X.20 X.21 . . X.22 X.23 . . X.24
- $A = -E(5)-E(5)^{4}$ = (1-ER(5))/2 = -b5

Bibliography

- [1] F. Ali, *Fischer-Clifford Theory for Split and Non-Split Group Extensions*, PhD thesis, University of Natal, Pietermaritzburg, 2001.
- F. Ali and J. Moori, Fischer-Clifford matrices of the non-split group extension 2⁶U₄(2), Quaest. Math. 31 (2008), no. 1, 27–36.
- [3] F.Ali and J. Moori, The Fischer-Clifford matrices and character table of a maximal subgroup of Fi₂₄', Algebra Colloq. 3 (2010), 389 - 414.
- [4] M. S. O. Almestady, Fischer Matrices for Generalized Symmetric Groups A Combinatorial Approach, PhD thesis, the University of Wales, 1998.
- [5] Z. Arad, Zeros in character tables of finite groups, Algebra Colloq. 1 (1994), 225 232.
- [6] M. Aschbacher, 3-Transposition Groups, Cambridge University Press, Cambridge, 1977.
- [7] Y. Berkovich, Finite groups with eight non-linear irreducible characters, Rend. Mat. Acc. Lincei 5 (1994), 141 148.
- [8] Y. Berkovich, Finite groups in which the degrees of non-linear constituents of some induced characters are distinct, Publ. Math. Debrecen 44 (1994), 225 234.
- [9] Y Berkovich, On induced characters, Proc. Amer. Math. Soc. 121 (1994), 679 685.
- [10] Y. Berkovich, Finite groups with small sums of degrees of some non-linear irreducible characters, J. Algebra 171 (1995), 426 - 443.
- [11] Y. Berkovich, E. Zhmud, *Characters of finite groups*. Part 1. Translations of Mathematical Monographs, **172**, Amer. Math. Soc., Providence, RI, 1998.
- [12] H. I. Blau and D. Chillag, On powers of characters and powers of conjugacy classes of a finite group, Proc. Amer. Math. Soc. 98 (1986), 7 - 10.
- [13] W. Bosma and J. J. Cannon, Handbook of Magma Functions, Department of Mathematics, University of Sydeny, November 1994.

- [14] R. Brauer, *Representations of finite groups*, Lectures on Modern Mathematics (T. L. Saaty, ed), J. Wiley and Sons (1963), 133 175
- [15] T. Breuer and K. Lux, The multiplicity-free permutation characters of the sporadic simple groups and their automorphism groups, Comm. Algebra 24 (1996), 2293 - 2316.
- [16] G. Butler, Computing the conjugacy classes of elements of a finite group, preprint.
- [17] G. Butler, An inductive schema for computing conjugacy classes in permutation groups, Math. Comp. 62 (1994), 363 - 383.
- [18] J. J. Cannon, An introduction to the group theory language CAYLEY, Computational Group Theory (M. Atkinson, eds), Academic Press, San Diego, 1984, 145 - 183.
- [19] R. W. Carter, Simple Groups of Lie Type, J. Wiley, New York, 1972.
- [20] A. H. Clifford, Representations induced in an invariant subgroup, Ann. of Math. 38 (1937), 533 - 550.
- [21] M. J. Collins, Representations and Characters of Finite Groups, Cambridge Studies in Advanced Mathematics 22, Cambridge University Press, Cambridge, 1990.
- [22] J. H. Conway A perfect group of order 8315553613086720000 and the sporadic simple groups, Bull. London Mth. Soc., 1(1969), 79-88.
- [23] J. H. Conway, R. T. Curtis, S. P. Norton, R. A. Parker and R. A. Wilson, Atlas of Finite Groups, Oxford University Press, Oxford, 1985.
- [24] C. W. Curtis and I. Reiner, Representation Theory of Finite Groups and Associative Algebras, Pure and Applied Mathematics XI, Interscience, New York, 1962.
- [25] H. Cuypers and J. I. Hall, The 3-transposition groups with trivial center, J. Algebra 178 (1995), 149 - 193.
- [26] M. R. Darafsheh and A. Iranmanesh, Computation of the character table of affine groups using Fischer matrices, London Mathematical Society Lecture Note Series 211, Vol. 1, C. M. Campbell et al., Cambridge University Press (1995), 131 - 137.
- [27] M. R. Darafsheh and A. Iranmanesh, Construction of the character table of the hyperoctahedral group, Riv. Mat. Pura Appl. 17 (1996), 71 - 82.
- [28] J. D. Dixon, The Structure of Linear Groups, Van Nostrand Reinhold Company, New York, 1971.
- [29] K. B. Farrmer, A survey of projective representation theory of finite groups, Nieuw. Arch. Wisk. 26 (1978), 292 - 308.

- [30] W. Feit, Characters of Finite Groups, W. A. Benjamin, New York, 1967.
- [31] W. Feit, Some consequences of the classification of finite simple groups, Proceedings, Symposia in Pure Mathematics 37, Amer. Math. Soc., Providence, Rhode Island, 1980.
- [32] B. Fischer, Finite Groups Generated by 3-Transpositions, Notes, Mathematics Institute, University of Warwick, 1970.
- [33] B. Fischer, Finite groups generated by 3-transpositions, Inventiones Math 13 (1971), 232 246.
- [34] B. Fischer, *Clifford matrizen*, manuscript (1982).
- [35] B. Fischer, unpublished manuscript (1985).
- [36] B. Fischer, *Clifford matrices*, Progr. Math. 95, Michler G. O. and Ringel C. M. (eds), Birkhauser, Basel (1991), 1 - 16.
- [37] B. Fischer, Character tables of maximal subgroups of sporadic simple groups -III, Preprint.
- [38] S. M. Gagola, Jr., An extension theorem for characters, Proc. Amer. Math. Soc. 83 (1981), 25
 26.
- [39] P. X. Gallagher, Group characters and normal Hall subgroups, Nagoya Math. J. 21 (1962), 223 - 230.
- [40] P. X. Gallagher, The number of conjugacy classes in a finite group, Math. Z. 18 (1970), 175 -179.
- [41] The GAP Group, GAP Groups, Algorithms and Programming, Version 4.4, Aachen, St Andrews, 2008, (http://www-gap.dcs.st-and.ac.uk/~gap).
- [42] S. P. Glasby, On the Faithful representations, of degree 2ⁿ, of certain extensions of 2-groups by orthogonal and symplectic Groups, J. Austral. Math. Soc. 58 (1995), 232 - 247.
- [43] D. Gorenstein, *Finite Groups*, Harper and Row Publishers, New York, 1968.
- [44] R. Gow, Some characters of affine subgroups of classical groups, J. London Math. Soc. 2 (1976), 231 - 238.
- [45] R. L. Griess, *The friendly giant*, Invent. Math. **69** (1982), 1 102.
- [46] R. M. Guralnick, Subgroups inducing the same permutation representation, J. Algebra 81 (1983), 312 - 319.
- [47] R.J. Haggarty and J.F. Humphreys, Projective characters of finite group, Proc. London Math. Soc. 36 (1975), 176 - 192.
- [48] M. Hall, Jr., The Theory of Groups, The Macmillan Company, New York, 1959.

- [49] D.G. Higman and C.C. Sims, A simple group of order 44352000 Math. Z. 105 (1968), 110-113.
- [50] D.G. Higman, On the simple group of D.H. Higman and C.C. Sims Illinois Journal of Mathematics 13 (1969), 74-80.
- [51] P. N. Hoffman and J. F. Humphreys, Projective representations of the symmetric groups, Clarendon Press, Oxford, 1992.
- [52] C. Holmes, Split extensions of abelian groups with identical subgroup structures, Contemp. Math. 33 (1984), 265 - 273.
- [53] D. F. Holt, The calculation of the Schur multiplier of a permutation group, Proc. London Math. Soc., Meeting on Computational Group Theory (Durham, 1982), Academic Press, London (1984), 307 -319.
- [54] D. F. Holt, A computer program for the calculation of a covering group of a finite group, J. Pure Applied Alg. 35 (1985), 287 - 295.
- [55] J. F. Humphreys, Projective character tables for the finite simple groups of order less than one million, Comm. Algebra (11) 7 (1983), 725 - 751.
- [56] J. F. Humphreys, A Course in Group Theory, Oxford University Press, Oxford, 1996.
- [57] B. Huppert, Eindliche Gruppen I, Springer, Berlin, 1967.
- [58] B. Huppert, Character Theory of Finite Groups, Walter de Gruyter, Berlin, 1998.
- [59] I. M. Isaacs, Characters of solvable and symplectic groups, Amer. J. Math. 95 (1973), 594 -635.
- [60] I. M. Isaacs, Character Theory of Finite Groups, Academic Press, San Diego, 1976.
- [61] I. M. Isaacs, Blocks with just two irreducible Brauer characters in solvable groups, J. Algebra 170 (1994), 487 - 503.
- [62] I. M. Isaacs and I. Zisser, Squares of characters with few irreducible constituents in finite groups, Arch. Math 63 (1994), 197 - 207.
- [63] G. James and M. Liebeck, *Representations and Characters of Groups*, Cambridge Mathematical Textbooks, Cambridge University Press, Cambridge, 1993.
- [64] C. Jansen, K. Lux, R. Parker and R. Wilson, An Atlas of Brauer Characters, London Mathematical Society Monographs New Series 11, Oxford University Press, Oxford, 1995.
- [65] G. Karpilovsky, On extensions of characters from normal subgroups, Proc. Edinburgh Math. Soc. 27 (1984), 7 - 9.

- [66] G. Karpilovsky, Projective representations of finite groups, Marcel Dekker, New York, 1985.
- [67] G. Karpilovsky, Group Representations: Introduction to Group Representations and Characters, Vol 1 Part B, North - Holland Mathematics Studies 175, Amsterdam, 1992.
- [68] M. Kitazume, Some non-split extensions of the orthogonal group $O_7(3)$, J. Algebra **224** (2000), 59 76.
- [69] M. Kitazume and S. Yoshiara, The radical subgroups of the Fischer simple groups, Preprint.
- [70] A. S. Kusefoglu, The second cohomology of finite orthogonal groups I, J. Algebra 56 (1979), 207 - 220.
- [71] W. Ledermann, *Introduction to Group Characters*, Cambridge University Press, Cambridge, 1977.
- [72] J. Leech, Notes on sphere packings, Canad. J. Math., 19 (1967), 251-257.
- [73] S. A. Linton and R. A. Wilson, The maximal subgroups of the Fischer groups Fi_{24} and Fi'_{24} , Proc. London. Math. Soc. **63** (1991), 113 - 164.
- [74] D. E. Littlewood, The Theory of Group Characters, Oxford University Press, Oxford, 1958.
- [75] R. J. List, On the characters of $2^{n-\epsilon} S_n$, Arch. Math. **51** (1988), 118-124.
- [76] R. J. List and I. M. I. Mahmoud, Fischer matrices for wreath products $G \le S_n$, Arch. Math. **50** (1988), 394-401.
- [77] K. Lux and H. Pahlings, Groups a Computational Approach, Cambridge University Press, New York 2010.
- [78] R. Lyons, Errata, Evidence for a new finite simple group, J. Algebra 20 (1972) 540-569 and J.
 Algebra 34 (1975) 188-189.
- [79] G. W. Mackey, Unitary representations of group extensions. I, Acta Math. 99 (1958), 265 -311.
- [80] M. K. Marshall, Numbers of conjugacy class sizes and derived lengths for A-groups, Canad. Math. Bull. 39 (1996), 346 - 351.
- [81] J. Moori, On the Groups G^+ and G of the forms $2^{10}:M_{22}$ and $2^{10}:\overline{M}_{22}$, PhD thesis, University of Birmingham, 1975.
- [82] J. Moori, On certain groups associated with the smallest Fischer group, J. London Math. Soc. 2 (1981), 61 - 67.
- [83] J. Moori, On the automorphism group of the group $D_4(2)$, J. Algebra 80 (1983), 216 225.

- [84] J. Moori, Action tables for the Fischer group \overline{F}_{22} , Proceedings of the 1987 Singapore Conference, Walter de Gruyter, Berlin New York, (1989), 417 435.
- [85] J. Moori, Classification of finite simple groups, South African Journal of Science 89 (1993), 29
 34.
- [86] J. Moori, Subgroups of 3-transposition groups generated by four 3-transpositions, Quaestiones Math. 17 (1994), 83 - 94.
- [87] J. Moori, On the affine subgroups of the symplectic groups, Article presented at the 9th Kwazulu Natal Conference, University of Durban-Westville, 1997.
- [88] J. Moori, Fischer-Clifford matrices and representation theory of group extensions, Article presented at the First Joint International Meeting of the AMS and the HKMS, December 13-16, 2000 Hong Kong, People's Republic of China.
- [89] J. Moori, *Representation Theory*, Lecture Notes, University of Natal, Pietermaritzburg.
- [90] J. Moori , Z.E. Mpono, The centralizer of an involutory outer automorphism of F_{22} , Math. Japonica **49** (1999), 93 113.
- [91] J. Moori , Z.E. Mpono, The Fischer-Clifford matrices of the group 2⁶:SP₆(2), Quaestiones Math. 22 (1999), 257-298.
- [92] J. Moori, Z.E. Mpono, Fischer-Clifford matrices and the character table of a maximal subgroup of \bar{F}_{22} , Intl. J. Maths. Game Theory, and Algebra **10** (2000), 1 12.
- [93] J Moori, Z E Mpono, The Fischer-Clifford matrices of the group 2⁶:SP(6,2), Quaest. Math.,
 22 (1999), 257 298.
- [94] J Moori , K Zimba, Permutation actions of the symmetric group S_n on the groups Z_m^n and $\overline{Z_m^n}$, Quaest. Math. **28** (2005), no. 2, 179–193.
- [95] A.O. Morris, Projective representations of finite groups, Proc. on Conference on Clifford Algebra, Matscience, Madras, 1971.
- [96] A.O. Morris, Projective representations of Weyl groups, J. London Math. Soc. 8 (1974), 125 -133.
- [97] A.O. Morris, Projective representations of exceptional Weyl groups, J. Algebra 29 (1974), 567 586.
- [98] A.O. Morris, Projective representations of reflection groups, Proc. London Math. Soc. 32 (1976), 403 - 420.
- [99] Z. E. Mpono, Fischer-Clifford Theory and Character Tables of Group Extensions, PhD thesis, University of Natal, Pietermaritzburg, 1998.

- [100] H. Nagao and Y. Tsushima, Representations of Finite Groups, Academic Press, San Diego, 1987.
- [101] G. Navarro, Characters with stable irreducible constituents, J. Algebra 172 (1995), 320 334.
- [102] J. Neubuser, An invitation to computational group theory, unpublished.
- [103] H. Pahlings, Computing with characters of finite groups, Acta Appl. Math. 21 (1990), 41 -56.
- [104] D. V. Pasechnik, Geometric characterization of the sporadic groups Fi₂₂, Fi₂₃ and Fi₂₄, J. Combin. Theory Ser. A 68 (1994), 100 114.
- [105] W. Plesken and B. Souvignier, Constructing rational representations of finite groups, Experiment. Math. 5 (1996), 39 - 47.
- [106] A. Previtali, Orbit lengths and character degrees in p-sylow subgroups of some classical Lie groups, preprint.
- [107] E.W. Read, Projective characters of the Weyl group of type F_4 , J. London Math. Soc. 8 (1974), 83 93.
- [108] E.W. Read, The linear and projective characters of the finite reflection groups of type H_4 , Quart. J. Math. Oxford **25** (1974), 73 -79.
- [109] E.W. Read, On projective representations of finite reflection groups of type B_l and D_l , J. London Math. Soc. **10** (1975) 129 -142.
- [110] B. G. Rodrigues, On the theory and examples of group extensions, MSc thesis, University of Natal, Pietermaritzburg 1999.
- [111] J. J. Rotman, An Introduction to the Theory of Groups, Allyn and Bacon, Inc., Boston, 1984.
- [112] R. B. Salleh, On the Construction of the Character Tables of Extension Groups, PhD thesis, University of Birmingham, 1982.
- [113] U. Schiffer, *Cliffordmatrizen*, Diplomarbeit, Lehrstul D Fur Matematik, RWTH, Aachen, 1995.
- [114] I. Schur, Uber die Darstellung der endlichen Gruppen durch gebrochene lineare Substitutionen,
 J. f
 ür Math. 127 (1904), 20 -50.
- [115] W. R. Scott, *Group Theory*, Prentice-Hall, Inc., Englewood Cliffs, New Jersey, 1964.
- [116] H. Shi, On the characters of symmetric group, Acta Math. Sinica 10 (1994), 74 85.
- [117] J. S. Shin, Properties of finite groups whose irreducible character degrees are primes, J. Korean Math. Soc. 31 (1994), 1 - 9.

- [118] C.C. Sims, The Existence and Uniqueness of Lyons Group, Gainesville Conf., University of Florida, 1972, 138 - 141.
- [119] C C. Sims, A presentation for the Lyons simple group. Computational methods for representations of groups and algebras, Progr. Math., Birkhuser, Basel, 173 (1999), 241 - 249.
- [120] N. S. Whitley, Fischer Matrices and Character Tables of Group Extensions, MSc thesis, University of Natal, Pietermaritzburg, 1994.
- [121] H. Wielandt, Finite Permutation Groups, Academic Press, New York, 1964.
- [122] R. A. Wilson, Is J_1 a subgroup of the Monster ?, Bull. London. Math. Soc., 18 (1986), 349-350.
- [123] R. A. Wilson, The local subgroups of the Fischer groups, J. London. Math. Soc. 36 (1987), 77 - 94.
- [124] R. A. Wilson, et al, Atlas of Finite Group Representation, http://brauer.maths.qmul.ac.uk/ Atlas/version 3, 2006.
- [125] R. A. Wilson, *The Finite Simple Groups*, Springer Verlag 2007.
- [126] K. Zimba, Fischer-Clifford Matrices of the Generalized Symmetric Group and some Associated Groups, PhD thesis, University of KwaZulu Natal, 2005.